### SUPPLEMENT USE, VITAMIN C INTAKE AND

#### **BREAST CANCER RISK IN UK WOMEN**

Jayne Hutchinson

Submitted in accordance with the requirements for the degree of PhD

The University of Leeds

School of Medicine

School of Food Science and Nutrition

December 2011

#### **DECLARATION**

I confirm that the work submitted is my own, except where work which has formed part of jointly authored publications has been included. My contribution and the contribution of the other authors to this work has been explicitly indicated below. I confirm that appropriate credit has been given within the thesis where reference has been made to the work of others.

Chapter 6, relating to vitamin C supplement taking and family history of cancers, incorporates a large part of a jointly authored published paper.<sup>1</sup> I was jointly responsible for the design of the analysis, along with Janet Cade, Victoria Burley and Darren Greenwood, and I undertook the analysis and interpretation. Darren Greenwood provided statistical advice. I was the lead author responsible for writing and submitting the manuscript, and incorporated comments from the above co-authors. James Thomas managed the creation of the UKWCS supplement databases, and also enabled the calculation of vitamin C intake from all supplements used. I undertook an extensive final cleaning of the participant supplement database for these supplements.

Chapter 10, relating to vitamin C intake from diary recordings and risk of breast cancer in the UK Dietary Cohort Consortium, incorporates work from a jointly authored published paper.<sup>2</sup> The following co-authors contributed to the design, development and establishment of the consortium of five cohorts from the Universities of Oxford, Cambridge, Leeds and UCL: Marleen Lentjes, Darren Greenwood, Victoria Burley, Janet Cade, Cristina Cleghorn, Diane Threapleton, Tim Key, Benjamin Cairns, Ruth Keogh, Christina Dahm, Eric Brunner, Martin Shipley, Diana Kuh, Gita Mishra, Alison Stephen, Amit Bhaniani, Gabor Borgulya and Kay Tee Khaw. I provided supplement data from the UKWCS and EPIC-Oxford cohorts, and checked and recoded UKWCS covariates for the pooled dataset used in this thesis. Marleen Lenties supplied EPIC-Norfolk supplement data. I was jointly responsible for the design of the vitamin C analysis, along with Janet Cade, Victoria Burley and Darren Greenwood, and I undertook the analysis and interpretation. Darren Greenwood provided statistical advice. I was the lead author responsible for writing and submitting the manuscript and the other co-authors were invited to review it. Marleen Lentjes, Janet Cade, Victoria Burley, Darren Greenwood, Ruth Keogh and Ben Cairns provided the majority of comments, most of which were incorporated into the manuscript.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement. © 2011 University of Leeds and Jayne Hutchinson.

## ACKNOWLEDGMENTS

There are many organisations and people to whom I am very grateful, and without their help I would have not been able to produce this thesis and achieve my aims of developing an understanding of and adding a little to this research area.

First of all I would like to thank my wonderful supervisors Janet Cade, Victoria Burley and Darren Greenwood who have provided excellent direction, advice and encouragement and who have been enthusiastic and motivating during my supervision meetings. Extra thanks go to Darren Greenwood for his invaluable help with the statistics. I am also grateful to Darren Daily for explaining Directed Acyclic Graphs as one way to understand the art of adjusting for confounders. Everyone in the Nutritional Epidemiology Department has been so friendly and welcoming, and their lunchtime seminars have been engaging and very informative. Thank you for the encouragement I have received from the PhD students at the Universities of Leeds and York, and also from Elaine at Bath. In addition to my supervisors, I am grateful to those at the Open University and the University of York, who in the past, encouraged me and helped me to develop an ability to write academically and gave me an understanding of health research.

I am grateful to the World Cancer Research Fund for funding the establishment of the UK Women's Cohort Study, thank you to the thousands of women who took part, and the many people at Leeds involved in the set-up of the study and data input which happened years before I arrived. Many thanks go to James Thomas, who managed the databases and who provided specific data for me from the supplement databases. I am also very grateful to the UK Dietary Cohort Consortium members for the use of their dataset, and to those who reviewed the manuscript resulting from my analysis. Thank you to Marleen Lentjes for supplying the Cambridge supplement data and to Tim Key for allowing me to photocopy the Oxford supplement diary data, and for the help I received there.

Thanks to all my friends who have helped to keep me sane as well as physically healthy whilst socialising on our runs, bike rides and during other activities in the last three years. I am so grateful to my husband, Peter, and my teenage son, Phil, for their understanding, support and encouragement, and for bearing with me during my preoccupation with this research. Thank you also to Peter for sorting out my hardware problems at home and for taking over the cooking and shopping partway through the PhD. Finally, a big thank you goes to Chris, Elaine and Peter for proof-reading.

## ABSTRACT

*Background:* When analysing relationships between breast cancer risk and vitamin C intake, few prospective studies have included vitamin C intake from supplements, or excluded general supplement users in dietary only analyses. For the first time these relationships are explored in UK women, and from diary recordings.

*Methods:* The UK Women's Cohort Study was used in prospective breast cancer risk analyses examining exposures from: general supplement use; fruit and vegetable intake; and dietary only vitamin C intake recorded at baseline using FFQs and additional questions for 33,000 women (~1,000 cases); vitamin C contained in supplements recorded by diaries at phase 2 for 11,000 women (239 cases); and total vitamin C intake from diet and supplement recorded by diaries in pooled UK nested case-control studies (851 cases 2727 controls).

*Results*: There was no evidence of dose-response relationships between breast cancer risk and vitamin C intake from diet, supplements or both, or from fruit and vegetable intake; risk estimates were non-significant and generally close to unity. There was some evidence that risks differed by menopausal status and supplement use. There were no significant associations for non-users of supplements or post-menopausal women by continuous estimate or intake category. Risks were raised for pre-menopausal women who were: frequent users of supplements containing low vitamin C (1-60mg/d) (HR=2.37; 95% CI: 1.32, 4.27; p=0.004); daily multivitamin users (HR=1.51; 95% CI: 0.90, 2.54); or general supplement users (HR=1.14; 95% CI: 0.91, 1.43), compared to non-users of these supplements. Women with a history of breast cancer were significantly more likely to be high dose vitamin C users ( $\geq 1000mg/d$ ).

*Conclusion:* There was no evidence of significant associations between vitamin C intake per se and breast cancer incidence, even at high doses. The increased risk for pre-menopausal women taking supplements containing low dose vitamin C may be due to other ingredients.

## **TABLE OF CONTENTS**

C	HAPTER	1	1		
1	Introduc	tion and Objectives	1		
	1.1 Ob	jectives	4		
C	HAPTER 2				
2	Supplem	nent use and vitamin C	7		
	2.1 Su	pplement use in the UK	7		
	2.2 Vit	amin C	10		
	2.2.1	Vitamin C recommended intake	11		
	2.2.2	Vitamin C bioavailability, absorption, transport and excretion	12		
	2.2.3	Characteristics associated with vitamin C plasma concentrations	13		
	2.2.4 cance	Potential mechanisms of vitamin C and other antioxidants in relatio	n to 14		
C	HAPTER	3	18		
3	Literatur	e review of breast cancer risks	18		
	3.1 Int	roduction	18		
	3.2 Cla	assification of breast cancer	19		
	3.3 Br	east cancer incidence, trends and survival	20		
	3.4 Ma	ajor breast cancer risk factors	23		
	3.4.1	Family history and genetics	23		
	3.4.2	Breast density	24		
	3.4.3	Previous diagnosis	25		
	3.4.4	Thoracic radiotherapy	25		
	3.4.5	Risk prediction tools	25		
	3.5 Ho	rmonal and related factors affecting risk of breast cancer	25		
	3.5.1	Hormone lifetime exposure	25		
	3.5.2	Exogenous hormones in HRT and contraceptive pills	26		
	3.5.3	Endogenous hormones	27		
	3.5.4	Pregnancy and lactation	29		
	3.5.5	Body fatness	30		

3	.5.6	Phy	sical activity	32
3	.5.7	Hori	monal exposure in utero and childhood development	32
3.6	Die	et and	breast cancer risk	33
3	.6.1	Diet	ary Patterns	34
	3.6	.1.1	Healthy and unhealthy dietary patterns	.35
	3.6	.1.2	Vegetarian compared to meat consumption dietary patterns	.35
3	.6.2	Foo	ds	36
	3.6	.2.1	Fruit and vegetables	.37
	3.6	.2.2	Dairy products	.39
3	.6.3	Mac	ronutrients and breast cancer risk	39
	3.6	.3.1	Fats	.39
	3.6	.3.2	Carbohydrates	.40
	3.6	.3.3	Fibre	.41
	3.6	.3.4	Alcohol	.42
3	.6.4	Micr	onutrients and breast cancer risk	42
	3.6 car	.4.1 ncer ri	WCRF systematic literature review of vitamin C intake and brea	ast .44
3.7 can	Up cer	dated	systematic literature review of vitamin C intake and risk of brea	ast 45
3	.7.1	Crite	eria for considering studies	45
	3.7	.1.1	Search strategy for identification of studies	.45
3	.7.2	Sea	rch results summary	47
	3.7	.2.1	Quality of studies	55
3	.7.3	Diet	ary vitamin C intake	56
	3.7	.3.1	Post-menopausal women	56
	3.7	.3.2	Pre-menopausal women	.58
	3.7	.3.3	ВМІ	.59
	3.7	.3.4	Menopausal status not specified	.59

3.7.4	Supplement only vitamin C intake	60
3.7.5	Total vitamin C intake	61
3.7.6	Conclusion	62
3.7 ne	7.6.1 How the current thesis may address areas where further research eded	is 62
CHAPTER	4	64
4 Methods		64
4.1 Stu	udy design, study populations and datasets	64
4.1.1	The UK Women's Cohort study (UKWCS)	65
4.1	.1.1 Baseline dataset	65
4.1	.1.2 Phase 2 'follow up' dataset	65
4.1	.1.3 Phase 2 nested case-control dataset	66
4.1	.1.4 Laboratory analysed dataset	67
4.1.2	The UK Dietary Cohort Consortium	67
4.1	.2.1 Oxford arm of EPIC	67
4.1	.2.2 Norfolk arm of EPIC	67
4.1	.2.3 The UKWCS	69
4.1	.2.4 The Whitehall II Study	69
4.1	.2.5 The National Survey of Health and Development	69
4.2 Etl	nical considerations	69
4.3 Me	asurement of cancer outcomes	69
4.3.1	Censor dates	70
4.4 Ex	clusions	70
4.5 As	sessment tools used for recording dietary and supplement intake in t	he
UKWCS		70
4.5.1	Baseline FFQ and Health and Lifestyle Questionnaire	70
4.5.2	Phase 2 Health and Lifestyle Questionnaire	72
4.5.3	Phase 2 Diary	74
4.5	5.3.1 Supplement intake captured from phase 2 diaries	74

4.5.3.2 Supplement ingredient database75
4.5.3.3 Cleaning vitamin C supplement data in the database of supplements taken by participants
4.5.3.4 Dietary intake captured from phase 2 diaries77
4.6 Assessment tools for dietary and supplement intake in other cohorts of the UK Dietary Cohort Consortium
4.6.1 Dietary intake recorded by diaries79
4.6.2 Supplement use recordings in EPIC-Oxford and EPIC-Norfolk
4.7 Statistical methods80
4.7.1 Time-to-event analyses 81
4.7.2 Conditional logistic regression analyses 82
4.7.3 Evaluating interactions83
4.7.4 Coding of menopausal status
4.7.5 Adjustments by potential confounders
4.7.5.1 Model building86
4.7.5.2 Assessment of potential confounders by Directed Acyclic Graphs87
4.7.5.3 Energy intake94
4.7.6 Sensitivity analyses
4.7.6.1 Excluding outliers e.g. for high dietary vitamin C intake
4.7.6.2 Adjustment for vitamin E96
4.7.6.3 Excluding or adjusting for women with a family history of cancer96
4.7.7 Samples sizes98
CHAPTER 5
5 Evaluation of diet and supplement vitamin C intake recordings in the UKWCS 100
5.1 Summary 100
5.2 Comparison of diary and questionnaire methods to assess vitamin C supplement use at phase 2
5.2.1 Introduction 102
5.2.2 Methods

	5.2.2	2.1	Subjects	103
	5.2.2	2.2	Dietary assessment methods	103
	5.2.2	2.3	Statistical analyses	103
5.2	2.3	Res	ults	104
	5.2.: supp	3.1 olem	Comparisons between vitamin C intake recorded by dia tents recorded by the phase 2 questionnaire	iry and
5.2	2.4	Disc	cussion	107
5.3 vege	Corr table	npari con:	isons relating to dietary vitamin C intake and servings of fr sumed at baseline	uit and 110
5.3	3.1	Intro	oduction	110
5.3	3.2	Meth	hods	111
	5.3.2	2.1	Subjects	111
	5.3.2	2.2	Dietary assessment methods	111
	5.3.2	2.3	Statistical analyses	111
5.3	3.3	Res	ults	116
	5.3.3	3.1	Comparisons between methods of recording fruit and vegetab	les.116
	5.3.3	3.2	Women with high discrepancies	121
5.3	3.4	Disc	cussion	122
5.4	Corr	npari	ison of vitamin C intake recorded by FFQ and by diary with bior	narkers
in a s	small	sam	ple of the UKWCS at phase 2	126
5.4	4.1	Intro	oduction	126
5.4	4.2	Meth	hods	127
	5.4.2	2.1	Subjects	127
	5.4.2	2.2	Dietary assessment methods	127
	5.4.2	2.3	Statistical analyses	127
5.4	4.3	Res	ults	128
5.4	4.4	Disc	cussion	132
5.5	Con	clusi	ion from evaluations	135

СН	IAPT	ER 6.		136	
6	6 Is vitamin C supplement use associated with a personal or family history of cancer				
and	and what characteristics predict vitamin C supplement use in UK women? A phase 2				
ana	alysis	s of the	e UKWCS	136	
	6.1	Sum	mary	136	
	6.2	Introd	duction	137	
	6.3	Meth	ods	138	
	6.4	Resu	ılts	139	
	6.5	Discu	ussion	147	
СН	APT	ER 7 .		151	
7	ls ge	neral	supplement use associated with breast cancer risk in the UKWCS?	151	
	7.1	Sum	mary	151	
	7.2	Introd	duction	152	
	7.3	Meth	od	153	
	7.	3.1 \$	Study population	153	
	7.	3.2 E	Exposure measurement	153	
	7.	3.3 8	Statistical analyses	154	
	7.4	Resu	ılts	155	
	7.5	Discu	ussion	163	
СН		ER 8.		166	
8	 Do м	uomen	who take supplements containing vitamin C in the LIK have an increa-	sed	
risk	c of b	reast	cancer? Phase 2 analyses of the UKWCS	166	
	8.1	Sum	mary	166	
	8.2	Introd	duction	167	
	8.3	Meth	ods	172	
	8	3.1 5	Subjects	172	
	8. 8	о. 20 г	Determining supplement use	172	
	0.	0.2 1	1. Vitamin Cusa from diany recordings	170	
		0.3.2	. I vitamin C use nom ulary recordings	172	
		8.3.2	.2 Multivitamin use from questionnaire recordings	174	
	8.	3.3 8	Statistical analyses	174	

8	.3.4	Excluding family history of breast cancer	175
8.4	Res	sults	176
8	.4.1	Breast cancer risk and use of supplements containing vitamin C	178
8	.4.2	Breast cancer risk and multivitamin use	183
8.5	Dis	scussion	185
СНАРТ	ER 9	9	191
9 Is bi intake? UKWC	reast Doe S	cancer incidence associated with dietary vitamin C or fruit and veg es supplement use modify these relationships? A baseline analysis	etable of the 191
9.1	Sur	mmary	191
9.2	Intr	roduction	192
9.3	Me	thod	194
9	.3.1	Study population	194
9	.3.2	Exposure measurement	194
	9.3	3.2.1 Fruit and vegetable intake	194
	9.3	8.2.2 Vitamin C intake	195
	9.3	8.2.3 Supplement use	195
9	.3.3	Statistical analyses	196
9.4	Re	sults	197
9.5	Dis	scussion	215
СНАРТ	ER 1	10	219
10 ls to diary da	otal v ata fr	vitamin C intake associated with breast cancer incidence? An analy	ysis of 219
10.1	Sur	mmary	219
10.2	Intr	roduction	220
10.3	Me	thods	221
1	0.3.1	Subjects	221
1	0.3.2	2 Case ascertainment and matching	221
1	0.3.3	B Dietary methods	223
1	0.3.4	Statistical analyses	224

10.3.4.1 Dietary and total vitamin C analyses
10.3.4.2 Supplement only vitamin C analyses
10.4 Results
10.4.1 Dietary vitamin C intake 225
10.4.2 Total vitamin C intake230
10.4.3 Supplement-only vitamin C intake
10.5 Discussion
CHAPTER 11
11 Summary discussion
11.1 Ideas for future research 244
11.1.1 UKWCS
11.1.2 Other
11.2 Public health issues 248
12 References
Appendix A
Summary of research findings relating to breast cancer risks from the 2007 WCRF report on food nutrition, physical activity and cancer prevention
Appendix B
Summary of research findings relating to supplement use from the 2007 WCRF report
on food nutrition, physical activity and cancer prevention
Appendix C
Search strategy for vitamin C and breast cancer
Appendix D
NICE guidelines used to identify women in the UKWCS who may be at raised risk or high risk of developing breast cancer
Appendix E
Poster and oral conference presentations by Jayne Hutchinson
Appendix F
Baseline combined Food Frequency Questionnaire and Health and Lifestyle Questionnaire

## List of tables

Table 1 Current recommended vitamin C intakes 11
Table 2 Studies identified by a search for articles published after the 2007 WCRF
report and 2008 WCRF continuous update
Table 3 The WCRF reported studies on vitamin C intake and breast cancer risk 50
Table 4 Factors controlled for in the vitamin C studies published since the 2007 and
2008 WCRF reports
Table 5 Factors controlled for in the studies included in the 2007 and 2008 WCRF      reports    53
Table 6 Characteristics of the five cohorts in the UK Dietary Cohort Consortium
included in analyses of vitamin C and breast cancer risk
Table 7 Percentage of supplements containing vitamin C in the UKWCS supplement
ingredient database that are multivitamins, antioxidants or single vitamin C
supplements, categorized by vitamin C content
Table 8 Mean dose of vitamin C by supplement type in the UKWCS supplement      databases      77
Table 9 Women who recorded daily use of vitamin C on the phase 2 questionnaire split
by diary vitamin C intake category
Table 10 Percentage of women in each vitamin C intake category who recorded daily
intake of vitamin C, antioxidants or multivitamins on the phase 2 questionnaire 105
Table 11 Kappa agreements between frequency of vitamin C intake captured via diaryand that captured via the phase 2 questionnaire
Table 12 Cross-tabulation of daily vitamin C supplement use recorded by questionnairewith daily intake of supplement containing vitamin C recorded by diaries106
Table 13 Cross-tabulation of daily vitamin C or antioxidant or multivitamin supplementuse recorded by questionnaire with intake of supplement containing vitamin C on 3 or 4diary days106
Table 14 Cross-tabulation of daily vitamin C or antioxidant supplement use recorded by
questionnaire and taking supplements containing vitamin C above 60mg for 3 or 4 diary days
Table 15 The main dietary sources of vitamin C at baseline in the UKWCS for all
women and for vegetarians separately116

Table 16 Spearman's correlations between fruit as servings and as grams, andbetween derived vitamin C intake119
Table 17 Spearman's correlations between vegetables as servings and as grams, andbetween derived vitamin C intake119
Table 18 Tabulation of fruit servings from cross-check answers and those derived fromthe FFQ excluding women with missing data120
Table 19 Kappa agreements between groupings of fruit consumed as servings, intakein grams and derived vitamin C intake120
Table 20 Tabulation of vegetable servings from cross-check answers and thosederived from the FFQ excluding women with missing data
Table 21 Kappa agreements between grouping of vegetable consumed as servings,intake in grams and derived vitamin C intake
Table 22 Characteristics of women with large discrepancies between methods ofmeasuring fruit and vegetable portions122
Table 23 Characteristics of women by dietary vitamin C intake recorded by FFQ 129
Table 24 Characteristics of women by dietary vitamin C intake recorded by diary 129
Table 25 Spearman's correlation coefficients between different measures of vitamin Cfor all UKWCS women who provided blood samples
Table 26 Spearman's correlation coefficients between different measures of vitamin C      for non-vitamin C supplement users in the sub-sample of UKWCS women who      provided blood    131
Table 27 Percentage increase in plasma concentrations of ascorbic acid and totalvitamin associated with a doubling of vitamin C from all foods recorded by diary and byFFQ132
Table 28 Characteristics associated with taking supplements containing any dose ofvitamin C and taking supplements containing high doses of vitamin C (1000mg orabove)141
Table 29 Odds ratios of taking supplements containing vitamin C: any dose; or 1000mg or more for UKWCS women who self-reported a personal or a family history of cancer
Table 30 Odds ratios of taking supplements containing vitamin C for a range of dosesfor UKWCS women who self-reported a personal history of cancer or a family history ofbreast cancer146

Table 31 Characteristics of supplement users and non-users at baseline in the UKWCS
Table 32 Breast cancer risks according to any supplement use at baseline in theUKWCS157
Table 33 Baseline characteristics of women in the LIKWCS according to supplement
use at baseline and phase 2: never users; inconsistent users; and consistent users 158
Table 34 Breast cancer risks according to any supplement use for never users,
inconsistent users and consistent users according to use at baseline and phase 2 of
the UKWCS
Table 35 Assessment of modifying effects of supplement use at baseline on breast
cancer risk using interaction terms for joint effects of supplement use and other
characteristics
Table 36 Studies analysing breast cancer risk by vitamin C supplement dose 168
Table 37 Methods and factors controlled for in studies of breast cancer risk and vitaminC supplement use169
Table 38 Characteristics of women at phase 2 in the UKWCS by vitamin C supplement
intake group177
intake group    177      Table 39 Breast cancer risks of women supplementing with vitamin C recorded by      diaries at phase 2 in the UKWCS      179
intake group177Table 39 Breast cancer risks of women supplementing with vitamin C recorded by diaries at phase 2 in the UKWCS179Table 40 Breast cancer risks of women supplementing with vitamin C recorded by diaries in the UKWCS, excluding women with a family history of breast cancer
intake group177Table 39 Breast cancer risks of women supplementing with vitamin C recorded by diaries at phase 2 in the UKWCS179Table 40 Breast cancer risks of women supplementing with vitamin C recorded by diaries in the UKWCS, excluding women with a family history of breast cancer180Table 41 Breast cancer risks of women supplementing with vitamin C recorded by diaries in the UKWCS, excluding women supplementing with vitamin C recorded by 180181
intake group177Table 39 Breast cancer risks of women supplementing with vitamin C recorded by diaries at phase 2 in the UKWCS179Table 40 Breast cancer risks of women supplementing with vitamin C recorded by diaries in the UKWCS, excluding women with a family history of breast cancer
177Table 39 Breast cancer risks of women supplementing with vitamin C recorded by diaries at phase 2 in the UKWCS
intake group
intake group177Table 39 Breast cancer risks of women supplementing with vitamin C recorded by diaries at phase 2 in the UKWCS.179Table 40 Breast cancer risks of women supplementing with vitamin C recorded by diaries in the UKWCS, excluding women with a family history of breast cancer
intake group177Table 39 Breast cancer risks of women supplementing with vitamin C recorded by diaries at phase 2 in the UKWCS179Table 40 Breast cancer risks of women supplementing with vitamin C recorded by diaries in the UKWCS, excluding women with a family history of breast cancer

Table 47 Characteristics of women at baseline in the UKWCS by category of dietary      vitamin C consumption
Table 48 Breast cancer risks by fruit and vegetable intake and dietary vitamin C intakeat baseline in the UKWCS201
Table 49 Assessment of modification by menopausal status, supplement use andfamily history of breast cancer on fruit and vegetable and dietary vitamin C intake andassociations with breast cancer risk in the UKWCS205
Table 50 Breast cancer risks by fruit and vegetable intake and dietary vitamin C intakeat baseline in the UKWCS by menopausal status206
Table 51 Breast cancer risks by fruit and vegetable intake and dietary vitamin C intakeat baseline in the UKWCS by supplement use207
Table 52 Breast cancer risks by fruit and vegetable intake and dietary vitamin C intakeat baseline in the UKWCS by supplement use in post-menopausal women
Table 53 Breast cancer risks by fruit and vegetable intake and dietary vitamin C intakeat baseline in the UKWCS by supplement use in pre-menopausal women
Table 54 Characteristics of the five cohorts from the UK Dietary Cohort Consortiumincluded in the analyses of vitamin C and breast cancer risk222
Table 55 Characteristics of women by fifth of dietary vitamin C intake derived from fooddiaries in the UK Dietary Cohort Consortium
Table 56 Breast cancer risks by dietary vitamin C intake recorded by diaries in the UKDietary Cohort Consortium
Table 57 Breast cancer risks by dietary vitamin C density intake recorded by diaries inthe UK Dietary Cohort Consortium
Table 58 Characteristics by case/control status & total vitamin C intake of women inEPIC-Oxford, EPIC-Norfolk and UKWCS nested case-control cohorts
Table 59 Breast cancer risks by total vitamin C intake recorded by diaries in EPIC- Oxford, EPIC-Norfolk and UKWCS cohorts of the UK Dietary Cohort Consortium 232
Table 60 Breast cancer risks according to vitamin C content of supplements recorded      by diaries in EPIC-Oxford, EPIC-Norfolk UK and UKWCS cohorts of the UK Dietary      Cohort Consortium
Table 61 The percentage of women in each supplement category from the three cohorts of the UK Dietary Cohort Consortium in the vitamin C supplement analyses 235

## List of figures

Figure 1 Flow chart of datasets, assessment tools & intake data available for use in risk analyses
Figure 2 Percentage of women reporting daily supplement use in the UKWCS phase 2 questionnaires by supplement type
Figure 3 Estimated European age-standardised Incidence and mortality rates in 2008 by EU country
Figure 4 European age-standardised breast cancer incidence and mortality rates for Britain between 1975-2008
Figure 5 Age-Specific breast cancer incidence rates for Britain, 1975-2008
Figure 6 Flow diagram of studies identified in current literature search
Figure 7 A section from the baseline FFQ relating to intake of seasonal fruit
Figure 8 Questions relating to supplements on the phase 2 questionnaire
Figure 9 Phase 2 diary supplement recording example provided to participants74
Figure 10 Phase 2 diary food and drink recording example provided to participants 78
Figure 11 Questions used to determine menopausal status (phase 2 questionnaire)84
Figure 12 Flow diagram of coding for menopausal status in the UKWCS
Figure 13 A Directed Acyclic Graph (DAG) of factors associated with dietary vitamin C intake and breast cancer
Figure 14 A Directed Acyclic Graph (DAG) of factors associated with supplement vitamin C intake and breast cancer
Figure 15 DAG linking eating patterns in childhood with the main exposure and outcome
Figure 16 Questions asked at baseline relating to family history of cancer
Figure 17 Questions asked at phase 2 relating to family history of cancer
Figure 18 Box whisper plots and histograms showing distributions of FFQ derived dietary vitamin C (mg/d), total fruit and vegetable intake (g/d), and total servings per day of fruit and vegetables
Figure 19 Histograms showing distributions of total fruit and vegetables servings per week recorded by cross-check question, and separate distributions for fruit and for vegetables

Figure 20 Bland-Altman plots of ratios between FFQ and cross-check question recording total fruit and vegetable servings for all women and for vegetarians only 117
Figure 21 Separate fruit and vegetable Bland-Altman plots of ratios between servings measured by FFQ and by cross-check question
Figure 22 Breast cancer time-to-event curve for supplement users split by socio- economic status
Figure 23 Adjusted restricted cubic spline models of breast cancer risks by dietary vitamin C intake before sub-grouping
Figure 24 Adjusted restricted cubic spline models of breast cancer risks by vitamin C intake for pre-menopausal and post-menopausal women
Figure 25 Adjusted restricted cubic spline models of breast cancer risks by vitamin C intake for non-supplement users
Figure 26 Adjusted restricted cubic spline models of breast cancer risks by vitamin C intake for supplement users
Figure 27 Adjusted restricted cubic spline models of breast cancer risks by nutrient density for non-supplement users and supplement users

## Abbreviations

AICR	American Institute for Cancer Research
BRAC1	Breast Cancer type susceptibility protein 1
BMI	Body Mass Index
BMJ	British Medical Journal
CNC	Centre for Nutritional Epidemiology in Cancer Prevention and Survival
DAG	Directed Acyclic Graph
DANTE	Diet and Nutrition Tool for Evaluation
DHA	Dehydroascorbic Acid
DNA	Deoxyribonucleic acid
DIDO	Diets in Nutrients Out
DINER	Diet into Nutrients for Epidemiological Research
EFSA	European Food Standards Authority
EPIC	European Prospective Investigation into Cancer
ER	Oestrogen Receptor
EU RDA	European Recommended Daily allowance
FFQ	Food Frequency Questionnaire
HR	Hazard Ratio
HRT	Hormone Replacement Therapy
IGF	Insulin-like Growth Factor
IQR	Inter-quartile range
NICE	National Institute of Clinical Excellence
NSHD	National survey of Health and Development
OR	Odds Ratio
PR	Progesterone Receptor
RCT	Randomised Control Trial
RDA	Recommended Daily Allowance
RNI	Recommended Nutrient Intake
ROS	Reactive Oxygen Species
RR	Risk Ratio
SHBG	Sex-Hormone-Binding Globulin
UKWCS	UK Women's Cohort Study
WACS	Women's Antioxidant Cardiovascular Study
WCRF	World Cancer Research Fund
WHI	Women's Health Initiative Study
WHO	World Health Organisation

#### **CHAPTER 1**

#### **1** Introduction and Objectives

Breast cancer is the most common cancer in women in the UK, and in many other western countries.<sup>3 4</sup> The cumulative risk of being diagnosed with breast cancer in the UK was 6% by the age of 65, and 11% over a lifetime, based on cancers diagnosed between 1994 and 1997.<sup>5</sup> The latest estimate of life-time risk increased to 12.5% (.i.e. 1 in 8) based on breast cancers diagnosed in UK women in 2008.<sup>6</sup>

Whilst age, genetics, hormonal and reproductive factors are established breast cancer risk factors, there is evidence that lifestyle factors such as diet, weight and exercise can also contribute to risk (as discussed in chapter 3).7 Initial research using retrospective recall methods indicated that fruit and vegetable intake probably reduced the risk of many cancers, including breast cancer.<sup>8</sup> Subsequently the 5-A-day fruit and vegetable initiative, based on the World Health Organisation (WHO) recommendations,<sup>9</sup> was introduced in the UK in 2001 to encourage healthy lifestyles with a view to reducing cancer risk in general, as well as the risk of other chronic diseases. Since then, however, an increasing number of null associations between breast cancer risk and prospective recordings of fruit and vegetable intake have been published; consequentially overall there is no convincing evidence of a protective effect.7 10-13 Similarly, studies of antioxidant intake have often produced conflicting results between observation methods.<sup>7</sup> Furthermore, a recent Randomised Control Trial (RCT) using several types of antioxidant supplements found no significant associations with breast cancer incidence.<sup>14</sup> Although media reporting of conflicting studies may have caused many people to ignore dietary advice relating to cancer risks.<sup>15</sup> many health conscious women remain interested in such information. For instance, a study of UK women who were sufficiently health conscious to attend mammography screening for breast cancer, reported that the majority were interested in obtaining dietary and exercise advice at these clinics; many of these women were interested in advice about how to reduce risk of serious diseases.<sup>16</sup>

In addition to maintaining healthy lifestyle behaviours, many health conscious women tend to take supplements, moreover levels of use in general continue to rise in the UK (section 2.1) and Europe, as well as in the USA. Even though the majority of women are likely to use supplements for reasons other than specifically reducing their cancer risk, research is needed to determine whether some supplement types may increase the risk of cancers. The 2007 World Cancer Research Fund (WCRF) report has systematically reviewed research on cancer risk and diet and supplement use.<sup>7</sup> Some

of these and more recent studies are discussed in section 3.6, particularly in relation to vitamin C which is a common ingredient in supplements and a common micronutrient in fruit and vegetables. The majority of research on supplements has been undertaken in the USA; however it is important to evaluate the effects of supplementation on different populations, with different diets and levels of food fortification. No published research has assessed the effect of supplementation on cancer risk in the UK population.

The aim of this current research is to explore associations between the risk of developing breast cancer in UK women and intake of vitamin C from diet and supplements, and dietary supplements in general. Associations with any supplement use (chapter 7) and with vitamin C supplementation (chapter 8) one of the most popular supplements, are assessed in relation to breast cancer risk. The analysis of vitamin C intake from all supplement types categorised in relation to recommended daily allowance and high doses of vitamin C (see section 2.2.1 for definition) leads onto a brief analysis of risks relating to the use of multivitamins (chapter 8). Total vitamin C intake, which includes both supplement and dietary intake, is assessed (chapter 10) as is dietary vitamin C intake sub-analysed by users and non-users of any supplements (chapter 9). The analysis of non-users provides a clearer picture of the effects of dietary vitamin C intake on breast cancer risk, without the influence of the variety of different supplement types taken by many of the women. Breast cancer risks in relation to fruit and vegetable intake, the main source of vitamin C, are also explored in chapter 9.

Since UK women generally obtain sufficient nutrients from their diet, it is hypothesised that women who maximise their intake of vitamin C by taking high doses of this watersoluble supplement on a daily basis may have a higher risk of breast cancer due to pro-oxidation mechanisms or to the reduction of beneficial apoptosis. These mechanisms are discussed in more detail in section 2.2.4. Conversely, women who have a low vitamin C intake may have an increased risk of breast cancer particularly if they have high levels of damaging free radicals and generally have a low antioxidant status.<sup>17</sup> Consequently, the analysis of a population with a wide intake range may produce a U-shaped risk curve. It is also possible that vitamin C or other micronutrients could increase breast density which has been linked to breast cancer risk.

The analyses use pre-gathered prospective cohort data incorporating a wide range of intakes enabling important vitamin C dose-response relations to be examined, which would not be possible using RCTs where only one or two doses can be evaluated at a time. The thesis is original since no research on any of the above associations has been conducted on the UK population. Furthermore, the total vitamin C analysis utilises both supplement and dietary vitamin C intake recorded by diary as opposed to

questionnaire methods used in previous research in other countries. Evaluation of vitamin C intake by diary method in comparison to the questionnaire method is covered in chapter 5. Fruit and vegetable intake is also evaluated here.

Chapter 6 explores cross-sectional associations between taking supplements containing vitamin C and a personal or family history of cancer, primarily to identify whether these women are more likely to frequently take high dose vitamin C. Women who reported a family history of breast cancer, who may have had different health behaviours and cancer risks compared to other women, were identified and excluded in some sensitivity analyses of risks. Unfortunately numbers were too low to power risk sub-analyses by family history or by those estimated to be genetically at raised risk or high risk of breast cancer (defined in section 4.7.6.3 and 0). Other potential confounding characteristics of women are discussed in sections 3 4.7.5 and 4.7.5.2.

All the analyses use data from the UK Women's Cohort Study (UKWCS), either from the baseline FFQs (35,000 women), phase 2 diaries (12,500 women) or phase 2 questionnaires. The relationship between the available data and the thesis objectives are shown in Figure 1 of this introduction. Further details of the cohorts and tools used to capture the intake data can be found in sections 4.1 to 4.6 of the methods chapter. Unfortunately total vitamin C intake (diet plus supplements) could not be analysed for the full UKWCS cohort members at either phase; at baseline details of supplement types used were not electronically captured, and at phase 2 dietary intake recorded from diaries were only electronically captured for a small nested case-control analysis. However, total vitamin C was analysed by pooling nested case-control data from the UKWCS and two other UK cohorts: EPIC-Norfolk and EPIC-Oxford which are part of the European Prospective Investigation into Cancer (EPIC) and are also discussed in the main methods chapter, chapter 4.

#### 1.1 Objectives

- 1. To evaluate vitamin C intake in UKWCS (Chapter 5):
  - To compare the frequency of vitamin C supplement use in the UKWCS recorded by diaries with that recorded by the follow-up phase 2 questionnaire.
  - b. To compare the baseline dietary vitamin C intake with fruit and vegetable intake in the UKWCS.
  - c. To compare ascorbic acid blood plasma levels in a sub-sample of the UKWCS with derived vitamin C intake from diary recordings and also from Food Frequency Questionnaires (FFQs).
- To determine what characteristics of women predict vitamin C supplement use at phase 2 in the UKWCS. To determine whether women who have a history of breast cancer are more likely to use high dose vitamin C supplements (Chapter 6).
- To determine whether there is an association between any supplement use and breast cancer risk at baseline in the UKWCS and whether the associations are modified by sub-factors. To determine whether the risk is different for long-term users: women who have taken supplements at both baseline and phase 2 of the UKWCS (Chapter 7).
- 4. To determine whether there is an association between vitamin C intake from supplements and the risk of breast cancer at phase 2 in the UKWCS, categorised in relation to the European Recommended Daily Allowance (60mg/d) and high dose (500mg/d) use (Chapter 8)
- To determine whether there is an association between dietary vitamin C intake and the risk of breast cancer at baseline in the UKWCS, and whether it is modified by supplement use of any type (Chapter 9).
- To determine whether there is an association between total vitamin C intake (supplements plus dietary) and the risk of breast cancer in the UK Dietary Cohort Consortium using pooled nested case-control data from UKWCS, EPIC-Norfolk and EPIC-Oxford (Chapter 10).



Figure 1 Flow chart of datasets, assessment tools & intake data available for use in risk analyses

The UKWCS and two of the other cohorts in the UK Dietary Cohort Consortium are some of the largest population-based prospective studies in the UK which were designed for examining associations between diet and chronic diseases, and are ideal studies to investigate the objectives above. Their large size increases the power of the analyses. Another major strength of these studies is their prospective nature which minimises reverse causality and recall bias (responder bias) which can affect results of retrospective case-control studies. Additionally the studies have captured many important health and lifestyle factors which may be potential confounders and can be adjusted for in the analyses. Furthermore, the majority of the analyses in this thesis use previously unexploited supplement data.

The UKWCS, used in all the analyses, has an extensive database of supplements used and covers a wider range of vitamin C intakes than most studies which have reported on similar objectives. Women from baseline and phase 2 in the UKWCS were followedup for cancer incidences for over 11 and 7 years respectively, which is comparable to or better than most studies reviewed in section 3.7.

The UKWCS, instigated through the World Cancer Research Fund, in general comprised of health conscious women. The results are therefore directly applicable to similar women; who would be interested to know how lifestyle choices could affect their risk of breast cancer, and who may be prepared to alter their behaviours. In addition, the UK Dietary Cohort Consortium of other UK studies provides results that are more generalised to the UK population as a whole, as well as providing an opportunity to examine total vitamin C intake in relation to breast cancer risk using diary data.

First, supplement use and vitamin C in particular will be discussed in chapter 2, and then a review of literature about life-style factors affecting breast cancer risk, including diet, will be described in chapter 3, before systematically reviewing articles which examine the influence of vitamin C intake on breast cancer risk. Chapter 4 then describes the methods used in this thesis, which is followed by the results chapters (5-10) and the summary discussion in chapter 11.

## **CHAPTER 2**

### 2 Supplement use and vitamin C

Supplements are proactively marketed in the developed world in a multi-billion pound industry, retailed through shops, mail order and the internet. One of the largest producers and consumers of supplements is the United States (US). Annual sales in the US since 2004 are estimated to have been over \$20 billion per year, about 1% of the US health expenditure, providing employment for about 200,000 people directly and over a quarter of a million indirectly.<sup>18</sup> The number of different types of supplements marketed in the US may be between 29,000-50,000.<sup>18</sup>

Both in the US and Europe supplements are classified as food rather than drugs. In the US the Food and Drug Administration (FDA) regulates the supplement industry, with powers to take action relating to quality, unsafe levels, and false claims or misleading labelling. Although US manufacturers are allowed to make general health claims, any claims relating to treatment, prevention or cure for a specific disease or condition would be considered illegal.<sup>19</sup> Regulation has been similar in Europe under the Food Supplements Directive 2002.<sup>20</sup> However, the European Food Safety Authority (EFSA) has implemented regulations for approving or rejecting general health claims made on foods and supplements based on whether these claims are independently verified by research studies.<sup>21</sup> Additionally, safe or tolerable upper limits have been set by the EFSA for many supplements,<sup>22</sup> but many high dose products do not state warnings on their packages as required.

#### 2.1 Supplement use in the UK

Sales of supplements in the UK in 2009 were estimated to be £396M.<sup>23</sup> Vitamin supplement use reported by UK women increased from 17% in 1986/1987 to 41% reported in 2008/9.<sup>24 25</sup> A 2008 UK survey reported users were most likely to be women above 55 years and of higher socioeconomic status.<sup>26</sup> An analysis of the UK Women's Cohort Study (UKWCS) found that users were significantly more likely to lead healthier lifestyles; to be more physically active; have a lower alcohol intake; a lower body mass index (BMI) and eat diets which met recommended dietary intakes.<sup>27</sup> Therefore they were less likely to need supplements than non-users.<sup>27</sup> Further support for this 'inverse supplement hypothesis' has been found in the UK,<sup>25 28 29</sup> and elsewhere.<sup>30-35</sup> Moreover, those classifying themselves as high strength supplement users in a recent UK survey were particularly health conscious.<sup>26</sup>

An exploration of reasons behind supplement use in a UKWCS sub-sample of 303 woman, incorporating the theory of planned behaviour and other potential predictors,<sup>36</sup> found supplement-taking was predicted by an individual's value of health and their perceived ability of supplements to reduce susceptibility to illness.<sup>37</sup> Supplement use was associated with a perception that supplements were particularly able to reduce development of colds, flu, arthritis and rheumatism, and also to have a protective effect on the development of heart disease, anaemia and menstrual problems.

General health and well-being was the most common reason (57%) given for taking supplements in the 2008 UK Random Location Omnibus Survey.<sup>26</sup> Ten percent of the 801 sample of both men and women were taking them to ward off colds, 11% of full time workers were taking them because they believed they had an unbalanced diet, 26% of over 45 year olds were taking them for a specific health reason, and 17% of women were taking them due to advice from a health practictioner.<sup>26</sup> High strength supplement users in particular considered the health implications of what they ate, actively looked for information about how to stay healthy and believed that they needed vitamin and mineral supplements to feel and stay healthy.<sup>26</sup> Additionally they were more likely to take other supplements than those not taking high strength supplements (52% vs. 43%). Thirty eight percent of those surveyed, an estimated 15% of the population, reported taking high strength supplements.<sup>26</sup> Sixteen percent of users bought supplements because of their high strength content; however only 68% of these users were fully aware that high doses may cause adverse side effects.<sup>26</sup> Only 6% of high strength users were likely to report they were in excellent health compared to 25% of non-users.<sup>26</sup>

The most commonly taken supplements in the 2008 UK survey were multivitamins (36%), cod liver oil (35%) and vitamin C (24%).<sup>26</sup> In the EPIC-Norfolk study only 4.7% of women took single supplement vitamin C, the most popular supplements being cod liver oil (24.5%) and multivitamins, with or without minerals (10.4%).<sup>38</sup> The percentages of women reporting daily use of supplements by questionnaire in the UK Women's Cohort Study (UKWCS), the main study in this thesis, are detailed by type in Figure 2. This shows vitamin C, the focus of this thesis, is one of the most commonly taken nutrients in supplements both as a single supplement and in combination with other nutrients.

**Figure 2** Percentage of women reporting daily supplement use in the UKWCS phase 2 questionnaires by supplement type





#### 2.2 Vitamin C

Vitamin C (C<sub>6</sub>H<sub>8</sub>O<sub>6</sub>) exists as ascorbic acid (the reduced state) and dehydroascorbic acid (DHA, the oxidised state) in both food and the human body. Humans are one of only a few species that cannot synthesis vitamin C, and therefore obtain this micronutrient from fruit, vegetable and other plant food. It is widely present in plant tissues, for instance in peppers, oranges, lemons, strawberries, grapes, brussel spouts, broccoli, cauliflower and tomatoes.<sup>39</sup> The storage, processing, and cooking of foods can reduce vitamin C content substantially therefore fresh or steamed produce is recommended. Although natural and synthetic forms are chemically identical and have similar bioavailability, the added benefits of other bioactive components from food are lost if vitamin C is obtained predominately from supplements.

Vitamin C has a number of functions in the body, and supplementation is often necessary in malnourished populations such as the chronically sick and the elderly, as well as in developing countries. Low doses of vitamin C, 10mg/day, are needed to prevent scurvy,<sup>40</sup> and doses of 20mg/day are necessary for optimum wound healing.<sup>41</sup> Symptoms of scurvy, which is now uncommon in developed countries, include bleeding gums, easy bruising, impaired wound and fracture healing, joint pains, fatigue and depression.<sup>39</sup> Vitamin C acts as a cofactor for enzymes in the synthesis of collagen, carnitine, tyrosine and the neurotransmitters by maintaining metal ions within the enzymes in a reduced state.<sup>42</sup> Collagen is the main protein found in body structures such as skin, bones, teeth, cartilage, tendons, blood vessels and eye lens and cornea. Carnitine is involved in the transport of fatty acids into mitochondria for energy production. There is evidence that vitamin C suppresses the oxidation of low density lipoproteins (LDL) which are implicated in atherosclerosis.<sup>43</sup> Through electron donation ascorbic acid acts as a reducing agent i.e. an antioxidant, by scavenging oxygen and nitrogen free radicals and also regenerates the fat-soluble vitamin E antioxidant (αtocopherol). The oxidised form of vitamin C, DHA, is recycled back to vitamin C by several enzyme systems within cells. Additionally vitamin C supplementation is found to increase the absorption of non-haem iron from plants, probably by reducing iron to its ferrous state (Fe<sup>2+</sup>).44

Pharmacological uses of vitamin C, though not necessarily evidence based, include protection against cardiovascular disease, reducing histamine, boosting immune function, aiding iron absorption, protecting against cataracts, as well as use with the aim of reducing symptoms of the common cold and prolonging survival in cancer patients.<sup>40</sup> Ascorbic acid (E300) and its esters are also added to food as browning inhibitors, reducing agents, flavour stabilisers, dough modifiers and colour stabilisers.<sup>42</sup>

Suggestions that vitamin C is able to reduce the incidence of colds have been unsubstantiated in randomised controlled trials.<sup>45</sup> Additionally, food and supplement manufacturers often make claims for vitamin C in relation to protection of cells from premature ageing, antioxidant activity, antioxidant content and antioxidant properties, protection of DNA, proteins and lipids from oxidative damaging. However, the EFSA has recently reported that no evidence has been provided to them to establish that having antioxidant activity/content and/or antioxidant properties exerts beneficial physiological effects on humans.<sup>46</sup> In future manufacturers should not make these claims.

#### 2.2.1 Vitamin C recommended intake

Table 1 shows current recommended vitamin C intakes, primarily to prevent scurvy and for optimum wound healing, and recommended safe upper levels set to minimise gastrointestinal disturbances.<sup>47</sup> Most supplements sold in the UK quote doses as a percentage of the EU Recommended Daily Allowance of 60 mg/day for adults. A meta-analysis indicated that daily intake of 60mg vitamin C would produce average plasma levels of ascorbic acid of 42.4 µmol/L.<sup>48</sup> A higher recommended daily allowance of 100 mg/d has been recommended to maintain average plasma concentrations of ascorbic acid at 50 µmol/L,<sup>48 49</sup> which has been suggested as optimum.<sup>50</sup>

	Recommended daily intake	Recommended safe upper levels
UK	RNI 40 mg/day <sup>51</sup>	None set: insufficient data <sup>52</sup>
Europe	RDA 60 mg/day <sup>53</sup>	None set: insufficient data <sup>22</sup>
US	RDA 75 mg/day for women	2000 mg/day <sup>47</sup>

Table 1 Current	recommended	vitamin	С	intakes
-----------------	-------------	---------	---	---------

RNI = Reference Nutrient Intake

RDA = Recommended Daily Allowance

RDA = EAR + 2 SD<sub>EAR</sub> (EAR: Estimated average requirement; SD<sub>EAR</sub>: Standard deviation of the EAR)

The most clearly defined side effect of taking large doses of vitamin C is abdominal pain and osmotic diarrhoea due to unabsorbed vitamin C being metabolised by intestinal bacteria.<sup>54</sup> Although vitamin C does not accumulate to toxic levels like fat soluble vitamins, high intake may increase the risk of renal stones in some at risk individuals due to the increased excretion of the vitamin C metabolite oxalate. A recent review of clinical trial evidence concluded that doses of 2000mg/day are safe for most adults,<sup>55</sup> other than those genetically at risk.<sup>56</sup> Individuals with genetic iron disorders are at risk of iron overload and development of haemochromatosis or thalassaemia if

they consume high doses of vitamin C since it aids the absorption of dietary non-haem iron.<sup>56</sup> Rebound scurvy has been reported anecdotally, and was thought to occur after abrupt cessation of long-term high intakes of vitamin C due to established accelerated metabolism or disposal of ascorbic acid, but the results appear to have been misinterpreted,<sup>41</sup> and systematic conditioning appears to be minimal.<sup>57</sup> Evidence shows that plasma ascorbic acid concentrations do fall in well-nourished individuals in such circumstances, but to within normal levels.<sup>57</sup>

In its 2004 review of adverse effects of vitamin C, the EFSA suggested an upper limit of 1000mg/day,<sup>58</sup> however both the EFSA and the UK Expert Group on Vitamins and Minerals currently believe there is insufficient data to set an upper limit.<sup>22 52</sup> The EFSA acknowledged there was conflicting and insufficient evidence to set limits relating to breast cancer and vitamin C intake and welcomed future research on its safety.<sup>22 58</sup>

#### 2.2.2 Vitamin C bioavailability, absorption, transport and excretion

The bioavailability of vitamin C is the effectiveness with which it is released from the source into the tissues of the body. Natural and synthetic vitamin C are chemically identical and the bioavailability of vitamin C from food sources and from supplements have been reported to be similar.<sup>59</sup> 'Slow release' or 'time release' vitamin C supplements and those which also contain bioflavonoids are promoted as having increased bioavailability, though evidence for this is lacking.

At low doses ascorbic acid is actively transported by sodium ions at the brush border membrane in the small intestine<sup>60</sup>. DHA, the oxidised form, diffuses into cells facilitated by glucose transporters, and is immediately reduced to ascorbic acid, creating a concentration gradient which drives the movement of the former.<sup>60</sup> At low doses of about 20mg/day absorption of vitamin C is about 98%;<sup>61</sup> at normal intakes of between 30-180mg/day absorption is about 70-95%.<sup>54,62</sup> Absorption falls to 50% at doses of 1.5g/day, 25% at 6g and 16% at 12g/day.<sup>63</sup> In plasma, vitamin C is transported mainly as water-soluble ascorbic acid, normal levels being about 30-70 µmol/L.

Vitamin C is stored in the body, although not to the same extent as fat soluble vitamins which have stable reserves and associated toxicities. Ascorbic acid is stored in white blood cells and is also distributed to all tissues of the body. Body stores can be more than 100 times greater than plasma content, the greatest concentrations being found in the adrenal and pituitary glands,<sup>64</sup> and the largest amount in the liver.<sup>60</sup> The ascorbic acid content of a range of leukocytes types generally reflect body stores, whereas plasma levels reflect recent vitamin C intake which fall more rapidly in depletion

studies. The maximum body pool of vitamin C has been estimated to be about 1500 mg.<sup>62</sup>

The bioavailability and therefore plasma levels of vitamin C as water-soluble ascorbic acid are determined by the rate of intestinal absorption and of excretion.<sup>60</sup> The relationship between vitamin C intake and plasma ascorbic acid is S-shaped; between doses of 30-100 mg/day plasma levels and increases linearly with increasing intake in healthy women, with plasma and circulating cell saturation occurring between 200-400 mg/day.<sup>65</sup> Excretion from the body in urine starts at intakes of about 100mg/day before tissue saturation is reached,<sup>61,54</sup> and continues to increase steadily if dosage is increased. However, oral doses of 500 mg and 1250 mg are not entirely excreted in urine by women;<sup>65</sup> most likely due to a decline in intestinal absorption at high doses.<sup>63</sup> Therefore, it appears that intakes as high as 200-400 mg/day are utilised in the body by healthy women, but whether these levels have optimum health effects is unknown. Urine excretion as well as absorption limitations means intakes above these levels may not be utilised and may be unnecessary.

Due to the body's limited ability to store vitamin C, measurements of plasma concentrations are thought to represent intake only in the preceding few weeks.<sup>66</sup> Another measurement problem is that it is unstable in blood and rapidly deteriorates if not stabilised by chemicals such as metaphosphoric acid and stored at low temperatures.<sup>67</sup>

# 2.2.3 Characteristics associated with vitamin C plasma concentrations

Low ascorbic plasma concentrations have been associated with older age, smoking,<sup>48</sup> high blood pressure and in women with high cholesterol.<sup>67</sup> Cross-sectional analyses of the EPIC-Norfolk study have reported inverse associations between plasma vitamin C and BMI, waist-to-hip ratio (WHR), diabetes, prevalent undiagnosed hyperglycemia and HbA1c plasma levels (a measure of blood glucose levels).<sup>68</sup> <sup>69</sup> Low plasma concentrations may reflect increased antioxidant requirements in those with higher oxidative stress, associated with abdominal obesity.<sup>68</sup> Furthermore, the increased blood glucose levels in those with hyperglycaemia and diabetes may inhibit the uptake of DHA by glucose transporters;<sup>70</sup> DHA may metabolize and be excreted if not taken up by cells. In addition, plasma vitamin C has been inversely associated with the inflammatory marker C-reactive protein (CRP),<sup>71</sup> though an ability to reduce inflammation is inconclusive.<sup>72</sup> In addition supplement use in general is also positively associated with ascorbic acid levels.<sup>67</sup> Lifestyle factors may explain the link between plasma vitamin C levels and adiposity, HbA1c and CRP plasma levels.<sup>68 69</sup> Indeed, high

levels of plasma vitamin C may be a marker for healthy diet, lifestyle and socioeconomic position across the life course which appear protective of a range of conditions; factors such as exercise, low fat and high fibre diets, housing conditions and education have been associated with increasing vitamin C plasma concentrations.<sup>73</sup>

Furthermore, in the UK low ascorbic acid plasma concentrations have been associated with increased mortality from all-causes, from cardiovascular disease (CVD), ischaemic heart disease in men and women, and cancer mortality in men but not women.<sup>67</sup> The latter difference between sexes may possibly be due to the lower baseline antioxidant status of men. The associations remained after excluding diabetics and cigarette smokers, and after excluding supplement users (except for ischemic heart disease in women). Unfortunately, it was not possible to adjust for socio-economic status (SES) or physical activity which were likely to confound the results.<sup>67</sup> Associations between cancer risk and ascorbic acid plasma concentrations are given in section 3.6.4.1.

# 2.2.4 Potential mechanisms of vitamin C and other antioxidants in relation to cancer

Sales of supplements such as vitamin C, vitamin E, vitamin A, selenium and other antioxidants have been promoted by focusing on their ability to eliminate free radicals which cause DNA damage and thereby increase risk of chronic diseases.<sup>74</sup> However, their mechanisms in relation to cancer have only been partially elucidated.<sup>75</sup> As discussed below antioxidants may exert different influences at different stages of cancer development. This can be split into three stages: initiation where DNA damage and repair occurs; promotion where cell proliferation, involving the mutation of tumour-suppressor genes and the formation of oncogenes, can be reversed by programmed cell death; and progression which leads to the change of cells from benign to malignant and involves angiogenesis (the increased supply of blood vessels to the tumour) and metastasis (the spread of the cancer to other areas of the body).<sup>76</sup>

Vitamin C, as ascorbic acid, can readily donate electrons to suppress activities of free radicals such as reactive oxygen species (ROS). High levels of ROS such as super oxide ( $O_2$ ), hydrogen peroxide ( $H_2O_2$ ), the hydroxyl radical (OH) and nitric oxide radical (NO) are thought to initiate cancer through oxidative damage to DNA. ROS are produced internally by the release of electrons from mitochondria, white blood cells, peroxisomes and the detoxification enzyme cytochrome P450.<sup>76</sup> It is possible that antioxidant supplementation may prevent DNA damage during the initial stage of cancer in individuals with high levels of ROS which are not eliminated by endogenous antioxidants such as superoxide dismutase.<sup>76</sup> <sup>77</sup> Salganik et al. (2001) suggest

populations may need screening to identify those with high and low innate levels of ROS in order to determine appropriate supplementation.<sup>77</sup> However, levels of 8-OHdG, a marker of DNA damage, do not appear to reduce with vitamin C supplementation, except for individuals with very low intakes.<sup>78</sup> Higher baseline levels of 8-OHdG have been found in individuals who had lower serum ascorbic acid levels, those who did not use antioxidant supplements, or had low fruit and vegetable intake, or did not exercise regularly.<sup>79</sup> However, two months of supplementation with vitamin C (500mg/day) or vitamin E (400 IU/day) in this US RCT did not affect levels of 8-OHdG.<sup>79</sup> These results may indicate that supplementation may need to be prolonged to be effective or that high vitamin C intake or high serum ascorbic acid levels may be a marker of a good diet and healthy life style rather than a true protective factor against the initiation of cancer.<sup>79</sup>

One study has found a moderate but significant increase in breast cancer risk in women who had short telomeres and who had low dietary and supplement intake of antioxidants.<sup>80</sup> Telomeres protect genes from degradation, but appear to be detrimentally shortened by oxidative damage; therefore it may be possible that antioxidant activity could protect telomere length.<sup>80</sup>

Conversely, antioxidants may decrease protective functions of ROS which are necessary to suppress the promotion or proliferation stage of cancer which leads to the formation of preneoplastic cells.<sup>76</sup> As part of our immune system, phagocytes require ROS to enable them to destroy invading microorganisms and cancer cells. Excess ROS induce beneficial apoptosis, programmed death of damaged cells, without which uncontrolled cell division may lead to the progression of cancer.<sup>29, 77, 81</sup> The inhibitive effects of antioxidants on programmed cell death may be one reason why some studies have not shown antioxidant supplementation,<sup>7</sup> or vegetarianism to be protective of cancer.<sup>82</sup> In individuals who are constantly exposed to carcinogens such as tobacco, antioxidants may eliminate ROS preventing beneficial apoptosis whilst the carcinogen is left to promote cancer cells.<sup>77</sup> This may explain why the antioxidant  $\beta$ -carotene, a form of vitamin A, increases the incidence of lung cancer in smokers, but decreases the incidence in non-smokers.<sup>783</sup>

On the other hand, vitamin C appears to beneficially modulate the expression of tumour suppressor genes such as p53 and p73 which initiate apoptosis of damaged cells.<sup>84</sup> Therefore the effect of vitamin C at the proliferation stage of cancer may depend upon the balance it offers between the detrimental elimination of ROS required for apoptosis and its beneficial promotion of suppressor genes.

Some anticancer drugs such as cis-platin generate ROS and eliminate cancer cells by mediating apoptosis. It has been reported that the antioxidant alpha-tocopherol, a form of vitamin E, can reduce the amount of ROS and apoptosis produced by cis-platin in breast cancer cells, thus reducing its effectiveness.<sup>77</sup>. Although vitamin C can regenerate vitamin E and may inhibit apoptosis itself, research on cell-lines have shown it can increase the sensitivity of cis-platin.<sup>85</sup> However, a recent systematic review of antioxidant supplementation during chemotherapy indicated in general they improved survival times and tumour responses, and also reduced toxicities to the chemotherapy.<sup>86</sup> Mixed results were produced using high dose vitamin C supplementation on terminal cancer patients.<sup>87-89</sup> Results of trials using intravenous vitamin C for cancer therapy also produced mixed results.<sup>39</sup>

It is unclear whether ascorbic acid and other antioxidants can inhibit processes in the final stage of cancer development involving angiogenesis and metastasis, although mechanisms have been proposed.<sup>84</sup> Since vitamin C promotes collagen formation this may help maintain the extracellular matrix and prevent tumour cells for spreading. On the other hand, collagen formation will support the growth of new blood vessels needed for tumours to develop.<sup>84</sup>

Vitamin C as ascorbic acid can also act as a pro-oxidant by reducing 'free' transition metals such as copper and iron, which via the Fenton reaction with hydrogen peroxide leads to the formation of highly reactive and damaging hydroxyl radicals (Fe(II) + H<sub>2</sub>O<sub>2</sub>  $\rightarrow$  Fe (III) + OH' +OH-).<sup>76 77 84</sup> Potentially high intakes of haem-iron in red meat plus ascorbic acid could initiate cancer;<sup>90</sup> indeed this combination was associated with an increase risk of lung cancer in the Iowa Women's Health Study.<sup>91</sup> However, since free iron or copper are not normally available in vivo because they are bonded to proteins such as ferritin, transferrin and metallothioneins, this hypothesis is controversial. Nevertheless, Kabet and Rohan (2007) propose that high levels of stored iron, particularly found in menopausal women, can be released by mechanisms involving infection, inflammation, high alcohol intake and oestrogen metabolites which via the Fenton reaction could increase breast cancer risk.<sup>92</sup> Conversely, since both iron and copper are more abundant in cancer cells than in normal cells, this pro-oxidant mechanism could be used to target the apoptosis of tumour cells, particularly since ascorbic acid has been found to mobilise copper ions.<sup>93 94</sup>

As discussed above, the overall effect of various mechanisms involving vitamin C or other antioxidants on the development of cancer in general is still unclear. With regard to specific cancers, vitamin C may reduce the risk of gastric cancer by suppressing the formation of nitrosamine compounds; however its effect is reduced when infection by H pylori is present.<sup>95 96</sup> A meta-analysis of RCTs, however, did not find a significant

protective effect of antioxidant supplements on gastric cancer.<sup>97</sup> The influence that vitamin C may have on breast cancer specifically is less clear since, as seen in the next chapter in section 3.5, this cancer is predominately influenced by hormones. Vitamin C may have a greater influence on non-hormonal cancers, including non-hormonal breast cancers. Nevertheless, as seen in section 3.4.2, vitamin C may affect breast density which is associated with breast cancer risk.
## **CHAPTER 3**

## 3 Literature review of breast cancer risks

## 3.1 Introduction

The main approach used for the time-to-event analyses in this thesis is based on the lifestyle model, and in particular on the assumption that lifestyles during mid-life may affect the risk of developing adult chronic diseases. Later chapters in this thesis use recordings of diet and/or supplement intake during one time period in mid-life to estimate their influence on cancer risk seven or more years later. Other lifestyle factors measured at mid-life which may affect breast cancer risk, such as exercise and alcohol intake, are discussed in sections 3.5 and 3.6 of this chapter and most of these are taken into account in the analyses. Some factors highlighted in the life course approach to chronic diseases which may also influence breast cancer risk can also be controlled for in the analyses, such as age at menarche or age of first birth; these are discussed in section 3.5. This combination of approaches has been used in the majority of studies evaluating diet and cancer risk discussed in section 3.6.

Cancer can take decades to develop and may be due to independent cumulative exposure effects or may be due to a chain of linked events. Within the life course approach there are models which emphasise critical periods such as gestation, childhood, adolescence, young adulthood as well as adult life.<sup>98</sup> Interaction effects between exposures in different critical periods may also occur.<sup>98</sup> Exposures during gestation, childhood and adolescence may, for instance, influence hormonal mechanisms linked to breast cancer risk, as discussed in section 3.5. Such data have not been gathered in studies used in the current analyses. Indeed, few longitudinal studies have recorded information from the pre-natal stage, though famine studies and animal studies may illuminate hypotheses based on pre-natal programming of adult cancer.<sup>99 100</sup> Long follow-up studies on humans are expensive, time consuming and prone to attrition, and as yet there is no intermediate breast cancer end-point that can be easily employed.

Cancer occurs both through genetic modification to the DNA sequence and epigenetic modifications, i.e. changes in gene expression which occur without altering the DNA sequence. DNA sequence mutations occur due to damage from external factors such as sunlight and radiation, and from internal factors. Damaging reactive oxygen species

(ROS) are produced by a variety of normal intracellular processes, some of which are mentioned in section 2.2.4. Furthermore, there is substantial evidence that ROS become elevated as a result of intense exercise<sup>101</sup> or hyperglycaemia<sup>102</sup>

Epigenetic changes are manifest in DNA methylation, histone (chromatin) modification and non-coding RNAs (ribonucleic acid). Such alterations are transmitted to daughter cells during cell division and appear to be involved in most steps of tumour progression and development.<sup>103</sup> Although currently unclear, maternal diet and other exposures *in utero* have the potential to cause epigenetic changes to the foetus, such as mammary gland development, which may increase susceptibility to breast cancer.<sup>99</sup> Twin studies show that epigenetic changes can also accumulate over the life course, with older twins showing greater differences in DNA methylation and histone acetylation levels.<sup>104</sup> Environmental and dietary factors such as tobacco smoke, ionizing radiation, UV radiation, alcohol and folic acid may contribute to the accumulation of epigenetic changes during aging which cause cancer.<sup>103</sup> Moreover, there is evidence that epigenetic changes originating *in utero* in animals can be passed on to future generations.<sup>105</sup> Unlike genetic changes, epigenetic changes are reversible which means new cancer prevention and treatment strategies could be developed in the future.<sup>103</sup>

This chapter will review the literature relating to breast cancer risks. Rather than reporting individual studies which can produce conflicting results it focuses in general on larger or more recent pooling studies and reviews, particularly systematic reviews and meta-analyses. Many articles referenced in this chapter appear to be influential, as evident from their large number of citations. Methodological issues including study design, measurement tools employed, measurement error, confounding factors, bias, length of follow-up and the initial health of study members, can all affect results. In relation to nutrition, conflicting results may arise due to dose effects, and to interactive and synergistic effects of nutrients. First, a brief description of breast cancer types and a summary of incidence rates, trends and survival are given.

## 3.2 Classification of breast cancer

Lobular carcinoma in situ (LCIS) and ductal carcinoma in situ (DCIS) are non-invasive or pre-invasive cancers, sometimes termed pre-cancerous, which have not spread elsewhere. Respectively they describe cancer cells present in the lobes in the breast glandular tissue, and cancer cells present in the milk ducts between the nipple and the lobes. LCIS does not cause symptoms and usually does not show up on mammograms.<sup>106</sup> If cancer cells do spread outside of the lobe it is called invasive lobular breast cancer. Similarly if the cancer cells have spread outside of the ducts it is called invasive ductal breast cancer; 70-80% of breast cancers diagnosed are of this type.<sup>107</sup> When cancer is also found in the lymph glands, such as those in the armpit or surrounding the breast, then cancer is likely to spread to other parts of the body via the lymphatic system; nevertheless this is still classed as early stage breast cancer. Secondary breast cancer, also called metastatic breast cancer or stage 4 breast cancer, is a term for cancer that has spread to other parts of the body.

Rare breast cancers include inflammatory breast cancers which occur when the lymph channels become blocked, and Paget's disease which starts at the nipple area.

Breast cancers are also classified according to the presence and type of hormone receptor or oncogene. If the breast cancer cells have oestrogen receptors present then the tumour is said to be oestrogen receptor positive (ER+). Other receptor statuses relate to the presence of progesterone receptors (PR) and Human Epidermal Growth factor Receptor 2 (HER2). Triple negative cancers do not have these receptors; this is often found in basal type cancers which are associated with women who have BRCA1 germ line mutations.

Treatment decisions are based on the TNM stage and the grade of each breast cancer case. The TNM stage classification system describes the size of the tumour or whether the cancer remains in situ (T1-4, Tis(DCIS), Tis(LCIS)), whether cancer has spread to the lymph nodes (N0-3), and whether it has spread anywhere else in the body i.e. metastasized (M0 or M1).<sup>108</sup> The grade of cancer describes whether it is a slow or fast growing cancer. Fast growing cancers are more likely to spread and return after treatment than slow growing cancers.

## 3.3 Breast cancer incidence, trends and survival

Breast cancer is the most common cancer in women and the second most common cancer overall in men and women worldwide.<sup>4</sup> Globally, there were an estimated 1,383,000 cases of malignant breast cancer in women in 2008, making up 10.9% of all cancer cases and 22.9% of all female cancers (excluding non-melanoma skin cancer).<sup>4</sup> The world age-standardised incidence rates (ASR) in 2008 for female breast cancer were 39 per 100,000 and the cumulative risk to the age of 74 or more was 4.1%.<sup>4</sup> Breast cancer incidence is substantially higher in more developed countries (except Japan) compared to less developed countries (ASR = 66.4 vs. 27.3 per 100,000). In Northern Europe in 2008 it was 84.0 per 100,000 and the estimated cumulative risk to the age of 74 was 8.9%.<sup>4</sup> Figures for North America were 76.7 per 1000,000 and 8.4%

respectively.<sup>4</sup> The higher incidence in developed countries may be partially due to better procedures to detect breast cancer cases earlier e.g. through surveillance, as well as better procedures to accurately record cases. Some African and Asian countries submitted no data at all to Globocan for the global cancer estimate, and their estimates were based on data from neighbouring countries,<sup>4</sup> though the accuracy and quality of the latter may not be good.

In 2008 there were an estimated 332,770 breast cancer cases in the European Union and 45,570 in the UK. The European age standardised rate (EASR) for breast cancer incidence in the European Union (EU) countries in 2008 was 103.7 per 100,000, and the EASR for the UK was 119.1 per 100,000; the UK has sixth highest rate of the EU countries (Figure 3).<sup>3</sup> In the UK, 31% of all female cancers were breast cancers, making it the most common cancer.<sup>3</sup> The lifetime risk in 2008 in the UK has been estimated as 1 in 8. <sup>6</sup> Figure 4 shows that breast cancer incidence rates have more than doubled in Britain since the mid 1970s.<sup>109</sup> Rates increased substantially after breast screening was introduced in 1988, which found previously undetected cancers.<sup>110</sup> HRT use during the 1990s is also believed to have contributed to the increased rates.<sup>111</sup> Figure 5 shows that the sudden increase in rates was mainly confined to the 50-64 age group who were invited to join the breast screening program.<sup>109</sup>

The estimated deaths (458,000) attributed to breast cancer in women globally in 2008 was substantially lower than incidence rates, being 12.5 per 100,000. Globally 13.7% of female cancer deaths were attributed to breast cancer, it was the most common death from cancer in females and the fifth most common overall in men and women.<sup>4</sup> In 2008 there was an estimated 332,800 breast cancer deaths in the European Union and 89,800 in the UK; 16.6% and 16.0% cancer deaths in the EU and UK related to breast cancer.<sup>3</sup>

The percentage of women still surviving five years after their first malignant invasive breast cancer diagnosis varies widely between countries. Survival rates at 5 years, age-standardised to the International Cancer Survival Standard weights, for women diagnosed between 1990-1994, ranged from 80% or over in North America, Sweden, Japan, Finland, and Australia to less than 60% in Brazil and Slovakia, and below 40% in Algeria.<sup>112</sup> Most European countries were in the 70-79% range, however the UK survival rate was 69.7%.<sup>112</sup> The five year survival rates have improved in England over the last thirty years, from 52% for women diagnosed in the early 1970s to 82% for women diagnosed between 2001-2006.<sup>109</sup> Five year survival rates are lower the later

the stage at diagnosis, ranging from 98% for very early stage to 18% for metastatic breast cancers in Europe.<sup>113</sup>



**Figure 3** Estimated European age-standardised incidence and mortality rates in 2008 by EU country<sup>3 109</sup>

**Figure 4** European age-standardised breast cancer incidence and mortality rates for Britain between 1975-2008<sup>109</sup>



Source: Cancer Research UK. Cancer Statistics - breast cancer http://info.cancerresearchuk.org/cancerstats/types/breast/incidence/



Figure 5 Age-Specific breast cancer incidence rates for Britain, 1975-2008<sup>109</sup>

Source: Cancer Research UK. Cancer Statistics - breast cancer http://info.cancerresearchuk.org/cancerstats/types/breast/incidence/

## 3.4 Major breast cancer risk factors

#### 3.4.1 Family history and genetics

Most women who have a family history of breast cancer do not develop cancer themselves. A 2001 meta-analysis estimated that women with one first degree relative (mother, sisters, daughters) who developed breast cancer have about double the risk of developing breast cancer compared to women with no affected relative.<sup>114</sup> Increased risks may be due to shared genetic and/or environmental factors. Risks are estimated to be nearly four-fold for women who have three or more affected relatives;<sup>114</sup> these cancers are more likely to be inherited. Risks were higher the younger the relatives were at diagnosis, particularly if diagnosis was before the age of 50.<sup>114</sup>

Risks are very high in women who have certain rare germ line mutations such as BRCA1, BRCA2, TP53, PTEN, STK11 and CDH1. These high-risk alleles are rare (~0.1% of the population are carriers) but account for about 16% of familial breast cancer risk,<sup>115</sup> and about 5% of all breast cancer cases. Women with BRCA1 and BRCA2 mutations have a 10- to 20-fold relative risk and a 30-60% risk of developing

breast cancer before the age of 60 compared to 3% in the general population.<sup>115</sup> 0 describes the criteria the NHS use for defining people at high risk. These genetic mutations are also linked to high risks of developing other cancers; for instance BRCA2 is also associated with ovarian and pancreatic cancer as well as prostate and breast cancer in men.<sup>116</sup>

Other rare mutations, such as ATM, BRIP1, CHEK2 and PALB2 confer moderate risks of two- to 4-fold of developing breast cancer; together these may only account for 2.3% of familial risk of breast cancer.<sup>115</sup> These are involved in biological pathways relating to BRCA1 and BRCA2 genes. Additionally, the genome-wide search of hundreds of thousands of single-nucleotide polymorphisms have found common low-risk alleles associated with increased breast cancer risks of less than two-fold. Although the frequency in the population of some of these alleles may be as much as 50%, those discovered so far are likely to account for only a small percentage relating to familial risk in total.<sup>115</sup> It is unknown how these may interact with the high-risk alleles.

#### 3.4.2 Breast density

Breast density, distinguished by non-fatty mammary tissue, is strongly and independently associated with increased breast cancer incidence, and shows a doseresponse relationship.<sup>117</sup> A 2006 systematic review estimated that dense breast tissue in more than 75% of the breast increases risk of breast cancer by over four fold compared to densities below 5%.<sup>117</sup> Mammographic breast density is also associated with other risk factors such as age, menopausal status, parity and body weight, and to a lesser extent family history, race, alcohol consumption and diet. <sup>118</sup> High saturated fat intake has been associated with lower breast density.<sup>119</sup> One study found an inverse relationship between breast density and fruit and vegetable intake, vitamin C intake, and also with olive oil.<sup>120</sup> Conversely, another study reported total vitamin C intake and supplement vitamin C intake was positively associated with breast density in premenopausal women, although not in post-menopausal women.<sup>119</sup> There is evidence that density can be altered using combined oestrogen-progesterone hormone therapy, although the evidence relating to the effects of endogenous hormones and growth factors is inconsistent.<sup>121</sup> Twin studies have demonstrated that breast density is heritable,<sup>121</sup> nevertheless it has been reported that carriers of BRCA1 and BRCA2 genes have a similar breast density to non-carriers.<sup>122</sup> It has been suggested that breast density should be incorporated into risk prediction models,<sup>123</sup> however it is unknown whether breast density is predictive in young women when high density breast tissue is more common.<sup>117</sup>

## 3.4.3 Previous diagnosis

Benign breast disease is a risk factor for malignant breast; women with proliferative breast disease without the production of abnormal cells (without atypia), have double the risk compared to the general population, whilst those with atypical hyperplasia have a more than four-fold increased risk.<sup>124</sup> Women with a previous *in situ* tumour (DCIS and LCIS) have double the risk of invasive breast cancer in the same breast compared to the general population.<sup>125</sup>

Previous invasive breast cancer can increase the risk of contra-lateral breast cancer by as much as 5-fold, although the spread of the initial cancer to the other breast within the first two years probably inflates this figure.<sup>126</sup> Contra-lateral breast cancer diagnosed two years after the primary breast cancer is linked to tumour size and age; being more common if the first cancer was diagnosed before the age of 40.<sup>126</sup>

## 3.4.4 Thoracic radiotherapy

Women who were treated with moderate to high dose chest radiation for childhood, adolescent or young adult cancer have a substantially increased risk (13- to 15- fold) of developing breast cancer at a young age.<sup>127</sup> Exposure to lower doses of radiation in infancy has also been associated with increased breast cancer risk; risks increased linearly with increasing dose.<sup>128</sup>

### 3.4.5 Risk prediction tools

Various models have been developed to estimate life-time risk of breast cancer in order to help clinicians decide whether surveillance is necessary for particular women; these include Gail-2; Claus Model; Claus Tables; BOADICEA; Jonker Model; Claus-Extended Formula, and Tyrer–Cuzick.<sup>129</sup> In addition to breast biopsy history, family history and/or genetic test results, parameters often included in these models are age of menarche; parity; age at first child; menopausal status; menopausal age; BMI; ovarian status; age at ovarian status; HRT use.<sup>129</sup> These will be discussed below. Whilst the models aim to calculate the risk for an individual, the risks used in the models for these various risk factors are based on population risks from epidemiology studies.

# 3.5 Hormonal and related factors affecting risk of breast cancer

### 3.5.1 Hormone lifetime exposure

Hormonal and reproductive factors such as early onset of menarche, late age of first birth, low parity and late menopause are established risk factors for breast cancer.<sup>130 131</sup>

Hormonally induced changes in mammary cells during pregnancy may provide protective effects.<sup>132</sup> In contrast, it is believed that early onset of menarche and late menopause increase risk via their influence on women's lifetime exposure to endogenous oestrogens. High energy diets during childhood have been associated with early menarche thereby increasing exposure and risk.<sup>133</sup> <sup>134</sup> Risk increases rapidly during the reproductive years but then increases at a slower rate during menopause, which is reflected in the reduced production of oestrogen by the ovaries.<sup>131</sup> Due to fluctuations in hormone levels in the menstrual cycle of pre-menopausal women, lifetime exposure to oestrogens cannot be measured, therefore hormonal and reproductive events, plus the use of contraceptive pills and Hormone Replacement Therapy (HRT) that contain them are employed as proxy covariates in risk analyses.

#### 3.5.2 Exogenous hormones in HRT and contraceptive pills

Oestrogen production from the ovaries diminishes as women approach the menopause, which can cause numerous problematic symptoms that can often be alleviated with Hormonal Replacement Therapy (HRT). However, there is substantial evidence that HRT increases breast cancer risk during its use, with increased risk disappearing about five years after cessation.<sup>135</sup> <sup>136</sup> It has been estimated that risk increases by 2·3% for each year of use in current or recent HRT users; this compares with an increased risk of 2.8% that would normally occur for each year that menopause is delayed.<sup>135</sup> The increased risk has been reported to be more pronounced in women with lower BMIs.<sup>135</sup> Post-menopausal oestrogen-plus-progesterone replacement therapy has been found to increase breast cancer risk to a greater extent than oestrogen alone replacement therapies;<sup>136</sup> <sup>137</sup> Additionally, there is evidence that mammographic breast density, which is linked to increased breast cancer risk, increases during HRT.<sup>138</sup>

In addition to oestrogen and progesterone, ovaries also produce testosterone in normal pre-menopausal women. Testosterone combined HRT has been used to treat women with removed ovaries, as well as to treat menopausal symptoms in normal post-menopausal women.<sup>139</sup> Adding testosterone to HRT has been found to inhibit the stimulatory effects of oestrogen on breast cell proliferation in monkeys.<sup>140</sup> However, there has been little research relating to women and this is inconclusive: although one study reported no increased breast cancer risk in women using the testosterone combined HRT,<sup>141</sup> another study reported increased risk.<sup>142</sup>

A modest increase in breast cancer risk was reported in a large meta-analysis of 54 studies relating to current or recent use of oral contraceptives (containing oestrogen

and progestin); this increased risk disappeared within 10 year of cessation.<sup>143</sup> There is evidence from an unadjusted meta-analysis of case-control studies that risk is more pronounced in multiparous women who used oral contraceptives before their first full term pregnancy than women who started use later.<sup>144</sup> Newer formulations of oral contraceptives have reduced oestrogen and progestin content; results of some studies indicate that the risk of breast cancer from the use of these may be lower than that from the older type of contraceptive.<sup>145</sup>

#### 3.5.3 Endogenous hormones

There is consistent research evidence to show that high serum concentrations of endogenous oestrogen are associated with increased breast cancer risk in postmenopausal women; additionally there is evidence that androgens, prolactin, insulin hormones and insulin-like growth factors (IGF) may also influence risk.<sup>146</sup> This section describes the evidence that links serum concentrations of these hormones to breast cancer risk, much of which is from seminal work on the pooling of thousands of cases in nine prospective studies. Pooling not only increases the power to detect smaller effect sizes, but in contrast to meta-analyses ensures uniformity in analysis methods. The following sections describe how these hormones are related to lactation, body fatness, physical activity and height which are also associated with breast cancer risk.

The pooled analysis of nine prospective studies showed that post-menopausal women with the highest fifth of oestradiol serum concentrations had twice the relative risk compared to the lowest fifth for this endogenous oestrogen.<sup>147</sup> Other prospective studies such as EPIC have reported similar results.<sup>132</sup> <sup>148</sup> <sup>149</sup> Free oestradiol levels, including non-SHBG bound oestradiol which were free to enter breast cells had a larger effect than total oestradiol on risk.<sup>147</sup> Evidence suggests that oestradiol and other oestrogens stimulate mammary cell proliferation, thereby promoting cancer.<sup>150</sup> Oestradiol has been found to have a strong and significant association with oestrogen receptor positive breast cancer in post-menopausal women in some studies,<sup>149</sup> <sup>151</sup> although in other studies the increase was not statically significant. <sup>148</sup> <sup>152</sup> Although there is some evidence of an increased risk relating to higher oestradiol levels in premenopausal women, it is inconclusive, probably as a result of large fluctuations in levels during the menstrual cycle.<sup>146</sup> Despite the fact that some genes are associated with an increase in oestrogen levels there is no evidence they have a large enough effect to significantly increase breast cancer risk.<sup>146</sup>

The pooling of nine prospective studies also found breast cancer risk was more than doubled for post-menopausal women with testosterone concentrations in the highest fifth compared to the lowest fifth.<sup>147</sup> Testosterone is more abundant in post-menopausal women than oestrogen, and is converted to oestradiol by aromatase; thereby indirectly increasing risk via increases in oestradiol. Nevertheless testosterone was found to be independently associated with breast cancer after adjustment for oestradiol, though rates were slightly attenuated.<sup>147</sup> A positive association was also seen in premenopausal women in four prospective studies.<sup>146</sup> Positive associations between other androgens with breast cancer have also been found.<sup>147</sup> In addition to being an oestrogen precursor, androgens can act directly on breast cells by binding to androgen receptors and may stimulate cell proliferation.<sup>153</sup> Conversely, other researchers propose that androgens such as testosterone are protective in respect to breast cancer risk,<sup>139 154</sup> since some studies found an inverse association; differences in results may depend on the methods for measuring testosterone, the calculation of free testosterone as well as oestrogen status.<sup>139</sup> Hypotheses have been proposed which may explain some, but not all of the inconsistencies in results: in the absence of oestrogens, for instance in post-menopausal women, androgens may stimulate growth of breast cancer cells via binding to oestrogen receptors; alternatively in the presence of high oestrogen concentrations, androgens may act as anti-oestrogens to inhibit oestrogen stimulation of growth of breast cancer cells via binding to androgen receptors.155 Furthermore, in vitro androgens have been shown to stimulate or inhibit breast cancer cell proliferation depending on the breast cancer cell line.<sup>155</sup> In post-menopausal cancer patients, high testosterone levels have been significantly associated with positive oestrogen receptor status, large tumour size and high BMI.<sup>156</sup> Although this supports the original androgen excess theory,<sup>156</sup> the association in cross-sectional studies could also be explained by the tumours' effect on androgen production. The mechanisms by which androgens affect breast cancer risk remain unclear.

Serum levels of sex-hormone-binding globulin (SHBG) have been inversely associated with breast cancer risk,<sup>147 157</sup> risk reduced by approximately a third in a high verses low analysis of nine pooled cohorts.<sup>147</sup> SHBG not only reduces the bioavailability of oestrogens and testosterone to cells, but may also act directly on breast cancer cells by inhibiting oestradiol induced cell proliferation.<sup>158</sup>

Prolactin, the luteotropic hormone, was associated with a moderate increase in breast cancer risk in a pooled analysis of two prospective studies.<sup>159</sup> No difference in risks between pre-menopausal and post-menopausal women was found, probably since prolactin declines only slightly during the menopause.<sup>159</sup> Risks were higher for women with ER positive tumours.

In a large pooled analysis of seventeen studies insulin-like growth factor-1 (IGF-1) showed a significant association with breast cancer risk, in particular oestrogen receptor positive cancers (OR = 1.38 (95% CI 1.14-1.68), for both pre-menopausal and post-menopausal women.<sup>160</sup> Results of an earlier meta-analysis showed associations were limited to pre-menopausal women,<sup>161</sup> however this analysis did not include four large recent studies included in the pooled analysis. The association in the pooled study was independent of levels of oestradiol and testosterone, and was not modified by these.<sup>160</sup> IGF-1 was positively associated with height and age at first pregnancy and inversely associated with age at menarche and years since menopause,<sup>160</sup> which are risk factors for breast cancer. Dietary factors associated with increased IGF-1 are dairy protein and calcium.<sup>162</sup>

Other reviews have suggested a possible link between insulin or C-peptide (a marker of pancreatic insulin secretion) and increased breast cancer risk, although some studies showed no association.<sup>163</sup> <sup>164</sup> Hyperinsulinemia has also been associated with increased breast cancer risk.<sup>165</sup> Insulin may affect tumour growth indirectly through increasing bioavailability of IGF-1 and decreasing sex-hormone-binding globulin (SHBG). Insulin and IGF-1 may also have a direct effect by increasing breast cancer cell proliferation, though this has only been demonstrated in vitro.<sup>158</sup>

Despite evidence that the combined oral contraceptive containing oestrogen and progesterone confers more risk than oestrogen alone, the evidence for the influence of endogenous progesterone on breast cancer risk has been inconsistent and does not provide support for an oestrogen-progesterone hypothesis.

Shift work may affect breast cancer risk through hormones levels; a meta-analysis showed that shift work increased breast cancer risk by about 50%.<sup>166</sup> It is hypothesised that reduction in melatonin, which occurs in artificial light, may increase levels of hormones such as oestrogen, additionally melatonin may have a direct effect on tumour suppression.<sup>166</sup>

#### 3.5.4 Pregnancy and lactation

In addition to a 7% reduction in breast cancer risk for every birth, a woman's risk is further decreased by 4.3% for every year in total that she breast-feeds.<sup>167</sup> This evidence is provided by a pooled analysis of 47 epidemiological studies from 30 countries.<sup>167</sup> The mechanisms involved remain unclear, however hormonal influences are likely to be involved, relating for instance to delayed ovulation and increased breast cell differentiation.<sup>168</sup>

#### 3.5.5 Body fatness

Body fatness is an established risk factor for post-menopausal women, whereas a probable inverse relationship with breast cancer exists for pre-menopausal women. <sup>7 169</sup> Several hypotheses involving hormones such as insulin, insulin-like growth factor and oestrogens that were discussed in previous sections have been proposed to explain the former. There is no clear mechanism, however, to explain the inverse relationship for pre-menopausal women. Waist circumference and waist to hip ratio are positively associated with breast cancer risk in postmenopausal women. 7 169 The 2008 WCRF continuous update report stated that abdominal fat is a probable cause of postmenopausal breast cancer, as is adult weight gain.<sup>169</sup> BMI, which is derived from readily obtainable height and weight measurements, is often used as a marker of body fat and is associated with breast cancer risk. The updated meta-analyses of cohort studies in the 2008 WCRF continuous update report showed a 7 per cent decreased risk per 5kg/m2 for premenopausal women and an 8% increased risk per 5kg/m<sup>2</sup> increase in BMI for postmenopausal women.<sup>169</sup> Obese post-menopausal women in particular (BMIs over 30 kg/m<sup>2</sup>) are at increased risk of breast cancer. Additionally, obese patients tend to have larger sized tumours, their cancers are more likely to metastasize due to lymph-node involvement and they generally have a poorer prognosis for survival.<sup>170</sup> <sup>171</sup> Weight loss is recommended for survivors and gastric bypass surgery improves cancer outcomes in obese patients, for both pre and postmenopausal women.<sup>172</sup> There is some evidence that women genetically at high risk could reduce their risk by avoiding weight gain post-menopausally.<sup>173</sup>

Levels of oestradiol are higher in obese post-menopausal women, which appear to explain why the risk of breast cancer is greater in post-menopausal women with high BMI; indeed controlling for oestradiol concentrations in a prospective study reduced the risk towards unity in the association between breast cancer and BMI.<sup>174</sup> The association between oestrogen production and obesity related breast cancers is further supported by the fact that these cancers are more often ER+,<sup>175</sup> and particularly in women who were not currently taking HRT. <sup>176</sup> Weight gain during adulthood is also associated with an increased risk of post-menopausal breast cancer,<sup>169</sup> and is also stronger in women who were not currently,<sup>176</sup> or previously using HRT,<sup>177</sup> or who had a later menarche.<sup>176</sup> Therefore the effects of weight gain on risk appear to be obscured in women who had early menarche or used HRT which would increase their cumulative exposure to oestrogen .<sup>176</sup>

Adipose tissue becomes the primary source of circulating oestrogens during menopause when the ovaries cease production. Aromatase in adipose tissue, mainly

found in breast, abdomen, thighs and buttocks, convert androgens to oestrogen.<sup>158</sup> Aromatase inhibitors are recommended to reduce peripheral production of oestrogens in women with advanced (or metastatic) breast cancer.<sup>178</sup>

Alternatively insulin resistance and IGF-1 concentrations associated with metabolic syndrome, together with the action of oestrogens may explain the link between breast cancer and obesity.<sup>158</sup> Since high levels of insulin are inversely related to SHBG, the bioavailability of oestrogens and testosterone are also likely to increase.<sup>175 179 180</sup> Intraabdominal adipose tissue, in particular, influences the production of insulin and other hormones; waist circumference being a reasonable measure of abdominal adiposity. Hyperinsulinemia may also promote the synthesis and activity of IGF-1.<sup>158</sup> However, despite IGF-1 being positively associated with weight in overweight women, lower levels of IGF-1 are found in obese women and therefore cannot explain the increased risk of breast cancer in obese women.<sup>160 181</sup> Insulin resistance and hyperinsulinemia are seen in the early stages of type-2 diabetes, and both hyperinsulinemia and type-2diabetes are associated with an increase in breast cancer risk.<sup>163 165</sup>

A third hypothesis proposes that various cytokines, polypeptides and hormone-like molecules secreted by adipocytes may influence risk via their autocrine, paracrine, and endocrine functions.<sup>158</sup> Since the majority of cells in the breast are adipocytes, their autocrine and paracrine activities are likely to affect breast epithelial cells. The adipokines secreted by adipocytes include leptin, adiponectin, necrosis factor-a (TNFa) and interleukin (IL-6); the latter two being a marker of inflammation along with Creactive protein. Furthermore, tumours stimulate the production of aromatase and oestrogens locally via the autocrine and paracrine actions of TNF-α and IL-6.158 Chronic inflammation has been implicated in various steps of tumour formation including cell proliferation, angiogenesis and metastasis, and suppression of apoptosis has been associated with increased risk of breast cancer.<sup>182</sup> In vitro evidence suggests leptin may also promote cancer in a variety of ways including breast cancer cell proliferation, and via effects on ER and insulin signalling.<sup>158</sup> Additionally, case-control studies have reported an association between leptin and breast cancer.<sup>164</sup> Leptin suppresses appetite, and although high levels are found in obese individuals it is likely they develop leptin resistance.<sup>158</sup> Adiponectin is inversely associated with insulin resistance and obesity, and is also inversely associated with breast cancer.<sup>183</sup> Inflammation, which is linked to abdominal obesity, is also associated with increased reactive oxidative species which can cause damage to DNA.

#### 3.5.6 Physical activity

There was sufficient evidence in the WCRF 2007 and 2008 reports to suggest that physical activity probably protects against breast cancer in post-menopausal women, whereas the evidence for pre-menopausal women is inconsistent and limited.<sup>7 169</sup> For post-menopausal women meta-analyses showed a 3% decrease in risk per 7 MET (Metabolic equivalent task) hours of recreational activity per week (equivalent to about 2 hours brisk walking).<sup>7</sup> Since physical activity is difficult to measure this may explain some of the inconsistencies in the evidence base. There are a number of biological mechanisms involving hormones whereby physical activity may reduce the risk of breast cancer in post-menopausal women, though they remain unclear.

In a recent review Neilson et al. proposed a biological model which links the influence of physical activity through BMI, oestrogens, insulin resistance and chronic inflammation, and also through less commonly considered biomarkers such as leptin.<sup>164</sup> Physical activity can reduce adiposity which will then result in reduced amounts of circulating oestrogen produced from the conversion of androgens by aromatase found in adipose tissue. Fat loss from exercise will also lead to a reduction of inflammatory markers, such as TNF- $\alpha$  and IL-6, and also leptin, which have been associated with breast cancer risk.<sup>158</sup> Physical activity also reduces insulin levels;<sup>184</sup> since this is inversely related to circulating SHBG the bioavailability of testosterone and oestradiol is also likely to decrease.<sup>175 179 180</sup> Additionally, regular physical activity can sustain insulin sensitivity and reduce the likelihood of developing type-2 diabetes,<sup>184</sup> which is associated with breast cancer.

#### 3.5.7 Hormonal exposure in utero and childhood development

Increases in birth-weight above 4000g compared to below 2500g are associated with increased breast cancer risk, and most consistently with pre-menopausal breast cancer.<sup>185</sup> A 2007 WCRF meta-analysis showed an 8% increase in risk per kg increase in birth weight.<sup>7</sup> Trichopoulos hypothesised that *in utero* exposure to oestrogen influences the risk of breast cancer,<sup>186</sup> because oestrogen levels are at their highest during pregnancy and high maternal levels have been linked to high birth-weight.<sup>187</sup> <sup>188</sup> IGF levels during pregnancy also substantially regulate foetal growth,<sup>189</sup> and may play a role in increased breast cancer risk.<sup>185</sup> These and other *in utero* factors may directly affect breast development, for instance by inducing epigenetic changes to the expression of oestrogen receptors, mammary stem cells or gland morphology.<sup>99</sup> Alternatively they may mediate breast cancer risk through other changes which increase risk in adult life; their mechanisms are as yet unclear.<sup>190</sup>

Birth length, adolescent height and adult height have been positively associated with breast cancer risk.<sup>191 192</sup> Meta-analyses from the 2007 WCRF report showed for every 5cm increase in adult height there is a 9% and 11% increased risk of pre-menopausal and post-menopausal breast cancer respectively <sup>7</sup> Factors that lead to adult height may influence breast cancer risk. Birth length, which is associated with adult height, may be a proxy for exposure to oestrogen and growth factors in utero. Birth length has also been associated with age at menarche; one study showed that girls who were long and light at birth had earlier menarche, which is associated with an increased risk.<sup>193</sup> Early growth spurts in adolescence are also positively associated with early menarche, though it is unclear whether early growth spurts are an independent factor in breast cancer risk.<sup>191</sup> Nutrition may influence age at menarche via rapid post-natal growth,<sup>193</sup> or through diet during childhood.<sup>133 134</sup> Despite being linked to earlier onset of menarche, body fatness during childhood has been inversely related to breast cancer in later life.<sup>194</sup> One hypothesis is that early breast differentiation induced via oestrogen by body fat may reduce breast malignancies later in life.<sup>195</sup> Additionally, as mentioned in section 3.5.3 IGF-1 concentrations have been positively associated with increased height and with breast cancer risk, but are inversely associated with age at menarche.<sup>160</sup> Furthermore, despite adult height being positively associated with breast cancer risk, taller women tend to have had a later menarche which can reduce breast cancer risk.<sup>191</sup> Despite some inconsistencies in these explanatory mechanisms, early life and adolescence are critical times for maturation of the hypothalamic pituitary ovarian axis, which regulates production of oestrogen and other hormones linked to breast cancer risk.<sup>191</sup>

## 3.6 Diet and breast cancer risk

The risk of breast cancer varies internationally; possible reasons for this, in addition to differing provisions for recording breast cancer incidence, include differences in genetics, environmental pollutants, and lifestyle factors such as reproductive events, exercise and diet. The observation that the risk of cancer for populations who migrate alters from their country of origin to their country of residence indicates that, at the population level, it is changes in lifestyle or environment which have a large influence on risk rather than genetics. Ecological evidence relating to breast cancer is apparent from women of Asian origin settled in the US whose breast cancer risk is substantially above that of their country of origin.<sup>196</sup> Given that one of the biggest changes in lifestyle for migrants is diet, this area of research has been given considerable attention.

On the whole the role of diet on cancer risk of individuals has been assessed using cohort or case-control studies, the latter being prone to recall bias and selection bias which may explain the frequent discrepancies in results between the two types of studies. Furthermore, self-reporting of food and supplement intake is prone to reporting bias in all types of studies, as well as measurement error, the latter being likely to attenuate any associations. Few dietary studies have used a RCT design, one weakness being the difficulty for randomised participants to adhere to new diets for a substantial length of time, although this is less of a problem with interventions using supplements. Conformity within the non-intervention group may also be poor, since motivated members may easily alter their dietary or supplement intake towards the intervention if this is known to them.

Several approaches have been used to assess the role of diet on cancer risk focusing on dietary patterns, individual or groups of foods, or individual nutrients. The majority of the analyses in this thesis uses the individual nutrient approach, assessing the impact of vitamin C from diet and supplements in chapters 8 to 10 using cohort designs. The effect of fruit and vegetable intake, the main source of vitamin C, is also analysed in chapter 9. Sections 3.6.4.1 and 3.7 of this chapter review the previous research on vitamin C. First the evidence of breast cancer risk in relation to dietary patterns and some foods, focusing in particular on fruit and vegetable intake, will be discussed in sections 3.6.1 and 3.6.2. Then evidence relating to macronutrients that have been linked to breast cancer risk will be discussed in section 3.6.3

#### 3.6.1 Dietary Patterns

The advantage of this over the individual food or nutrient approach is that dietary patterns incorporate interactions between foods or nutrients which are likely to occur in the body. Because of the difficulty of characterising dietary patterns, research in this area is not as extensive as other approaches. A dietary pattern constitutes a variety of dietary components which may typify the intake of a population or sub-population. Some of those that have been examined are traditional diets such as the Mediterranean and the Asian diets, which are viewed as healthy in contrast to the Western diet. The Mediterranean diet, characterised by seafood, poultry, fresh vegetables and olive oil, has been associated with a lower risk of breast cancer in US women,<sup>197</sup> and in pre-menopausal women in the UKWCS, though the latter was non-significant.<sup>198</sup> Associations between a variety of healthy diet indices, based on healthy eating guidelines, and breast cancer risk have also been assessed.<sup>199</sup> The alternate Mediterranean Diet Score (aMDS) and also the Alternate Healthy Eating Index (AHEI) and the Recommended Food Score (RFS) have been inversely associated with risk of post-menopausal oestrogen receptor negative (ER-) breast cancers.<sup>200</sup> Inverse

associations between the Diet Quality Index-Revised (DQI-R) and also the Canadian Healthy Eating Index (CHEI) and BRCA cancers indicate that some dietary patterns may be protective for women at high risk.<sup>201</sup>

#### 3.6.1.1 Healthy and unhealthy dietary patterns

Some studies have used factor analysis, principle components analysis or reduced rank regression to derive healthy and unhealthy dietary patterns; although these methods are less subjective than using diet scores, an element of subjectivity is still required during categorisation. A recent systematic review and meta-analysis of 18 case-control and cohort studies found a small decreased risk in breast cancer for women in the highest compared to those in the lowest categories of prudent/healthy dietary patterns.<sup>202</sup> No association was found for Western/unhealthy dietary patterns when both case-controls and cohorts were included, although a significant and modest increased risk was seen when only case-control studies were use; these however are prone to recall and selection bias.<sup>202</sup> The Western unhealthy dietary patterns were characterised by high intake of red and/ or processed meat, refined grains, potatoes, sweets and high-fat dairy. Conversely the prudent/healthy dietary patterns was typified by high intake of fruit, vegetables, poultry, fish low-fat dairy and whole-grains.<sup>202</sup>

#### 3.6.1.2 Vegetarian compared to meat consumption dietary patterns

Compared to meat consumption diets, vegetarian diets are low in energy and protein and have been associated with lower levels of IGF-1;<sup>203</sup> increased levels of the latter have been positively associated with breast cancer risk as discussed in section 3.5.3.<sup>160</sup> Vegetarian diets also avoid the high temperature production of heterocyclic amines produced during the frying, grilling or barbecuing of muscle meats which are thought to increase cancer risk.<sup>204</sup> On the other hand, it is possible that nitrates and nitrites found in vegetables may be converted in the body into carcinogenic nitrosamines compounds.<sup>205</sup> These are also formed in meat during curing.<sup>205</sup>

There is evidence that Mediterranean diets are anti-inflammatory;<sup>206</sup> and as mentioned in section 3.5.5 this may influence breast cancer risk. In particular, there is some evidence that magnesium, dietary fibre, omega-3, polyunsaturated fatty acids, monounsaturated fatty acids, flavonoids, and carotenoids from food are associated with decreased levels of inflammatory markers in serum.<sup>206</sup> Additionally, there is evidence from cross-sectional studies that vitamin C is associated with the anti-inflammatory marker C-reactive protein, however there are conflicting reports from RCTs.<sup>7172</sup>

A dietary pattern high in consumption of raw vegetables and olive oil, determined via factor analysis in the Italian ORDET cohort, was found to be protective against breast

cancer, particularly relating to HER-2 positive beast cancers or lean women of BMI below 25 kg/m<sup>2,207</sup> Although vegetables are major components of the Mediterranean and prudent/health dietary patterns which are inversely associated with breast cancer, there was no evidence in the EPIC-Oxford cohort that a vegetarian diet free of meat or fish reduces the risk of breast cancer in UK women.<sup>208</sup> The non-vegetarian comparison group in this cohort also had relatively high intakes of fruit and vegetables and this may have reduced the ability to find a significant association. There was also no evidence of a significant association between breast cancer risk and vegetarians in the fully adjusted model for pre or post-menopausal women in a UKWCS analysis of dietary patterns, when comparing them to women who ate red meat at least once a week.<sup>209</sup> Vegetarians were defined pragmatically as women who consumed red meat, poultry or fish less than once a week. In contrast, an earlier UKWCS analysis using a stricter definition of vegetarians who ate no meat, reported that vegetarians had a lower breast cancer rate than women who had medium to high intakes of red or processed or total meat.<sup>82</sup> However in the analysis of pre-menopausal women, low meat consumers were found to have a 32% lower risk than vegetarians, even after adjusting for the different characteristics of these groups.<sup>82</sup> The authors suggest that the vegetarian group may consist of a larger proportion of women genetically at high risk of breast cancer who started a vegetarian diet in the hope that it may be protective.<sup>82</sup> Alternatively, vegetarians may lack dietary components that are protective, particularly if some foods are consumed in moderation. Dose-response relations relating to fruit and vegetable intake are mentioned in section 3.6.2.1 on food groups.

In the UKWCS, compared to women who ate red meat at least once a week, a dietary pattern of eating fish at least once a week (excluding the consumption of red or white meat once a week or more) was inversely associated with breast cancer incidence.<sup>209</sup> A reduced risk of 40% was found for post-menopausal women, but there were no associations relating to pre-menopausal women.

#### 3.6.2 Foods

On the whole the WCRF 2007 report on Food, Nutrition, Physical Activity and the Prevention of Cancer focused on food rather than dietary patterns or individual nutrients, and reasoned that micronutrients may be simply markers for particular foods which contain other constituents that are associated with cancer development and progression.<sup>7</sup> Despite extensive systematic literature reviews and meta-analyses limited conclusions were reached about the influence of the foods considered in the WCRF 2007 report (summarised in Appendix A) and in the 2008 continuous update for breast cancer; evidence was limited due mainly to conflicting results from different

studies.<sup>7 169</sup> Whilst retrospective case-control studies have produced evidence of protective effects for some foods, results from cohorts are often mixed. There was, however, convincing evidence that alcohol increased breast cancer risk in both preand post-menopausal women and suggestive evidence that fats increased risk in postmenopausal women. A review of prospective observational studies by Michels et al. (2007) relating to intake of fruit and vegetables, dairy products and macronutrients such as fats and carbohydrates found no consistent, strong or statistically significant evidence of an association between diet and breast cancer, other than through being overweight and gaining weight.<sup>11</sup> The evidence relating to breast cancer risk and the amount of fruit and vegetables and dairy products consumed are discussed in more detail here. Dietary constituents such as fats, carbohydrate, fibre and alcohol are discussed in the following macronutrient section.

#### 3.6.2.1 Fruit and vegetables

Theoretically, there are various mechanisms by which fruit and vegetables could reduce cancer risk. As low energy-dense foods, they may help to reduce weight gain and obesity, which as seen in section 3.5.5 has been linked to increased breast cancer risk. Additionally their fibre content may help to regulate and increase the excretion of endogenous oestrogen,<sup>210</sup> thereby reducing exposure to one of the main hormones associated with breast cancer development (section 3.5.3). A diverse intake of fruit and vegetables can also provide a variety of vitamins and mineral including carotenoids, folate, vitamin C, D, E and selenium. Other bioactive compounds such as phytochemicals found in fruit and vegetables are non-essential components for humans but provide colour, flavour and preservation. Many of these bioactive compounds are believed to protect against cancer, for instance via their antioxidant properties. Unfortunately, storage and cooking processes usually reduce the nutrient content of fruit and vegetables, particularly in the case of vitamin C which is destroyed by heating or exposure to air by chopping; additionally, it is lost into the cooking water.<sup>39</sup> However, processing can also increase the bioavailability of some nutrients for absorption within the intestine; pureeing and adding oil increases the bioavailability of fat soluble lycopene found in tomatoes,<sup>211</sup> and chopping some foods can release enzymes that help the formation of other nutrients. Growing and transport conditions in addition to the timing of harvesting can also alter the nutrient content of foods.

The 2007 WCRF report judged that there was sufficient evidence to show that high intakes of fruit and non-starchy vegetables probably reduce the risk of cancers of the digestive track, such as mouth, larynx, oesophagus and stomach.<sup>7</sup>. However, a recent EPIC analysis found only a very small inverse association between intake of total fruits and vegetables and total cancer risk.<sup>212</sup> Furthermore, there is no conclusive evidence

they are associated with breast cancer, whether this be grouped as total fruit, total vegetables or fruit and vegetables.<sup>7</sup> <sup>11-13</sup> Meta-analyses which included mainly retrospective case-control studies reported a reduction in breast cancer risk with increasing intake of total fruit and vegetables by as much as 25%.<sup>8 213</sup> However, results of a subsequent pooled analysis of cohort studies showed that fruit and vegetable consumption was not significantly associated with breast cancer risk.<sup>12</sup> The susceptibility of case-control studies to recall bias and selection bias may explain the discrepancy in results. Furthermore, the pooling of studies is superior to meta-analyses since it ensures a standardised approach to coding, grouping of intake categories, analysis and adjustment by confounders. Nevertheless, a limitation of both designs is that dietary measurement may vary systematically between studies, even if similar measurement tools are used. One aspect of measurement error for the FFQs is that recorded intake of fruit and vegetable tends to increase with increasing numbers of fruit and vegetable items on the FFQs. This was evident in the pooling of eight studies from Northern Europe and North America where fruit and vegetable items varied 4-fold between the studies.<sup>12</sup> A pooling of eight European cohort studies by EPIC attempted to overcome differences in FFQs by calibrating the FFQs used in all the studies with the same 24-hour recall instrument; furthermore the measurement errors of the tools were likely to be independent.<sup>13</sup> This analysis, however, also found no significant associations, although this may have been limited by the median follow-up of only 5.4 years.<sup>13</sup> A strength of this pooling project was the wide range of fruit and vegetable intakes found throughout Europe, which increased the power to detect a significant result.

The EPIC pooling project also examined specific vegetables in relation to breast cancer risk, but found no inverse associations.<sup>13</sup> Subgroups examined were: leafy vegetables, fruiting vegetables, root vegetables, cabbages, mushrooms and garlic and onions. Other groups of vegetables were examined in the pooling project of studies from Northern Europe and North America, but no associations were observed for green leafy vegetables, eight botanical groups and 17 specific fruit and vegetables.<sup>12</sup> Although preparation of brassica/cruciferous vegetables (broccoli, cauliflower and cabbage) produces isothiocyanates and indole-3-carbinol which may be anti-carcinogenic, evidence of a protective association between breast cancer and this type of vegetables has only been observed in case-control studies.<sup>214</sup>

Daily grapefruit intake, which is high in vitamin C content, was associated with a 30% increased risk of post-menopausal breast cancer in a US cohort study, where higher risks were observed for non-users of HRT <sup>215</sup> However this was not supported by

results of a second study.<sup>216</sup> Grapefruit, nevertheless, has been found to increase plasma oestrogen concentrations,<sup>217</sup> possibly through the inhibition of the CYP3A4 enzyme which is involved in the metabolism of oestrogen.

There are indications from cohort studies that the relationship between fruit and vegetable intake and breast cancer risk may be modified by hormonal receptor status. Fruit and vegetable intake, and healthy dietary indexes which recommend high intakes of fruit and vegetables amongst other items, have been inversely associated with risk of oestrogen receptor negative (ER-) breast cancers.<sup>200 218</sup> In a cohort study of African-American women significant associations between a prudent diet and specifically between frequent vegetable intake and oestrogen/progesterone receptor negative (ER-PR-) breast cancers have been observed.<sup>219 220</sup> There is also evidence that high consumption of raw vegetables can potentially reduce HER-2 positive beast cancers by up to 75%.<sup>207</sup>

#### 3.6.2.2 Dairy products

Dairy products have been associated with IGF-1 levels in children in the UK.<sup>221</sup> US milk used in dairy products is likely to have higher levels of IGF-1 since, unlike milk in the UK and Europe, it is contaminated with bovine growth hormones which increase levels of IGF-1, as well as being fortified with vitamin D.<sup>11</sup> As discussed in section 3.5.3, IGF is positively associated with breast cancer,<sup>160</sup> therefore associations between breast cancer and dairy products in US studies may differ from European studies. However, in one review no differences in results were found from cohort studies by region;<sup>11</sup> many studies showed an inverse association between dairy products or milk and breast cancer risk, some being significant.<sup>11 169</sup> The 2008 WCRF continuous report, however, judged that no conclusions could be reached on whether dairy products influenced breast cancer risk, or on whether vitamin D did so either.<sup>169</sup>

#### 3.6.3 Macronutrients and breast cancer risk

#### 3.6.3.1 Fats

Fat intake has been the focus of many studies. In the 2007 WCRF report and its detailed breast cancer systematic literature review, meta-analyses of case-control studies showed weak but significant increases in breast cancer risk with increasing total fat, saturated fat intake or mono-unsaturated fat intake.<sup>7</sup> Conversely, inconsistent results were found for cohort studies, though a meta-analysis of 4 cohort studies showed a weak increased risk for saturated fat in post-menopausal women.<sup>7</sup> Since then a large EPIC pooling study of over 300,000 women from European cohorts

reported a significant 13% increased risk of breast cancer with increased saturated fat intake.<sup>222</sup>

In comparisons of dietary tools used, an initial analysis of EPIC-Norfolk found a 22% significant increase in breast cancer risk per quintile increase in energy adjusted saturated fat intake using 7-day diary data, but not using FFQ data when analysing the same women.<sup>223</sup> Bingham et al. (2003) suggest that measurement error may be the reason why many studies using FFQs find no associations. <sup>223</sup> Similarly the nested case-control analysis of the Women's Health Initiative Study (WHI) found a significant positive association using 4-day diaries and a non-significant association using FFQs.<sup>224</sup> Results of a recent pooled analysis of additional UK studies using diary data, however, did not support these findings.<sup>225</sup> Although diary recordings may be more detailed, they are not without their limitations, the main one being their short-term episodic nature. Additionally, since they are very time consuming, participants may alter their diet in order to record less items, furthermore the quality of recording may reduce over the diary time period; nevertheless this is the same for both cases and controls.

An RCT using an intervention group target of reducing dietary fat intake by 20% and increasing fruit and vegetable intake to at least five portions, and grain intake to at least six portions, did not find a significant reduction in breast cancer incidence overall after an average follow-up of 8 years.<sup>226</sup> Although there was a significant difference in fat intake between the groups, very few in the intervention group achieved the target of reducing fat intake by 20%. However, women in the intervention group who were in the highest quartile for fat intake at baseline showed a 20% significantly lower risk of breast cancer compared to similar women in the non-intervention group.<sup>226</sup> The intervention group also had a lower risk of developing progesterone receptor negative breast cancer. A recent review of studies of intentional weight loss supports the hypothesis that intentional weight loss can substantially reduce cancer risk.<sup>227</sup> Weight loss is associated with decreased endogenous oestrogen levels and increased levels of serum levels of sex-hormone-binding globulin (SHBG),<sup>227</sup> which have been consistently linked to breast cancer risk in menopausal women (section 3.5.3). Weight loss is also linked to decreased inflammatory markers,<sup>227</sup> which as discussed in section 3.5.5, is another mechanisms by which body fatness may be positively associated with breast cancer risk.

#### 3.6.3.2 Carbohydrates

Carbohydrates as an energy source are available as complex polysaccharides such as starch found in breads, rice, peas, beans, root vegetables, potatoes and bananas, or as simple carbohydrates i.e. sugars which are found in many processed foods and drinks, and also in fruit and vegetables.

A meta-analysis of cohort studies in the 2007 WCRF report, produced evidence of borderline significance of a positive association between carbohydrate intake and breast cancer risk in post-menopausal women (pooled RR per 50g/d=1.09; 95% CI: 1.00, 1.18).<sup>7</sup> The four studies included in the 2008 WCRF/AICR continuous update report relating to energy from carbohydrates and breast cancer risk provided insufficient evidence of an association.<sup>169</sup> Nevertheless, there was evidence of a positive association between percentage energy from carbohydrates with a high glycemic index (GI) and breast cancer risk in an Italian prospective study, not split by menopausal status.<sup>228</sup> Results for low GI food were not significant.<sup>228</sup> Glycemic Index (GI) is a measure of the effect of carbohydrates on plasma blood sugar levels, and glycemic load (GL) is the measure per 100g serving. Simple and refined carbohydrates, such as white bread, release sugar quickly and have a high GI, whereas whole grain products tend to have lower GI. Although GI and GL are positively associated with insulin levels, and insulin resistance and hyperinsulinemia has been positively associated with breast cancer incidence,<sup>163</sup> research into GI and GL have produced null or unexpected results with breast cancer incidence.<sup>11 229 230</sup>

#### 3.6.3.3 Fibre

Fibre, the indigestible content of foods such as non-starch-polysaccharide carbohydrates which are present in some fruit and vegetable, cereals and legumes may also influence risk, via possible affects on hormone levels.<sup>231</sup> However the weak inverse association with fibre found in the 2008 WCRF meta-analysis of cohort studies did not reach significance for post-menopausal women (pooled RR=0.96,95% CI: 0.91, 1.01).<sup>169</sup> Results for pre-menopausal women were also inconclusive in the WCRF continuous update.<sup>169</sup> However there was some evidence of a reduction in risk with higher fibre intakes in an earlier analysis of pre-menopausal women in the UKWCS (high versus low RR= 0.48, 95% CI: 0.24, 0.96); fibre from cereals and possibly fruit appeared protective.<sup>231</sup> A very recent meta-analysis of 10 cohort studies, which grouped together both pre- and post-menopausal women, reported a 11% reduction in breast cancer risk from total dietary fibre which was significant.<sup>232</sup> Evidence, however, was limited about whether specific food sources of fibre, i.e. cereal, fruit or vegetable, were protective.<sup>232</sup>

#### 3.6.3.4 Alcohol

The majority of cohort and case-control studies report a positive association between alcohol intake and breast cancer risk.<sup>7</sup> Meta-analyses and pooling studies estimate the risk to increase linearly by 7-10% per 10mg/d increase in ethanol intake.<sup>7 169</sup> This evidence was judged as convincing in the 2007 WCRF report and in the 2008 WCRF continuous update report for both pre- and post-menopausal breast cancers.<sup>7 169</sup> A few cohort studies have reported increased risk for hormone receptor positive tumours but not for ER-PR- breast cancer with increasing alcohol intake,<sup>233-235</sup> indicating that alcohol interferes with oestrogen metabolism. Additionally, it has been observed in some studies that high folate intake attenuates the association between alcohol intake and breast cancer risk.<sup>236</sup>

#### 3.6.4 Micronutrients and breast cancer risk

Initial research using retrospective recall methods in case-control studies indicated that fruit and vegetable intake probably reduced the risk of many cancers, including breast cancer.<sup>8</sup> Subsequently the identification of the active component in fruit and vegetables has been the focus of many studies. Micronutrients such as vitamins, minerals and antioxidants in these foods have been considered to be potentially protective against cancer, indeed the meta-analyses of retrospective studies indicated some micronutrients were protective.<sup>8</sup> Possible protective mechanisms of vitamin C and other antioxidants have been put forward, which have been discussed in section 2.2.4.

Furthermore, the 2007 second WCRF report judged there was convincing or probable evidence from prospective studies that some micronutrients were associated with specific cancers (summarised in Appendix B).<sup>7</sup> For instance, it reported that the antioxidant selenium was associated with a reduced risk of lung, stomach and prostate cancer,<sup>7</sup> although more recent results from the SELECT trial do not support the latter.<sup>237</sup> Conversely, there was important convincing evidence that supplementation with the antioxidant  $\beta$ -carotene increased the risk of lung cancer in smokers.<sup>238</sup> However, prospective studies assessing dietary and/or supplement intake in relation to breast cancer incidence have produced many null or conflicting results. The 2007 WCRF report judged that the evidence was limited or non-conclusive, therefore no conclusion could be reached regarding the intake of micronutrients and breast cancer risk.<sup>7</sup> The 2008 WCRF continuous update report for breast cancer which reviewed studies published to the end of 2008 also made the same judgement.<sup>169</sup>

In relation to antioxidants and breast cancer risk most prospective studies have reported no associations for carotene, vitamin A or vitamin E intake,<sup>239-246</sup> although the Nurses Health Study reported protective effects for vitamin A, vitamin E intake and  $\alpha$ -

and  $\beta$ -carotene for pre-menopausal women with a family history of breast cancer.<sup>169 247</sup> Similarly there was no overall convincing evidence of associations between the intake of vitamin C and breast cancer risk; these studies are discussed in sections 3.6.4.1 and 3.7.

In general, the 2007 WCRF report recommended that nutritional needs should be met through diet alone, when possible, rather than through supplementation. Furthermore, most studies published since the WCRF reported no reduced risk of cancer from micronutrient supplementation or reported evidence of harm.<sup>14 248-254</sup> A meta-analysis of four antioxidant RCTs published to October 2007 showed no association between antioxidant supplementation and breast cancer risk (RR=1.0; 95% CI: 0.90, 1.11).<sup>255</sup> A more recent factorial designed RCT based in the US using daily 500mg vitamin C, 600IU Vitamin E every other day and 50mg beta carotene every other day reported no effects of these on breast cancer incidence as well as on total cancer.<sup>14</sup> The French SU.VI.MAX trial used low dose antioxidant supplements (120mg vitamin C; 33iu vitamin E; 100µg selenium) and reported a reduced risk of total cancer in men but not women, which may be due to the lower baseline antioxidant status of men.<sup>256</sup> Breast cancer risk was not reported in this trial.

Most studies evaluating the relationship between supplement use and breast cancer risk have focused on individual supplements. Only a few studies mentioned in the 2007 WCRF review examined general supplement use in relation to breast cancer and these were case-control studies which may be prone to selection and recall bias.<sup>257-259</sup> No associations were found in the Danish and US studies reported in the 2007 review,<sup>257</sup> whereas a significant protective effect of supplements on breast cancer was found in a Taiwanese study (OR=0.40, 95% CI: 0.3, 0.7). Taiwanese women may generally have lower intake levels of nutrients than western women and therefore may require supplementation to reduce their risk.

A recent meta-analysis of five cohort studies found no evidence of a significant association between multivitamin use and breast cancer risk.<sup>260</sup> This included a Swedish study which reported an association between multivitamin use and increased risk of breast cancer.<sup>261</sup> The authors suggest that folic acid in the multivitamins may have promoted cancer, since there is no mandatory folic acid fortification of food in Sweden. Conversely, the US does fortify food with folic acid and the four US studies have not found an increased risk of breast cancer with multivitamin use.<sup>250 251 262 263</sup>

## 3.6.4.1 WCRF systematic literature review of vitamin C intake and breast cancer risk

Over thirty-five studies published to the beginning of 2006 relating to vitamin C intake and breast cancer risk were systematically reviewed in the 2007 WCRF report;<sup>7</sup> most were retrospective case-control studies, 12 were prospective cohort studies and 5 of these were nested case-control studies (the individual cohorts are detailed in Table 3 in section 3.7.2). The meta-analysis of 14 retrospective case-control studies of dietary vitamin C intake provided significant evidence of a protective association for breast cancer not split by menopausal status (RR=0.88, 95% CI: 0.84, 0.92 per 100mg/day), though heterogeneity was high ( $l^2 = 85\%$ ) between the studies. Meta-analyses of premenopausal (RR=0.90; 95% CI: 0.82, 0.98 per 100mg/day), and post-menopausal analyses (RR=0.86; 95% CI 0.79-0.93 per 100mg/day), supported this, as did the analyses comparing high verses low intake categories. In contrast, as seen later in Table 3 in section 3.7.2, cohort results were inconclusive; this was the case for analyses assessing dietary vitamin C only,239 240 242 243 245-247 264-267 supplement vitamin C only,<sup>241 242 245-247 267</sup> or total vitamin C intake.<sup>239 241 242 245 247</sup> In the 2007 WCRF report no significant associations were produced for dietary vitamin C in the post-menopausal women meta-analyses of three cohort studies (HR=1.15; 95% CI: 0.92, 1.43 per 100mg/d), <sup>7 245 246 265</sup> or in the high versus low intake analysis of two US studies.<sup>242 247</sup>

One study included in the WCRF cohort meta-analyses did find substantial evidence that breast cancer risk increased with increasing vitamin C intake in relation to dietary only (RR=2.06; 95%Cl 1.45, 2.91, per 100mg/d), supplementation only (RR=1.06; 95%Cl: 1.01,1.13) and total vitamin C intake (RR=1.08; 95%Cl: 1.02, 1.15).<sup>245</sup> This Danish nested case-control study is compared with the other studies later in this chapter.

A prospective nested case-control study included in the 2007 WCRF report assessed the relationship between vitamin C plasma levels and breast cancer and found no association.<sup>268</sup> Whether or not plasma levels are a better indicator of vitamin C intake than data from FFQs or diaries is discussed in section 5.4 in the evaluation chapter.

Differences in findings between study types may be due to recall bias in the retrospective case-control studies or to differences between assessment methods; FFQs were used in the cohort studies and interview-administered diet histories were used mainly in retrospective case-control studies. No relevant studies were published between the search for the 2007 WCRF report and that for the 2008 WCRF continuous update, although results are now clearly tabled in the latter.<sup>169</sup> The studies from the

WCRF reports are tabled and compared in section 3.7 of this thesis with the studies found from the new literature search undertaken in 2011 for this thesis.

# 3.7 Updated systematic literature review of vitamin C intake and risk of breast cancer

The purpose of this systematic review was to search for and evaluate cohort studies and RCTs relating to vitamin C intake and breast cancer incidence in women with no previous breast cancer. Only studies that were published since the beginning of 2006 until June 2011, and that were not included in the 2007 or 2008 WCRF, report were considered. New case-control studies were not evaluated since it was clear from previous meta-analyses that these types of studies show an inverse association between retrospectively reported vitamin C intake and decreased breast cancer risk; these findings may be due to recall bias.

## 3.7.1 Criteria for considering studies

#### **Types of studies**

Included: Cohorts, prospective nested case-control studies and RCTs. Reported in English.

Excluded: Retrospective case-control studies, cross-sectional studies

#### Types of participants

Included: women whose dietary or supplement vitamin C intake has been estimated, or who were part of an RCT vitamin C supplementation study and whose cancer status was known at time of dietary recording/intervention and at censor date. No age limit, any country

Excluded: women with previous breast cancer (i.e. cancer patients or survivors)

#### Types of exposure

Included: dietary or supplement vitamin C intake estimated using dietary assessment tools such as FFQs and diaries or intake provided in the intervention of a RCT supplementation study

#### Types of outcome

Included: incident invasive breast cancer

#### 3.7.1.1 Search strategy for identification of studies

OVID MEDLINE (Jan 2006-June 2011) was systematically searched for relevant articles using the key words which may appear in the title, subject headings or abstract

of articles. These key words were: *vitamin C; ascorbate; ascorbic acid; antioxidant* in conjunction with *vitamin; micronutrients* in conjunction with *fruit and vegetable; breast* in conjunction with *cancer, neoplasm, tumour, malignant, carcinoma and adenocarcinoma.* The search was restricted to extract articles relating to diet only studies using key words: *food, diet, intake, nutrient, supplement, nutrient surveys, dietary supplement, and dietary record.* It was also restricted to include only those articles published during and after the last year of the WCRF 2007 search i.e. from the beginning of 2006 to the beginning of June 2011. Search criteria were also applied to restrict the search to epidemiological studies using terms recommended by the BMJ.<sup>269</sup> The full search strategies can be found in 0.

The initial search criteria found 62 articles, six of which were duplicates. The title, subject headings and abstracts for all the articles were downloaded into a reference database and labelled according to their relevance. Eight articles at this stage appeared relevant, relating to the study of the association between breast cancer risk and vitamin C intake.<sup>244 270-276</sup> Five were case-control studies and were excluded,<sup>271 272</sup> <sup>274-276</sup> two of these were sub-analysed by genotype.<sup>272 276</sup> This left three cohort studies for the main review,<sup>244 270 273</sup> which are tabled in Table 2.

Articles citing or cited by relevant articles which were identified above or in the 2007 WCRF report were manually screened to determine whether they were also relevant. No further cohort studies were found; however, one RCT was identified which analysed vitamin C supplementation and breast cancer risk, though 'breast cancer' did not appear in the title or abstract.<sup>14</sup> Figure 6 shows the flow diagram of the studies included in this current review. A limitation of the search criteria was that articles with breast cancer', but not 'breast cancer' had been included in the title or abstract. Furthermore, articles that did not contain key words related to vitamin C in the title or abstract may have been missed. As described at the end of 0 some work was undertaken to check the latter, but no additional articles that were found met the full inclusion criteria.

Later, during a follow-up manual search for citations, a meta-analysis of vitamin C and breast cancer risk was found which was published online in July 2011. This review by Fulan et al. (2011), which had searched for publications to March 2011, found no studies in addition to those found in the search undertaken for this thesis.<sup>277</sup> Furthermore, it did not report finding one study (Roswall et al. 2010)<sup>273</sup> which was found in the current search. Nevertheless, since this recently published review had searched the PubMed, Embase and Cochrane databases, it was decided not to extend the current search to other databases.<sup>277</sup>

#### 3.7.2 Search results summary

The four recent studies identified in the current search relating to breast cancer risk and vitamin C intake are summarised in this section and shown in Table 2. In later sections these are compared with the studies that were included in the 2007 WCRF report and the 2008 WCRF continuous update which are shown in Table 3.<sup>7 169</sup>

Two out of the three cohort studies retrieved undertook separate analysis for dietary vitamin C intake, supplement intake and also diet plus supplement i.e. total vitamin C intake in postmenopausal women.<sup>270 273</sup> Both of these studies also sub-analysed by hormone-receptor status of the cancers, the results of which are discussed in section 3.7.3.1.3. One was an analysis by Cui et al. (2008) of the Women's Health Initiative study, a large US cohort following 84,805 women over an average of 7.6 years with 2879 cases which did find an increased risk of breast cancer with intake.<sup>270</sup> The other study by Roswall et al. (2010) followed 26,224 Danish women over a median of 10.6 years with 1072 cases, and did not find any evidence of an association with vitamin C intake.<sup>273</sup> A previous analysis of this Danish cohort by Nissen et al. (2003) found an increase in breast cancer risk with dietary, supplement and total vitamin C intake, however this was a nested case-control study and therefore was subject to potential selection bias of the controls, and incidences were followed-up for a shorter period.<sup>245</sup>

The third study retrieved was an EPIC pooled analysis of cohort data from 10 European countries following about 520,000 women over a median of 8.8 years resulting in 7502 cases.<sup>244</sup> Only dietary vitamin C was considered in this analysis. It was substantially larger than any previous study undertaken relating to breast cancer and vitamin C intake, and had sufficient power to sub-analyse by supplement use, alcohol consumption, hormone use and smoking status, as well as by menopausal status. To account for differences in questionnaire design and follow-up procedures between the cohorts, a stratum method of Cox proportional hazards was used. This did not find any significant associations in the main pre- and post-menopausal sub-analyses.

The RCT, the Women's Antioxidant Cardiovascular Study (WACS) mentioned earlier in section 3.6.4, was based in the US and had a factorial design with intervention groups using 500mg vitamin C daily, 600IU Vitamin E every other day and 50mg beta carotene every other day for 9.4 years. Women who were at high risk of cardiovascular disease, post-menopausal and over 40 were recruited. The study reported no effects of the supplements, either individually or together, on total cancer, including breast cancer incidence or mortality.<sup>14</sup>

Figure 6 Flow diagram of studies identified in current literature search



Author, year	Study	Design Baseline	Follow up & cancer	Meno- pausal	Vitamin C risk ratios	Additional sub-group
		date & age	incidence	status		analyses
Roswall	Diet, Cancer	Prospective	Median	Post	High vs. Low intake $1.11 (0.98, 1.40)$ p = 0.28	ER and PR status
$2010^{273}$		IN=20,224	10.0915		$1.11 (0.06 - 1.40) p_{trend} = 0.36$	HIStology
2010	Denmark	Age 50-64	1072 00365		Supplement $0.96 (0.77-1.21) p_{trend} = 0.41$	
Nagel et	EPIC	Prospective	Median		High vs. Low intake	Alcohol
al 2010 <sup>244</sup>	Europe wide	pooled cohorts	8.8yrs	Pre	Dietary $1.12 (0.92-1.46) p_{trend} = 0.37$	consumption,
	•	N=520,000	7502 cases		Non-supplement users	hormone use,
		1992-2000			1.15 (0.88-1.51) p <sub>trend</sub> = 0.69	smoking status
		Age 35-70				
				Post	Dietary $0.98 (0.87-1.11) p_{trend} = 0.79$	
					Non-supplement users	
	Momon's	Droopostivo		Deet	$0.92 (0.77-1.10) p_{trend} = 0.69$	ED and DD status
$2009^{270}$	Women's	cohort	AV 7.0915	Post	Total $1.18(1.04, 1.24)$ p $-0.000$	ER and PR status
2008	Initiative	N-84 805	2019 Cases		Dietary $1.06 (0.92-1.22) \text{ p}_{\text{trend}} = 0.009$	
	US	1993-1998			Supplement 1 16 (1 04-1 30) $p_{trend} = 0.03$	
	00	Age 50-79				
Lin et al	Women's	RČT	Av 9 yrs	Post	500mg/d supplement group	600 IU/d vitamin
2009 <sup>14</sup>	Antioxidant	Factorial design	135 cases		compared to placebo group	E
	Cardio-	N=7,627			1.11 (0.87–1.41)	50/2d β-carotene
	vascular Study	Age >=40				
	US					

**Table 2** Studies identified by a search for articles published after the 2007 WCRF report and 2008 WCRF continuous update

Author, year	Study	Design Baseline date & age	Follow up & cancer incidence	Meno- pausal status	Vitamin C risk ratios	Additional sub-group analyses
Cho et al. 2003 <sup>239</sup>	Nurses' Health Study II US	Prospective cohort N=90,655 1991-1999 Age BL 26-46	8yrs: 714 cases	Pre	High vs. low intake Total 0.98 (0.75-1.21) p <sub>trend</sub> =0.72 Diet 1.30 (1.00-1.69) p <sub>trend</sub> =0.16	Smoking status
Nissen et al. 2003 <sup>245</sup>	Danish Diet & Cancer Cohort	Prospective nested case- control 1993-1997 Age 50-64	Mean 4.7yrs: 418 Cases 394 control Both diet & sup: 228 Cases 246 controls	Post	Per 100mg/d (vit C sup users) Total 1.08 (1.02-1.15) Diet 2.06 (1.45-2.91) Sup 1.06 (1.01-1.13)	Non-vit C supplement users =1.54 (0.08-2.96)
Zhang et al. 1999 <sup>247</sup>	Nurses' Health Study US	Prospective cohort 1980 N=83,235 Age 33-60	14yrs: 2697 Cases Pre 784 Post 1913	Pre Post Not split	High vs. low intake Total 1.01 (0.81-1.26) p <sub>trend</sub> =0.59 Diet 1.01 (0.81-1.26) p <sub>trend</sub> =0.82 Total 0.99 (0.85-1.14) p <sub>trend</sub> =0.77 Diet 1.06 (0.91-1.22) p <sub>trend</sub> =0.57 Supplement >1300mg/d vs. none 1.04 (0.77-1.42)	HRT use, Family history of breast cancer, Alcohol
Kushi et al. 1996 <sup>242</sup>	Iowa Women's Health Study US	Prospective cohort 1986 N=34,387 N=18,910(sup) Age 55-69	Av 7yrs: 879 Cases (sup 570)	Post	High vs. low intake Total 0.88 (0.70-1.11) p <sub>trend</sub> =0.46 Diet 1.06 (0.77-1.47) p <sub>trend</sub> =0.88 Sup 0.77 (0.50-1.17) p <sub>trend</sub> =0.20	Supplement only
Graham et al. 1992 <sup>265</sup>	New York State Cohort US	Prospective cohort 1980 N=17401 Age 50-107	8yrs 344 Cases	Post	High vs. low intake Diet 0.81 (0.59-1.12) p <sub>trend</sub> =0.20	

**Table 3** The WCRF reported studies on vitamin C intake and breast cancer risk

Table continued	ممالم ببنهم المغبية مالم		ممسط لممرم مناميهم	
Table continued -	reported studies	on vitamin C	intake and brea	st cancer risk

Author,	Study	Design	Follow up &	Meno-	Vitamin C risk ratios	Additional
year		Baseline	cancer	pausal		sub-group analyses
		date & age	incidence	status		
Li et al. 2005 <sup>266</sup>	Shanghai Breast	Prospective Nested case-	Cases 130 (who also had	Not split	High vs. low Diet 0.8 (0.2 - 2.6) Dtrond=0.6	
	examination	control	proliferative			
	Trial	1995-2001	breast conditions)			
		Age 35-60+	Controls 1070			
Horn-	California	Prospective	2 years	Not split	High vs. low	
Ross et	Teachers Study	Lonort 1005-1006	Cases 711		Diet 1.1 (0.8 - 1.3) $p_{trend}=0.5$	
2002 <sup>240</sup>		Age 21-103				
Michels	Swedish	Prospective	130 months	Not split	High vs. low	BMI: _>25 vs. ≤25
et al.	Mammography	Cohort	Cases 1271		Diet 0.94 (0.78 -1.14) p <sub>trend</sub> =0.99	kgm <sup>-2</sup> & high
2001	Cohort	Age 40-76				linoleic acid intake
Verho-	The Netherlands	Prospective	4.3 years	Not split	High vs. low	Low/High
even et	Cohort	Nested case-	Cases 650	(but	Diet 0.77 (0.55 -1.08) p <sub>trend</sub> =0.08	polyunsaturated
al. 1997 <sup>246</sup>	Study on Diet	control	Controls 1066	likely	Vit C supplement yes vs. no	fatty acids
1001	and Cancer	1986 Ago 55-69		post)	1.06 (0.79-1.43)	
Rohan	Canadian	Prospective	6 vears	Not split	Diet per 119 mg/d	
et al.	National Breast	Nested case-	Cases 519		0.99 (0.83 -1.17)	
1993 <sup>267</sup>	Screening Study	control	Controls 1182		Supplement >250mg/d vs. none	
		1982			1.46 (1.05-2.01) p <sub>trend</sub> =0.98	
Giovann-	Nurses' Health	Age 40-59 Prospective	2 vears	Not split	High vs. low	
ucci, et	Study	Nested case-	Cases 392	Not opin	Diet 1.27 (0.82 -1.95) p <sub>trend</sub> =0.25	
al.	2	control	Controls 786		, , , , , , , , , , , , , , , , , , , ,	
1993		1986				
Liveter	Noneerlineer	Age 30-55	0	Nist and		
Hunter	Study	Cohort	o years Cases 1439	NOT SPIIT	Tign VS. IOW   Total 1.03 (0.87-1.21) p0.67	
1993 <sup>241</sup>	Cluby	1980	00363 1403		Supplement $>1300 \text{mg/d} \text{ vs. none}$	
		Age 34-59			1.12 (0.75-1.69) p <sub>trend</sub> =0.98	

Author, year	Exclusions	Hormonal factors	Dietary factors	Other factors controlled for
		controlled for	controlled for	
Roswall et al. 2010 <sup>273</sup>	Cancer diagnosis No lifetime menstruation Missing covariates and micronutrient intake	HRT Parity Age at first birth	Alcohol intake Total intake of vitamin E and B- carotene Dietary or supplement vitamin C intake mutually adjusted	Age (used as the underlying time scale for hazard ratio rather than adjusted for it as a covariate) BMI Education
Nagel et al. 2010 <sup>244</sup>	Missing non-dietary data Top & bottom 1% energy intake	Age at menarche Age at full-term birth Parity Hormone therapy	Energy from protein and carbohydrate Saturated fatty acids (SFA) MFA, PFA intake Alcohol intake	Age (1 year intervals) Centre (stratified) Weight Height Smoking status Physical activity Education
Cui et al. 2008 <sup>270</sup>	History of breast cancer Implausible total energy intake <600>5000kcal/d	Age at menarche Age at first live birth Parity Age at menopause Contraceptive use HRT use Hysterectomy Bilateral oophorectomy	Energy intake Alcohol intake Dietary folate intake Dietary vitamin C	Age Ethnicity Educational level Smoking Family history of breast cancer History of benign breast disease
Lin et al. 2009 <sup>14</sup>	Self-reported cancer diagnosis (other than skin) Unwilling to avoid vitamin A,C or E use	Randomised	Randomised	Randomised

**Table 4** Factors controlled for in the vitamin C studies published since the 2007 and 2008 WCRF reports

Author, year	Exclusions	Hormonal factors controlled for	Dietary factors controlled for	Other factors controlled for
Cho et al. 2003 <sup>239</sup>	Post menopausal. Cancer (other than skin) Implausible total energy intake Missing >70 items on FFQ	Age at menarche Oral contraceptive use Age at first birth Parity Menopausal status	Calorie intake Animal fat Alcohol	Age Height BMI Family history of breast cancer History of benign breast disease Smoking status
Nissen et al. 2003 <sup>245</sup>	Previous cancer diagnosis Pre-menopausal Missing items on FFQ Missing potential confounders on lifestyle questionnaire	Age at first birth Parity HRT duration	Intake of vitamin A & E (tested for total energy intake) Alcohol intake	Cases and Controls matched on age School education History of benign breast disease BMI
Zhang et al. 1999 <sup>247</sup>	Previous cancer diagnosis Implausible total energy intake <500 or >3500kcal/d Missing >10 FFQ blank	Age at menarche Age at first live birth Parity Age at menopause HRT use	Alcohol intake	Age Height Length of follow up BMI & BMI @18 yrs Weight change from 18yrs History of breast cancer in mother or sister
Kushi et al. 1996 <sup>242</sup>	Post menopausal. Full or partial breast removal Cancer (other than skin). Missing >30 items on FFQ Extreme total energy intake	Age at menarche Age at first live birth Parity Age at menopause	Energy intake Alcohol intake	Age Educational attainment BMI @ BL & 18yrs Waist/hip ratio Family history of breast cancer among first degree relatives
Graham et al. 1992 <sup>265</sup>	Previous cancer diagnosis			Age Education

**Table 5** Factors controlled for in the studies included in the 2007 and 2008 WCRF reports
Author, year	Exclusions	Hormonal factors	Dietary factors	Other factors controlled for
		controlled for	controlled for	
Li et al. 2005 <sup>266</sup>	Energy intake >4000 kcal/d		Fruit & vegetable intake	Age Matched on year of interview
Horn-Ross et al. 2002 <sup>240</sup>	Previous cancer diagnosis Energy intakes <600 or >5000 kcal/d	Age at menarche Age at first live birth Menopausal status	Energy intake	Age Race Family history of breast cancer Physical activity BMI
Michels et al. 2001 <sup>243</sup>	Previous cancer diagnosis (other than skin) Energy intakes <417 or >3729 kcal/d Missing or unreasonable covariate data	Parity Age at first live birth	Energy intake Intake of alcohol, fibre and mono-sat fatty acids	Age (5 year categories) Family history of breast cancer Height BMI Education
Verhoeven et al. 1997 <sup>246</sup>	Baseline self-reported cancer (other than skin) at baseline Incomplete FFQ (>60 items blank)	Age at menarche Age at menopause Age at first live birth Parity	Energy intake Alcohol intake Vitamin A	Age History of benign breast disease History of breast cancer in mother or sister
Rohan et al. 1993 <sup>267</sup>	History of breast cancer No mammogram in proceeding 12 months	Age at menarche Surgical menopause Age at first live birth	Energy intake	Matched on age (+/- 1yr, screening centre, date of enrolment) Age Years of education Family history of breast cancer History of benign breast disease
Giovannucci, et al. 1993 <sup>264</sup>	Energy intakes <500 or >4000 kcal/d			Matched on age
Hunter et al. 1993 <sup>241</sup>	Follow-up cases not confirmed by records	Age at menarche Age at first live birth Menopausal status	Energy intake Alcohol intake	Age (5 year categories) Length of follow-up History of breast cancer in mother or sister BMI History of benign breast disease

Table continued - Factors controlled for in the studies included in the 2007 and 2008 WCRF reports

#### 3.7.2.1 Quality of studies

The overall quality of the three cohort studies was good, with only some limitations. Quality was very good in relation to outcome measures which used methods employed by previous cohort studies: in two of the recent cohorts breast cancer incidence was provided from cancer registries;<sup>244</sup> <sup>273</sup> and in the other cohort and RCT, self-reported cases were independently assessed from medical records.<sup>270</sup> Follow-up for the RCT was 93% complete. The average length of follow-up was between 7.6 and 10.6 years which is comparable with the previous cohort studies, the longest being the analyses of the Nurses' Health Study (14 years). A limitation, however, is that these follow-up times may not be long enough to detect the development of cancer which can take decades.

Quality was good in terms of controlling for confounding factors: similar to the majority of previous studies the cohorts adjusted for many known hormonal and non-hormonal factors linked to breast cancer as discussed earlier in this chapter, and seen in Table 4. There was additional adjustment for some dietary factors though these varied between cohorts and from previous cohorts. All of these, and previous studies, adjusted for alcohol intake and the majority adjusted for energy intake though this was not done in the main analysis of the Danish study;<sup>273</sup> the previous nested case-control analysis of this cohort made adjustment for energy in sensitivity analysis.<sup>245</sup> (A discussion of reasons for controlling for energy intake can be found in section 4.7.5.3.) The Danish study also adjusted for other antioxidants examined. In the supplement-only analyses adjustment was made for dietary vitamin C in the Women's Heath Initiative and the Danish studies,<sup>245 270 273</sup> but other supplement-only analyses did not do this.<sup>242 247 267</sup> Despite the adjustment for a large number of factors, residual confounding is still likely. The RCT was superior in this respect since potential confounding factors were randomised between intervention and non intervention group. However, there is no information on whether steps were taken to avoid subversion by staff during randomisation. Furthermore, some women in the non-intervention group may have consumed vitamin C supplements since they are readily available, even though they would have been asked to avoid using them.

There were some weaknesses in relation to exposure which is common to all cohort studies in this research area: the vitamin C dietary and supplement intake of participants of all the cohorts may not be very representative of women in the populations they were recruited from. Women who were prepared to complete the FFQs (which in general comprised of between 100-200 items) were likely to be more health conscious that the average woman, and therefore more likely to have a higher intake of vitamin C. Since all cohorts used FFQs, the ascertainment of intake is

reasonably comparable between cohorts though there were some differences between types used. However, there are limitations with the self-reporting nature of FFQs, the general ability of the FFQ to record average intake and the conversion of food intake into nutrient intake, some of which are discussed further in chapter 5. In relation to exposure the RCT was limited to 500mg/d supplement intake, meaning it may be less able to detect the most effective dose than cohorts. Compliance, defined as having taken at least two thirds of the study supplements, was 76% at 4 years and 68% at 8 years (average73%). Outside use of the supplements for at least 4 days per month was not different between the supplement and the placebo group, which ranged from 2% to 13% at both 4 years and 8 years.<sup>14</sup>

The results from the four recent studies for breast cancer risk in relation to vitamin C intake are discussed separately in the following sections by dietary, supplement and total intake (Table 2) and are compared to results of studies that were included in the 2007 WCRF report and the 2008 WCRF continuous update (Table 3).<sup>7 169</sup>

# 3.7.3 Dietary vitamin C intake

#### 3.7.3.1 Post-menopausal women

All three of the recent cohort studies found no evidence of a significant association between dietary vitamin C intake and post-menopausal breast cancer risk.<sup>244 270 273</sup> As seen in Table 2 risk ratios for high versus low intake for all three studies showed weak but non-significant increases in risk. There was also no significant evidence of trends across the intake groups neither in the three studies, nor for the risk per 100mg/d increases calculated in two of the studies where risk ratios were close to unity.<sup>244 273</sup>

Similarly, three earlier analyses found no evidence of significant associations for postmenopausal women in the New York State cohort, the Iowa Women's Health Study and the Nurses' Health Study.<sup>242 247 265</sup> The analysis of women aged 55-69 in the Netherlands Cohort Study, likely to be of post-menopausal status, reported a small non-significant reduction in risk with increasing intake of dietary vitamin C (high versus lowest fifth OR= 0.77; 95% CI: 0.55, 1.08) and a significant trend in intake.<sup>246</sup>

In contrast, Nissen et al. (2003) in an earlier nested case-control analysis of the Danish Diet, Cancer and Health cohort (418 case and 394 controls) did find a significant doubling in risk per 100mg/d increase in intake for post-menopausal women (OR=2.06; 95% CI: 1.45, 2.91).<sup>245</sup> Unlike other studies mentioned, this was restricted to women who consumed vitamin C through both diet and supplements; women who did not consume vitamin C supplements were excluded.<sup>245</sup> This nested case-control study followed women for 4.7 years; a shorter period than the other studies. Unlike the other

studies in the WCRF meta-analysis, Nissen et al. (2003) adjusted for intake of the other antioxidants vitamin A and E. Other possible reasons for the difference between the results of two analyses of the Danish cohort could be due to selection bias of the controls in the earlier nested case-control study, or that both cases and controls may not have been representative of the whole cohort, or that the more recent analysis used age as the underlying time scale for the hazard ratio calculation rather than adjusted for it as a covariate.<sup>245 273</sup>

This risk ratio of 2.06 from the Danish study was used in the 2007 WCRF cohort metaanalysis of dietary vitamin C intake for postmenopausal women,<sup>7</sup> even though the other studies in the meta-analysis did not restrict their analysis to women who consumed vitamin C through both diet and supplements. Combined with the New York State Cohort and the Netherland Cohort,<sup>246 265</sup> a non-significant increase in risk per 100mg/d increase in dietary intake for post-menopausal women was produced in the metaanalysis (RR=1.15; 95% CI: 0.92, 1.43).<sup>7</sup> Updating the meta-analysis to include recent studies may produce results closer to unity; the risk ratio was close to unity for the EPIC pooling project, and since this was a very large study it would have a much greater weighting than the other cohorts. Indeed, the recent meta-analysis of dietary vitamin C by Fulan et al. (2011) produced non-significant odds ratios close to unity for post-menopausal women (RR=0.97; 95% CI: 0.81, 1.06). However this calculation included case-control studies as well as cohort studies.<sup>277</sup> Although it included the recent large EPIC pooling project and the Women's Health Initiative,<sup>244 270</sup> Fulan et al. (2011) had not included the Roswall et al. (2010) Danish study<sup>273</sup> in any of their metaanalyses.<sup>277</sup> Rather than continuous estimates, the Fulan et al. (2011) pooled high versus low results of the studies even though the cut off points for each study were different; this was an additional limitation of the meta-analyses.

#### 3.7.3.1.1 Non-supplement users

Supplement users in general have healthier behaviours than non-users,<sup>27</sup> therefore excluding all types of supplement users from analyses will reduce potential confounding and provide a clearer picture of associations between dietary vitamin C intake and risk. The exclusion of all supplement users in the EPIC pooling project, which had sufficient power to undertake sub-analyses, did not produce any evidence of associations and the hazard ratios did not change substantially in the pre or post-menopausal analyses.<sup>244</sup>

Excluding vitamin C supplement users in the Danish nested case-control study reported by Nissen et al. (2003) did reduce the association substantially from a doubling in risk which became non-significant (OR=1.54; 95% CI: 0.80, 2.96 per

100mg/day).<sup>245</sup> The more recent analysis of the Danish study, however, did not exclude supplement users from the analysis.<sup>273</sup> An additional analysis of the Iowa Women's Health Study excluded women who took antioxidant supplements A,C and E and this increased the point estimate in the high versus low calculation from a non-significant decrease (HR=0.88; 95% CI: 0.70, 1.11) to a non-significant increase in risk (HR=1.06; 95% CI: 0.77, 1.47). Unlike the EPIC study all supplement users were not excluded.

#### 3.7.3.1.2 HRT use

In the EPIC pooling project, post-menopausal women using exogenous hormones who were in the highest fifth of dietary vitamin C intake had a lower breast cancer risk than those in the lowest fifth (HR=0.88; 95% CI: 0.72, 1.07), though this was not significant. However there was evidence of a significant trend of decreased risk per 100/mg increases in intake, though the hazard ratio was close to unity and it was only marginally significant (HR=0.989; 95% CI 0.979, 0.999;  $p_{trend}$  0.05).<sup>244</sup> This does not provide strong evidence of a protective effect. No other study appears to have undertaken this sub-analysis

#### 3.7.3.1.3 Hormone receptor status

Both the Women's Health Initiative and the Danish Study undertook sub-analyses by hormone-receptor status of the cancers.<sup>270</sup> <sup>273</sup> They found no association between dietary vitamin C intake and breast cancer risk of ER+PR+, ER-PER-, ER+PR- or ER-Pr+ cancers. <sup>270</sup> <sup>273</sup> Results of analyses with supplement intake and total intake are found in sections 3.7.4 and 3.7.5.

#### 3.7.3.2 Pre-menopausal women

Only one of the recent studies analysed dietary vitamin C intake for pre-menopausal women which was the pooled EPIC study of 10 European cohorts; this found no evidence of a significant association.<sup>244</sup>

Results included in the 2007 WCRF report from the first and second phase of the Nurses' Health Study produced some conflicting results; the first phase analysis found no associations,<sup>247</sup> however the second phase reported significant evidence of an increased risk between high and low intake (mean 200 vs. 69mg/d; HR=1.30; 95% CI: 1.00, 1.69) but there was no significant trend, the highest risk being in the middle intake group.<sup>239</sup>

Pre-menopausal women with a family history of breast cancer in the analysis of the Nurse' Health Study reported by Zhang et al (1999) did have a substantially reduced risk (high vs. low intake HR=0.37; 95% CI: 0.17, 0.80), however the number of cases in this sub-analysis was low (90) which may have influenced the results.<sup>247</sup>

#### 3.7.3.2.1 Smoking status

The recent EPIC study and the Nurse' Health Study II were sub-analysed by smoking status.<sup>239 244</sup> Although Negal et al. (2010) mentioned that the results of their study suggests pre-menopausal former and never smokers have a higher risk of breast cancer at higher vitamin C intakes, neither study has produced strong evidence that smoking status modifies risk.<sup>239 244</sup>

#### 3.7.3.3 BMI

None of the recent studies examined whether BMI modified the relationship with dietary vitamin C and breast cancer incidence. However the RCT did assess this for total cancer incidence and cancer death in relation to supplement intake; this is mentioned below in section 3.7.4.<sup>14</sup>

As tabled in the 2008 WCRF continuous update report,<sup>169</sup> a Swedish study of 59,036 women attending Mammography screening produced evidence of an inverse association between breast cancer incidence and dietary vitamin C intake in women who were overweight (>25kg/m<sup>2</sup>) (high vs. low HR=0.61; 95% CI: 0.45, 0.82; p<sub>trend</sub> 0.004).<sup>243</sup> Conversely there was an increase in risk with increased intake for women of normal or lower BMI ( $\geq$ 25kg/m<sup>2</sup>) (high vs. low HR=1.27 95% CI: 0.99, 01.63; p<sub>trend</sub> 0.02). This analysis was not split by menopausal status, though the women were aged between 40-76 years old therefore the majority would have been post-menopausal at the time the FFQs were completed.<sup>243</sup>

### 3.7.3.4 Menopausal status not specified

None of the three recent cohorts report risk un-stratified by menopausal status. As listed in the 2008 WCRF continuous update for breast cancer, there were six previous cohorts which reported risk un-stratified by menopausal status. <sup>240</sup> <sup>243</sup> <sup>246</sup> <sup>264</sup> <sup>266</sup> <sup>267</sup> As seen in Table 3 none of these reported a significant association between dietary vitamin C intake and breast cancer risk. The recent meta-analyses by Fulan et al. (2011) for cohort studies which was un-stratified by menopausal status produced a relative risk close to unity (RR=1.01; 95% CI: 0.95, 1.08) comparing high versus low intake; the heterogeneity between cohorts was low.<sup>277</sup>

One study (not tabled) compared dietary vitamin C intake in adolescence with breast cancer incidence between the ages of 40-65, and found no evidence of an association. <sup>278</sup> This nested case-control analysis of the Nurse's Health Study gathered the vitamin C data retrospectively; possible recall bias, and also the short 24 item food questionnaire were limitations of the study.

# 3.7.4 Supplement only vitamin C intake

Mixed results were produced in the three recent studies (the Women's Health Initiative, the Danish Diet and Cancer Cohort and the Women's Antioxidant Cardio-vascular Study RCT),<sup>14 270 273</sup> as well as in previous studies that assessed supplement only vitamin C intake.<sup>242 245 247 267</sup>

There was no evidence that supplementation with 500mg/d vitamin C had an effect overall on breast cancer incidence in the US double-blind RCT of initially cancer free post-menopausal women over the age of 40.<sup>14</sup> Additionally, in sub-analysis by BMI no evidence of significant risks of breast cancer were found. However, the sub-group analysis of women with normal or low BMI (<25kg/m<sup>2</sup>) in the vitamin C supplementation group had a significantly increased risk of cancer death (RR=2.00; 95%CI: 1.12, 3.58), whereas there was no evidence of a significant association for overweight women (BMI  $\geq$ 25kg/dm<sup>2</sup>).<sup>14</sup> The author believes the results may be a chance finding due to small numbers involved (44 supplement group: 17 placebo group). An alternative explanation is that normal weight women may have a lower level of reactive oxygen species (ROS) than overweight women; ROS required for apoptosis of damaged cells may be suppressed by antioxidants in normal women.

Recently Cui et al. (2008) reported a significant trend ( $p_{trend} = 0.03$ ) of increasing risk with increasing vitamin C intake from supplements for post-menopausal women in the Women's Health Initiative; there was also significant evidence of a weak positive association in the high versus low intake calculation (HR=1.16; 95% CI: 1.04, 1.30).<sup>270</sup> Conversely, in the recent Danish analysis Roswall et al. (2010) reported no evidence of associations for post-menopausal women.<sup>273</sup>

However, in the Danish nested case-control analysis, Nissen et al. (2003) did report a significant but weak increased risk with increasing vitamin C intake from supplements (OR=1.06; 95% CI: 1.01,1.13).<sup>245</sup> Kushi et al. (1996) however, reported no significant associations in the Iowa Women's Health Study.<sup>242</sup> Neither was there evidence of an association in the Nurses' Health Study for pre-menopausal, post-menopausal or total women.<sup>241 247</sup> Information about the doses examined in these studies can be found later, in chapter 8. Only the Danish nested case-control study and the Iowa Women's Health Study were included in the 2007 WCRF meta-analysis per 100mg/d increase in supplement vitamin C which produced a risk ratio close to unity (RR=0.99; 95% CI: 0.98, 1.01).<sup>7</sup>

Rohan et al. (1993) reported evidence of a moderate increase in breast cancer risk in the Canadian Breast Screening Study for women consuming more than 250mg/d (OR=1.45: 95%CI: 1.05, 2.01) compared to women with no vitamin C intake from

supplements.<sup>267</sup> Results split by post-menopausal and other women were not statistically significantly different.<sup>267</sup>

In comparisons between total vitamin C users and non-vitamin C users no associations with breast cancer risk were found in the Netherland Cohort Study.<sup>246</sup> Duration of use was also not associated with risk in the Nurses' Health Study.<sup>247</sup>

Sub-analyses by hormone-receptor status for the recent Danish study and the Women's Health Initiative produced no evidence of significant associations for vitamin C intake from supplements. <sup>270 273</sup> No other studies have examined this relationship by hormone-receptor status.

The results of the above studies are detailed further by dose category in chapter 8 which determines whether breast cancer incidence in the UKWCS is associated with consumption of supplements containing vitamin C.

# 3.7.5 Total vitamin C intake

Two of the recent studies assessed total vitamin C intake.<sup>270 273</sup> A significant trend ( $p_{trend} = 0.009$ ) of increasing risk by increasing total vitamin C intake was found in the Women's Health Initiative for post-menopausal women; there was also evidence between high versus low intake of a significant weak positive association (HR=1.18; 95% CI: 1.04,1.34).<sup>270</sup> However, in the Danish study there was no significant trend ( $p_{trend} = 0.38$ ) and although the high versus low intake risk ratio showed a weak increase in risk this was not significant (HR=11; 95% CI: 0.88-1.40).<sup>273</sup>

Nissen et al. (2003) in the earlier nested case-control analysis of the Danish cohort found significant evidence of a moderate increase in risk comparing high versus low intake (OR=1.69: 95% CI: 1.12, 2.57) and also a weak increase in risk per 100mg/d increase in intake for post-menopausal women (OR=1.09: 95% CI: 1.07, 1.39).<sup>245</sup>

The Women's Health Initiative <sup>270</sup> was the most elderly cohort (average 64 years, oldest 79) of all the previous cohorts assessing total intake <sup>239</sup> <sup>242</sup> <sup>245</sup> <sup>247</sup>; the positive associations could indicate that high intake of vitamin C may promote the progression of cancer in older people or at later stages of the disease. In contrast, the Nurses' Health Study, an equally large study, which had the longest follow-up (average 14 years) found no association with total vitamin C intake for women below the age of 60.<sup>247</sup>

# 3.7.6 Conclusion

The three recent cohorts and one RCT published since the 2007 WCRF report<sup>7</sup> produced mixed results; though on the whole they showed a tendency towards a weak increase in breast cancer risk with increased vitamin C intake, this was not statistically significant. These results, together with those from previous cohort studies detailed in the systematic literature review of the 2007 WCRF report and the 2008 WCRF continuous update report,<sup>169</sup> do not provide substantial evidence of associations between breast cancer risk and vitamin C intake. This is the case whether the relationship is examined as dietary intake, supplement intake or total intake.

# 3.7.6.1 How the current thesis may address areas where further research is needed

As seen from section 3.7.4 of this review, mixed results have been produced in analyses relating to vitamin C intake from supplements and breast cancer risk. The UKWCS phase 2 data provides an opportunity to assess the association between breast cancer incidence and vitamin C intake from supplements for this thesis. In chapter 8 this is assessed in relation to the European Recommended Daily Allowance (60mg/d) and high dose (500mg/d) use, as well as by continuous intake. Characteristics of women that predict vitamin C supplement use at phase 2 in the UKWCS are described first and an analysis determines whether women who have history of breast cancer are more likely to take high dose vitamin C supplements (Chapter 6).

As mentioned in section 3.7.3.1.1, previously very few studies have excluded supplement users from the dietary vitamin C analyses; these women generally have different health behaviours from non-supplement users, so their exclusion reduces confounding. The baseline phase of the UKWCS provides an opportunity within this thesis to sub-analyse dietary vitamin C intake derived from FFQ recordings by supplement users (58% of women) and non-supplement users (42%).

Other sub-analyses may be enlightening. Few of the previous cohort studies have subanalysed by pre-menopausal status, one reason being there are less pre-menopausal than post-menopausal women. This thesis produces results by pre-menopausal status; however the power to find an effect may be limited in some analyses. Sub-analysis by hormone-receptor status could not be done since this information was not available for the UKWCS. Even though there is evidence that smoking status is associated with plasma C vitamin concentrations,<sup>48</sup> so far only the Nurses' Health Study II and the large EPIC pooling project have sub-analysed the relationship with dietary vitamin C by smoking status.<sup>239 244</sup> However, since only 11% were current smokers in UKWCS, there was limited power for sub-analysis, and this was not undertaken. HRT sub analysis has been undertaken in only one previous study,<sup>244</sup> and BMI has been subanalysed in two studies.<sup>14 243</sup> Similar sub-analyses have not been undertaken in the UKWCS for this thesis, but with accumulating breast cancer cases there may be sufficient numbers in the future for these sub-analyses.

As mentioned in section 3.7.2.1, and discussed further in chapter 5, there are some limitations with the general ability of FFQs to record average vitamin C intake which have been used by the three recent cohort studies and previous cohorts. The UK Dietary Cohort Consortium, described in the methods chapter, provides an opportunity to assess vitamin C intake in relation to breast cancer risk using an alternative measurement method: diary data. It also provides an opportunity to assess total intake which has only be analysed previously in 5 cohorts (by 7 analyses). The current analysis and results of this are discussed in chapter 10 of this thesis.

The next chapter describes the methods relating to the UKWCS and the UK Dietary Cohort Consortium, and also describes the cleaning of vitamin C supplement data and discusses the choice of covariates used.

# **CHAPTER 4**

# 4 Methods

# 4.1 Study design, study populations and datasets

All the analyses in this thesis used pre-gathered data from some of the largest population-based prospective studies in the UK which were designed to assess associations between diet and chronic diseases. Figure 1 in chapter 1 illustrates the datasets and the dietary tools used in the analyses. The majority of these analyses used previously unexploited vitamin C supplement data, results of which are reported in chapters 6, 8 and 10. Data from the UK Women's Cohort Study (UKWCS) was used in all the analyses, and was part of the consortium of UK nested case-control studies described below in section 4.1.2 and chapter 10.

For the first time these datasets have been used to explore breast cancer risk in relation to:

- any supplement use, and also dietary vitamin C intake from FFQs (split by supplement users and non-users) at baseline in the UKWCS
- vitamin C contained in supplements at phase 2 of the UKWCS
- total vitamin C intake (from diet and supplements) from diary records in the UK Dietary Cohort Consortium

The phase 2 UKWCS data also provided an opportunity to determine whether women who have a history of breast cancer were more likely to take high dose vitamin C supplements. Note that total vitamin C intake could not be assessed using the full UKWCS cohort because both dietary and supplement vitamin C intake had not been captured electronically at either baseline or phase 2. Instead, pooling of nested case-control data was required.

A major strength of these studies is that their prospective nature minimised recall bias and responder bias, which can affect results of retrospective case-control studies. Selection bias, however, may have been present in the prospective nested casecontrol studies, though this may have been minimal.

# 4.1.1 The UK Women's Cohort study (UKWCS)

#### 4.1.1.1 Baseline dataset

The UK Women's Cohort Study was formed from about 500,000 responders to a direct mail survey from the World Cancer Research Fund (WCRF), targeted towards women living in England, Wales and Scotland.<sup>279</sup> The overall response rate to the initial mailing was 17%, and 75% of these were willing to participate further.<sup>279</sup> Sixty one thousand women aged between 35-69 years old were then invited to take part in the study. Since the cohort was originally designed to compare disease incidence in vegetarians, fisheaters and red meat eaters, all eligible women who stated they were vegetarians or non-red meat eaters were asked to take part, but only a portion of the red meat eating majority were invited to do so. For each vegetarian, the next non-vegetarian in the list aged within 10 years of the vegetarian was selected for the cohort.<sup>279</sup> This structure provided a wide range of fruit and vegetable intakes, and as well vitamin C intakes, and helped to identify the effects of high consumption. The type of women recruited via the WCRF willing to complete long forms, in addition to the inclusion of a large proportion of vegetarians, meant that in general the cohort was likely to be more health conscious than UK women on average. This reduces the generalizability of the results.

In total 35,372 women (58% of those invited) provided data for the baseline; between 1995 and 1998 these women completed a 217-item Food Frequency Questionnaire (FFQ) and also additional demographic, health and lifestyle questions, totalling 24 pages (see 0).<sup>279</sup> The cohort was mainly white, well-educated, middle-class, middle-aged, married women.<sup>279</sup> Although 28% stated they were vegetarian, a more accurate but pragmatic definition classified 18% as vegetarian if they ate meat or fish less than once a week. At baseline 62% took some type of dietary supplement; however, the details of supplements taken were not captured electronically.

Baseline UKWCS dietary vitamin C data derived from the FFQ is evaluated in section 5.3. The risk of breast cancer risk in relation to any supplement use at baseline is reported in chapter 7 and in relation to dietary vitamin C intake at baseline is reported in chapter 8.

### 4.1.1.2 Phase 2 'follow up' dataset

All the initial participants were re-contacted between 1999 and 2004, on average four years after recruitment; 14,172 (40%) completed a follow-up health and lifestyle questionnaire similar to that used at baseline and 12,453 (35%) also completed a four day food diary.

Answers to questions on the health and lifestyle questionnaire relating to 17 different types of supplement were electronically captured for all phase 2 women (see section 4.5.2). More detailed supplement use was recorded in the 4-d diaries at phase 2 (see section 4.5.3). The cleaning of the vitamin C supplement data derived from supplement use detailed in the 4-d diary recordings is explained below in section 4.5.3.3. Phase 2 UKWCS supplement vitamin C data is evaluated in section 5.2 and the breast cancer risk in relation to supplement vitamin C intake at phase 2 is reported in chapter 8.

Unfortunately, the effects of dietary vitamin C intake could not be assessed for the whole cohort at phase 2 because FFQs were not included at phase 2 and in addition the dietary intake entered in diaries was only captured electronically for a small proportion of women in the nested case-control datasets (see section 4.1.1.3).

# 4.1.1.3 Phase 2 nested case-control dataset

Diaries provide an alternative method of measuring dietary intake; comparisons with FFQs relating to vitamin C intake are discussed in chapters 5 and 10. However, coding of diaries requires a large amount of resources. To reduce the amount of coding needed to examine the risk of breast cancer in relation to dietary intake recorded by diaries it had been necessary to create a nested case-control dataset within the phase 2 cohort. This initial UKWCS dataset comprised roughly 200 women from the total cohort women who had developed breast cancer after completing the diaries and these were matched to four or five controls. To increase the power of the analyses, the UKWCS nested case-control dataset, comprising 186 cases and 785 controls, was pooled the with four other UK nested case-control data by the Centre for Nutritional Epidemiology and Cancer Prevention (CNC) based in Cambridge to form the original breast cancer dataset of the UK Dietary Cohort Consortium. More information is given in section 4.1.2 about this consortium, and chapter 10 reports the results of breast cancer risk in this consortium in relation to dietary and total vitamin C intake recorded by diaries.

Note that, due to potential selection bias, 144 cases and 583 matched controls from the UKWCS, which were coded using DINER (Data into Nutrients for Epidemiological research), a system developed at Cambridge, were dropped from the dataset used for the published manuscript.<sup>2</sup> These controls were considered not to be a random selection since they were more likely to be meat-eaters and less likely to be vegetarians than the cohort as a whole. This did not change the overall results from those reported in chapter 10.

# 4.1.1.4 Laboratory analysed dataset

An additional 303 UKWCS women who completed phase 2 diaries and questionnaires also provided blood samples for laboratory analysis of micronutrient levels in a prior study.<sup>280</sup> This dataset was previously used to assess the impact of high non-starch polysaccharide intake on serum micronutrient concentrations.<sup>280</sup> Information on these women and the methods used to collect the plasma vitamin C data can be found in section 5.4.2.1 and 5.4.2.3. As described in section 5.4, this dataset was used within this thesis to compare plasma vitamin C concentrations with vitamin C intake derived from FFQs and with intake derived from diary data.

# 4.1.2 The UK Dietary Cohort Consortium

Breast cancer nested case-control data was pooled by the Centre for Nutritional Epidemiology and Cancer Prevention (CNC) at Cambridge to explore associations between breast cancer risk and dietary intake recorded by diary data from the five prospective UK cohorts described below. Information about the number of cases to controls, the mean time to diagnosis and the number of diary days for each cohort can be observed in Table 6. Data from all five cohorts were used in the analysis of dietary vitamin C intake and breast cancer risk. Only EPIC Oxford, EPIC Norfolk and UKWCS case-control datasets had detailed vitamin C intake from supplements and were pooled for the total vitamin C analyses. The risk of breast cancer risk in relation to dietary intake and total intake (supplement and dietary vitamin C) recorded by diary is reported in chapter 10.

# 4.1.2.1 Oxford arm of EPIC

A nested case-control group of women was taken from the EPIC Oxford study which was originally established to compare fish eaters, meat eaters, vegetarians and vegans.<sup>281</sup> In total, 65,429 men and women aged 20-79 years had been recruited during the 1990s using several methods: by direct mailing of the general public using GP listings in Oxfordshire, Buckinghamshire and Greater Manchester; by mailing members of The Vegetarian Society of the UK and also all surviving participants in the Oxford Vegetarian Study<sup>282</sup>; and by a small portion by GPs recruiting patients attending GP surgeries in Scotland. This resulted in a wide variation in intakes of major nutrients over the whole cohort.

### 4.1.2.2 Norfolk arm of EPIC

A nested case-control group of women was taken from the EPIC-Norfolk study which was made up of 23,003 men and women aged 45-75 years recruited between 1993 and1997 from people registered with 35 Norfolk GPs.<sup>283</sup>

Table 6 Characteristics of the five cohorts in the UK Dietary Cohort Consortium included in analyses of vitamin C and breast cand									
Cohort	Participants	Diary days	Years when food diary completed	Last follow up date	Mean time to diagnosis of cases	Cases	Controls	Mean(sd) dietary vit C intake	Mean(sd) total vit C intake
EPIC-Norfolk	General population in Norfolk	7 days	1993-1998	31.12.2006	6.0 yrs	365	1329	91 (50)	118 (167)
EPIC-Oxford	General population and vegetarians in the UK	7 days	1993-1998	31.12.2004	3.5 yrs	194	194	111 (61)	233 (436)
UK Women's Cohort Study (UKWCS)	Middle aged women in the UK	4 days	1999-2003	31.12.2006	2.4 yrs	186	785	117 (61)	239 (361)
Whitehall II	Civil servants in the UK	7 days	1991-1993	30.09.2005	7.8 yrs	70	275	101 (51)	_a 

31.12.2006 10.8 yrs

36

Table 6 Char ast cancer risk

<sup>a</sup>Whitehall and NSHD did not have detailed diary data of vitamin C intake from supplements

5 days

1989

Nationally reprehensive

cohort of women who were born in one week

in March 1946 in

Scotland.

England, Wales and

National

(NSHD)

Survey of

Health and Development а

66 (37)

144

# 4.1.2.3 The UKWCS

The UKWCS women who were included in the UK Dietary Cohort Consortium have previously been described in section 4.1.1.3.<sup>279</sup>

# 4.1.2.4 The Whitehall II Study

A nested case-control group of women was taken from the Whitehall II study which recruited 10,308 male and female civil servants aged 35–55 years working in the London offices of 20 Whitehall departments in 1985–88.<sup>284</sup> People from a wide range of grades and salaries were recruited, including clerical and office support staff, middle-ranking executives, and senior administrative grades. The cohort was originally established to determine the extent to which psychosocial factors at work and outside account for social class differences in mortality and morbidity.

# 4.1.2.5 The National Survey of Health and Development

A nested case-control group of women was taken from the National Survey of Health and Development (NSHD); consisting of a nationally representative cohort of women born during one week in March 1946 in England, Wales and Scotland.<sup>285</sup>

# 4.2 Ethical considerations

Ethical approval was granted at the initiation of the cohorts; 174 local research ethics committees were contacted and permission to carry out the baseline UKWCS study was obtained.<sup>279</sup> Participants had consented to the use of information gathered at baseline, future phases and from cancer registries for research purposes provided that confidentiality was maintained.

# 4.3 Measurement of cancer outcomes

The cancer outcome in the analyses was incident malignant breast cancers. These were identified by codes 174 and C50 of the 9<sup>th</sup> and 10<sup>th</sup> versions respectively of the International Statistical Classification of Diseases, Injuries and Causes of Death (ICD). Cancer diagnoses are registered under ICD codes by local cancer registries and collated by the National Health Service Central Register (NHSCR). Cancer and death registrations for women in UKWCS are extracted quarterly by the NHSCR and made available to the University of Leeds at least once a year. This information is then linked to the UKWCS identification codes and any personal identifying information is deleted before the rest is made available for data analysis. At this time the distribution of the cancer dates are checked to ensure that a typical number of cases have been recorded and that a large number are not missing, particularly in the most recent months.

Research showed that the NHSCR for England and Wales missed about 10% of all incident cases of malignant disease, relating to diagnosis in 1971–1989.<sup>286</sup> However, incident cases for the analyses in this thesis were gathered after this time, and may well be more accurate. Cancer registration is currently about 18 months or more behind cancer incidence; the latest cancer incident dates used for the time-to-event analysis in this thesis were July 2008.

# 4.3.1 Censor dates

The censor date (follow-up date) for the UKWCS baseline dataset used in this study was 01/01/2008. The median time to cancer incidence or time to censor date from the questionnaire was received was 11.2 years.

At phase 2 of the UKWCS the censor date for the dataset used was 01/07/2008. The median number of years follow up was 7.4 years from phase 2 diary dates.

The breast cancer censor dates and the median follow-up dates differ for each cohort in the UK Dietary Cohort Consortium; these are shown in Table 6.

# 4.4 Exclusions

In all time-to-event analyses women were excluded who had any registered malignant cancer (except for skin cancer) prevalent at FFQ date (for UKWCS baseline analyses) or at diary date (for UKWCS phase 2 and UK dietary Cohort Consortium analyses). Women diagnosed with breast cancer within 6 months after the FFQ or diary dates were also excluded. This was done to ensure that latent disease not formally diagnosed was not present; otherwise disease suspected by participants could have influenced their dietary habits.

Women with extremely high or low total energy intake (more than 6000kcal and less than 500kcal/d) were excluded in the baseline UKWCS dietary vitamin C time-to-event analyses.

# 4.5 Assessment tools used for recording dietary and supplement intake in the UKWCS

# 4.5.1 Baseline FFQ and Health and Lifestyle Questionnaire

At baseline the mean daily dietary vitamin C intake for all women who had completed a FFQ had been previously calculated from the list of simple and mixed food items on the FFQ using DANTE (Diet and Nutrition Tool for Evaluation), a Microsoft Access program developed by the University of Leeds Nutritional Epidemiology Group. The vitamin C

intake for each food item had been calculated by multiplying an assumed standard portion in grams by the number of portions consumed per day, then multiplying the product by the grams of vitamin C contained in 100g of the food item as listed in The Royal Society of Chemistry Food tables, version 5 and supplements in the 5<sup>th</sup> Edition of McCance and Widdowson.<sup>287</sup> This was then converted into mg/d. Standard portions had been calculated by averaging three data sources: portion sizes from a pilot study of the weighed food diaries of vegetarians and from published 1993 and 1994 national survey values.<sup>288</sup>

The baseline FFQ, shown in 0, was developed from one used by EPIC UK studies,<sup>289</sup> <sup>290</sup> by adding extra vegetable composition dishes to accommodate the higher proportion of vegetarians in the UKWCS to produce a 217 item FFQ <sup>279</sup>. Thirty one vegetables (excluding potatoes), 19 fresh fruit and five dried fruit items were listed in the FFQ. Nine of the fresh fruit were seasonal; no canned or frozen fruits were specifically listed. For each item participants chose from 10 pre-coded frequency of consumption categories ranging from never to 6 or more times per day, on average over the past year (an example is given in Figure 7). Vitamin C content was also derived from mixed food items listed including casseroles, lasagnes and curries. An evaluation of fruit and vegetable intake is given in section 5.3.3.1.

FOODS AND AMOUNTS	HOW OFTEN HAVE YOU EATEN THESE FOODS IN THE LAST 12 MONTHS?									
	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
SEASONAL FRUIT			120.00	1.2						
How often have you eaten these fruits, when they are in season?										
Apricots	0	1	2	3	4	5	6	7	8	9
Melon	0	1	2	3	4	5	6	7	8	9
Nectarines	0	1	2	3	4	5	6	7	8	9
Peaches	0	1	2	3	4	5	6	7	8	9
Plums	0	1	2	3	4	5	6	7	8	9
Raspberries	0	1	2	3	4	5	6	7	8	9
Red currants/Black currants	0	1	2	3	4	5	6	7	8	9
Rhubarb	0	1	2	3	4	5	6	7	8	9
Strawberries	0	1	2	3	4	5	6	7	8	9

#### Figure 7 A section from the baseline FFQ relating to intake of seasonal fruit

Please estimate how often you eat the following foods, and please answer every question.

# PLEASE PUT A TICK(✓) ON EVERY LINE

Whether women were supplement users or non-users at baseline had been electronically captured from yes/no responses to the question:

Q29 Do you take any vitamins, minerals, fish oils or other food supplements?

Women who did not answer this question but recorded information about the type of supplements taken on the questionnaire were also classed as supplement users. Full details of the supplements, however, were not captured electronically, thus vitamin C supplement use at baseline was not available.

# 4.5.2 Phase 2 Health and Lifestyle Questionnaire

Dietary vitamin C intake could not be calculated from the 20 page Health and Lifestyle questionnaire at phase 2 since it did not include a full FFQ.

General supplement use at phase 2 question could be ascertained from yes/no responses to the following questions:

Q14 Have you taken any vitamins, minerals, fish oils or fibre or other food supplements in the last year Q15 Do you presently use any dietary supplements?

General supplement use at phase 2 for the analyses in chapters 6 & 9 was determined using answers from Q15. Additionally, if participants did not answer yes to Q15 but provided further details of the type of supplements taken weekly or more frequently, then these women were designated as being current supplement users.

If participants answered yes to Q15 they were asked to note which supplements they took from a list of 17 and to indicate how often they used them. As seen in Figure 8 these included vitamin C, selenium, iron and antioxidants; although antioxidants, vitamin A and vitamin E were not listed separately. Frequencies of use were categorised as: more than daily; daily; weekly; monthly; and less than monthly. For each of the different types of supplements the percentages of women taking them daily were calculated, these percentages are displayed in Figure 2 in section 2.1.

Figure 8	Questions	relating to	sunnlements	on the nh	ase 2 questionnaire
i iguic o	Questions	rolating to	Juppionionio	on the ph	

5. Do you presently use dietary supplements         Yes       1       No       2         If yes, please indicate which supplements you use. Please look at the list of dietary supplements & ist in the box below how often you use them. See example       3       = Vitamin B12         I       Vitamin C       2       = Vitamin B complex       3       = Vitamin B12         I       Folic Acid       5       = Antioxidants       6       = Multivitamins with minerals         I       = Folic Acid       5       = Antioxidants       6       = Multivitamins with minerals         I       = Folic Acid       5       = Antioxidants       9       = tron         I0       = Calcium       11       = Selenium       12       = Zinc         I3       = Other Minerals       14       = Fish Oils (Cod / halibut liver oils)       15       = Evening Primose/Starflower Oils         I6       = Garlic supplements       I7       = Other supplements       Monthly       Less than once a monte once a day         I       1       2       3       4       5       =         I6       1       2       3       4       5       =         I6       1       2       3       4       5       = <tr< th=""><th><ul> <li>b. Do you presently use dietary supplements Yes 1 No 2</li> <li>yes, please indicate which supplements you use. Please look at to t in the box below how often you use them. See example</li> <li>= Vitamin C 2 = Vitamin B complex</li> <li>= Folic Acid 5 = Antioxidants</li> <li>= Multivitamins without minerals 8 = Other single vitamins</li> <li>D = Calcium 11 = Selenium</li> </ul></th><th>the list of dietary 3 = Vitamin B12 6 = Multivitamins 9 = Iron 12 = Zinc</th><th>v supplements &amp;</th></tr<>	<ul> <li>b. Do you presently use dietary supplements Yes 1 No 2</li> <li>yes, please indicate which supplements you use. Please look at to t in the box below how often you use them. See example</li> <li>= Vitamin C 2 = Vitamin B complex</li> <li>= Folic Acid 5 = Antioxidants</li> <li>= Multivitamins without minerals 8 = Other single vitamins</li> <li>D = Calcium 11 = Selenium</li> </ul>	the list of dietary 3 = Vitamin B12 6 = Multivitamins 9 = Iron 12 = Zinc	v supplements &
5. Do you presently use dietary supplements Yes $1$ No $2$ Yes, please indicate which supplements you use. Please look at the list of dietary supplements & st in the box below how often you use them. See example = Vitamin C 2 = Vitamin B complex 3 = Vitamin B12 = Folic Acid 5 = Antioxidants 6 = Multivitamins with minerals = Hultivitamins without minerals 8 = Other single vitamins 9 = Iron 0 = Calcium 11 = Selenium 12 = Zinc 13 = Other Minerals 14 = Fish Oils (Cod / halibut liver oils) 15 = Evening Primose/Starflower Oils 16 = Garlic supplements 17 = Other supplements Supplement number How often do you take the Dietary Supplement – please tick $\checkmark$ More than Daily Weekly Monthly Less than once a day e.g. 3 1 2 $\checkmark$ 3 4 5 1 2 3 4 5 1 3 2 3 4 5 1 3 2 3 4 5 1 3 2 3 4 5 1 4 5 1 5	Yes       1       No       2         yes, please indicate which supplements you use. Please look at 1         t in the box below how often you use them. See example         = Vitamin C       2         = Folic Acid       5         = Multivitamins without minerals       8         = Calcium       11	the list of dietary 3 = Vitamin B12 6 = Multivitamins 9 = Iron 12 = Zinc	v supplements &
Yes       No       "         yes, please indicate which supplements you use. Please look at the list of dietary supplements & st in the box below how often you use them. See example $3 = Vitamin B 12$ = Vitamin C $2 = Vitamin B complex$ $3 = Vitamin B 12$ = Folic Acid $5 = Antloxidants$ $6 = Multivitamins with minerals$ = Multivitamins without minerals $8 = Other single vitamins$ $9 = Iron$ $0 = Calcium$ $11 = Selenium$ $12 = Zinc$ $3 = Other Minerals$ $14 = Fish Oils (Cod / halibut liver oils)$ $15 = Evening Primrose/Starflower Oils$ $6 = Garlic supplements$ $17 = Other supplements$ Less than once a month once a day $monce a day$ $10 = 2$ $3$ $4$ $5$ $e.g. 3$ $1$ $2$ $3$ $4$ $5$ $1$ $2$ $3$ $4$ $5$ $1$ $2$ $3$ $4$ $5$ $1$ $2$ $3$ $4$ $5$ $1$ $2$ $3$ $4$ $5$	Yes       No         yes, please indicate which supplements you use. Please look at to to the box below how often you use them. See example         = Vitamin C       2         = Vitamin C       2         = Folic Acid       5         = Multivitamins without minerals       8         > = Calcium       11	the list of dietary 3 = Vitamin B12 6 = Multivitamins 9 = Iron 12 = Zinc	v supplements &
yes, please indicate which supplements you use. Please look at the list of dietary supplements & st in the box below how often you use them. See example         = Vitamin C       2       = Vitamin B complex       3       = Vitamin B12         = Folic Acid       5       = Antioxidants       6       = Multivitamins with minerals         = Multivitamins without minerals       8       = Other single vitamins       9       = Iron         0       = Calcium       11       = Selenium       12       = Zinc         3       = Other Minerals       14       = Fish Oils (Cod / halibut liver oils)       15       = Evening Primrose/Starflower Oils         16       = Garlic supplements       17       Other supplements       15       = Evening Primrose/Starflower Oils         16       = Garlic supplements       17       Other supplements       15       = Evening Primrose/Starflower Oils         16       = Garlic supplements       1       2       3       4       5         9       = Iron       1       2       3       4       5         1       2       3       4       5       5         1       2       3       4       5       5         1       2       3       4       5 <td>yes, please indicate which supplements you use. Please look at it in the box below how often you use them. See example         = Vitamin C       2         = Folic Acid       5         = Multivitamins without minerals       8         = Calcium       11</td> <td>the list of dietary 3 = Vitamin B12 6 = Multivitamins 9 = Iron 12 = Zinc</td> <td>v supplements &amp;</td>	yes, please indicate which supplements you use. Please look at it in the box below how often you use them. See example         = Vitamin C       2         = Folic Acid       5         = Multivitamins without minerals       8         = Calcium       11	the list of dietary 3 = Vitamin B12 6 = Multivitamins 9 = Iron 12 = Zinc	v supplements &
yes, hease inductor motor portuge in the box below how often you use them. See example         as in the box below how often you use them. See example         a Vitamin C       2 = Vitamin B complex       3 = Vitamin B12         a Folic Acid       5 = Antioxidants       6 = Multivitamins with minerals         a Folic Acid       5 = Antioxidants       8 = Other single vitamins       9 = Iron         a Multivitamins without minerals       8 = Other single vitamins       9 = Iron         a Calcium       11 = Selenium       12 = Zinc         3 = Other Minerals       14 = Fish Oils (Cod / halibut liver oils)       15 = Evening Primrose/Starflower Oils         6 = Garlic supplements       17 = Other supplements       15 = Evening Primrose/Starflower Oils         6 = Garlic supplements       17 = Other supplements       velockly       Monthly       Less than once a month once a day         e.g. 3       1       2       3       4       5       1         1       2       3       4       5       1       1       2       3       4       5         1       2       3       4       5       1       1       2       3       4       5       1         e.g. 3       1       2       3       4       5       1	yes, please indicate inforceppendix tin the box below how often you use them. See example         = Vitamin C       2         = Folic Acid       5         = Multivitamins without minerals       8         = Calcium       11	<ul> <li>3 = Vitamin B12</li> <li>6 = Multivitamins</li> <li>9 = Iron</li> <li>12 = Zinc</li> </ul>	s with minerals
= Vitamin C2= Vitamin B complex3= Vitamin B12= Folic Acid5= Antioxidants6= Multivitamins with minerals= Multivitamins without minerals8= Other single vitamins9= Iron0= Calcium11= Selenium12= Zinc3= Other Minerals14= Fish Oils (Cod / halibut liver oils)15= Evening Primrose/Starflower Oils6= Garlic supplements17= Other supplements15= Evening Primrose/Starflower Oils6= Garlic supplementsDailyWeeklyMonthlyLess than once a month once a daye.g. 312345123451123451123451123451123451123451123451123451123451123451123451123451123451123451123451123451 <tr< td=""><td>= Vitamin C       2       = Vitamin B complex         = Folic Acid       5       = Antioxidants         = Multivitamins without minerals       8       = Other single vitamins         0 = Calcium       11 = Selenium</td><td><ul> <li>3 = Vitamin B12</li> <li>6 = Multivitamins</li> <li>9 = Iron</li> <li>12 = Zinc</li> </ul></td><td>s with minerals</td></tr<>	= Vitamin C       2       = Vitamin B complex         = Folic Acid       5       = Antioxidants         = Multivitamins without minerals       8       = Other single vitamins         0 = Calcium       11 = Selenium	<ul> <li>3 = Vitamin B12</li> <li>6 = Multivitamins</li> <li>9 = Iron</li> <li>12 = Zinc</li> </ul>	s with minerals
$=$ Vitamin C $2 = Vitamin B complex$ $3 = Vitamin B 12$ $=$ Folic Acid $5 = Antioxidants$ $6 = Multivitamins with minerals$ $=$ Multivitamins without minerals $8 = Other single vitamins$ $9 = Iron$ $0 = Calcium$ $11 = Selenium$ $12 = Zinc$ $3 = Other Minerals$ $14 = Fish Oils (Cod / halibut liver oils)$ $15 = Evening Primose/Starflower Oils$ $3 = Other Minerals$ $14 = Fish Oils (Cod / halibut liver oils)$ $15 = Evening Primose/Starflower Oils$ $6 = Garlic supplements$ $17 = Other supplements$ $15 = Evening Primose/Starflower Oils$ Supplement more a day $e.g. 3$ $1$ $2 \checkmark$ $1$ $2$ $3$ $4$ $5$ $1$ $2$ $3$ $4$ $5$ $1$ $1$ $2$ $3$ $4$ $1$ $2$ $3$ $4$ $1$ $2$ $3$ $4$ $1$ $2$ $3$ $4$	= Vitamin C     2     = Vitamin B complex       = Folic Acid     5     = Antioxidants       = Multivitamins without minerals     8     = Other single vitamins       0 = Calcium     11     = Selenium	<ul> <li>3 = Vitatility B12</li> <li>6 = Multivitamins</li> <li>9 = Iron</li> <li>12 = Zinc</li> </ul>	s with minerals
= Folic Acid5 = Antioxidants6 = Multivitamins with minerals= Multivitamins without minerals8 = Other single vitamins9 = Iron $a = Calcium$ 11 = Selenium12 = Zinc $a = Calcium$ 14 = Fish Oils (Cod / halibut liver oils)15 = Evening Primrose/Starflower Oils $a = Other Minerals$ 14 = Fish Oils (Cod / halibut liver oils)15 = Evening Primrose/Starflower Oils $a = Calcium$ 17 = Other supplements15 = Evening Primrose/Starflower Oils $a = Garlic supplements$ How often do you take the Dietary Supplement – please tick / More than once a dayMore than once a mont once a day $a = 0$ 12345 $a = 0$	= Folic Acid     5     = Antioxidants       = Multivitamins without minerals     8     = Other single vitamins       > = Calcium     11     = Selenium	<ul> <li>6 = Multivitamins</li> <li>9 = Iron</li> <li>12 = Zinc</li> </ul>	s with minerals
$= Multivitamins without minerals 8 = Other single vitamins 9 = Iron$ $= Calcium 11 = Selenium 12 = Zinc$ $3 = Other Minerals 14 = Fish Oils (Cod / halibut liver oils) 15 = Evening Primrose/Starflower Oils 6 = Garlic supplements 17 = Other supplements 17 = Other supplements \frac{Veekly}{Nore than} \frac{Valke the Uetary Supplement - please tick \cdot}{Nore take monoce a day} e.g. 3 \frac{1}{2} \sqrt{2} \sqrt{3} \frac{4}{2} \sqrt{5} \frac{5}{2} e.g. 3 \frac{1}{2} \sqrt{2} \sqrt{3} \frac{4}{2} \sqrt{5} \frac{5}{2} 11 = 2 \sqrt{3} \sqrt{4} \sqrt{5} \sqrt{5} 11 = 2 \sqrt{3} \sqrt{4} \sqrt{5}$	= Multivitamins without minerals 8 = Other single vitamins = Calcium 11 = Selenium	9 = Iron 12 = Zinc	
= Multivitamins without minerals8= Other single vitamins9= Iron $0 = Calcium$ 11 = Selenium12 = Zinc $3 = Other Minerals$ 14 = Fish Oils (Cod / halibut liver oils)15 = Evening Primrose/Starflower Oils $6 = Garlic supplements$ 17 = Other supplements15 = Evening Primrose/Starflower Oils $6 = Garlic supplements$ 17 = Other supplements15 = Evening Primrose/Starflower OilsSupplement numberHow often do you take the Dietary Supplement – please tick /More than once a dayDailyWeeklyMonthlyLess than once a monthe.g. 312345123451123451123451123451123451123451123451123451123451123451234512345123451234512345123451234512345123 <td>= Multivitamins without minerals 8 = Other single vitamins = Calcium 11 = Selenium</td> <td>9 = Iron 12 = Zinc</td> <td></td>	= Multivitamins without minerals 8 = Other single vitamins = Calcium 11 = Selenium	9 = Iron 12 = Zinc	
0 = Calcium $11 = Selenium$ $12 = Zinc$ $3 = Other Minerals$ $14 = Fish Oils (Cod / halibut liver oils)$ $15 = Evening Primrose/Starflower Oils$ $6 = Garlic supplements$ $17 = Other supplements$ $17 = Other supplements$ Move often do you take the Dietary Supplement – please tick /More than once a dayDailyWeeklyMonthlyLess than once a monthe.g. 312345	) = Calcium 11 = Selenium	12 = Zinc	
$a = CalcumH = Contenta = Other Minerals14 = Fish Oils (Cod / halibut liver oils)15 = Evening Primrose/Starflower Oils6 = Garlic supplements17 = Other supplements17 = Other supplementsSupplementMore than once a dayMore than once a daye.g. 312 \checkmark34512345123451234512345123451234512345123451234512345$	J = Calcium		
3 = Other Minerals14 = Fish Oils (Cod / halibut liver oils)15 = Evening Primrose/Starflower Oils6 = Garlic supplements17 = Other supplementsSupplement numberHow often do you take the Dietary Supplement – please tick / More than once a day0More than once a dayWeeklyMonthlyLess than once a mont once a monte.g. 31234512345123451234512345123451234512345123451234512345123451234512345			
17 = Other supplements         Bow often do you take the Dietary Supplement – please tick /         More than once a day       Daily       Weekly       Monthly       Less than once a mont once a mont         e.g. 3       1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5	3 = Other Minerals 14 = Fish Oils (Cod / halibut liver oils)	15 = Evening Prin	mrose/Starflower Oil:
17 = Other supplementsSupplement numberHow often do you take the Dietary Supplement – please tick / More than once a dayMore than once a dayDailyWeeklyMonthlyLess than once a mont once a monte.g. 31234512345123451234512345123451234512345123451234512345123451234512345			
Supplement numberHow often do you take the Dietary Supplement – please tick / More than once a dayMore the Dially Daily WeeklyMonthly Monthly 4Less than once a month 5e.g. 31234512345123451234512345123451234512345123451234512345123451234512345	6 = Garlic supplements 17 = Other supplements		
Supplement numberHow often do you take the Dietary Supplement – please tick /More than once a dayDailyWeeklyMonthlyLess than once a monthe.g. 3123451234561345661 <th>and the second second</th> <th>A LEAST</th> <th></th>	and the second	A LEAST	
numberMore than once a dayDailyWeeklyMonthlyLess than once a monthlye.g. 31 $2 \checkmark$ 34512345512345512345512345512345512345512345512345512345512345512345512345512345512345	Supplement How often do you take the Dietary Supple	ement – please	tick 🗸
More than once a day       Daily once at work once a month once a month         e.g. 3       1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5         1       2       3       4       5	number Doilu Wookly	Monthly	Less than
e.g. 31 $2 \checkmark$ 3451234512345123451234512345123451234512345123451234512345	More than Daily Weekly	Monthly	once a mont
e.g. 3     i     i       1     2     3     4       1     2     3     4       1     2     3     4       1     2     3     4       1     2     3     4       1     2     3     4       1     2     3     4       1     2     3     4       1     2     3     4       1     2     3     4       1     2     3     4		4	5
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	e.g. 3	4	5
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	1 2 3	4	5
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1 2 3	4	5
1     2     3     4     5       1     2     3     4     5       1     2     3     4     5       1     2     3     4     5		4	5
1     2     3     4     5       1     2     3     4     5       1     2     3     4     5			5
1 2 3 4 5 1 2 3 4 5	1 2 3	4	5
1 2 3 4 5	1 2 3	4	5
	1 2 3	4	5
1 2 3 4 5	1 2 3	4	5
2 3 4 5		4	5

# 4.5.3 Phase 2 Diary

# 4.5.3.1 Supplement intake captured from phase 2 diaries

At phase 2 over 13,000 participants completed food diaries for 4 days with a separate page for each day to record individual supplements taken: Brand; Name; Amount taken; Dosage (the example provided to participants is shown below in Figure 9). Between 1999 and 2004, upon return of the completed diaries, this information was captured onto a Microsoft Access database of supplements taken by participants, and was matched at entry via a drop-down menu against supplements listed in a supplement ingredient database. This enabled the allocation of ingredient amounts to the supplements taken.

After cleaning (as explained in section 4.5.3.3), the daily vitamin C intake from all supplement types was calculated for each participant, and was averaged across the total number of diary days that vitamin C was taken.

٦

Figure 9 Phase 2 diary supplement recording example provided to participants

Supplements			
Please list any today, giving a Please enclose this diary back	v vitamins, minera is much information e the packaging of to us.	ls or other food s on as possible. f the supplement	supplements taken ts when you return
Brand	Name	Amount and form of supplement	Strength or tick if empty packet is enclosed
1. Healthcrafts	Multivitamin with iron and calcium	(see empty conta 1 tablet	iner enclosed) ✓
2. Boots	Evening Primrose	1 capsule	1000mg per capsule
3.			
4.			
5.			
6.			

# 4.5.3.2 Supplement ingredient database

The supplement ingredient database is a Microsoft Access database created by the Nutritional Epidemiology department at the University of Leeds between about 1999 and 2004. In total there were 3,996 different marketed supplement types listed in the database and each was given a separate supplement identification code. This contained brand name, supplement descriptions, ingredient composition and units (normally milligrams (mgs) for vitamin C) which were obtained from product labels provided by participants, suppliers' websites or provided directly from manufacturers upon request. A small proportion of vitamin C supplements contained ascorbate buffered with metal irons such as magnesium, calcium, zinc or potassium; the milligrams for the whole compound had been recorded in the ingredient database and a conversion factor was later applied to establish the ascorbate content (this could be found online or calculated using relative atomic mass of the elements). No adjustment was made for 'slow release' or 'time release' vitamin C supplements and those containing bioflavonoids which are promoted as having increased bioavailability, although evidence for this is lacking. Additionally it was assumed that the bioavailability of vitamin C in tablets, capsules, powder and liquid form was the same.

The supplement ingredient database was inspected to determine the common types of supplements containing vitamin C in relation to the recommended daily allowance (EU RDA = 60mg/day<sup>53</sup>) and in relation to 500mg doses which are usually classed as high dose. As summarised in Table 7, a large proportion (46%) of supplements containing vitamin C up to the EU recommended daily allowance (EU RDA 60mg/day <sup>53</sup>) are branded as multivitamins in the UK. Supplements which include more than the EU RDA but less than 500mg/day are often named multivitamins (26%) or antioxidants (14%). Vitamin C supplements containing high doses of 500mg or more are usually single ingredient (68%), and only a small amount are name as antioxidants (6%) or multivitamins (1%).

Table 7 Pe	ercentage o	f supp	oleme	ents containing	vitamin C	in the	UKWCS	suppleme	ent
ingredient	database	that	are	multivitamins,	antioxida	ants o	r single	vitamin	С
supplemen	ts, categoriz	zed by	vitar	nin C content					

	Supplements categorised by vitamin C content					
	≤60mg/d	≥500mg/d				
Branded as	N=561	N=408	N=219			
Multi %(n)	46% (256)	26% (105)	1% (3)			
Antioxidants <sup>a</sup> %(n)	13% (72)	14% (56)	6% (13)			
Vitamin C (only) <sup>b</sup> %(n)	3% (16)	10% (40)	68% (150)			

<sup>a</sup>or selenium with vitamins ACE

<sup>b</sup> names indicated supplements contain vitamin C only

# 4.5.3.3 Cleaning vitamin C supplement data in the database of supplements taken by participants

Although details of supplements recorded by participants in the 4-day diaries had already been entered into the database, over 15% of all supplement entries had no ingredients or doses allocated to them because no supplement identification code had been selected during entry. From participants' descriptions it appeared that the majority of these supplements did not contain vitamin C. In addition, it was unknown what proportion of supplement identification code allocations had to be deduced at entry because participants did not fully describe supplements used, or how many of these may have been incorrect. (Note that EPIC-Norfolk reported only 11% of their participants' supplement descriptions were direct matches to their supplement ingredient database at the point of coding and a substantial amount of allocations (41%) required an assumption<sup>38</sup>).

Any supplements listed in the UKWCS database of supplements taken by participants that were likely to contain vitamin C were checked and cleaned. Entries which indicated supplements containing vitamin C were found by systematically searching the participant database for the following descriptions: 'vitamin C', 'vitamin', 'ascorb', 'anti', 'A C E', 'glucosamine', multi', 'meno', 'A-Z', '50+', 'gold'. The participant's description captured in the database was compared with the description of the supplement allocated from the ingredient database; if different these were compared with the entries in the physical diaries. Over 400 diaries (3% of the total diaries) were checked as part of the cleaning process.

New fields in the database were created for corrected doses of vitamin C, or corrected or new supplement identification codes. New codes 9001-9007 were created to input mean doses for seven types of commonly taken supplements likely to contain vitamin C when dose and/or brand information was missing (see Table 8). The mean doses were calculated over the whole database by averaging complete entries which had supplement descriptions and identification codes relevant to these common supplements, and these were weighted by the number of women taking these types. (Note that EPIC-Norfolk also created generic supplement identification codes using weighted means, though these were more numerous and more specific<sup>38</sup>).

After this initial cleaning, the allocated vitamin C amount per day per participant was calculated by the database manager using the Access databases. This was checked and then uploaded into STATA. This diary-gathered data was then compared with participants' questionnaire answers relating to frequency of vitamin C use. The diaries of women who had no vitamin C dose captured were checked, if questionnaire answers

indicated they took vitamin C supplements daily. This cross-checking of assessment methods identified a further 400+ diaries which were manually checked; this resulted in vitamin C supplement intake recorded in diaries being identified and captured for an extra 87 women. (Detailed results of further comparisons between the two assessment tools are given in section 5.2).

	Mean		
Supplement type	vitamin C	New	Observations
	dose	code	corrected
Selenium plus vitamins A, C and E	90mg	9001	48
Antioxidants	153mg	9002	19
Multivitamins	75mg	9003	71
Multivitamins and minerals	71mg	9004	62
Multivitamins and iron	57mg	9005	34
Vitamin C (no strength indicated)	609mg	9006	65
Vitamin C high strength (suggested by	743mg	9007	16
description*)			

 Table 8 Mean dose of vitamin C by supplement type in the UKWCS supplement databases

\*Note: 'high strength': those described as 'with bioflavonoids' or 'time released' or slow released', or if most of the suppliers products were 500mg or over.

A final check of 63 diaries (half a percent of the phase 2 dataset), randomly selected by computer and inline with the National Audit Office statistical sampling guidance,<sup>291</sup> indicated that 95% (+/-5%) of diaries have supplement usage correctly captured in STATA (both correct number of days and satisfactory doses). A 5% error has the potential to slightly reduce a true relative risk of 2 to an observed relative risk of 1.9.<sup>292</sup>

### 4.5.3.4 Dietary intake captured from phase 2 diaries

At phase 2 in the UKWCS, food and drink intake had also been recorded in the 4-d diaries. The front pages of the booklet detailed instructions on the information required, and included suggestions about using household measures such as tablespoons and cups if weight scales were not available. Participants were asked to describe food and drink consumed each day as shown in the example in Figure 10, and to list recipes and details of ready meals used and the portion consumed by the participant. Due to resource constraints only about a 1000 cases and controls in total were coded into the DANTE system developed at Leeds, or later into the DINER system developed at

Cambridge. These produced a dietary vitamin C intake per day per participant. DANTE had previously been validated against DINER using 100 diaries which showed excellent agreement; geometric mean energy intake from DANTE was just 2% lower (95% CI, 0%, 5%) than from DINER.<sup>293</sup>

In addition, dietary intake of micronutrients was also calculated by DANTE for 274 women who had plasma micronutrient levels laboratory analysed. Vitamin C intake derived from the UKWCS FFQ and also the phase 2 diaries are compared with plasma vitamin C concentrations in section 5.4

Figure 10 Phase 2 diary food and drink recording example provided to participants

Time of food or drink	Description of food or drink consumed (include brandname where possible)	amount
7.15 am	Filter Coffee	1 cup (200ml)
	semi-skimmed milk	3 tablespoons
7.30 am	Sainsbury's orange juice, un-sweetened	1 glass (150ml)
	Sainsbury's Bran flakes	40g
	semi-skimmed milk	180ml
10.30am	Plain chocolate digestives (large biscuits)	2
	Earl Grey tea (weak) no milk	1 cup
11 am	banana (medium sized)	95g
11.30 am	London herb company Lemon Zester tea	1
12.10 pm	Local bakery's wholemeal bread un-sliced loaf (cut thickly)	1 slice 47g
	Tesco sunflower margarine	thinly spread
	home-made mushroom risotto (see recipe)	About 1/3 of recipe
	green seedless grapes	32g
	Cox's Orange Pippin apple (medium)	82g
	Sainsbury's wholemilk fruit yoghurt (150g)	1 pot
	London herb company sweet berry tea	1
2pm	Warburton's Carrot cake – with cream cheese topping (see nutritional information)	1 slice – 75g (on packet)

# EXAMPLE DAY - UP TO LUNCH

# 4.6 Assessment tools for dietary and supplement intake in other cohorts of the UK Dietary Cohort Consortium

# 4.6.1 Dietary intake recorded by diaries

In the EPIC studies, the diaries allowed for the recording of the description, preparation and amount of foods eaten at main meals, snacks and between meal times over seven consecutive days. Detailed instructions on the information required in describing and quantifying each type of food or drink were printed on the front pages of the booklet.<sup>283</sup> Seventeen sets of colour photographs were included to help the participants describe the portion size of the food they consumed, as small, medium or large. The diaries were coded using DINER.

The Whitehall II study also used 7-d diaries, containing instructions and pages to record foods eaten during seven time periods (before breakfast, breakfast, mid-morning, lunch, tea, evening meal, later evening). Fifteen sets of black and white food photographs were included, showing three portion choices for a common food item.<sup>294</sup>

For the National Survey of Health and Development (NSHD) a 5–d diary was used. The instructions contained 66 examples of how to describe foods in terms of their preparation and quantity. The main part of the diary comprised of seven spaces for each day in which to record meals and between-meal snacks, a reminder section about any other snacks and drinks and a space in which to write recipes.<sup>295</sup> Fifteen sets of black and white food photographs were included, showing three portion choices for a common food item. The diaries were coded using DIDO.

# 4.6.2 Supplement use recordings in EPIC-Oxford and EPIC-Norfolk

Both cohorts used similar diaries to record supplement intake. Participants were asked to record their supplement use in a 7-day diary by answering an open-ended question, however this only provided room to record one supplement per day.

Please name any vitamins, minerals or other food supplements taken on each day of last week. Headings: Brand name; Amount taken In a later version of the diary more space was given for participants to list a number of supplements used and an example was provided.<sup>38</sup>

Please name any vitamins, minerals or other food supplements taken on each day of last week. Please write down all the details from each packet/container, and enclose label(s) giving ingredients and individual amounts where possible.

Headings: Brand; Name; Amount taken per day – number of pills, capsules or teaspoons; Tick box(es) to show which day(s) supplements was taken last week -M T W T F S S.

Using data from diary records and their supplement ingredient database, EPIC-Norfolk calculated the average vitamin C intake from supplements over the seven days of the diary for each participant in the nested case-control dataset. These averages were provided by Cambridge for pooling with the UKWCS and EPIC-Oxford.

The supplement use diary recordings for each participant in the EPIC-Oxford nested case-control dataset were photocopied and transferred to a spreadsheet. If details given did not state doses, then vitamin C amounts were allocated based on information from the UKWCS ingredient database. Although this database was established at the end of the period the Oxford diaries were completed, many of the supplements in the UKWCS ingredient database would have been available in this earlier period. The EPIC-Norfolk database was not available for use for this task.

# 4.7 Statistical methods

This section details the statistical methods used that are common to many of the following chapters containing results; for example time-to-event analysis, the selection of covariates for adjustment, coding of menopausal status, exclusion of outliers in sensitivity analysis, etc. Although conditional logistic regression is only used in one of the following chapters (10), it is appropriate to explain this here as an alternative to Cox's regression analysis.

Kappa statistics and Bland Altman plots are explained in chapter 5 which are used to evaluate diet and supplement vitamin C recordings in the UKWCS. In addition restricted cubic splines are explained in chapter 9 which are used to determine whether there are non-linear relationships between dietary vitamin C intake and breast cancer risk.

All statistical analyses used Stata SE version 10 and all tests calculated two sided p values and 95% confidence intervals.

# 4.7.1 Time-to-event analyses

Time-to-event analyses (also known as survival analyses) using Cox proportional hazards models were used in chapters 7 to 9 to determine whether dietary or supplement vitamin C intake were associated with breast cancer risk. The analyses measure the time from FFQ or diary completion date to breast cancer incidence or the date the participant was lost to follow-up because they died of other causes or, for women who did not develop breast cancer, the time to censor date (as detailed in section 4.3 for each dataset).<sup>296</sup> This enables the analysis to deal with recruitment which occurs at different times throughout the study. Therefore before running any risk analyses, a variable containing time to incidence or to censor date for each participant was created in Stata from FFQ completion dates for the baseline analyses, and for diary completion dates in the phase 2 analyses. Additionally a variable was required which flagged cancer incidence after the FFQ / dairy completion dates. As stated earlier women who were diagnosed with any cancer (except skin) before FFQ / diaries were completed, were excluded.

The proportion of women who did not have breast cancer at any time since recruitment can be plotted in a Kaplan-Meier curve. Two or more groups of women can be compared using this technique as shown in Figure 22 in chapter 7

This method produces an approximate estimate of a true time-to-event curve if both of the following assumptions hold:

- 1. Participants who are censored have the same risk of subsequently developing breast cancer than those who were not censored
- 2. The observations are independent of each other

Hazard ratios for time-to-event analyses in this thesis were calculated using Cox regression, also known as proportional hazards regression. Hazard ratios compare the hazard of one group with another group e.g. supplement users with non-users. In simple terms the hazard is the probability that an individual in the study with no prevalent breast cancer is diagnosed with breast cancer at any given time. The distribution of the hazard over the length of the study is called the hazard function. The hazard can change over the study period so to be more precise, the hazard function is the instantaneous risk (i.e. the probability over a small interval of time) of breast cancer for an individual given that the individual has been breast cancer free up to a particular time.<sup>297</sup> The analysis uses a conditional likelihood estimation procedure where at each point in time that an event occurs the value of the exposure variable, e.g. supplement use, for a participant who has incident breast cancer is compared to the value of exposure variable for all those without the disease.<sup>296</sup> Although the Cox regression

method makes no assumptions about the shape of the curve, the hazard ratio between the two groups should be a constant proportion over time. This proportional hazard assumption was tested using the cumulated hazard by plotting the minus log of minus log time-to-event against log of time. The curves should approximately be a similar distance apart all the way along or they should at least not cross. In addition there should be about ten cases per variable in the model.<sup>298</sup>

# 4.7.2 Conditional logistic regression analyses

The risk of developing breast cancer in the pooled nested case-control studies was calculated using conditional logistic regression. In this type of logistic regression method breast cancer cases are only compared to controls within the same matched set, making it conditional on matching.<sup>296</sup> Controls were randomly selected from women with matching variables who were still breast cancer free at the end of the study, representing the population at risk of the disease. They were matched within study centre by month of diary completion ( $\pm$  3 months or as close as possible), so that follow-up times were comparable, thereby dealing with the varying recruitment dates. They were also matched on age ( $\pm$  3 years). As an aid to this analysis each matched set had been numbered and this had been added to the dataset as a variable. It was also possible to control for variables other than those controlled through matching by adjustment in the model. Using between two and four controls for matching increased the likelihood that controls were representative of the cohorts.

Wacholder et al. (1992)<sup>299</sup> explained that overmatching, which can cause bias or reduce efficiency (the ability to demonstrate expose-outcome relationships) compared to an unmatched study, can occur in a number of situations if a matching variable is: 1) an intermediate on the causal pathway 2) is a surrogate for or a consequence of the outcome; or 3) is closely related to the exposure variable but not related to the outcome. It is unlikely that any of the matched variables would cause overmatching. For instance, age does not meet the first two points and, although age is related to breast cancer risk, as seen in the analysis of the nested case-control study in chapter 10 it is not closely related to dietary vitamin C intake.

# 4.7.3 Evaluating interactions

An interaction occurs between the effects of two exposures if the effect of one exposure on the outcome varies with the effect of the other. This is also called effect modification; for instance menopausal status has been found to modify the effect of an exposure on the incidence of breast cancer, examples of which can be found in the literature review chapter. Additionally, since supplement users are likely to have different health behaviours than non-users this variable was tested for interactions and split into strata in datasets when possible.

Other variables suggested as modifiers were defined *a priori*: BMI; amount of exercise; family history of cancer; and dietary group. Socio-economic status and HRT use was considered later. Chapter 7 tests whether these modify the effect of supplement use at baseline on breast cancer risk and analyses incorporating interaction terms were reported. However, these variables were not tested in relation to levels of vitamin C intake due to limited power and the possibility of spurious findings from multiple testing.

# 4.7.4 Coding of menopausal status

Menopausal status used in the time-to-event analyses relates to status at the time of exposure (dietary or supplement intake assessment) rather than time of diagnosis or censor date, due to difficulties of determining status later. This means that some women in the datasets who developed cancer when they were post-menopausal have been defined as pre-menopausal at baseline or phase 2 assessments. The questions used at baseline and at phase 2 to determine menopausal status are shown in Figure 11. The flow diagram in Figure 12 shows the rules used for coding menopausal status in the UKWCS for this thesis, which is similar to that used in the UK Dietary Cohort Consortium. For women who did not specify on the questionnaires whether they still had periods, it was assumed they were post-menopausal after the age of 50. The mean age (46 years) at which women were assumed to be post-menopausal after removal of ovaries was based on research by Farquhar 2005.<sup>300</sup> A peri-menopausal category was not used for the UKWCS analyses.

Figure 11 Questions used to determine menopausal status (phase 2 questionnaire)

50. Have you ever used oral contraceptives (the pill)?	Yes 1 No 2
If Yes, how old were you when you first used the pill?	years old
For how long altogether did you use the pill?	years
Are you currently using the pill?	Yes 1 No 2
If no, how old were you when you last used it?	years old
51. How many "natural" menstrual periods have you had Do not count bleeding while using the pill or HRT (Ho	in the last 12 months? In the Replacement Therapy)
None 1 1 to 3 2 4 to 5 3 6 to 9	10 or more 5
Not applicable (using the Pill or HRT or currently pregnant)	,
If none, how old were you when you had your last "natural" Do not count bleeding while using the pill or HRT (Hormon	" menstrual period? year e Replacement Therapy).
52. Have you ever used HRT (hormone replacement the	rapy for menopause)? Yes 1 No 2
If Yes, how old were you when you first used HRT?	years old
Are you currently using HRT?	Yes 1 No 2
If no, how old were you when you last used HRT?	years old
53. Have you ever had a hysterectomy (womb removed, Age at time of operation	)? Yes 1 No 2 years old
54. Have you had an operation to remove one or both your ovaries? Yes 1	No 2 Don't know 3
If yes, how old were you?	years old
Were one or both ovaries removed? One 1 Bo	oth <sup>2</sup> Don't know <sup>3</sup>

84



Figure 12 Flow diagram of coding for menopausal status in the UKWCS

# 4.7.5 Adjustments by potential confounders

Unlike RCTs, where behaviours and characteristics which may influence the risk of developing breast cancer are randomised between intervention groups, in cohort studies these potential confounders can seriously affect the results of risk analyses if they are not statistically controlled for. Therefore their selection was carefully considered for this thesis.

# 4.7.5.1 Model building

Potential confounders are variables that may be associated with both the exposure of interest i.e. vitamin C and the outcome variable i.e. breast cancer incidence. Univariate analyses were performed to examine the association of each potential confounder with breast cancer incidence and with vitamin C intake. Variables that were significantly associated with vitamin C as well as breast cancer incidence in the datasets were considered for adjustment. Those that did not meet this criteria but where there was strong prior evidence from previous studies, as detailed in section 2.2.3 and chapter 3, suggesting they were confounders, they were still considered for inclusion. These potential confounders were assessed by visual methods using diagrams called Directed Acyclic Graphs as detailed in section 4.7.5.2. Covariates which appear to be on the causal pathway between vitamin C intake and breast cancer should not be included, since these mediators are not true confounders and controlling for them would attenuate the effect of vitamin C.<sup>301</sup>

Confounders that had a substantial proportion of missing observations were not included in the main adjusted analyses (e.g. waistline had 7500 missing entries at baseline), particularly if other closely related variables were available e.g. BMI. For instance socio-economic status rather than education was controlled for at baseline since it had fewer missing observations. Important confounders which had a high proportion of missing variables in the UK Dietary Cohort Consortium analyses were considered for inclusion in sensitivity analyses i.e. cumulative breast feeding and age at menarche.

### 4.7.5.1.1 Assessing linearity with outcome

Continuous variables, such as age and BMI were tested to determine whether they had a linear or non-linear relationship with cancer incidence. As categorical variables, there were no assumptions about the shape of the relationship between the variable and the outcome. The null hypothesis that the relationship was linear, was tested using the Likelihood Ratio Test (LRT) by comparing the log likelihood from the model using the continuous variable with the log likelihood with the model using the variable split into categories (e.g. BMI: underweight; normal; overweight; obese).<sup>296</sup> For BMI the LRT was significant meaning there was evidence against the null hypothesis and BMI was better modelled as a non-linear categorical variable which was subsequently used in the adjusted analyses. This was not the case for age.

#### 4.7.5.1.2 Excluding highly correlated variables

If two highly correlated variables are included in an adjusted model they are likely to produce an underestimation of their association with the outcome, and are said to be collinear.<sup>296</sup> If included together neither may appear to be associated with the outcome even if one or both are strongly associated with it.<sup>296</sup> For instance it was inappropriate to adjust for both BMI and weight, or BMI and waist size, or weight and waist size since the Pearson's correlation between these (0.67 or higher) showed they were highly correlated. However since most of the variables were categorical it was difficult to determine whether the categorical variables were too highly related. Nevertheless, large increases in the standard error estimate in the risk analyses for two or more variables may indicate a problem.<sup>296</sup>

#### 4.7.5.2 Assessment of potential confounders by Directed Acyclic Graphs

Directed Acyclic Graphs (DAGs) can provide a visual and logically rigorous way of summarising causal links between exposures and disease outcome, and can be used to check whether sufficient confounders have been selected for adjustment. Figure 13 and Figure 14 show causal pathways representing the first step in using the theory of DAGs to identify potential confounders relating to dietary and supplement vitamin C intake in unselected populations. The majority of these have been identified *a priori* from previous studies. All variables shown in Figure 13 and Figure 14, apart from 'health consciousness', were derived in the UKWCS datasets.

The arrows in the DAGs show the direction of influence from one variable to another, with the main causal pathway from exposure of interest to outcome highlighted in red. As explained by Greenland,<sup>302,303</sup> confounding is present if an association remains between the exposure of interest and outcome after removing all exposure effects from the exposure of interest. Therefore exposure effects of vitamin C should be removed on the diagram; on Figure 13 this includes removing the arrow from dietary vitamin C exposure to 'supplement use' in addition to removing the red arrow to breast cancer incidence. It can be argued that if low dietary vitamin C intake significantly promoted supplement use then the latter would mediate the effect of dietary vitamin C on breast cancer incidence; adjusting for it could create bias rather than reduce it. On the contrary, the odds of supplement use were found to increase significantly with

increasing dietary vitamin C intake; being health conscious was likely to have increased both.

Using these diagrams confounders can be detected by finding common influences on exposure and outcome. Identifying a minimum set of confounders for adjustment was done by tracing open backdoor paths from the outcome (breast cancer incidence) to the exposure (vitamin C intake); to aid their detection the first arrows in these backdoor paths were coloured blue in Figure 13. Paths containing colliders are not used for selecting confounding variables; these are closed backdoor paths where the path enters and exits a variable as follows:



An example of a collider is illustrated below; here pill use cannot have an effect on the other two variables which are conditions that existed before pill use occurred.



The Figure 13 and Figure 14 show that all open backdoor pathways pass through the variable 'Health consciousness'; this variable, therefore, is the main common link between exposure and outcome. Given this and the fact it is not a collider, according to the DAG theory it would be sufficient in adjusted analyses to control for this one variable only.<sup>303</sup> Unfortunately, this latent variable was not readily available in the datasets, though it could have been created using factor analysis. Nevertheless, for each open backdoor path which passes through this variable there are other more readily derived variables that can be used for adjustment instead; additionally there is no direct path from this to the outcome.

As explained by Greenland,<sup>303</sup> at least one variable on each open backdoor path from the outcome breast cancer incidence to the exposure, should be controlled for in adjusted analyses. In these diagrams every variable linked directly by a blue line to the outcome can be traced back to vitamin C on open backdoor paths, therefore each of these variables could be used in adjustments, assuming there is evidence of their effects on breast cancer. For example, 'cumulative breastfeeding' is linked directly to the outcome, and can be traced back through 'education', 'health consciousness' and 'fruit and vegetable intake' to 'dietary vitamin C intake'. It is questionable, though, whether smoking status has a direct effect on breast cancer and whether it should be included as a confounder. However, a meta-analyses has reported smoking increases breast cancer risk in women with NAT2 slow acetylator genotypes, which occurs in about 50% of Causaians.<sup>304</sup>

Greenland posits that adjusting for more than the minimally sufficient set of confounders may cause confounding.<sup>303</sup> Thus in most cases it will be unnecessary to adjust for more than one variable on each open backdoor path. For instance it appears that education should not be included as a confounder in addition to 'parity' and 'cumulative breast feeding'. Furthermore, one can question whether 'BMI' 'weight', waist size' and 'total energy intake' should all be included as confounders since one open backdoor back passes through them all. A discussion about this is included in section 4.7.5.1.2.

Socio-economic status (SES) has been linked to health behaviours in the UK,<sup>305</sup> and women of higher SES are generally more health conscious. Conversely women of higher SES are likely to have fewer children which would increase their risk of breast cancer. However, it can be argued that SES should not be also included as a confounder since other variables on open backdoor pathways containing SES could control for its indirect effects. There may be possible biological mechanisms, however, by which low SES may have an effect on breast cancer risk, for instance via stress responses caused by unfairness and inequality. Although these have not been specifically researched in relation to breast cancer, low control in the workplace was not associated with increased breast cancer risk in the Nurses Health Study.<sup>306,307</sup> Nevertheless to avoid residual confounding SES has been included as a confounder in the UKWCS adjusted analyses when education has not been included.


Figure 13 A Directed Acyclic Graph (DAG) of factors associated with dietary vitamin C intake and breast cancer



Figure 14 A Directed Acyclic Graph (DAG) of factors associated with supplement vitamin C intake and breast cancer



Figure 15 DAG linking eating patterns in childhood with the main exposure and outcome

Additional variables which were considered for adjustment that were not included in Figure 13 and Figure 14 DAGs were height, age at menarche and history of hysterectomies and oophorectomies. It has been established that height is associated with breast cancer.<sup>7</sup> Additionally, fat free mass has been independently associated with breast cancer incidence.<sup>308</sup> Eating patterns in childhood and adolescence influence age of menarche, height and fat free mass, therefore these variables may confound the main causal pathway between adult dietary intake and breast cancer via a variable relating to the maintenance of eating habits established in childhood (see Figure 15). Since height and age of menarche reflect childhood nutrition as well as the action of growth factors, they are often used as proxies for the early nutritional environment. Additionally, since height and FFM are correlated, height may be used as a proxy for fat free mass (FFM) which is not readily available. Note that the UK Dietary Cohort Consortium made the decision to use height and weight as a covariates rather than BMI which was used in other analyses in this thesis.

#### 4.7.5.2.1 Other unmeasured variables relating to health consciousness.

There are various unmeasured factors which influence health consciousness or health behaviours. These include attitudes, norms, intentions and perceived behavioural control as presented in Fishbein and Ajzen's (1975) Theory of Planned Behaviour.<sup>36</sup> This model, together with three additional predictors of intentions (self-identity, health value and perceived susceptibility) and beliefs, underlying attitudes, intentions and perceived control has been used to predict supplement use in 303 UKWCS participants.<sup>37</sup> Results showed that intentions were the major predictor of dietary supplement use, and in turn these were influenced by behavioural beliefs: an individual's value of health and their perception that supplements might reduce susceptibility to illness.<sup>37</sup> Additionally, participants also believed media influenced their behaviour.

Personality is also likely to influence health consciousness, health behaviours and health outcomes. Indeed out of the Big Five personality dimensions (Neuroticism, Extraversion, Openness to Experience, Conscientiousness and Agreeableness)<sup>309</sup> Neuroticism and Conscientiousness, and its facets, have been shown to influence health behaviours and health outcomes.<sup>310-312</sup> These have not been measured for the UKWCS datasets used in this thesis, however a questionnaire has been designed to measure them in a sub-sample of UKWCS women.

Since many health behaviours have been adjusted for in the UKWCS it would be unnecessary to also control for these other factors that influence health behaviour unless they have a direct influence on risk themselves. It is possible that the level of neuroticism could have a direct biological effect on the development of cancer.

## 4.7.5.3 Energy intake

Energy intake had been calculated from FFQ listings at baseline and from diary intake in the phase 2 nested case-control dataset, though this was not available for the full phase 2 dataset. There are various methods for controlling for energy intake, as discussed by Willet (1998).<sup>301</sup> A decision was made to produce two sets of adjusted risk analysis results relating to dietary vitamin C intake: using absolute vitamin C intake adjusted by energy intake; and the other using nutrient density intake (vitamin C intake/ metric mega joule energy intake), which was then adjusted further by energy intake in the risk analyses.

Nutrients such as dietary vitamin C intake tended to be weakly correlated with total energy intake even though they do not contribute energy;<sup>301</sup> this is because total food intake correlates with both. If it is hypothesised that vitamin C intake relative to body size has a biological influence on breast cancer risk, rather than absolute vitamin C intake, then given that energy intake tends to reflect body size, an adjustment for total energy intake would be appropriate. This assumes that a specific amount of vitamin C would have less effect on a larger, high energy consuming person than a smaller person.<sup>301</sup> These assumptions, however, cannot be applied to smaller people who are very physically active and are high energy consumers; for such women an analysis of absolute amounts of vitamin C may be more appropriate. Alternatively, adjusting for BMI or weight and physical activity in addition to energy intake may be suitable. Furthermore, if total energy intake is associated with breast cancer then an association, therefore adjustment for energy intake would be warranted.

An alternative approach to control for energy intake is through nutrient densities,<sup>301</sup> which are calculated by dividing total vitamin C dietary intake by total energy intake. Although its use tends to reduce between-person variations it may result in a nutrient density variable that is inversely related to breast cancer, if energy intake is related to breast cancer. Willet states that this inverse relationship with disease outcome is more likely when the nutrient is weakly correlated with total energy intake; furthermore he found a weak correlation between energy intake and vitamin C (Pearson's r=0.28).<sup>301</sup> Therefore, creating a vitamin C nutrient density variable would not necessarily control for energy intake. Willet suggests this can be overcome by adjusting nutrient density analyses by total energy intake.

To avoid over-adjustment, the intake of fat, meat and other specific energy containing foods were not adjusted for. However alcohol, which contains energy and for which there is convincing evidence of a positive association with breast cancer risk,<sup>7</sup> was adjusted for in addition to total energy intake in the analyses in this thesis. Some analyses were adjusted by a dichotomous variable which split the dataset into alcoholic drinkers and those drinking less than once a week or not at all; this may avoid over-adjustment of the energy contained in alcohol.

# 4.7.6 Sensitivity analyses

A variety of sensitivity analyses were carried out on the datasets e.g. excluding outliers or excluding women with a family history of breast cancer; details of the methods for these are given below in sections 4.7.6.1 and 4.7.6.3. Since sensitivity analyses were not decided *a priori*, they were not consistently used across the different datasets; therefore further sensitivity analyses on some of the results could be carried out later. For instance in addition to those excluded in other chapters, women with a family history of cancer at baseline could be excluded in sensitivity analyses for chapter 7 on general supplement use and breast cancer risk. Plans to exclude self-reported cancer in sensitivity analyses were not carried out. Risk results reported in chapter 8 on vitamin C supplementation did exclude self-reported prevalent cancers, however since self-reported cancers could be inaccurate, or could have been non-malignant, it may have been better to have included these women in the main analysis and excluded them in sensitivity analyses.

# 4.7.6.1 Excluding outliers e.g. for high dietary vitamin C intake

Various methods for dealing with outliers have been suggested and researched.<sup>313 314</sup> Many are only applicable to test univariate data or assume data is normally distributed or are designed to check for one outlier only. Grubbs method uses mean and standard deviations to calculate outliers,<sup>315</sup> however since these can be influenced by outliers themselves this univariate method may not be reliable. A simple alternative method which does not use the mean and does not assume a normal distribution is the use of box-plots.<sup>313</sup> This defines the lower and upper thresholds for outliers as 1.5 times the inter-quartile range (the upper quartile i.e. 75% percentile less the lower quartile i.e. 25% percentile). Figure 18 in section 5.3 shows that many dietary vitamin C data points in the UKWCS baseline dataset lie beyond the lines of the upper threshold. This method has been used in sensitivity analyses in this thesis to exclude outliers.

However, some outliers calculated using the inter-quartile range method may occur due to natural variation rather than error.<sup>313</sup> There is evidence that multivariate methods which used Mahalanobis distance calculations produced better agreement with

opinions of experts in the subject matter examined. <sup>313</sup> Unlike box-plots these methods do not sort all the data into ascending order but calculate the distance from each point to the geometric centre of all the scattered points; the processing time, however, is long for large datasets. Sensitivity analyses for dietary vitamin C intake were undertaken omitting outliers calculated using this type of multivariate method developed by Hadi and applied in Stata v10.<sup>316,317</sup>

# 4.7.6.2 Adjustment for vitamin E

The amount of vitamin E consumed from supplements recorded in the UKWCS phase 2 diaries was adjusted for in a sensitivity analysis of breast cancer risk and vitamin C intake from supplements in chapter 8. Vitamin E from supplements was cleaned and the mean daily intake was calculated in a similar manner to that described for vitamin C supplement data in section 4.5.3.3. Diary recordings of multivitamins which contained both vitamin C and E had already been cleaned earlier. The intake of vitamin A or other antioxidants in other supplements was not cleaned so could not be adjusted for.

Intake of vitamin E, along with vitamin A, which may affect cancer risk via their antioxidant or other properties, had been controlled for in the Danish nested case-control study which assessed the relationship between vitamin C intake and breast cancer risk.<sup>245</sup> No other previous studies had adjusted for these. By supplying electrons vitamin C reduces and regenerates the fat soluble vitamin E radical back to its active antioxidant form. Indeed, *in vitro* experiments have shown that vitamin E can act as a pro-oxidant in the absence of co-antioxidants such as vitamin C.<sup>49</sup> However, this may mean vitamin E may be on the causal pathway between vitamin C intake and breast cancer risk, and perhaps adjusting for vitamin E could create bias.

#### 4.7.6.3 Excluding or adjusting for women with a family history of cancer

Since women with a family history of breast cancer may have a higher risk of breast cancer and may also be more likely to take supplements, these women could be excluded or adjusted in sensitivity analyses to provide better risk estimates for women without a family history. Since many women with a family history are not genetically at raised risk of breast cancer an attempt was made using the detailed family history data at phase 2 to identify women who may be classed at raised or higher risk (see section 4.7.6.3.2). The data at baseline, however, was not detailed enough to estimate those at raised or high risk of breast cancer.

A review of 14 studies that evaluated both positive and negative self-reported family history cancers by comparing them with cancer registry records found that patient-

reported family cancer histories for first-degree relatives are accurate and valuable for breast and colon cancer risk assessments.<sup>318</sup> Negative family history reports for ovarian cancer was not so good, although the prevalence of this is low within families.<sup>318</sup>

## 4.7.6.3.1 UKWCS Baseline

At baseline a family history of breast cancer was defined as having a first degree relative; mother, father, brothers or sisters who had breast cancer. No information was given about aunts or uncles. Participants were asked to provide more details, but this was often incomplete. Since some participants had stated that a first degree relative had cancer but did not specify what type of cancer, additional adjustments/exclusions were made in sensitivity analyses for women who had a first degree relative with any type of cancer.

# Figure 16 Questions asked at baseline relating to family history of cancer

	Yes	1 N	lo 🛄 2	Don't Know
If yes, please give details				
				8111
If you have brothers and/or sister	s,have they ever suffered from	cancer (	or heart a	ittack/heart
If you have brothers and/or sister disease?	s,have they ever suffered from	cancer (	or heart a	attack/heart Don't Know
If you have brothers and/or sister disease? If yes, please describe details	s,have they ever suffered from Yes 🗌	cancer ( <sup>1</sup> N	or heart a lo 🗌 ²	ittack/heart Don't Know

#### 4.7.6.3.2 UKWCS Phase 2

At phase 2 a family history of breast cancer was defined as having a blood relative diagnosed with breast cancer as reported by participants at phase 2 in the questionnaire; they were questioned about cancers in mothers, fathers, sisters, brothers, aunts, uncles. Unlike at baseline, participants were specifically asked which family members had breast cancer and their age at diagnosis (see Figure 17 compared to Figure 16); therefore the breast cancer family history information provided by participants at phase 2 was likely to be more accurate.

35. We would like to find out about your family's history of cancer. Please look at the list of different types of cancer listed below and document whether or not any of your family members have ever had any of these types of cancer. Please include the age at diagnosis if known. If you are adopted or if your parents remarried, please give details of blood relatives only. Codes for Cancer Breast 1. 2 Lung Skin 3 Stomach 4. Colon & rectum 5 Ovary 6. 7. Cervix Uterus 8. 9. Pancreas Prostate 10. Code Mother Father Sister Uncle Brother Aunt E.g. 10 172 E.g. 1 ✓ 61

Based on information from the NICE report on Familiar breast cancer issued October 2006,<sup>319</sup> it was possible to estimate a small percentage of the phase 2 UKWCS women who may have been at raised risk (2.1%) or high risk (0.8%) of developing breast cancer (see 0). Unfortunately the NICE guidelines could not be followed in some areas; therefore the categories are only estimates. In this report women with a 10-year risk of 3–8% between the ages of 40–49 or a lifetime risk of 17% or greater but less than 30% were classed as having raised risk of developing breast cancer. Women with a 10-year risk of greater than 8% between the ages of 40–49 or a lifetime risk of 30% or greater were classed as at high risk; it included those with a 20% or greater chance of a faulty BRCA1, BRCA2 or TP53 gene in the family.

# 4.7.7 Samples sizes

Sample sizes were calculated *a priori*. There were sufficient numbers of women in the whole of the UKWCS phase 2 cohort exposed to high doses of vitamin C supplementation greater or equal to 500mg/day when sub-analysed into pre/post-menopausal status in phase 2 of the UKWCS to detect relative risks of 1.3 or above in breast cancer risk analyses (e.g. with 404 pre-menopausal women detectable HR = 0.797 or 1.278).

Assumptions:

Average follow-up time = 7.5 years Median time on the control treatment = 7 years Power of study = 90%, Level of significance = 5%

## Figure 17 Questions asked at phase 2 relating to family history of cancer

98

However, in case-control analyses of total vitamin C intake from phase 2 UKWCS diaries with only 200 breast cancer cases it would only be possible to detect odds ratios for breast cancer of 0.486 or 1.804 in women exposed to vitamin C above 60mg/day supplements relative to unexposed subjects (see assumptions below). Furthermore there would be only 45% power to detect odds ratios of 1.4 for breast cancer in those taking vitamin C above 60mg/day relative to those not taking these. Moreover, the power would reduce further if the breast cancer cases were split into pre-menopausal and post-menopausal, or other sub-analyses were performed, or the exposure level was increased. To detect odds ratios of 1.4 with 90% power, 657 case patients with five matched controls per case would be required, meaning the UK Dietary Cohort Consortium studies had to be consolidated in order to sufficiently power the case-control analyses for total vitamin C intake.

Assumptions:

Number of matched breast cancer cases = 200

Ratio of matched controls per case = 5

Probability of vitamin C exposure above 60mg/day from supplements among controls = 0.20

Correlation coefficient for exposure between matched cases and controls = 0.1Power of study = 90%, Level of significance = 5%

The next chapter evaluates the recording of dietary and supplement vitamin C intake in the UKWCS, then the following chapter determines whether women in the UKWCS with a history of breast cancer are more likely to take vitamin C supplements. The subsequent four chapters assess the relationships between breast cancer risk in UK women and general supplement use, vitamin C supplement use, dietary vitamin C intake and finally total vitamin C intake.

# **CHAPTER 5**

# 5 Evaluation of diet and supplement vitamin C intake recordings in the UKWCS

# 5.1 Summary

*Objective*: Since measurement error of diet can attenuate or exaggerate associations between dietary intake and disease incidence and reduce the power to find these, it is important to evaluate how well dietary tools record dietary exposure. The objective of this chapter was to evaluate vitamin C intake derived from diet and supplement recordings in the UKWCS using a variety of assessment methods.

# Methods:

Firstly, the frequency of taking supplements containing vitamin C recorded by diary was compared to the frequency of vitamin C supplement use recorded in the phase 2 questionnaire by 11,184 UKWCS participants. This was evaluated using kappa statistics and tabulations.

Secondly, fruit and vegetable servings derived from FFQ recordings were compared with summary cross-check questions recorded concurrently by 33,521 UKWCS baseline participants. Agreement between fruit and vegetable intake and vitamin C intake derived from the FFQ were also assessed. Spearman's correlations, tabulations, Bland-Altman plots and kappa statistics were used in the evaluations.

Thirdly, vitamin C intake derived from recordings from diaries and FFQs were compared to plasma ascorbic acid and plasma total vitamin C concentrations for 273 women who provided a blood sample at phase 2. This was evaluated using Spearman's correlations and regression analysis.

# Results:

Daily vitamin C supplement use reported on the questionnaire captured most of the women recording high dose vitamin C (>500mg) by diary in the UKWCS, the majority of which was likely to be single ingredient supplements. The inclusion of daily reporting of multivitamin/mineral use, as well as vitamin C use, on the questionnaire, produced substantial agreement with diary recordings of supplements containing any vitamin C dose taken on three or four of the four diary days.

Although there were fair to moderate associations between the two methods of recording fruit and vegetable servings using Spearman's correlations and kappa agreement methods, high discrepancies were seen using Bland-Altman's plots. The associations and agreement between cross-check fruit and vegetable servings and FFQ derived dietary vitamin C was also fair to moderate, nevertheless the agreement between grams of fruit and vegetable recorded by FFQ and derived dietary vitamin C was good. Women who were vegetarians, or not supplement users or on a low fat or slimming diet or eating small portions were significantly more likely to have high discrepancies in vegetable reporting between the two methods. Non-graduates and women from lower SES groups were more likely to report high discrepancies for either fruit or vegetables. Correcting FFQ derived vitamin C using the cross-check question data appeared inappropriate.

In the final evaluation, FFQ derived dietary vitamin C correlations with plasma ascorbic acid, and also with dietary vitamin C calculated from diary recordings, were weak to moderate but in line with previous studies. Although the derived values of vitamin C consumed may be inaccurate, the relative ranking of individuals for the full UKWCS cohort based on a similar FFQ may be acceptable for assessing associations between vitamin C dietary intake and breast cancer risk.

*Conclusion:* The comparisons between FFQs, diary and plasma concentrations were in line with those from other studies and were considered suitable for further analysis. These results highlighted some weaknesses in methods used in the UKWCS and other studies for recording food and supplements containing vitamin C, which should be taken into account when interpreting results of associations between vitamin C intake and breast cancer risks in the UKWCS.

# 5.2 Comparison of diary and questionnaire methods to assess vitamin C supplement use at phase 2

# 5.2.1 Introduction

One aim of this thesis was to explore the associations between vitamin C contained in supplements and breast cancer risk, results of which are provided in chapter 8. To explore dose-response relationships, however, it is important to include vitamin C from all supplements taken, not only from single ingredient vitamin C supplements. As seen in section 4.5.3.2 of the methods chapter, a large proportion of supplements containing vitamin C up to the EU recommended daily allowance (EU RDA 60mg/day <sup>53</sup>) are branded as multivitamins in the UK. Supplements which include more than the EU RDA but less than 500mg/day are often named as multivitamins or antioxidants, whereas vitamin C supplements containing high doses of 500mg or more are usually single ingredient. Some of the previous studies which have examined associations between vitamin C supplementation and breast cancer appear not to have used an ingredient database to take account of the amount of vitamin C in all supplements, although doses for multivitamins appear to have been allocated in some studies.<sup>242 267 270</sup>

Other than checks described in section 4.5.3.3 which were made whilst cleaning the supplement database, unfortunately it was not possible to assess whether the dose of vitamin C had been correctly assigned to supplements recorded in the UKWCS phase 2 diaries. In this section, however, an attempt was made to verify the frequency of intake recorded in the diaries with that recorded in questionnaires, and to determine whether the agreement depended on the type of supplement recorded in the diaries and their recorded or derived vitamin C amount. Daily intake of vitamin C was assessed since this may have a greater influence on cancer risk than less regular intake because, as mentioned in chapter 2, excess vitamin C supplements on the phase 2 health and lifestyle questionnaires only if they had taken single supplement vitamin C, or if vitamin C was specified in the name. On the other hand it was expected that women would not indicate they took vitamin C supplements on the questionnaire if they took multivitamins or antioxidants, and other supplements which contained vitamin C but did not mention it in the name.

This analysis compares daily vitamin C use recorded by the phase 2 health and lifestyle questionnaire with supplements containing vitamin C which were recorded in the 4-day diary at phase 2 in the UKWCS.

# 5.2.2.1 Subjects

At phase 2 of the UKWCS, 12,453 women completed a health and lifestyle questionnaire and a 4-day food diary, both of which included sections on supplement use. Women who were to be excluded from the related time-to-event analyses in chapter 8 because of their cancer status were also excluded in this evaluation, leaving 11,184 women for the analysis.

# 5.2.2.2 Dietary assessment methods

For each of the four diary days, the women were required to record supplement brand, name, amount taken and dosage of any supplement taken (as shown in section 4.5.3.1). This information was electronically captured and matched against a database of supplement descriptions and ingredient composition obtained from product labels provided by participants, suppliers' websites or directly from manufacturers. The cleaning of the supplement usage database is detailed in section 4.5.3.3.

Additionally, the frequency of use of 17 types of dietary supplements had been captured from the phase 2 health and lifestyle questionnaire (as shown in section 4.5.2). In addition to vitamin C, the types listed included antioxidant supplements, multivitamins, and multivitamin plus mineral supplements. Frequencies were categorised as: more than daily; daily; weekly; monthly; less than monthly.

# 5.2.2.3 Statistical analyses

Descriptive statistics of the 11,184 women at phase 2 are included in chapter 8 which examines associations between vitamin C supplement use and breast cancer risk.

The agreement between the two methods of capturing frequency of vitamin C use, i.e. diaries and phase 2 questionnaire, was compared using kappa statistics.<sup>320</sup> Agreements were interpreted as suggested by Landis and Koch detailed in section 5.3.2.3.<sup>321</sup> The definition of frequent use of supplements containing vitamin C was relaxed to three or more of the four diary days, to allow for women who may have accidentally missed completing this section on one of the diary days. Tabulations between the two methods were also produced, including tabulations with vitamin C dose split between1-60mg/d; >60<500mg/d; ≥500mg/d. These are categories used in chapter 8 on vitamin C supplement use and breast cancer risk.

# 5.2.3 Results

# 5.2.3.1 Comparisons between vitamin C intake recorded by diary and supplements recorded by the phase 2 questionnaire

When vitamin C supplement use recorded on three or four of the four diary days were split into categories by dose at zero, 60 mg and 500 mg cut-off points, about a third of users were in each intake category. As observed in Table 9, of women who recorded on the phase 2 questionnaires they took daily vitamin C supplements, 18% recorded no use of supplements containing vitamin C in the diaries on 3 or more diary days, about a third took supplements containing less than 500mg/d vitamin C on 3 or more diary days, and half recorded in the phase 2 diaries they took supplements containing 500mg/d vitamin C or more on 3-4 of the diary days.

**Table 9** Women who recorded daily use of vitamin C on the phase 2 questionnaire split

 by diary vitamin C intake category

Vitamin C supplement use	Daily use of vitamin C on phase 2 questionnaires				
categories by dose (taken on	Spiit i				
3 or 4 of the four diary days)	Yes	No	Total		
No vitamin C	351 (18%)	7042 (76%)	7393 (66%)		
1-60mg/d	200 (10%)	1293 (14%)	1493 (13%)		
>60<500mg/d	418 (22%)	817 (9%)	1235 (11%)		
≥500mg/d	947 (50%)	116 (1%)	1063 (10%)		
Total	1916 (100%)	9268 (100%)	11184 (100%)		

However, from the perspective of diary recordings, 2210 (56%) of the 3791 diary vitamin C supplement users documented use of supplements containing less than 500mg/d vitamin C in the diary but did not record vitamin C use in the questionnaire. As seen in Table 10 women (89%) who took 500mg/d vitamin C on three or four diary days also recorded on the phase 2 questionnaire that they took daily vitamin C supplements. As found from the inspection of the ingredient database in section 4.5.3.2., the majority of these supplements were likely to be single supplement vitamin C. About a third of women in the medium and high vitamin C intake categories reported on the questionnaire they took antioxidants. Forty percent of women in the highest vitamin C category also recorded the frequent use of multivitamins on the questionnaire, compared to 72% of women in the less than or equal to 60mg/d vitamin C category.

		Percentage of women in each vitamin C					
		intake category who recorded daily intake					
Vitamin C supp	plement use recorded	on pha	ase 2 questionn	aire of:			
by diary split in	to categories by dose	Vitamin C	Antioxidant	Multivitamin			
(taken on 3 or	4 of the four diary days)	N=1916	N=1099	N=2714			
No vitamin C	N= 7393 (66%)	5%	3%	8%			
1-60mg/d	N= 1493 (13%)	13%	10%	72%			
>60<500mg/d	N= 1235 (11%)	34%	34%	53%			
≥500mg/d	N= 1063 (9.5%)	89%	28%	40%			

**Table 10** Percentage of women in each vitamin C intake category who recorded daily intake of vitamin C, antioxidants or multivitamins on the phase 2 questionnaire

Using kappa statistics, the measure of agreement between the two methods of capturing frequency of vitamin C supplement usage at phase 2 varies between fair to substantial depending upon what questionnaire data is included in the calculation (Table 11).

**Table 11** Kappa agreements between frequency of vitamin C intake captured via diary and that captured via the phase 2 questionnaire

Questionnaire: Supplement	Supplement containing	Kappa	Agreement
taken daily	vitamin C from 4 days of	(95%CI)	321
	diary keeping		
Vitamin C <sup>a</sup>	4 out of 4 days (daily)	0.39 (0.37-0.41)	Fair
Vitamin C	at least 3 out of 4 days	0.41 (0.39-0.43)	Moderate
Vitamin C or multivitamin <sup>d</sup>	at least 3 out of 4 days	0.70 (0.69-0.71)	Substantial
Vitamin C or multivitamin or	at least 3 out of 4 days	0.71 (0.70-0.72)	Substantial
antioxidant <sup>e</sup>			
Vitamin C or antioxidant <sup>b</sup>	>60mg, 3 out of 4 days	0.61 (0.60-0.62)	Substantial
Vitamin C	≥500mg, 3 out of 4 days	0.59 (0.58-0.59)	Moderate
Multivitamin/ mineral <sup>c</sup>	1-60mg, 3 out of 4 days	0.41 (0.40-0.42)	Moderate

<sup>a</sup>ticked takes vitamin C daily, or more than daily on the phase 2 questionnaire

<sup>b</sup>ticked takes vitamin C or antioxidant daily, or more than daily on the questionnaire

<sup>c</sup>ticked takes multivitamin or multivitamin and mineral daily, or more than daily on the questionnaire

<sup>d</sup>ticked takes vitamin C, multivitamin or multivitamin and mineral daily, or more than daily on the questionnaire

<sup>e</sup> ticked takes vitamin C, multivitamin or multivitamin and mineral daily, or more than daily,

Only fair agreement (kappa=0.39; 95%CI: 0.37, 0.41) occurs when comparing those women who have ticked daily vitamin C supplement usage on the questionnaire with those shown to have taken four days of supplements containing vitamin C during diary completion (Table 11). As observed in Table 12 77.7% of recordings are in the expected quadrant. According to the kappa statistics, the agreement increases if the comparison is relaxed to usage on three or four of the four diary days. The agreement becomes substantial if daily multivitamin use indicated on the questionnaire is also included in the comparison (kappa=0.70; 95%CI: 0.69, 0.71). This kappa statistic alters

only slightly if daily antioxidant supplement use on the questionnaire is then also included in the comparison ((kappa=0.71; 95%CI: 0.70, 0.72); as seen in Table 13 86.6% are in the expected quadrant). Excluding women who were ill or on a weight reducing diet during diary recording, and who may have taken vitamin C or antioxidants or multivitamins during this time, perhaps on a temporary basis, does not alter the kappa statistic (kappa=0.71; 95%CI: 0.70, 0.72).

**Table 12** Cross-tabulation of daily vitamin C supplement use recorded by questionnaire

 with daily intake of supplement containing vitamin C recorded by diaries

Questionnaire:	Diary: daily use of supplements containing vitamin C						
Daily vitamin C	(4 out of 4 diary days)						
use	No	Yes	Total				
No	7332 (65.5%)	1936 (17.3%)	9268 (82.8%)				
Yes	556 (5.0%)	1360 (12.2%)	1916 (17.2%)				
Total	7888 (70.5%)	3296 (29.5%)	11184 (100%)				

Kappa = 0.39 (0.37-0.41)

**Table 13** Cross-tabulation of daily vitamin C or antioxidant or multivitamin supplement use recorded by questionnaire with intake of supplement containing vitamin C on 3 or 4 diary days

Questionnaire:	Diary: use of supplements containing vitamin C							
Daily vitamin C or	on 3 or 4 diary days							
antioxidant or								
multivitamin use	No	Yes	Total					
No	6497 (58.1%)	604 (5.4%)	7101 (63.5%)					
Yes	896 (8.0%)	3187 (28.5%)	4083 (36.5%)					
Total	7393 (66.1%)	3791 (33.9%)	11184 (100%)					

Kappa = 0.71 (0.70-0.72)

**Table 14** Cross-tabulation of daily vitamin C or antioxidant supplement use recorded by questionnaire and taking supplements containing vitamin C above 60mg for 3 or 4 diary days

Questionnaire:	Diary: use of supplements containing vitamin C above 60mg							
Daily vitamin C or	on 3 or 4 diary days							
antioxidant use	No Yes Total							
No	8054 (72.0%)	647 (5.8%)	8701 (77.8%)					
Yes	832 (7.5%)	1651 (14.7%)	2483 (22.2%)					
Total	8886 (79.5%) 2298 (20.5%) 11184 (100%)							

Kappa = 0.61 (0.60-0.62)

Substantial agreement also occurs in comparisons between supplements containing over 60 mg/d of vitamin C recorded by diary on three or four days and daily vitamin C supplement or antioxidant usage indicated on the questionnaire ((kappa=0.61; 95% CI: 0.60, 0.62); 86.7% are in the expected quadrant in Table 14). Similar results are produced in comparisons between supplements containing 500mg or more of vitamin C recorded by diary on three or four days and daily vitamin C supplement usage indicated on the questionnaire (kappa=0.59; 95% CI: 0.60, 0.62). Similar results are also produced after excluding women who completed the diary between the winter months, November to end of February, who may only take antioxidants or the higher doses of vitamin C throughout the winter or when they develop winter colds (Kappa=0.58; 95% CI: 0.56, 0.58)).

However, there was only moderate agreement when recording of supplements containing low amounts of vitamin C (60mg/d or less) on three or four diary days were compared with daily multivitamin recordings on the questionnaire (kappa=0.41, 95%CI: 0.40, 0.42).

# 5.2.4 Discussion

Only moderate agreement occurred when comparing those women who ticked daily vitamin C supplement usage on the questionnaire with those shown to have taken supplements containing vitamin C on three or four days of the 4-d diary. This was unsurprising given the difference seen in the various tabulations. For instance, over a third of women consuming supplements containing vitamin C as per diary recordings were consuming ones less than or equal to 60mg, and the majority of women in this category recorded daily multivitamin use rather than vitamin C or antioxidant use on the questionnaire. The results clearly show that responses to a simple question about vitamin C supplements containing vitamin C. Therefore, it was not possible to verify the frequency of diary recorded supplements containing vitamin C using only the vitamin C question on the questionnaire. Only after taking account of the reporting of daily use of multivitamin/minerals as well as vitamin C on the questionnaire, was substantial agreement reached between diary and questionnaire reports.

However, the results indicate that previous studies which have examined associations between vitamin C supplementation and breast cancer risk and allocated doses for single supplement vitamin C, multivitamins and antioxidants may have picked up the majority of supplements used containing vitamin C.<sup>242 267 270</sup> Nevertheless, the dose estimates that were allocated may have been somewhat inaccurate if no ingredient database was used. Furthermore, as seen from the inspection of the UKWCS

ingredient database in section 4.5.3.2, the majority of supplements in the medium vitamin C dose range are types other than those branded as vitamin C, multivitamins and antioxidants; these, however, may be less commonly consumed.

Although women recording supplements containing high doses of vitamin C (500mg/d or more) in the diaries usually reported vitamin C supplement on the phase 2 questionnaire, only half the women who reported the latter appeared to take high doses. Therefore, confirmative answers to vitamin C use on the questionnaire were not representative of high dose use in the diary, indicating that questionnaires should not be used for this purpose.

Although the majority of women (72%) using supplements containing doses of 60 mg/day or below (the EU recommended allowance) appear to report multivitamin use on the questionnaire, the overall kappa agreement between this diary category with recordings of daily use of multivitamins or multivitamins with minerals on the questionnaire was fairly poor. There are likely to be two reasons for this: the average multivitamin in the UK supplement data base contains over 70mg therefore some multivitamins will contain more than 60mg; and many women in the higher dose categories may also be taking low dose multivitamins so may have ticked the vitamin C and a multivitamin box on the questionnaire.

The questionnaire relates to a longer time period than the 4-d diary; this is likely to have lowered the agreement between the two tools. It is not clear from the section in the questionnaire whether the period of assessment for each supplement type listed relates to current use or use in the last year. The 4-d diary recordings are more likely to represent short-term episodic intake, particularly because high dose vitamin C is promoted for reducing seasonal common colds. Despite this, the agreements between the methods did not change when women who completed the diaries in the winter months were excluded. Neither did the agreement between methods change when women were excluded who were ill or on a weight reducing diet, and who may have been concerned that they lacked nutrients during this time and taken supplements temporarily. Furthermore, vitamin C supplementation was found to have the highest long-term use stability of any nutrient in a US study which compared supplement data collected by two questionnaires that were on average 2.4 years apart (kappa=0.64; 955 CI: 0.60, 0.67).<sup>322</sup> A similar agreement was found comparing 24-hr recall methods with the second questionnaire data, but comparisons with diary data were not undertaken.322

A previous study compared information transcribed from supplement packaging (the gold standard) with data from self-administered questionnaire, producing kappa

statistics for supplement type as 0.67 (0.53, 0.89) for vitamin C, and 0.68 (0.52, 0.84) for once a day multivitamins.<sup>323</sup> The questionnaires in this study requested more detail than those of the UKWCS, and the UKWCS diary recordings may not be as accurate as the gold standard, nevertheless some of the current results were in line with these results. The Spearman's correlation coefficients for dose of vitamin C in all supplements was 0.76 for this study.<sup>323</sup>

In the UKWCS the diary and questionnaire were completed within a short time period of each other; although this increases the likelihood that supplements were recorded on either tool, it may have also created recall bias and produced overstated comparisons.<sup>324</sup> Nevertheless, anomalies remain in the UKWCS data that may be due to recording or input error, which cannot be explained by changes in supplement use owing to seasonal differences, illnesses or weight-reducing diets. Despite the time that had been spent allocating supplement types from the ingredient database to supplements recorded in the diaries, and cleaning these, the methods used in recent research on the EPIC-Norfolk cohort to allocate vitamin C dose could potentially produce a more accurate vitamin C intake consumed from supplements.<sup>38</sup>

Despite diary assessment taking place over a narrow four day period, overall these results indicate that, after checking and cleaning supplements containing vitamin C, the validity of daily vitamin C supplement use is good when compared with the questionnaire assessment of vitamin C and multivitamin supplement use. This and the results of the final spot check indicate the data is acceptable for use in further analyses.

# 5.3 Comparisons relating to dietary vitamin C intake and servings of fruit and vegetable consumed at baseline

# 5.3.1 Introduction

Since measurement error of dietary exposure could distort the associations between vitamin C intake and breast cancer incidence in the UKWCS, it is important to evaluate the recording of food containing vitamin C in the UKWCS. At baseline self-reported FFQs were used to record diet since the cost and ease of administering them to over 35,000 women were lower than other methods. Additionally, the FFQs estimate intake over 12 months, being superior in this respect compared to diaries or biomarkers. This section focuses on the evaluation of UKWCS baseline FFQ recordings of fruit and vegetables, the main source of vitamin C.

Over-estimation of fruit and vegetables by FFQs is a known problem which has been previously assessed using data from 6,572 women at baseline in the UKWCS.<sup>288</sup> If estimated intake was systematically inflated for most women in the cohort then this may not impact on cancer risk ratios. However, the earlier analysis found that vegetarians and those on slimming diets were slightly more likely to have a large discrepancy between FFQ fruit and vegetable recordings and total portions of fruit and vegetables consumed given in two cross-check questions.<sup>288</sup> Women with higher education levels were less likely to over-estimate. The authors suggest that average portions of vegetables consumed in mixed dishes may have been over-estimated and socially desirable answers may have been given. Possible errors in FFQ derived fruit and vegetable or vitamin C intakes can be adjusted by applying a correcting factor for each subject using cross-check summary questions which ask for total fruit and vegetable portions consumed per day or per week.<sup>288 325-327</sup>

Expanding on the initial analysis of 6,572 women,<sup>288</sup> this baseline analysis of the full UKWCS cohort will compare fruit and vegetable intake derived from FFQ recordings with concurrent recordings by cross-check questions. Agreement between fruit and vegetable intake and vitamin C intake derived from the FFQ will also be assessed. The suitability of using the cross-check questions to correct for possible over-estimation of vitamin C derived from fruit and vegetables reported in the UKWCS FFQs will be discussed.

# 5.3.2 Methods

# 5.3.2.1 Subjects

General information about the recruitment of the 35,372 women who completed the baseline questionnaire is provided in section 4.1.1.1 of this thesis. Women who were to be excluded from the subsequent time-to-event analyses in chapter 9 were also excluded in this evaluation. This left 33,521 middle-aged women from the UKWCS who completed the baseline FFQ for this evaluation analysis.

# 5.3.2.2 Dietary assessment methods

At baseline the mean daily dietary vitamin C intake had been calculated from the list of simple and mixed food items on the FFQ. This FFQ was developed from one used in EPIC UK studies,<sup>289 290</sup> by adding extra vegetable composition dishes to accommodate the higher proportion of vegetarians in the UKWCS to produce a 217 item FFQ.<sup>279</sup> Thirty one vegetables (excluding potatoes), 19 fresh fruit and five dried fruit items were listed in the FFQ. Nine of the fresh fruit were seasonal; no canned or frozen fruit were specifically listed. As seen in section 4.5.1, for each item participants chose from 10 pre-coded frequency of consumption categories ranging from never to 6 or more times per day, on average over the past year. The cross-check questions were on a separate section of the baseline questionnaire:

Q7 How many servings of vegetable or vegetable containing dishes (excluding potatoes) do you usually eat each week?

Q11 How many servings of fruit or fruit containing dishes do you usually eat each week?

# 5.3.2.3 Statistical analyses

Descriptive statistics of the 33,521 women at baseline are included in chapter 9 on associations between dietary vitamin C intake and breast cancer risk.

After exclusions, histograms and box plots were produced of dietary vitamin C and total fruit and vegetable intake (see Figure 18). All graphs were skewed to the right and also showed twelve women with much higher intakes. No obvious errors were found between answers provided on the questionnaire and data electronically captured for these women. However, comparisons between the completion of the FFQ lists of fruit and vegetables and the number of servings given in the cross-check questions indicated that some women with very high intakes may have completed some sections incorrectly. For all but one participant, it was not obvious which parts were completed incorrectly, therefore no corrections or further exclusions were made at this stage.

The distributions of the variables were skewed, and so the median intakes and interquartile ranges (IQR) were calculated as well as the means for fruit and vegetables and for derived vitamin C intake split by fruit, vegetables, juice and potatoes. The Wilcoxon sign-rank test, for paired non-parametric data, was used to compare servings of fruit or vegetables measured by FFQ and by cross-check question for each woman. The correlations between total fruit and vegetable servings and also intake as grams and derived vitamin C were calculated using Spearman's correlation coefficients for nonparametric data (using pair-wise deletion of observations with missing values). This method is less influenced by outliers than Pearson's product moment correlations. The Spearman's method ranks data for each variable and calculates the Pearson's correlation between the ranks which is the number of standard deviations that one variable changes for a standard deviation change in the other. Unfortunately, correlations between two variables can be strong even if one is consistently much higher than the other variable, making the technique misleading.<sup>328</sup> Correlations measure the strength of the association between two variables but, unlike the Bland-Altman technique, cannot assess the extent of the agreement between them. The Bland-Altman technique can only be used to compare two methods with the same underlying units of measurement such as fruit and vegetable servings; the ratio of servings derived from FFQs to servings reported by the cross-check question was plotted against the mean of the two measures using log scales.<sup>328 329</sup> More specifically, this was achieved by plotting the exponential of the difference in the two logtransformed measures against the exponential of the mean of the log-transformed variables on log scaled graphs. The ratio was used instead of the difference between the two measures normally used in Bland-Altman plots, since the differences increased substantially as the servings increased. The ratios between the two variables should be relatively small for the agreement between the two measurements to be considered good. The 731 women who did not complete either cross-check question were excluded from the plot.

In a similar manner to previous research on the smaller sample of UKWCS, <sup>288</sup> the characteristics of women with FFQ derived servings that were more than five times larger than the cross-check answers were tabled.

For comparisons of variables with different underlying units, the variables were categorised into 5 groups of ascending intake and the agreement between the categorisation of participants using the two methods was compared using kappa statistics;<sup>320</sup> a technique normally used to measure agreement between two raters or measure reproducibility between two time points. Unweighted and weighted kappa statistics, K and K<sub>w</sub>, were produced.

For instance, the kappa statistic compared the observed portion of agreement between fruit intake in grams and fruit servings with the portion of agreements that would be expected by chance. Given that 5 categories were used, the chance of misclassification was quite high which tended to decrease the value of kappa. Rather than assigning no agreement, classification occurring in adjacent categories was counted as partial agreement and was weighted accordingly. The following linear weighting was used:

$$K_{wi} = 1 - (i/(n-1))$$

where *i* is the number of categories of difference between the two methods for each participant, and *n* is the number of categories. This gave weightings of: 1; 0.75; 0.5; 0.25, and 0 depending on whether the classifications differed by none, 1, 2, 3, or 4 categories respectively.

Agreements were interpreted as suggested by Landis and Koch (1977):<sup>321</sup>

- K =0 none, no better than chance
- K < 0.2 slight agreement
- $K \ge 0.21 \le 0.40$  fair agreement
- $K \ge 0.41 \le 0.60$  moderate agreement
- K ≥0.61 ≤0.80 substantial/good agreement
- K >0.8-1.00 very good agreement

**Figure 18** Box whisper plots and histograms showing distributions of FFQ derived dietary vitamin C (mg/d), total fruit and vegetable intake (g/d), and total servings per day of fruit and vegetables



Outliers shown in box plots but not in histograms

**Figure 19** Histograms showing distributions of total fruit and vegetables servings per week recorded by cross-check question, and separate distributions for fruit and for vegetables



# 5.3.3 Results

The median intake in grams (IQR) per day derived from the FFQ was 264g (164-399) for fruit, 281g (195-393) for vegetables, and 565g (210-400) for total fruit and vegetable including juices. The median total vitamin C derived from the FFQs was 156mg (114-210) for all women and 160mg (117-218) for vegetarians, a difference of only 3%. Mean totals were 171.2mg (sd 86.3) and 178.0mg (sd 93.9), respectively. Vegetarians were defined from the FFQ as eating meat or fish less than once per week. Dietary vitamin C intake calculated from juice, fruit, vegetables and potatoes are shown in Table 15 for all women and for vegetarians, who comprised of 19% of this dataset. The main source of vitamin C was from fruit and vegetables, making up 75% for all women and only slightly more for vegetarians at 76% (as calculated from means). Juice made up 19% of the total dietary vitamin C intake derived from the FFQ.

Table	15	The	main	dietary	sources	of	vitamin	С	at	baseline	in	the	UKWCS	for	all
womer	n ar	d for	veget	arians s	eparately	/									

	All v	vomen	FFQ defined vegetarians			
Vitamin C	Median	Mean	% of	Median	Mean	% of
from	(IQR) mg/d	(sd) mg/d	total <sup>a</sup>	(IQR) mg/d	(sd) mg/d	total <sup>a</sup>
Juice	24.1 (5.1-56.5)	32.4( 38.5)	19%	22.1 (5.1-56.6)	33.0 (41.5)	19%
Fruit	48.9 (28.3-77.2)	61.6 (56.7)	36%	51.3 (29.7-81.3)	65.5 (64.0)	37%
Vegetable	57.7 (36.9-85.9)	66.3 (43.3)	39%	60.8 (39.7-88.8)	69.5 (44.6)	39%
Potatoes	14.0 (9.2–19.2)	15.1 (9.4)	9%	12.4 (7.51-18.3)	13.8 (10.1)	8%

<sup>a</sup>Percentage of totals are based on means (these add up to slightly more than 100% due to some errors in splitting vitamin C between sources)

The median (IQR) daily servings of fruit and vegetables reported on the cross-check questions were 1.4 (1-2) and 1.4 (1-2), whereas the median servings derived from the FFQ were much higher: 4.3 (2.7-6.5) and 4.7 (3.2-6.6). The medians from the cross-check questions are significantly different from the FFQ servings (P<0.001). For total fruit and vegetables from the cross-check questions and from the FFQ the mean servings were 2.8 (sd 1.8) and 10.6 (sd 6.1) respectively. These averages excluded women with missing data. Of the 33,521 women, 731 (2%) had missing data for both cross-check questions, 4122 (12%) had missing responses for fruit only and 1397 (4%) for vegetables.

# 5.3.3.1 Comparisons between methods of recording fruit and vegetables

There were some anomalies relating to juice in the comparisons for fruit. Some participants may have included fruit juice in their responses to cross-check questions, others may not. Up to one portion of fruit juice per day was allowed in the calculation of number of fruit servings derived from the FFQ, which had been previously calculated.<sup>330</sup>Juice was not included in 'fruit in grams' and in 'vitamin C from fruit', both

derived from the FFQ. Comparisons relating to vegetables did not include potatoes, which made up 8-9% of total vitamin C intake. Participants were asked not to include potatoes in the vegetable cross-check question.

The log-transformed Bland-Altman plots in Figure 20 show that for 95% of women the fruit and vegetable servings per day derived from the FFQ exceeds the cross-check question recordings by between 1.2 and 11.9 times. The mean ratios and 95% confidence limits were similar for vegetarians (Figure 20) and non-vegetarians (not shown) and for fruit and vegetables separately (Figure 21).

**Figure 20** Bland-Altman plots of ratios between FFQ and cross-check question recording total fruit and vegetable servings for all women and for vegetarians only



**Figure 21** Separate fruit and vegetable Bland-Altman plots of ratios between servings measured by FFQ and by cross-check question





Despite the large reported discrepancies found above there was moderate correlation with Spearman's ranked correlations coefficients between the methods for recording servings for fruit ( $r_s$ =0.54) and for vegetables ( $r_s$ =0.40). There was also moderate correlation between the cross-check answers and the FFQ derived fruit or vegetable in grams or as vitamin C as seen in Table 16 and Table 17. The correlation was strong between servings (or grams) of fruit or vegetables derived from the FFQ and the vitamin C from fruit calculated from the FFQ, and similarly for vegetables ( $r_s$ =0.91 and 0.86).

 Table 16
 Spearman's correlations between fruit as servings and as grams, and between derived vitamin C intake

Comparison variables for f	Spearman's <sup>a</sup>	Correlation <sup>b</sup>	
X-check fruit serving <sup>c</sup>	FFQ fruit servings <sup>d</sup>	0.54 (p<0.001)	Moderate
X-check fruit serving	FFQ fruit (g) <sup>e</sup>	0.61 (p<0.001)	Moderate
X-check fruit serving	FFQ vit C from fruit <sup>f</sup>	0.54 (p<0.001)	Moderate
FFQ fruit servings	FFQ vit C from fruit	0.91 (p<0.001)	Strong
FFQ fruit (g)	FFQ vit C from fruit	0.89 (p<0.001)	Strong

<sup>a</sup>Spearmans's ranked correlation coefficients for non-parametric data

<sup>b</sup>Given strength of correlation is based on classification of Peason's correlation coefficients

<sup>c</sup>Fruit servings consumed, obtained from cross-check questions

<sup>d</sup>Fruit servings consumed, derived from the FFQ

<sup>e</sup>Grams of fruit consumed, derived from the FFQ

<sup>f</sup>Dietary vitamin C intake from fruit, derived from the FFQ

**Table 17** Spearman's correlations between vegetables as servings and as grams, and between derived vitamin C intake

Comparison variables for v	Spearman's	Correlation	
X-check veg serving	FFQ veg servings	0.40 (p<0.001)	Moderate
X-check veg serving	FFQ veg (g)	0.40 (p<0.001)	Moderate
X-check veg serving	FFQ vit C from veg	0.37 (p<0.001)	fair
FFQ veg servings	FFQ vit C from veg	0.86 (p<0.001)	Strong
FFQ veg (g)	FFQ vit C from veg	0.90 (p<0.001)	Strong

See notes to Table 16

Alternatively, using the Kappa method to measure the extent of the agreement, the fruit serving cross-check responses were split into five roughly equal groups and cross-tabulated against fruit servings derived from the FFQ split similarly (Table 18). Only 35% of participants were classified in the same fifths producing a kappa statistic of 0.19, of slight agreement (Table 19). However, 74% were classified in the same or adjacent fifths producing a weighted Kappa of 0.36 of fair agreement. (The kappa weighted method is explained in section 5.3.2). At the extremes, 9% were classified in groups greater than two groups difference. As seen in Table 19 the weighted Kappa

was similar for comparisons between fruit serving cross-check responses and vitamin C from fruit derived from the FFQ ( $K_w$ =0.37). As expected the latter's agreement with fruit as grams derived from the FFQ was substantial ( $K_w$ =0.71).

Cross-check	FFQ derived fruit servings per week								
fruit group	1 (0-16)	2 (17-25)	3 (26-34)	4 (35-49)	5 (50 +)	Total (%)			
(servings/week)	20%	20%	18%	21%	21%	100%			
1 (0-5)	3,165	1,257	695	445	286	5,848 (20)			
2 (6-7)	1,458	1,742	1,312	1,156	849	6,517 (22)			
3 (8-11)	668	1,350	1,265	1,211	851	5,345 (18)			
4 (12-14)	345	1,047	1,309	1,603	1,626	5,930 (20)			
5 (15 & above)	145	530	882	1,674	2,526	5,757 (20)			
Total	5,781	5,926	5,463	6089	6,138	29,397 (100)			

**Table 18** Tabulation of fruit servings from cross-check answers and those derived from

 the FFQ excluding women with missing data

**Table 19** Kappa agreements between groupings of fruit consumed as servings, intake

 in grams and derived vitamin C intake

Comparison variables		Kappa (SE)	Agreement <sup>321</sup>
X-check fruit servings	FFQ fruit servings	0.19 (0.0029)	Slight
		0.36 (0.0040) <sup>w</sup>	Fair
X-check fruit servings	FFQ fruit (g)	0.23 (0.0029)	Fair
		0.42 (0.0040) <sup>w</sup>	Moderate
X-check fruit servings	FFQ vit C from fruit	0.19 (0.0029)	Slight
		0.37 (0.0040) <sup>w</sup>	Fair
FFQ vit C from fruit	FFQ fruit (g)	0.48 (0.0029)	Moderate
		0.71 (0.0040) <sup>w</sup>	Substantial

<sup>w</sup>Weighted to taken into account observations allocated to adjacent groups (see section 5.3.2) SE= standard error

Women with missing fruit cross-check serving data were excluded from all the above kappa calculations

It proved difficult to split the vegetable serving cross-check responses into five roughly equal groups due to their uneven distribution. When these were cross-tabulated against vegetables servings derived from the FFQ split similarly (Table 20) only 29% of participants were classified in the same fifths producing a Kappa statistic of 0.11, of slight agreement (Table 21). However, 72% were classified in the same or adjacent fifths producing a weighted Kappa of 0.25 of fair agreement. At the extremes 9% were classified in groups greater than two groups difference. The weighted Kappa was similar for comparisons between vegetable serving cross-check responses and vitamin C from vegetables derived from the FFQ ( $K_w$ =0.24). The latter's agreement with vegetables as grams derived from the FFQ was substantial ( $K_w$ =0.72).

		3		5		
Cross-check	FFQ derived vegetable servings per week					
vegetable group	1 (0-18)	2 (19-28)	3 (29-38)	4 (39-54)	5 (55+)	Total (%)
(servings/week)	17%	23%	22%	23%	15%	100%
1 (0-6)	2,340	1,553	1,001	621	252	5,767 (18)
2 (7)	1,530	2,000	1,642	1,473	791	7,436 (23)
3 (8-11)	982	1,830	1,703	1,529	790	6,834 (21)
4 (12-14)	574	1,411	1,771	2,154	1,628	7,538 (24)
5 (15 & above)	171	635	1,061	1,486	1,194	4,547 (14)
Total	5,597	7,429	7,178	7,263	4,655	32,122 (100)

**Table 20** Tabulation of vegetable servings from cross-check answers and those derived from the FFQ excluding women with missing data

 Table 21 Kappa agreements between grouping of vegetable consumed as servings, intake in grams and derived vitamin C intake

Comparison variables		Kappa (SE)	Agreement <sup>321</sup>
X-check veg servings	FFQ veg servings	0.11 (0.0028)	Slight
		0.25 (0.0038) <sup>w</sup>	Fair
X-check veg servings	FFQ veg (g) <sup>e</sup>	0.11 (0.0028)	Slight
		0.25 (0.0038) <sup>w</sup>	Fair
X-check veg servings	FFQ vit C from veg <sup>f</sup>	0.11 (0.0028)	Slight
		0.24 (0.0038) <sup>w</sup>	Fair
FFQ vit C from veg	FFQ veg (g)	0.49 (0.0028)	Moderate
		0.72 (0.0038) <sup>w</sup>	Substantial

<sup>w</sup>Weighted to taken into account observations allocated to adjacent groups (see section 5.3.2)

SE= standard error

Women with missing vegetable cross-check serving data were excluded from all the above kappa calculations

The Kappas produced for comparisons of total fruit and vegetables were similar (not shown).

# 5.3.3.2 Women with high discrepancies

Fruit servings derived from the FFQ were more than double the responses for crosscheck questions for 78% of women who completed the cross-check questions, similarly for 82% of women relating to vegetable servings. FFQ derived servings were more than five times larger than the cross-check answers for about a quarter of the women (23% of the all women reporting fruit and 24% reporting vegetables, and 21% and 28% respectively for FFQ defined vegetarians). The characteristics of women with these large discrepancies are shown in Table 22. On average, high discrepancy fruit and vegetable reporters had higher median FFQ servings per day compared to other women but lower median servings per day reported on the cross-check questions compared to others (Table 22). These women, including the FFQ defined vegetarians amongst them, had a median of one serving per day for fruit and for vegetables as per the cross-check question. FFQ derived servings were seven times that of the crosscheck answers for over half of the high discrepancy reporters, and for about 10% of total women.

	High discrepancy		Other women	
	Fruit	Veg	Fruit	Veg
	reporting	reporting	reporting	reporting
FFQ servings, median (IQR)	6.4 (4.3-9.8)	6.7 (5.2-8.8)	4.0 (2.6-5.8)	4.1 (2.9-5.7)
X-check servings per day,	1.0 (0.6-1.4)	1.0 (0.7-1.3)	1.4 (1-2)	1.4 (1-2)
median (IQR)				
Age years, mean (sd)	52.4 (9.3)	51.6 (9.2)	51.6 (9.2)	52.0 (9.3)
BMI kgm <sup>-2</sup> , mean (sd)	24.5 (4.3)	24.4 (4.3)	24.4 (4.3)	24.4 (4.3)
Typical food portions <sup>a</sup> -small	1033 (27%)	1089 (26%)	2783 (73%)	3105 (74%)
medium	5122 (23%)	5814 (24%)	17109(77%)	18492(76%)
large	659 (22%)	778 (24%)	2397 (78%)	2523 (76%)
Vegetarian - Yes	1188 (21%)	1679 (28%)	4342 (79%)	4294 (72%)
No	5695 (24%)	6075 (23%)	18174(76%)	20076(77%)
Low fat diet - Yes	1829 (24%)	2324 (28%)	5932 (76%)	6113 (72%)
No/missing	5054 (23%)	5430 (23%)	16584(77%)	18257(77%)
Slimming diet -Yes	505 (21%)	724 (27%)	1924 (79%)	1930 (73%)
No/missing	6378 (24%)	7030 (24%)	20592(76%)	22440(76%)
Socio-economic status -1	2437 (21%)	2794 (22%)	9011 (79%)	9660 (78%)
2	2951 (24%)	3294 (25%)	9374 (76%)	10124(75%)
3	1354 (26%)	1508 (26%)	3798 (74%)	4207 (74%)
Educated to degree - Yes	1481 (19%)	1736 (21%)	6150 (81%)	6476 (79%)
No	5402 (25%)	6018 (25%)	16366(75%)	17894(75%)
Supplement user - Yes	2572 (23%)	2798 (23%)	8540 (77%)	9371 (77%)
No	4223 (23%)	4855 (25%)	13788(77%)	14781(75%)

 Table 22 Characteristics of women with large discrepancies between methods of measuring fruit and vegetable portions

<sup>a</sup>Self-categorised portions

High discrepancies in vegetable reporting were more likely for women who were vegetarian, or on a low fat or slimming diet (p<0.001), or eating small portions (p=0.01) or not supplement users (p<0.001), despite there being no differences in BMI on average between them and other women. There was no such clear pattern between diet and high discrepancy fruit reporting, though eating small portions (p<0.001) and not being on a slimming diet (p=0.001) was significantly associated with discrepancies. Non-graduates and women from lower SES were more likely to report high discrepancies for either fruit or vegetables (p<0.001).

# 5.3.4 Discussion

The majority of the dietary vitamin C estimated from FFQ recordings at baseline in the UKWCS came from fruit and vegetables (75%) and the remaining was from juice and potatoes. This compares with an estimate of 62% vitamin C from fruit and vegetables using 24-hour recall methods for 264 women from the UKWCS.<sup>331</sup> In the current analysis, Spearman's correlations and weighted kappa statistics confirmed that FFQ

derived grams of fruit and vegetables had strong agreement with total derived dietary vitamin C. However the corresponding non-weighted kappa statistics showed only moderate agreement; this may reflect the fact that some women may consume some types of fruit and vegetables that are more vitamin C rich than others. Furthermore, the accuracy of vitamin C derived from diet records is limited by the accuracy of food tables that are used to convert recordings into amounts of vitamin C consumed, in addition to limitations relating to the allocation of portion weight and recording of frequency of consumption. The derivation methods used for the UKWCS have been explained in section 4.5.1 and elsewhere.<sup>288</sup> A review found that the inclusion of portion-size questions produced lower correlation coefficients specifically for vitamin C, than if standard portions were used.<sup>332</sup> Although the UKWCS FFQ included a question about portion sizes, it was not used to derive nutrient values; a standard portion size was used.

The vitamin C intake estimates for vegetarians in this UKWCS baseline dataset were only 3% more than for non-vegetarians. Women in this cohort were generally more health conscious than the general population and may be likely to consume high intakes of fruit and vegetables, whether or not they are vegetarians. Additionally, they may have given socially desirable answers they believed were related to the aims of the research.<sup>333</sup>

Spearman's correlation method using relative rankings showed moderate agreement between the cross-check answers and other variables; however other methods such as the Bland-Altman and the Kappa statistics showed weaker agreement. In line with the initial analysis of 6,572 women at baseline in the UKWCS,<sup>288</sup> the current results confirmed there were considerable inconsistencies between the two variables representing fruit and vegetable servings. The initial analysis showed 15% of participants had differences of more than five times between fruit serving derived from the FFQ and those reported on the cross-check questions. With regard to vegetables, 26% of women reported a difference of more than five times. <sup>288</sup> These current results showed a similar percentage for vegetables (25%) and a much higher percentage for fruit (23%).

Reasons why most FFQ derived servings were much higher than cross-check responses could be because participants may have misunderstood how to complete the FFQ; they may have ticked too many items listed on the FFQ or over-stated their frequency of consumption. For instance, from checks of questionnaires of women with very high fruit and vegetable intake it was found that one woman had ticked every fruit and vegetable listed on the FFQ as being consumed once or 2-3 times per day, yet for the cross-check response she stated vegetable servings consumed were only 14 per

123

week, adding 2 per day as clarification. Research has shown that in comparisons of FFQs using 20, 33 or 40 fruit and vegetable items, those with more items lead to higher reporting of fruit and vegetables, ranging from a mean intake of 23.8 to 38.8 portions per week.<sup>334</sup> By comparison the UKWCS FFQ had 55 fruit and vegetable items in total and a mean total of 10.6 portions per day equating to 74 portions per week, which indicates an over-estimation of intake. Some discrepancies may have occurred due to counting vegetables twice; once in mixed dishes and once in the separate listings.<sup>335</sup> On the other hand, many women may have misread the cross-check questions as servings per day instead of per week. Indeed servings per day calculated from crosscheck answers were over seven times lower than FFQ servings for more than 10% of the women. The fact that high discrepancy fruit and vegetable reporters had both a higher median FFQ serving per day compared to other women but lower median servings per day reported on the cross-check questions compared to others, supports all explanations given above. Nevertheless the histograms in Figure 19 clearly show that a large proportion of women reported intake in multiples of seven, indicating that they correctly reported intake per week rather than per day. However, since the crosscheck answers for many of these women, including the vegetarians amongst them, were on average much lower than the national average intake of fruit and vegetables at the time, <sup>336</sup> this indicates that many of them may have misinterpreted the cross-check question as servings per day rather than servings per week. On the other hand, the median FFQ intake of high discrepancy reporters was well above the national average calculated from a sample of 6000 households in 1998 (4.25 portions based on a total average of 340g of fruit and vegetable intake per day which included fruit juice and excluded potatoes).<sup>336</sup> However, this difference could be due to the women in the UKWCS being highly health conscious rather than it being an over-estimation.

Possible errors in fruit and vegetables and vitamin C intake derived from FFQs can be attenuated by applying a correcting factor for each subject using answers to summary cross-check questions as demonstrated in previous studies.<sup>288</sup> <sup>325-327</sup> For the 6,572 women in the initial UKWCS analysis, each person's intake was adjusted by dividing the response to the cross-check question on the UKWCS questionnaire by the sum of frequencies from the FFQ, for each food group, as shown below:

# <u>Number of servings/week from cross-check question</u> Number of servings/week from individual food items on the FFQ

After applying the weightings the mean vitamin C intake reduced by 44% from 151 to 85 mg/d.<sup>288</sup> One study found that the correction of FFQ derived vitamin C improved its Pearson's correlation coefficients with 4-d diet records.<sup>326</sup> Another study found that

rankings remained relatively unchanged after corrections which reduced vitamin C intake by 20%.<sup>325</sup>

The above weightings could be used to adjust FFQ derived vitamin C used in the analyses of associations between breast cancer risk and vitamin C intake in the UKWCS. However, it was decided not to apply this weighting, since it was unclear whether the cross-check answers reflected true intake or at least true ranking of fruit and vegetable servings any better than the FFQ in the UKWCS. Similar concerns were expressed by Amanatidis et al. (2001) who also warned against comparing corrected or uncorrected FFQ figures with national target intakes.<sup>325</sup>

A review of studies found that FFQ recordings of vegetable intake had weak correlations (r=0.39) with other assessment methods, whereas fruit produced somewhat higher correlations (r=0.49).<sup>337</sup> Since then the Norfolk-EPIC study reported their FFQ recorded fruit and vegetable intake was about double that recorded by 7-day diaries.<sup>338</sup> Given that the UKWCS FFQ is based on the Norfolk-EPIC FFQ but has more fruit and vegetable items, its average intake may be more than double that recorded by diaries. Comparisons with biomarkers, or weighed food records, are considered appropriate methods of evaluating dietary assessment methods, but neither was employed at baseline in the UKWCS, and therefore could not be used to evaluate fruit and vegetable intake or vitamin C intake derived from the UKWCS FFQ. However, in the next section vitamin C derived from a similar FFQ is compared with biomarkers and food diary records at phase two on a small sample of women who provided blood samples.
# 5.4 Comparison of vitamin C intake recorded by FFQ and by diary with biomarkers in a small sample of the UKWCS at phase 2

#### 5.4.1 Introduction

Biomarkers are used to assess the nutrient status of individuals, and have also been used to investigate associations between nutrient status and disease risk. In relation to plasma vitamin C and breast cancer risk, no evidence of an association was fund in a nested case-control study,<sup>268</sup> and conflicting evidence has been produced in retrospective case-control studies.<sup>339-341</sup>

Biomarkers have also been used as objective measures to assess the validity of dietary measurement tools, although their usefulness in this respect is questionable for some nutrients; since the absorption and storage of vitamin C is limited, particularly above 400mg/d, <sup>65</sup> vitamin C plasma levels are unlikely to reflect dietary vitamin C intake well. Previously 4-d diary and 24hr records of estimated vitamin C intake have been compared with biomarkers in a small sample of women from the UKWCS.<sup>331</sup> Results of the regression analysis showed that plasma ascorbic acid was a good indicator of previous vitamin C intake from all foods sources but not a good proxy for specific food groups such as fruit and vegetables recorded by these dietary methods. Conversely, an EPIC-Norfolk study found plasma vitamin C correlated weakly but significantly with fruit and vegetables recorded by FFQ.<sup>342</sup> In a systematic review of dietary assessment methods, vitamin C intake estimated from FFQs correlated weakly with biomarkers, the mean correlation coefficient being only 0.28.343 However, in an earlier review, Cade et al. (2004) found correlations with biomarkers in 15 studies were somewhat higher at r=0.46.<sup>337</sup> The large difference between reviews may be due to the small number of studies involved, the assay methods used to measure biomarkers, or the quality weighting that studies were assigned in the later review.

Diaries have been the reference method most often used to validate FFQs; these have shown moderate correlations for vitamin C of r=0.50 in two reviews,<sup>337 343</sup> and slightly higher in a third review (r=0.59).<sup>332</sup>

This analysis will compare three methods of measuring vitamin C intake in a small subsample of women from the UWKCS, measured in close proximity and derived from: the 217 item FFQ; the 4-d diary; and blood samples. The resulting correlation coefficients and regression analysis will be compared with previous studies.

#### 5.4.2 Methods

#### 5.4.2.1 Subjects

Phase 2 UKWCS participants who lived within a 30 mile radius, or one hour drive of Leeds, were contacted between 3 and 5 years after baseline and a nurse arranged to take blood samples from 273 of these women.<sup>280</sup> The women also completed a 4-d diary and a FFQ which were returned to the nurse when the blood samples were collected. Dietary vitamin C intake was derived from the 4-d diaries and also from the FFQ, and vitamin C supplement dose was also recorded in the FFQs.

#### 5.4.2.2 Dietary assessment methods

Blood samples were collected into lithium heparin (8ml) after an overnight fast. They were kept cool and within 2 hours of collection were separated and prepared for storage at -70°C.<sup>280</sup> Plasma ascorbic acid and total vitamin C (ascorbic acid and dehydroascorbic acid) was analysed by high-performance liquid chromatography,<sup>344</sup> in the Division of Pathological Sciences, Department of Clinical Medicine at the University of Leeds. The food listings FFQs used for this dataset were almost identical to the FFQ used at baseline. The diaries used were the ones used in the full phase 2 dataset.

#### 5.4.2.3 Statistical analyses

Characteristics of women grouped into fifths according to self-recorded vitamin C intake were produced. The strength of the associations between plasma ascorbic acid or total vitamin C (which includes dehydroascorbic acid as well as ascorbic acid) and vitamin C intake derived from FFQs and diary records at phase 2 was calculated using Spearman's pairwise correlation coefficients, since the data was not normally distributed. The analysis was repeated for non-vitamin C supplement users; i.e. those women who did not record either frequency or dose of vitamin C supplements taken. Spearman's correlation between vitamin C intake derived from phase 2 FFQs and diary records was also calculated. Spearman's method was used as opposed to Pearson's method for parametric data because the vitamin C intake derived from phase 2 FFQs and diary records in this dataset was not normally distributed. In the pairwise correlation coefficient calculations, pairs are excluded if one of the two variables is missing. Due to vitamin C's role in the regeneration of vitamin E and in the absorption of iron and in the Fenton reaction,<sup>76</sup> correlations between plasma ascorbic acid or total vitamin C and plasma iron and plasma vitamin E concentration were also assessed.

Regression analysis was used to quantify the percentage increase in plasma ascorbic acid and plasma total vitamin C when vitamin C consumed from all foods were doubled, as derived from diary or from FFQ recordings. For regression estimates to be

valid, it is assumed the residuals (.i.e. the difference between actual values and the estimated line of best fit) are normally distributed and have uniform variance. These assumptions were examined using histograms of residues, and q-q plots of residues against the predictor variable. Since these did not follow normal distribution patterns, it was decided to log-transform the vitamin C variables for the regression analysis. The analysis of all women was adjusted for vitamin C supplement use and energy intake, and the sub-analysis by vitamin C supplement taking was adjusted for energy intake.

Although recommended,<sup>337</sup> Bland-Altman plots were not used to assess correlations between plasma concentrations and FFQs and diaries because of the different units used between assessment methods (Nevertheless, standardized units, such as standard deviations, could have been compared).

#### 5.4.3 Results

The mean vitamin C derived from diary recordings was 128 mg compared to 191 mg from FFQs. When women were split into fifths according to dietary vitamin C intake recorded by diary and by FFQ, their mean plasma ascorbic acid and plasma total vitamin C recordings increased with increasing fifths (Table 23 & Table 24). On average women were older and more likely to self-report being a vegetarian in each fifth of increasing vitamin C intake. There were no clear patterns for BMI or vitamin C supplement use.

As observed in Table 25, the estimates of dietary vitamin C intake from diaries and from FFQs correlated weakly with plasma ascorbic acid and plasma total vitamin C with diaries showing stronger correlations (r=0.35 and 0.34 vs. 0.26). When vitamin supplement dose recorded in the FFQ was added to the dietary FFQ value then correlations with plasma total vitamin C were 0.36 and 0.35. Correlations of plasma concentrations were even weaker with fruit and vegetable portions recorded by questionnaire, being higher for fruit than vegetables (0.20 and 0.21 vs. 0.08 and 0.13). Correlations between plasma ascorbic acid or total vitamin C and plasma iron and plasma vitamin E were also poor (0.09-0.18). In general the correlations were higher when 115 women (42%) were excluded who recorded vitamin C supplement use in either the frequency or dose section of the FFQ (Table 26). Dietary vitamin C recorded by diary for all women had a substantial correlation with that recorded by FFQ (r=0.49, Table 25). For 21 women (12%) taking supplement doses of 500mg or more vitamin C, substantial correlations were also found between supplement vitamin C intake and plasma ascorbic acid or plasma total vitamin C, being 0.60 and 0.48 respectively.

	Dieta					
	lowest	2nd fifth	3rd fifth	4th fifth	highest	Total
Dietary vitamin C, mean mg (sd)	85.6 (20.7)	135.2 (11.2)	175.2 (10.8)	214.7 (14.2)	340.8 (104)	190.8 (99.5)
Plasma ascorbic acid, µg ml <sup>-1</sup> (sd)	10.5 (5.4)	11.5 (3.5)	12.6(3.8)	11.8 (3.3)	14.0 (4.5)	12.1 (4.3)
Plasma vitamin C, μg ml <sup>-1</sup> (sd)	12.1 (5.0)	13.4 (3.2)	14.1 (4.1)	13.7 (3.1)	15.6 (4.6)	13.8 (4.2)
Age, mean (sd)	49.8 (8.5)	48.8 (8.0)	49.9 (9.2)	50.3 (8.4)	53.8 (9.2)	50.5 (8.8)
BMI, mean (sd)	25.7 (5.6)	23.2 (3.1)	24.2 (4.0)	24.5 (3.9)	24.9 (4.3)	24.5 (4.3)
Vit C supplement use, n (%)	22(42)	22(42)	20 (37)	28 (53)	22 (41)	114 (43)
Vegetarian self-reported, n (%)	17 (34)	18 (35)	17 (33)	20 (38)	22 (42)	94 (37)

 Table 23 Characteristics of women by dietary vitamin C intake recorded by FFQ

 Table 24 Characteristics of women by dietary vitamin C intake recorded by diary

	Dietar					
	lowest	2nd fifth	3rd fifth	4th fifth	highest	Total
Dietary vitamin C, mean mg (sd)	42.5 (13.6)	75.6 (9.2)	106.2 (10.2)	146.1 (13.4)	266.8 (122)	127.9 (95.7)
Plasma ascorbic acid µg ml <sup>-1</sup> (sd)	10.3 (5.5)	11.4 (3.9)	11.7 (2.8)	13.0 (3.8)	13.9 (4.2)	12.1 (4.3)
Plasma vitamin C µg ml <sup>-1</sup> (sd)	11.8 (5.0)	13.3 (3.4)	13.2 (2.9)	15.1 (3.4)	15.6 (4.1)	13.8 (4.1)
Age, mean (sd)	48.7 (8.0)	49.2 (8.5)	50.4 (8.4)	51.0 (8.2)	53.6 (10.0)	50.6 (8.7)
BMI, mean (sd)	25.1 (4.5)	24.2 (4.4)	24.8 (4.8)	24.2 (4.6)	24.3 (3.4)	24.5 (4.4)
Vit C supplement use, n (%)	27 (52)	19 (37)	20 (38)	23 (44)	25 (47)	114 (44)
Vegetarian self-reported, n (%)	13 (26)	18 (38)	15 (29)	23 (44)	23 (44)	92 (37)

	Plasma		Diary	FFQ		Questionnaire		
	Ascorbic	Ascorbic Total		Dietary Total		Fruit	Veg	Total
	acid	vit C	vit C	vit C	vit C	portions	portions	portions
Plasma ascorbic acid	1							
Plasma total vitamin C	0.76	1						
Diary dietary vitamin C	0.35	0.34	1					
FFQ dietary vitamin C	0.26	0.26	0.49	1				
FFQ total vitamin C	0.36	0.35	0.33	0.79	1			
Questionnaire fruit portions	0.20	0.21	0.39	0.43	0.31	1		
Questionnaire veg portions	0.08	0.13	0.23	0.27	0.24	0.47	1	
Total fruit & veg portions	0.17	0.19	0.34	0.43	0.36	0.86	0.82	1
Plasma iron	0.15	0.18	0.09	0.05	0.11	-0.01	0.02	0.05
Plasma vitamin E	0.16	0.09	-0.01	0.14	0.25	-0.06	0.05	0.06

**Table 25** Spearman's correlation coefficients between different measures of vitamin C for all UKWCS women who provided blood samples

Numbers in bold are statistically significant p<0.05

(N varies from 259 to 272)

	Plasma		Diary	FFQ	Questionnaire		re
	Ascorbic Total		Dietary	Dietary	Fruit	Veg	Total
	acid	vit C	vit C	vit C	portions	portions	portions
Plasma ascorbic acid	1						
Plasma total vitamin C	0.73	1					
Diary dietary vitamin C	0.46	0.49	1				
FFQ dietary vitamin C	0.36	0.35	0.51	1			
Questionnaire fruit portions	0.33	0.31	0.41	0.46	1		
Questionnaire veg portions	0.15	0.16	0.27	0.31	0.55	1	
Total fruit & veg portions	0.25	0.22	0.32	0.46	0.88	0.86	1
Plasma iron	0.13	0.24	0.04	0.06	-0.03	0.00	0.05
Plasma vitamin E	0.10	0.01	-0.05	0.18	0.09	-0.03	0.10

**Table 26** Spearman's correlation coefficients between different measures of vitamin C for non-vitamin C supplement users<sup>a</sup> in the sub-sample of UKWCS women who provided blood

Numbers in bold are significant

<sup>a</sup>Not indicated as vitamin C supplement users on the questionnaire

(N varies from 148 to 157)

The regression analyses showed that plasma ascorbic acid concentrations increased significantly by 19% when total vitamin C from all food sources recorded by diary was doubled (Table 27). When vitamin C supplement users were excluded this rose to 23%, but for the sub-analysis for non-vitamin C users the increase was only 15%. Similar results were found for recordings by FFQ. Increases in total plasma vitamin C concentrations were not as large; an increase of 14% occurred when total vitamin C from all food sources recorded by diary were doubled. When vitamin C supplement users were excluded this rose to 19%. Again, similar results were found for recordings by FFQ. The R<sub>2</sub> values ranged from 0.07 for vitamin C supplement users to 0.26 for women who did not take vitamin C supplements, indicating that plasma concentrations were not explained well by FFQ and diary recordings, particularly for supplement users.

 Table 27 Percentage increase in plasma concentrations of ascorbic acid and total vitamin associated with a doubling of vitamin C from all foods recorded by diary and by FFQ

	Plasma	Plasma
	ascorbic acid	vitamin C
	% increase (95%CI)	% increase (95%CI)
Total women (N=262) <sup>a</sup>		
Diary dietary vitamin C intake	18.9 (13.1, 25.0)	13.6 (9.3, 18.0)
FFQ dietary vitamin C intake	19.9 (12.0, 28.3)	14.4 (8.6, 20.4)
Excluding vitamin C supplement users (	N=148) <sup>b</sup>	
Diary dietary vitamin C intake	23.0 (15.3, 31.3)	18.5 (12.8 24.4)
FFQ dietary vitamin C intake	26.9 (16.4, 38.4)	18.9 (11.2, 27.3)
Vitamin C supplement users (N=114) <sup>b</sup>		
Diary dietary vitamin C intake	14.5 (5.7, 24.0)	8.8 (2.3, 15.6)
FFQ dietary vitamin C intake	12.5 (1.1, 25.2)	9.1 (0.8, 18.2)

<sup>a</sup>Adjusted for energy intake and supplement taking

<sup>b</sup>Adjusted for energy intake

#### 5.4.4 Discussion

The FFQ vitamin C correlations with plasma ascorbic acid and also with diary derived dietary vitamin C for the UKWCS were weak (ranging from 0.26 to 0.51) but in line with the mean coefficients between these measurement methods calculated in two reviews.<sup>337 343</sup> However, the UKWCS results are different from many of the energy adjusted correlations between plasma ascorbic acid and vitamin C intake for women derived from EPIC FFQs from a variety of European countries; these correlations ranged from 0.04 in Greece, to 0.29 in Italy, to 0.65 in Spain.<sup>345</sup> As observed in a previous EPIC-Norfolk study,<sup>346</sup> UKWCS diary derived vitamin C intakes correlated better with plasma ascorbic acid than FFQ data when they were recorded in close

temporal proximity. Both plasma concentrations and diaries reflect relatively short-term intake compared to FFQs which usually assess the previous 12 months' intake. However, in the previous study, correlations with biomarker levels re-measured several years later were found to be similar for diaries and FFQs. <sup>346 347</sup> In the current study, although significant, correlations between plasma ascorbic acid and total fruit and vegetables portions were even weaker than those with FFQ derived vitamin C intake; reflecting the fact that fruit and vegetables, excluding juice and potatoes, make up only about two thirds of total vitamin C consumptions, as observed in section 5.3.3. Similarly, the weak correlations between ascorbic acid and plasma iron and plasma vitamin E were expected to be greater due to ascorbic acids' role in the regeneration of vitamin E, in the absorption of iron and in the Fenton reaction.<sup>76</sup>

The most recent review found no difference in correlations for vitamin C between FFQs and diaries which recorded seven or more days' intake and those recording less than seven days.<sup>343</sup> However another review observed lower correlations for diaries recording less than seven days and these were more in line with the 4-d UKWCS diary correlations than the longer diaries (0.55 vs. 0.63).<sup>332</sup> Additionally, the most recent review found no large differences in correlations between studies that included vitamin supplements and those that did not;<sup>343</sup> however, this result has been questioned due to inconsistencies found in classifying studies as those including supplements.<sup>348</sup> Contrary to this, correlations for the UKWCS were substantially stronger when women taking supplements were excluded. Indeed dietary only intake is likely to correlate more strongly with plasma levels when the latter does not contain vitamin C from supplements. Surprisingly though, correlations in the UKWCS were the strongest between supplement vitamin C intake and plasma ascorbic acid for women taking high doses (>=500mg/d); such doses are unlikely to be absorbed or stored by the body and plasma levels are likely to fall within hours due to excretion. This may be a spurious result due to low numbers analysed. In contrast, the previous UKWCS analysis reported that plasma levels of ascorbic acid reached a plateau with dietary intakes around 60mg/d,<sup>331</sup> which is much lower than the average dietary vitamin C intake of the UKWCS, and lower than the 400mg/d plateau reported by Levine et al. (2001).65

In one review correlation coefficients between FFQs and references measures were higher in general if participants were allowed to describe their own portion sizes (r=0.54).<sup>337</sup> However another review found that portion-size questions produced lower correlation coefficients specifically for vitamin C (0.60 vs. 0.68).<sup>332</sup> Although a general question about portion sizes (small, medium or large) was asked in the UKWCS, it was not used to calculate the weight of fruit and vegetables consumed and derived vitamin C intake. The review also observed lower vitamin C correlations for FFQs developed

from the EPIC FFQ (as was the case for the UKWCS FFQ) than ones developed from the Willet or Block FFQs (0.53 vs. 0.67 and 0.63).<sup>326 332 345 349</sup> The EPIC based FFQ on average contained substantially more food items than the other two FFQs (154 vs. 113 and 100), but in general FFQs containing 200 items (as was the case for the UKWCS FFQ) correlated slightly better for vitamin C with non-biomarker reference measures than those containing 150 or 100 (r=0.64 vs. 0.60 vs. 0.56).<sup>332</sup>

The results of the regression analysis, which measures the extent of agreement between methods, were inline with those from the earlier study of the UKWCS after excluding women who were vitamin C supplement users.<sup>331</sup> For vitamin C supplement users in the current study, the increase in plasma ascorbic acid from a doubling of dietary intake recorded by diary or FFQ was about half that of the non-users, since their plasma levels were likely to be generally higher due to intake from supplements.

As mentioned in section 2.2.2, there are a number of physiological processes that determine plasma ascorbic acid concentrations meaning they do not simply reflect intake. Plasma ascorbic levels reflect the effects of digestion, absorption, uptake, utilisation, metabolism, excretion, homeostatic mechanisms, intake of other nutrients as well as the influence of health-related behaviours and the presence of chronic illness. Therefore, plasma records naturally will not correlate well with FFQ, diary or 24-hr recordings of intake. Since the human body has limited capacity to absorb and store water-soluble vitamin C,<sup>65</sup> excess is excreted within hours of consumption. Given this, plasma vitamin C biomarkers are likely to reflect short-term intake. In general, FFQs, including the UKWCS FFQ, estimate the previous 12 months' intake; therefore they are unlikely to correlate well with short-term measures of vitamin C as reported in an earlier UKWCS assessment where second samples were taken after 18 months.<sup>331</sup>

Derived values of vitamin C consumed may be inaccurate, however the relative ranking of individuals, important in the risk analyses and observed from the correlations, were acceptable compared to previous studies, although weak. None of the measurement methods of recording vitamin C intake in the UKWCS are ideal and all have strengths and weaknesses; biomarker and diary recordings tend to reflect short-term intake and although FFQs cover longer-term intake and are easier to administer at lower costs, they are limited by their retrospective nature and by problems reporting portion sizes and fruit and vegetable consumption. Nevertheless, the general difficulties in accurately measuring vitamin C consumed and correctly ranking the cohort individuals by intake, needs to be taken into account when interpreting the risk analysis results in the following chapters. Measurement error of intake can lead to substantial reduction of calculated disease-exposure relative risks.<sup>350</sup>

# 5.5 Conclusion from evaluations

The comparisons between FFQs, diary and plasma concentrations in all three sections in this chapter showed some weaknesses in methods used in the UKWCS for recording food and supplements containing vitamin C. However, on the whole the results were in line with those from other studies and were considered suitable for further analyses. Nevertheless, the weaknesses found should be taken into account when interpreting results of associations between vitamin C intake and breast cancer risks in the UKWCS.

In the following chapters, breast cancer risk will be assessed in relation to

- Any supplement use (chapter 7)
- Vitamin C contained in supplements at phase 2 of the UKWCS (chapter 8)
- Dietary vitamin C intake from FFQs (split by supplement users and non-users) at baseline in the UKWCS (chapter 9)
- Total vitamin C intake (from diet and supplements) from dairy records in the UK Dietary Cohort Consortium (chapter 10)

First, in the next chapter, the characteristics of women which predict vitamin C supplement use at phase 2 in the UKWCS will be determine and also whether women who have a history of breast cancer are more likely to use high dose vitamin C supplements.

# **CHAPTER 6**

# 6 Is vitamin C supplement use associated with a personal or family history of cancer and what characteristics predict vitamin C supplement use in UK women? A phase 2 analysis of the UKWCS

## 6.1 Summary

*Objectives:* To determine whether frequent vitamin C supplement use is associated with a personal or family history of cancer and to determine what characteristics predict vitamin C supplement use in UK women.<sup>1</sup>

*Methods:* This cross-sectional analysis of 12,453 women aged between 37-79 years from phase 2 of the UKWCS examines the odds of taking supplements containing vitamin C as recorded in 4-day food diaries, based on lifestyle characteristics and cancer history self-reported by questionnaire.

*Results:* Women frequently taking supplements containing vitamin C, compared to those who did not, had healthier behaviours, including higher consumption of fruit and vegetables. Frequent high dose vitamin C takers (>=1000mg) had a higher socioeconomic status, visited alternative practitioners more often than family or private doctors, and were more likely to be ex-smokers, and to drink little or no alcohol than women who did not take high doses. Women who self-reported having had cancer (OR=1.33; 95% CI: 1.00, 1.76) or specifically breast cancer (OR=1.70; 95% CI: 1.14, 2.55), or reported a family history of cancer (OR=1.16 (95% CI: 0.95, 1.41)) or breast cancer (OR=1.26; 95% CI: 1.01, 1.58) had increased odds of being frequent high dose users after adjusting for socio-demographic and health behaviours.

*Conclusion:* High dose vitamin C use by UK women was associated with healthier behaviours and a history of breast cancer.

# 6.2 Introduction

Despite lack of evidence of benefits,<sup>7 351</sup> vitamin supplement use reported by UK women increased from 17% in 1986/1987 to 41% reported in 2008/9.<sup>24 25</sup> A recent UK survey showed users are most likely to be women above 55 years and of higher socioeconomic status.<sup>26</sup> An analysis of the UK Women's Cohort Study (UKWCS) found that users were significantly more likely to lead healthier lifestyles: to be more physically active; have a lower alcohol intake; a lower body mass index (BMI) and eat diets which met recommended dietary intakes.<sup>27</sup> Therefore they were less likely to need supplements than non-users.<sup>27</sup> Further support for this 'inverse supplement hypothesis' has been found in the UK,<sup>25 28 29</sup> and elsewhere.<sup>30-35</sup>

Moreover, those classifying themselves as high strength supplement users in the 2008 UK Random Location Omnibus Survey were particularly health conscious.<sup>26</sup> They were more likely to consider the health implications of what they ate, to actively look for information about how to stay healthy and to believe that they needed vitamin and mineral supplements to feel and stay healthy.<sup>26</sup> Thirty eight percent of the 801 people surveyed reported taking high strength supplements,<sup>26</sup> and only 6% of these were likely to report they were in excellent health compared to 25% of non-users.<sup>26</sup>

Vitamin C is one of the most commonly used supplements in the UK,<sup>26 352</sup> and is often taken at high doses, however little is known about the characteristics of women who frequently take them. Although many users may take vitamin C supplements to boost their immune systems, suggestions that vitamin C is able to reduce the incidence of colds, have been unsubstantiated in randomised controlled trials.<sup>45</sup> It is unknown whether users take vitamin C supplements for other health reasons, such as reducing cancer risk or cancer reoccurrence. Despite some clear evidence of an association between plasma vitamin C levels and reduced mortality from all causes in men and women, and also reduced cancer mortality in men,<sup>67</sup> there is limited evidence to suggest that supplementation with vitamin C is associated with reduced risk of mortality or cancer.<sup>7 351</sup> The Vitamins and Lifestyle (VITAL) study in the US found no overall association between vitamin C supplement-taking (>=150mg) and prevalent cancer in a cross-sectional analysis,<sup>353</sup> despite general supplement use being widespread in cancer survivors in the US, particularly in breast cancer survivors.<sup>354</sup> However, a US study of women physicians showed those with pre-existing breast cancer were more likely to take vitamin C supplements than women who were free from breast cancer.<sup>31</sup>

An EPIC study reported that UK health-conscious women were more likely to be supplement users if they had prevalent cancer than if they were cancer free.<sup>352</sup> However, no study has examined the relationship between vitamin C supplement use in

UK women and a personal or family history of cancer. The main aim of this analysis was to determine whether women in the UKWCS who reported having had cancer, or reported a relative who had cancer, were more likely to use vitamin C supplements than those who did not report these histories. Breast cancer was the principle focus of the analyses. A further aim, which has not been examined in UK women, was to determine which characteristics and health behaviours predicted vitamin C supplement use, particularly at high doses of 1000mg/d or more.

# 6.3 Methods

UKWCS baseline data was gathered between1995-1998 from 35,367 women who completed a 217-item Food Frequency Questionnaire (FFQ).<sup>279</sup> Further details are found in section 4.1.1.1. At baseline 62% of participants took some type of dietary supplement.

All the initial participants were re-contacted at phase 2 between 1999 and 2004, on average 4 years after recruitment; 12,453 (35%) completed a follow-up health and lifestyle questionnaire and a 4-day food diary. For each day, the diaries requested supplement brand, name, amount taken and dosage of any supplement taken. This information was matched against a database of supplement descriptions and ingredient composition as explained in section 4.5.3.1 of the methods chapter. The average daily vitamin C intake contained in all supplement types was calculated for the total number of diary days vitamin C was taken.

Univariate logistic regression was applied to the phase 2 data to determine which participant characteristics predicted frequent supplement use in two different classifications of users: those taking any dose of vitamin C (yes/no); and those taking high doses of vitamin C (>=1000mg, yes/no). These were compared to women not frequently taking 'any' or 'high' doses respectively. Intakes above 1000mg/d have been linked to adverse effects, particularly gastrointestinal disturbance.<sup>52 355</sup> This level is more than 15 times the recommended daily allowance (EU RDA =  $60 \text{ mg/day}^{53}$ ) normally found in multivitamins. Frequent use of supplements containing vitamin C in this study was defined as taking on at least three out of the four diary days. As described in section 5.2, this agreed well with daily use of vitamin C and multivitamin supplements reported on the phase 2 questionnaire, which was completed concurrently with the diary. Socio-demographic and health related lifestyle variables that were significantly associated with either any dose or high intake were all included in a logistic regression model for mutual adjustment. Variables that remain significant should be predicative of vitamin C use over and above the effects of the other variables in these multivariable models.

Social class and marital status variables were derived from answers to the questionnaire used at baseline since this information was not provided at phase 2. All other variables were taken from responses to the phase 2 questionnaire: BMI (kg/m<sup>2</sup>); smoking status; level of physical activity; parity; drinking alcohol less than once a week, red meat servings; total fruit and vegetable servings; frequency of visits to doctors and alternative practitioners. Vigorous activity was defined as activity causing shortness of breath, rapid heart rate and sweating. Attendance at routine health checks was not significantly associated with vitamin C supplement use, and therefore was excluded from the models.

These variables, excluding visits to doctors and alternative practitioners, were used in logistic regression analyses to adjust the odds of women with a family or personal history of cancers taking any or high doses (>=1000mg) of vitamin C. Additional analyses were performed at doses above or equal to 250mg; 500mg; and 2000mg. Given that vitamin C supplements are more likely to be taken in winter because well-publicised research had linked it to reduced duration of the seasonal common cold,<sup>45</sup> sensitivity analyses were performed to assess the robustness of results to weighting the analyses by the inverse of the probability of being sampled in each season.

All information relating to family or personal history of cancers was reported by the participant at phase 2. They were asked whether or not family members (blood relatives only) ever had the following cancers: breast, skin, lung, colon and rectum, ovary, stomach, cervix, ovary, pancreas, or prostate. The cancer history of first and second degree relatives was used to identify women potentially at raised or high risk of hereditary breast cancer (further information on determining this is given in section 4.7.6.3.2 of the methods chapter). It was unknown whether affected relatives were on the same side of the family, therefore this could only approximate to the guidelines provided by the UK's National Institute for Health and Clinical Excellence (NICE).<sup>319</sup> Participants were also asked to report their own history of disease, including whether they had previously been told they had a diagnosis of one of the cancers listed above.

# 6.4 Results

Thirty-four percent (4242) of women frequently took supplements containing any dose of vitamin C, on at least three of the four diary days, and 5% (579) frequently took high doses of 1000mg or more. Twenty-seven percent (1165) of those frequently taking any dose and 52% (299) taking high doses of vitamin C took four or more types of supplements. Furthermore, 82% of users taking any dose and 86% of the high dose users took some type of supplement at baseline, on average four years earlier.

After mutual adjustment, significant lifestyle predictors of frequently taking supplements containing either high dose or any dose of vitamin C were eating more than five servings of fruit and vegetables per day; eating less servings of red meat; and visiting an alternative practitioner more often than women not frequently taking these supplements (Table 28). Odds of visiting an alternative practitioner four or more times in the last 12 months were substantially greater for high dose takers compared to any dose takers (OR=2.84; 95% CI: 2.20, 3.66 vs. OR=1.75; 95% CI: 1.51, 2.03). Additionally, the odds of taking supplements containing any dose of vitamin C were significantly higher in women who exercised vigorously more than three times a week (OR=1.52; 95% CI: 1.23, 1.8); or were aged 45 years or more; of intermediate social class; divorced; childless; frequent visitors to their GP; or leaner. Significant predictors of frequent high dose taking were being an ex-smoker when compared with never smokers (OR=1.25; 95% CI: 1.02, 1.53), drinking alcohol less than once a week (OR=1.37; 95% CI: 1.12, 1.67) and being of high socio-economic status compared to low status (OR=1.45; 95% CI: 1.06, 2.00).

	Any doco		>-1000mg	n
Characteristics	OR $(95\% \text{ CI})^{a}$	P value	$OR (95\% CI)^{a}$	value
Age (years) <sup>b</sup>		i valuo		Value
<45	1	0.07	1	0.3
45-54	1.20 (1.03, 1.41)	0101	1.11 (0.81, 1.54)	0.0
55-64	1.26 (1.07, 1.48)		0.85 (0.60, 1.20)	
65 and above	1.23 (1.03, 1.47)		0.91 (0.62, 1.34)	
Social class <sup>c</sup>			,	
High	1	0.1	1	0.04
Intermediate	1.10 (1.01, 1.21)	•••	0.96 (0.78, 1.17)	
Low	1.07 (0.94, 1.22)		0.69 (0.50, 0.94)	
Marital status <sup>c</sup>				
Married/living together	1	0.4	1	0.9
Divorced/separated	1.31 (1.14, 1.51)		1.25 (0.94, 1.66)	
Widowed	0.95 (0.78, 1.16)		1.14 (0.72, 1.80)	
Single	0.86 (0.72, 1.03)		0.88 (0.61, 1.28)	
Had children				
Yes	1	0.001	1	0.09
No	1.24 (1.11, 1.39)		1.23 (0.97, 1.56)	
Body mass index				
(BMI kg/m²) <sup>b</sup>				
Underweight (<18)	1.03 (0.72, 1.46)		1.07 (0.53, 2.15)	
Normal (18-24.99)	1	0.08	1	0.6
Overweight (25-30)	0.90 (0.82, 0.99)		0.87 (0.69, 1.09)	
Obese (>30)	0.93 (0.80, 1.07)		1.11 (0.81, 1.54)	
Smoking status <sup>6</sup>				
Never smoked	1	0.4	1	0.02
Ex-smoker	1.07 (0.98, 1.17)		1.25 (1.02, 1.53)	
Current smoker	0.91 (0.75, 1.00)		1.19 (0.79, 1.81)	
Drinks alconol more than once				
a week	4	0.4	4	0.004
Yes		0.1		0.001
	1.07 (0.98, 1.17)		1.37 (1.12, 1.67)	
No wookly optivity	1	-0.001	1	0 000
NO WEEKIY ACTIVITY		<0.001		0.006
	1.10 (0.95, 1.41)		0.94(0.00, 1.40)	
Vigorous >-3/wook	1.10(0.90, 1.40) 1.52(1.23, 1.80)		0.90(0.01, 1.07) 1.36(0.85, 2.10)	
Sorvings of rod most oston	1.52 (1.25, 1.69)		1.30 (0.03, 2.19)	
per week <sup>c</sup>				
None	1	~0.001	1	~0.001
1-3		<b>\0.001</b>		<b>NO.001</b>
4 or more	0.73(0.72, 0.07) 0.61(0.54, 0.68)		0.00(0.35, 0.03) 0.48(0.35, 0.65)	
Servings of fruit and yeg eaten	0.01 (0.01, 0.00)		0.10 (0.00, 0.00)	
per dav <sup>c</sup>				
<=2	1	<0.001	1	0.01
3-5	1.21 (1.08. 1.37)		1.11 (0.84. 1.48)	
More than 5	1.45 (1.26, 1.67)		1.40 (1.02. 1.92)	
	· · · · · · · · · · · · · · · · · · ·		· · · · /	

**Table 28** Characteristics associated with taking supplements containing any dose of vitamin C and taking supplements containing high doses of vitamin C (1000mg or above)

	Any dose		>=1000mg	р
Characteristics	OR (95% CI) <sup>a</sup>	P value	OR (95% CI) <sup>a</sup>	value
Number of visit to doctors in				
last 12mths <sup>c</sup>				
None	1	<0.001	1	0.9
1-4	1.27 (1.12, 1.42)		0.98 (0.76, 1.25)	
>4	1.45 (1.26, 1.67)		0.98 (0.72, 1.33)	
Number of visit to alternative				
practitioner in last 12mths <sup>c</sup>				
None	1	<0.001	1	<0.001
1-4	1.41 (1.23, 1.61)		1.77 (1.35, 2.31)	
>4	1.75 (1.51, 2.03)		2.84 (2.20, 3.66)	
Number of participants in the	10161		10161	
models				
and the second s	مطمام المعقدا ألمعا طعان		Valala natia	

Table continued: Characteristics associated with taking supplements containing any dose of vitamin C and taking supplements containing high doses of vitamin C (1000mg or above)

<sup>a</sup>Mutually adjusted for the other variables listed above, OR =Odds ratio

<sup>b</sup>p for trend over the continuous variable

<sup>c</sup>p for trend over categories

Table 29 shows that, after adjustment, frequent high dose use of vitamin C remained significantly associated with a personal history of any cancer (OR=1.33; 95% CI: 1.00, 1.76) and any hormone related cancer (OR=1.68; 95% CI: 1.16, 2.43); specifically breast cancer (OR= 1.70; 95% CI: 1.14, 2.55). Additionally, frequent high dose use was significantly greater for women with a family history of breast cancer (OR=1.26; 95% CI: 1.01, 1.58) and appeared more likely, though non-significant, in women with a family history of any cancer (OR=1.16; 95% CI: 0.95, 1.41), any hormone related cancer (OR=1.19; 95% CI: 0.98, 1.46), and pancreatic cancer (OR=1.44; 95% CI: 0.94, 2.21). Taking any dose of vitamin C was significantly associated with a family history of cancer of the uterus (OR=1.38; 95% CI: 1.10, 1.74). These results were very similar when the analysis was weighted to take into account differential sampling in each season. For instance, the seasonally weighted odds of being a high dose user with a personal history of breast cancer was OR = 1.61 (95% CI: 1.07, 2.41), or with a family history of breast cancer was OR = 1.29 (95% CI: 1.03, 1.62) or with a family history of pancreatic cancer was OR = 1.53 (95% CI: 0.99, 2.36). However, the result for a personal history of total cancers was not significant (OR = 1.30; 95% CI: 0.97, 1.73) after seasonally weighting.

It may be observed in Table 30 that the odds of taking vitamin C increased with increasing doses above 500mg for women who had any family member with a history of breast cancer or who had a personal history of breast cancer. For the latter, ORs were 1.09 (95% CI: 0.78, 1.52) at >=500mg, 1.70 (95% CI: 1.14, 2.55) at >=1000mg and 2.36 (95% CI: 1.00, 5.56) at intakes of 2000mg or above. A similar pattern occurs for those with a personal history of cancer of the uterus or cervix, and was also seen in the total analyses of any cancer or any hormone related cancer. The small numbers of women in some of the categories, however, may have produced spurious results. Although the odds of having a mother or sister with breast cancer or potentially being at raised risk of this cancer increased at higher intakes, these were not statistically significant.

			Any Dose: N	=4242 (34%)	>=1000mg:	N =579 (5%)
			Unadjusted	Adjusted <sup>a</sup>	Unadjusted	Adjusted <sup>a</sup>
Туре с	of cancer	N <sup>b</sup>	OR(95% CI)	OR(95% CI)	OR(95% CI)	OR(95% CI)
Persor	nal history of					
Any ca	ancer	1268	1.14 (1.01, 1.29)	1.12 (0.97, 1.28)	1.31 (1.02, 1.68)	1.33 (1.00,1.76)
Any ho	ormone	642	1 11 (0 94 1 31)	1 08 (0 89 1 31)	1 50 (1 09 2 08)	1 68 (1 16 2 43)
	Simone	072	1.11 (0.54, 1.51)	1.00 (0.03, 1.01)	1.00 (1.00, 2.00)	1.00 (1.10, 2.40)
	Breast	523	1.13 (0.94,1.36)	1.10 (0.89, 1.35)	1.53 (1.08, 2.18)	1.70 (1.14, 2.55)
	Uterus	75	0.85 (0.52, 1.39)	0.77 (0.44, 1.34)	1.78 (0.77, 4.12)	1.97 (0.77, 5.02)
	Ovarian	60	1.29 (0.77, 2.17)	1.28 (0.71, 2.33)	1.35 (0.60, 3.07)	0.84 (0.20, 3.51)
Any cancer	non-hormone	584	1.16 (0.98, 1.40)	1.11 (0.91, 1.34)	1.16 (0.80, 1.70)	1.05 (0.68, 1.60)
Carlooi	Skin	324	1.14 (0.91, 1.43)	1.04 (0.81, 1.34)	0.85 (0.48, 1.49)	0.71 (0.36, 1.39)
	Cervix	190	1.26 (0.94, 1.69)	1.20 (0.86, 1.66)	2.03 (1.22, 3.36)	1.70 (0.94, 3.05)
	Colon Rectum	63	1.19 (0.71, 1.98)	1.30 (0.73, 2.30)	1.07 (0.34, 3.44)	0.98 (0.24, 4.10)
Family	history of					
Any c	ancer	7,259	1.08 (1.00, 1.16)	1.04 (0.96, 1.13)	1.15 (0.97, 1.36)	1.16 (0.95, 1.41)
Any ho	ormone cancer	3,629	1.09 (1.01, 1.18)	1.09 (0.99, 1.19)	1.16 (0.97, 1.38)	1.19 (0.98, 1.46)

**Table 29** Odds ratios of taking supplements containing vitamin C: any dose; or 1000mg or more forUKWCS women who self-reported a personal or a family history of cancer

ny hormone cancer	3,629	1.09 (1.01, 1.18)	1.09 (0.99, 1.19)	1.16 (0.97, 1.38)	1.19 (0.98, 1.46
Breast Prostate Ovarian	2,370 958 423	1.06 (0.97, 1.17) 1.09 (0.95, 1.25) 1.10 (0.90, 1.35)	1.04 (0.94, 1.16) 1.13 (0.97, 1.32) 1.12 (0.90, 1.41)	1.23 (1.00, 1.51) 1.04 (0.76, 1.41) 1.07 (0.69, 1.70)	1.26 (1.01, 1.58 1.09 (0.77, 1.51 1.09 (0.66, 1.79
Uterus	380	1.41 (1.14, 1.73)	1.38 (1.10, 1.74)	1.08 (0.68, 1.73)	1.11 (0.66, 1.87

Table continued: Odds ratios	of taking supplements co	ontaining vitamin C: a	any dose; or 1000mg	or more or
for UKWCS women who self-	reported a personal or a	family history of canc	cer	

		Any Dose: N	=4242 (34%)	>=1000mg: N =579 (5%)		
		Unadjusted	Adjusted <sup>a</sup>	Unadjusted	Adjusted <sup>a</sup>	
Type of cancer	N <sup>b</sup>	OR(95% CI)	OR(95% CI)	OR(95% CI)	OR(95% CI)	
Family history of						
Any non-hormone	5,227	1.07 (0.99, 1.16)	1.03 (0.95,1.12)	1.03 (0.87, 1.22)	1.04 (0.86, 1.25)	
Lung	2,066	1.06 (0.96, 1.17)	1.00 (0.89, 1.11)	1.07 (0.86, 1.33)	1.00 (0.78, 1.29)	
Colon/Rectum	1,608	0.96 (0.86, 1.08)	0.98 (0.86, 1.11)	0.97 (0.76, 1.25)	1.08 (0.82, 1.43)	
Stomach	1,300	1.02 (0.90, 1.15)	1.01 (0.88, 1.16)	0.91 (0.69, 1.21)	0.97 (0.71, 1.33)	
Skin	957	1.01 (0.88, 1.16)	0.97 (0.83, 1.13)	0.88 (0.64, 1.23)	0.86 (0.60, 1.24)	
Pancreas	455	1.13 (0.93, 1.37)	1.11 (0.90, 1.38)	1.41 (0.96, 2.08)	1.44 (0.94, 2.21)	
Cervix	311	1.03 (0.81, 1.31)	1.04 (0.79, 1.36)	0.68 (0.36, 1.28)	0.74 (0.38, 1.46)	

<sup>a</sup>Adjusted for BMI, age, social class, marital status, children, smoking status, level of physical activity, low alcohol consumption, red meat servings, total fruit and vegetable servings.

<sup>b</sup>Total numbers with history of cancer

	Frequent use of supplements containing vitamin C doses greater or equal to							
-		250mg	•••	500mg		1000mg		2000mg
		OR(95% CI) <sup>a</sup>		OR(95% CI) <sup>a</sup>		OR(95% ČI) <sup>a</sup>		OR(95% ČI) <sup>a</sup>
	N <sup>b</sup>	N=1,448 (12%)	N <sup>b</sup>	N=1,195(10%)	N <sup>b</sup>	N=579 (5%)	N <sup>b</sup>	N==92(1%)
Personal history								
Any cancer	159	1.06 (0.87, 1.30)	131	1.03 (0.83, 1.29)	74	1.33 (1.00,1.76)	19	2.86 (1.64, 4.98)
Any hormone cancer	81	1.04 (0.79, 1.39)	69	1.08 (0.80, 1.46)	43	1.68 (1.16, 2.43)	12	3.50 (1.75, 7.01)
Breast cancer	68	1.10 (0.81, 1.49)	56	1.09 (0.78, 1.52)	36	1.70 (1.14, 2.55)	8	2.36 (1.00, 5.56)
Uterus	8	0.99 (0.45, 2.22)	8	1.25 (0.56, 2.78)	6	1.97 (0.77, 5.02)	3	8.64 (2.52, 29.6)
Ovarian	7	0.50 (0.15, 1.62)	7	0.64 (0.20, 2.06)	3	0.84 (0.20, 3.51)	1	2.75 (0.37, 20.8)
Any non-hormone cancer	69	0.98 (0.73, 1.30)	56	0.97 (0.71, 1.33)	31	1.05 (0.68, 1.60)	8	2.52 (1.19, 5.32)
Skin	34	0.79 (0.53, 1.20)	26	0.74 (0.47, 1.19)	13	0.71 (0.36, 1.39)	2	1.08 (0.26, 4.49)
Cervix	32	1.43 (0.93, 2.21)	29	1.60 (1.03, 2.52)	17	1.70 (0.94, 3.05)	4	3.14 (1.10, 8.94)
Colon Rectum	5	0.69 (0.24, 1.94)	3	0.41 (0.10, 1.72)	3	0.98 (0.24, 4.10)	2	7.20 (1.62, 32.1)
Eamily history of breast cancer								
Any family member	299	1.15 (0.99, 1.34)	244	1.10 (0.93, 1.30)	129	1.26 (1.01, 1.58)	27	1.69 (1.01, 2.83)
Mother or sister	163	1.13 (0.93, 1.37)	129	1.04 (0.84, 1.29)	67	1.16 (0.87, 1.55)	15	1.55 (0.81, 2.96)
Respondent at raised risk	32	1.11 (0.73, 1.68)	25	1.04 (0.66, 1.65)	15	1.31 (0.73, 2.32)	4	2.03 (0.62, 6.56)
Respondent at high risk <sup>c</sup>	9	0.67 (0.30, 1.47)	8	0.71 (0.31, 1.65)	4	0.69 (0.22, 2.23)		

**Table 30** Odds ratios of taking supplements containing vitamin C for a range of doses for UKWCS women who self-reported a personal history of cancer or a family history of breast cancer

<sup>a</sup>Adjusted for BMI, age, social class, marital status, children, smoking status, level of physical activity, low alcohol consumption, red meat servings, total fruit and vegetable servings. Comparison group = all respondents not taking stated dose.

<sup>b</sup>Total numbers with a history of cancer listed taking doses specified.

<sup>c</sup>Insufficient numbers at higher doses.

# 6.5 Discussion

The frequent use of supplements containing any dose or high doses of vitamin C in the UKWCS was associated with healthier lifestyle behaviours, and therefore supports the inverse supplement hypothesis, as seen in analyses of any supplement-taking in the UK or elsewhere.<sup>25 28-35</sup> Women taking either high (>=1000mg per day) or any dose of vitamin C were more likely to consume over five servings of fruit and vegetables, the main dietary source of vitamin C.<sup>67</sup> This is consistent with evidence from studies of any supplement-taking,<sup>27 28 33</sup> and US studies of vitamin C supplement-taking,<sup>30 33</sup> and suggests that many high dose vitamin C takers are less likely to need them. Furthermore, in-line with US findings, UKWCS vitamin C takers were likely to eat less meat.<sup>31</sup> They also exercised vigorously more often, supporting previous research linking activity to supplement use.<sup>27-29 32-35</sup> Distinguishing characteristics of high dose vitamin C takers in the UKWCS were being an ex-smoker, drinking alcohol less than once a week and being of high socio-economic status. These characteristics were not significant predictors of using supplements containing any dose of vitamin C in the UKWCS, however they have been positively associated with taking any type of supplement in other studies.<sup>26 29</sup> Additionally, the current results indicated high dose vitamin C users relied more on alternative practitioners rather than family or private doctors. Healthy behaviours associated with vitamin C supplement use are likely to reduce health risks, therefore these behaviours identified should be considered for adjustment in longitudinal studies of risks.<sup>33</sup>

Despite controversy surrounding evidence of benefits of high dose vitamin C supplementation for prolonged cancer survival,<sup>87-89</sup> our results showed prior to seasonal weighting women with any type of cancer were more likely to be high dose vitamin C supplement-takers than women with no history of cancer. Since antioxidants can potentially reduce the effectiveness of anti-cancer drugs,<sup>77 356</sup> patients should be encouraged to discuss their supplement use with their doctors in order to avoid contraindications. For some cancer patients supplement use may be a coping behaviour and a way of taking control.<sup>37 357</sup> Similar health related behaviours may also occur in women with concerns about risk of developing cancer: for instance women who attended mammography have also been positively associated with supplement use in the US.<sup>33</sup> Likewise, women attending UK breast screening clinics had similar characteristics to supplement users in the UKWCS and wanted diet and exercise advice to be provided at these clinics.<sup>16</sup> Doses below 1000mg/d and any dose of vitamin C, however, were not significantly associated with total cancer in the UKWCS; this has been also observed in some US studies.<sup>353 358</sup>

To the best of our knowledge this is the first UK study to analyse associations between vitamin C supplement use and specific prevalent cancers, and therefore the first to report significant associations of frequent high dose vitamin C taking (=>1000mg/day) in UK women with a personal or family history of breast cancer. This supports findings that US women physicians with breast cancer were more likely to take vitamin C.<sup>31</sup> Furthermore, our results show the odds of having a history of cancer increased at higher doses (>=2000mg). However, whilst US research found that women at high risk of breast cancer and with inconclusive genetic test results were significantly more likely to take any supplements, the increased odds of taking high doses of vitamin C in the UKWCS for women with increased risk of hereditary breast cancer or those having mothers or sisters with breast cancer were not significant.<sup>359</sup> Our results may be due to low numbers and lack of power. In general, a history of non-hormone related cancer did not appear to be associated with vitamin C supplement-taking in the UKWCS, nevertheless associations with a personal history of cervical cancer remained significant at some doses after adjustments, including adjustment of socio-economic status which is known to be linked with this cancer.<sup>360</sup>

In relation to prevention of cancer and other chronic diseases, the 5 A Day fruit and vegetable initiative based on the World Health Organisation (WHO) recommendations could have influenced antioxidant supplement sales at the time of the phase 2 UKWCS follow-up.<sup>9</sup> Proactive marketing by supplement companies would have also increased sales. The 1997 WCRF report, nevertheless, stated that supplements were probably unnecessary and unhelpful for reducing cancer risk.<sup>8</sup> Whilst the recent 2007 WCRF report found no probable or convincing evidence that vitamin C supplementation affects cancer risk there was evidence of an increased or decreased risk with some other supplement types (Appendix B), though this was usually from studies of high-risk groups.<sup>7</sup> In summary the 2007 report states that it is unwise to recommend widespread supplement use for cancer prevention since effects cannot be confidently predicted in the general population.<sup>7</sup> Indeed high doses of some supplements, including vitamin C may promote cancer,<sup>77</sup> although doses above 400mg of this water soluble vitamin are likely to be excreted in healthy women.<sup>65</sup> Its antioxidant properties may reduce DNA damage by reactive oxygen species during the initial stage of cancer particularly in individuals with high levels of ROS.<sup>77</sup> This antioxidant property, however, may decrease beneficial apoptosis, the ROS induced programmed death of damaged cells,<sup>81</sup> and thereby lead to the progression of cancer particularly in individuals with low levels of ROS.<sup>77</sup> Vitamin C may also act as a pro-oxidant creating highly reactive and damaging hydroxyl radicals via the Fenton reaction in the presence of iron.<sup>76</sup> However this hypothesis is controversial since free iron is normally unavailable in vivo.<sup>76</sup> Apart from a family history of breast cancer and a moderate but non-significant association

148

with a family history of pancreatic cancer our results indicate that UK women were probably not taking high dose vitamin C supplements as a preventive measure due to a family history of cancer in general. Since cancer of the pancreas has a poor prognosis, some women with this family history may have been more motivated to take high doses of vitamin C supplements.

It is unknown why vitamin C supplements were taken by women in the UKWCS. Given that only a relative small proportion of the UK population are advised by the medical profession to take supplements for health reasons,<sup>26</sup> some health conscious UK women may be self-treating with vitamin C. Alternatively, women with or without cancer may take supplements under the belief supplements can make them feel better generally or increase immune function.<sup>361</sup> Additionally, due to the cross-sectional nature of the study the direction of cause and effect cannot be determined; it is unknown whether vitamin C has been taken to prevent or manage cancer symptoms or whether vitamin C has caused cancer.

Lack of information to determine whether cancer developed before or after frequent vitamin C supplement use started is a limitation of this study. Although cancer registry data was available for the cohort we suggest it is women's self-reported perceived health, whether accurate or not, that influences their supplement-taking behaviour. Self-reporting of supplement descriptions, for only four days by diary were limitations of the study. Whilst 4-day diaries are capable of capturing daily, and near daily intake, in reality some three to four day diary recordings may represent spasmodic rather than frequent intake. Nevertheless, our results in section 5.2 of the evaluation chapter show that substantial agreement was found between these frequent diary recordings and daily use recorded concurrently by questionnaire. Although the number of years of supplementation was not collected in either the diary or questionnaire, and no further follow-up was conducted, the majority of vitamin C users (82%) were taking a supplement of some type on average four years earlier at baseline. An additional problem was the wide variety of formulations of supplements used which made coding difficult. Whilst high dose vitamin C supplements were unlikely to contain other micronutrients,<sup>35</sup> our results show that consistent with other research,<sup>26</sup> women taking high doses were highly likely to take other supplements. Therefore vitamin C use may be a marker for the intake of other supplements.

Our study capitalises on the large sample size of the UKWCS, substantial numbers of women frequently taking vitamin C (34%) and as well as the wide variety of characteristics and cancers recorded. However, the small numbers of women in some of the categories, particularly those shown in Table 30 and especially those taking

2000mg/day or more, may have produced spurious results. Another limitation of our study is that UKWCS participants were more health conscious than the general population and therefore not representative of the whole UK population. Differences in characteristics between frequent takers and non-frequent takers in the UKWCS may not be as pronounced as that found in the general population.

Our research may help to identify high-dose users, such as ex-smokers, low alcohol drinkers and women with a history of breast cancer who could be made aware of inconsistencies in evidence relating to suggested benefits, and of warnings relating to high strength supplements.<sup>362</sup> Furthermore, patients should be encouraged to discuss their supplement use with their doctors to avoid contraindications.<sup>81 356</sup> Finally, additional research is needed to establish the effects of both supplement and dietary vitamin C intake on cancer initiation and development. Breast cancer risks for UK women in relation to vitamin C intake are reported in chapter 8, 9 and 10. First, breast cancer risk in relation to general supplement use is explored in the next chapter.

# **CHAPTER 7**

# 7 Is general supplement use associated with breast cancer risk in the UKWCS?

# 7.1 Summary

*Objectives:* To determine whether general supplement use is associated with breast cancer risk at baseline in the UK Women's Cohort Study (UKWCS) and whether use modifies risks in different groups of women. In addition breast cancer risks for women who were users at both baseline and phase 2 will be determined.

*Methods:* Hazard ratios were calculated for 33,138 middle-aged women in the UKWCS in relation to any supplement use recorded by questionnaire at baseline. The median follow-up period was 11.2 years with 982 registered breast cancer incidences. Risks for 12,917 of these women, who also completed the phase 2 questionnaires on average 4.4 years later, were analysed in relation to consistent supplement use. Median follow-up for these women was 11.6 years from baseline during which 414 breast cancers occurred. Adjustments were made for potential confounders measured at baseline. Interactions between baseline supplement use and other characteristics of the women were tested.

*Results:* There was no evidence of associations with breast cancer risk for baseline supplement users (HR=1.00; 95% CI: 0.87, 1.16) compared to non-users, or for women who were still taking supplements at phase 2 (HR=0.96; 95% CI: 0.74, 1.26) compared to never-users. There were also no significant risks for pre- (HR=1.14; 95% CI: 0.91, 1.43) or post-menopausal women (HR=0.89; 95% CI: 0.73, 1.08) for baseline users or similarly for consistent users. Significant interactions were found between baseline supplement use and menopausal status (p=0.04), socio-economic status (p=0.01) and vigorous exercise (p=0.05). Increasing healthier behaviours were apparent from never users, to inconsistent users, to consistent users.

*Conclusion:* There were no overall associations in this UK cohort between breast cancer risk and general supplement use, including women who were users on average 4.4 years later. However, the results indicate supplement users may have different risks from non-users depending on their menopausal status, their socio-economic status or level of physical activity.

151

# 7.2 Introduction

Due to the unregulated increased use of supplements by UK women, which has risen from 17% in 1986/1987 to 41% reported in 2008/9,<sup>24 25</sup> it is important to establish whether or not general supplement use is associated with detrimental effects, such as an increased risk of breast cancer. Indeed the 2007 WCRF review clearly states that supplements are not recommended for cancer prevention, furthermore it reports no convincing evidence that supplements protect against breast cancer.<sup>7</sup> Most studies evaluating the relationship between supplement use and breast cancer risk have focused on individual supplement types or ingredients. Only a few studies mentioned in the 2007 WCRF review examined general supplement use in relation to breast cancer and these were case-control studies which may be prone to selection bias and recall bias.<sup>257-259</sup> No associations were found in the Danish and US studies reported in the 2007 review,<sup>257 259</sup> whereas a significant protective effect of supplements on breast cancer was found in a Taiwanese study (OR=0.40, 95% CI: 0.3, 0.7). Furthermore, there is a lack of evidence on whether general supplement use modifies factors linked to breast cancer risk; supplement use could be beneficial for some groups of women but detrimental for others.

Additionally, no study has assessed whether consistent use of supplements in general is associated with breast cancer risk. Only one study has analysed the consistency of general supplement use in relation to cancer, though this was for total cancer mortality and no significant associations were found for women who were consistent users at three survey points over 11 years.<sup>363</sup> Nevertheless, in this German cohort, consistent users had a healthier diet compared to inconsistent users as well as non-users.<sup>364</sup> As reported in the previous chapter, 6, supplement users are more likely to lead healthier lifestyles, for instance they are more likely to be physically active, have a lower BMI and eat diets which meet recommended dietary intakes. However, there is no research on whether consistent users are more health conscious than inconsistent users in the UK.

The current study prospectively examines the relationship between breast cancer incidence and use of any type and amount of supplements at baseline in the UKWCS, and also compares the risk for women who are still using any supplements at phase 2 with those who are inconsistent users and those who are never-users. It also describes characteristics of consistent and inconsistent users. Furthermore, it capitalises on the large sample size of the UKWCS at baseline by testing for interactions between supplement use at baseline and women's characteristics and their associations with breast cancer risk.

# 7.3 Method

## 7.3.1 Study population

General information about the recruitment of the UKWCS baseline population used in this analysis is provided in section 4.1.1.1 of the methods chapter. Of the 35,372 women in the UKWCS who completed the baseline questionnaire, 34,958 (99%) provided information about whether or not they took supplements. Women with any prevalent malignant cancers recorded in the cancer registries before baseline questionnaire date and women diagnosed with breast cancer within 6 months after the questionnaire date were excluded. To be consistent with other baseline analyses that will be presented in this thesis, women with extremely low or high total energy intake (more than 6000kcal and less than 500kcal) were also excluded. This left 33,138 women for the time-to-event analysis. Over the median follow-up period of 11.2 years there were 982 incident breast cancers which were registered by the censor date 01/01/08.

Risks for 12,917 of these women who also completed supplement questions on the UKWCS phase 2 questionnaires on average 4.4 years after baseline, were analysed in relation to consistent supplement use. Consistent users are defined in this analysis as women who were taking any type of supplement at both baseline and phase 2. The same types of exclusions were made as above. Over a median follow-up of 11.6 years from baseline there were 414 incident breast cancers registered to censor date 01/07/08.

#### 7.3.2 Exposure measurement

General supplement use at baseline was determined by questionnaire using yes/no answers to:

Q29 Do you take any vitamins, minerals, fish oils or other food supplements?

Supplement use at phase 2 was determined by questionnaire using yes/no answers to: Q15 Do you presently use any dietary supplements?

Supplements had previously been defined on the phase 2 questionnaire as vitamins, minerals, fibre, fish oils or other food supplements.

Additionally, if participants did not answer yes to the above questions but provided details of any type of supplements taken, regardless of amount taken, then these women were designated as being general supplement users. As a result an extra 3083 women were coded as general supplement users at baseline, and also an extra 846

women at phase 2 were coded as users who detailed they took supplements weekly or more frequently on the phase 2 questionnaire (see sections 4.5.1 and 4.5.2). These numbers are before exclusions mentioned in 7.3.1.

#### 7.3.3 Statistical analyses

Characteristics of users and non-users at baseline were described using means and percentages; significant differences between means were established using t-tests and significant differences between categories were established using chi squared tests. Women who had also completed the phase 2 questionnaire were split into three categories: Never users at baseline and phase 2; inconsistent users (use at only one collection point); Consistent users (use at both baseline and phase 2). Characteristics of these women were described using means and percentages. Significant trends for means across groups from never users to inconsistent users to consistent users were calculated using linear regression followed by tests for linear hypotheses.

Cox proportional hazard ratios were used to estimate breast cancer risk relating to supplement use of women in the UKWCS. The proportional hazard assumption was checked using graphs of log-log curves. Covariates used for adjustment were based on those previously identified as risk factors for breast cancer which had also previously been associated with supplement use, as illustrated in the DAG in Figure 14 in section 4.7.5.2. These were age; BMI (underweight, normal, overweight, obese); parity (none, 1-2, 3 or more); estimated cumulative breast feeding (wks); age at menarche; minutes sweating exercise per week; contraceptive pill use (never, past, current); HRT use (never, past, current); alcohol intake (g/day); smoking status (never, past, current); food group (meat, oily fish, fish, vegetarians); total energy intake; and also where appropriate supplement use (yes, no); menopausal status (postmenopausal; premenopausal) and socio-economic status (high: professional and managerial; intermediate; low: routine and manual). Menopausal status was determined at baseline based on the answers to this questionnaire (see section 4.7.4 including Figure 12). Covariates were derived from the health and lifestyle part of the baseline questionnaire apart from total energy intake, total alcohol intake and food groups which in turn had been derived from the FFQ part of the baseline questionnaire.

To investigate whether supplement use at baseline modified the effects of women's characteristics on breast cancer risks, adjusted likelihood ratio tests for overall interactions were performed between supplement use status and each of the following characteristics: menopausal status; HRT use; socio-economic status (SES); educational level; BMI; level of exercise; and allocated food group. These are factors that have been associated with breast cancer risk, many of which are discussed in the

literature review chapter, 3, and/or in section 4.7.5.2 of the methods chapter. The hazard ratios shown in Table 35 include interaction terms for joint effects of supplement use status and each characteristic, which were adjusted for the same covariates in the other time-to-event analyses. Linear combinations of coefficients were used to calculate the hazard ratios for the users after running the main Cox proportional hazard command (using the Stata lincom command).<sup>365</sup>

# 7.4 Results

Of the 33,138 total women, 62% percent were supplement users at baseline and 54% of the 12,917 women followed-up at phase 2 had taken supplements at both recording points, and were classed as consistent users. Twenty-four percent of women at phase 2 were inconsistent users (had taken supplements at either recording point but not at both) and 21% were never users. Of the 8,896 (69%) women who were currently taking supplements at phase 2 as many as 6,992 (79%) were classed as consistent users.

There were significant differences between general supplement users and non-users at baseline for the majority of characteristics listed in Table 31 except for age at menarche, smoking status, socio-economic status and family history of breast cancer. In particular users were more likely to have a higher fruit and vegetable intake (mean 652 vs. 591 g/d), a lower meat intake (mean 59.8 vs. 72.9 g/d), more likely to exercise vigorously (15.9 mins/d vs. 14.1 mins/d), but less likely to engage in breast feeding (22.6 vs. 25.1 cumulative weeks) than non-users. Users were slightly less likely to have a degree than non-users. There were similar differences in the prospective comparison in Table 33 which showed increasingly healthier behaviours from never users at baseline and phase 2, inconsistent users, to consistent users at phase 2 (e.g. fruit and vegetable intake respectively was 603 vs. 627 vs. 676 g/d).

There was no difference in breast cancer incidence between supplement users and non-users for total women either in the unadjusted or adjusted analyses (HR=1.00; 95% CI: 0.87, 1.16) For post-menopausal women hazard ratios for users were lower and for pre-menopausal hazard ratios were raised, compared to non-users (HR=0.89; 95% CI: 0.73, 1.08 and HR=1.14; 95% CI: 0.91, 1.43 respectively), but they were not significant (Table 32). For women who reported taking any supplement at both baseline and phase 2 similar patterns of non-significant risks were found; these also were not significant (Table 34).

	N=12678	N=20460	р
	(38%)	(62%)	
Age (years) mean (sd)	51.5 (9)	52.3 (9)	<0.001
BMI (kg/m <sup>2</sup> ) mean (sd)	24.8 (4)	24.2 (4)	<0.001
Age at menarche (years)	12.8 (2)	12.8 (2)	0.8
Parity mean (sd)	1.9 (1)	1.8 (1)	<0.001
Cumulative breast feeding (weeks) mean (sd)	25.1 (31)	22.6 (34)	<0.001
Vigorous exercise (mins/d) mean (sd)	14.1 (28)	15.9 (30)	<0.001
Total energy intake (cal/d) mean (sd)	2323 (710)	2360 (717)	<0.001
Alcohol intake (g/day) mean (sd)	9.0 (11)	8.5 (10)	<0.001
Total meat intake (g/day) mean (sd)	72.9 (63)	59.8 (60)	<0.001
Total fruit & veg (g/day) mean (sd)	591 (329)	652 (354)	<0.001
Never smoked (%)	58.2	57.5	0.2
Never used HRT (%)	70.0	66.5	<0.001
Never used pill (%)	31.1	32.6	0.008
Socio-economic status (%)			
Higher	39.2	39.0	0.8
Middle	42.3	42.7	
Lower	18.5	18.3	
Education level, (%)			
Degree	26.9	25.7	0.05
Family history of breast cancer (%)	7.5	7.7	0.4

Table 31 Characteristics of supplement users and non-users at baseline in the UKWCS  $% \left( {{{\rm{S}}} \right) = 0} \right)$ 

Any supplements use	Cases/	Unadjusted		Adjusted <sup>a</sup>	
at baseline	Non-cases	HR	(95% CI)	HR	(95% CI)
Total women					
Non-users	364/12314	1		1	
Users	618/19842	1.04	(0.92, 1.18)	1.00	(0.87, 1.16)
Post-menopausal Non-users Users	218/6061 353/10514	1 0.91	(0.76, 1.08)	1 0.89	(0.73, 1.08)
Pre-menopausal Non-users Users	146/6253 265/9328	1 1.21	(0.99, 1.48)	1 1.14	(0.91, 1.43)

Table 32 Breast cancer risks according to any supplement use at baseline in the UKWCS

<sup>a</sup>Adjusted for baseline covariates: age, BMI (grouped), socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hrs exercise sweating per week, contraceptive pill use, HRT use, alcohol intake, smoking status, food group, total calories

	Never <sup>a</sup>	Inconsistent <sup>∞</sup>	Consistent <sup>c</sup>	p <sub>trend</sub>
	users	users	users	
	N=2756	N=3169	N=6992	
	(21%)	(24%)	(54%)	
Age (years) mean (sd)	52.1 (9)	51.2 (9)	52.6 (9)	<0.001
BMI (kg/m <sup>2</sup> ) mean (sd)	24.4 (4)	24.4 (5)	23.9 (4)	<0.001
Age at menarche (years) mean (sd)	12.8 (2)	12.8 (2)	12.8 (2)	0.2
Parity mean (sd)	1.4 (1)	1.4 (1)	1.3 (1)	0.002
Cumulative breast feeding (weeks)	28.5 (39)	26.1 (37)	24.4 (35)	<0.001
mean (sd)				
Vigorous exercise (mins/d) mean (sd)	13.4 (26)	13.9 (23)	16.6 (31)	<0.001
Total energy intake (cal/d) mean (sd)	2341 (674)	2366 (709)	2364 (694)	<0.001
Alcohol intake (g/day) mean (sd)	8.9 (13)	8.6 (10)	8.0 (10)	<0.001
Total meat intake (g/day) mean (sd)	67.0 (62)	60.5 (59)	53.0 (56)	<0.001
Total fruit & veg (g/day) mean (sd)	603 (312)	627 (337)	676 (347)	<0.001
Never smoked (%)	63.4	61.0	61.3	0.2
Never used HRT (%)	72.0	69.6	65.3	0.007
Never used pill (%)	34.2	28.4	33.0	0.03
Socio-economic status (%)				0.8
Higher	41.9	41.0	41.0	
Middle	41.7	41.7	42.7	
Lower	16.4	16.7	16.3	
Education level, (%)				
Degree	31.0	30.0	27.2	0.001
Family history of breast cancer (%)	8.0	7.7	8.2	0.7

**Table 33** Baseline characteristics of women in the UKWCS according to supplement use at baseline and phase 2: never users; inconsistent users; and consistent users

<sup>a</sup>Never users: women who reported no supplement use at baseline and at phase 2

<sup>b</sup>Inconsistent users: women who reported supplement use on only one questionnaire, either baseline and at phase 2

<sup>c</sup>Consistent users: women who reported supplement use at both baseline and at phase 2

Any supplements use at	Cases/	Unadjusted		Adjusted <sup>a</sup>	
baseline or phase 2	Non-cases	HR	(95% CI)	HR	(95% CI)
Total women Never users at BL & P2 Inconsistent users Consistent users	85/2671 99/3070 230/6762	1 1.00 1.05	(0.75, 1.34) (0.82, 1.35)	1 0.92 0.96	(0.67, 1.26) (0.74, 1.26)
Post-menopausal Never users at BL & P2 Inconsistent users Consistent users	54/1389 54/1438 130/3667	1 0.95 0.89	(0.65, 1.39) (0.65, 1.23)	1 0.87 0.80	(0.57, 1.33) (0.56, 1.15)
Pre-menopausal Never users at BL & P2 Inconsistent users Consistent users	31/1282 45/1632 100/3095	1 1.13 1.32	(0.71, 1.79) (0.88, 1.98)	1 1.01 1.16	(0.63, 1.63) (0.77, 1.76)

**Table 34** Breast cancer risks according to any supplement use for never users,inconsistent users and consistent users according to use at baseline and phase 2 ofthe UKWCS

<sup>a</sup>Adjusted for baseline covariates: age, BMI (grouped), socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hrs exercise sweating per week, contraceptive pill use, HRT use, alcohol intake, smoking status, food group, total calories

In the test for interactions, supplement use was found to modify the effect of socioeconomic status (SES), vigorous exercise, as well as menopausal status on breast cancer incidence after adjustment for potential confounders and inclusion of interaction terms for joint effects of supplement use status and each characteristic. Table 35 shows that pre-menopausal users tend to have a slightly higher breast cancer risk than non-users whereas post-menopausal users have a slightly lower risk than non-users. though individual risk estimates are not significant. Table 35 also shows that supplement use has little effect on breast cancer risk for women of high SES, but risks are different for middle and low SES groups depending on whether or not they used supplements. Non-users in the lowest SES group had a higher breast cancer risk (HR=1.52; 95% CI: 1.11, 2.07) than users in the lowest SES group (HR=0.97; 95% CI: 0.72, 1.31), when compared to women of high SES who did not take supplements. However, supplement users in the middle SES group had higher risks (HR=1.23; 95% CI: 0.98, 1.54) than non-users in the middle SES group, though these estimates were not statically significant. The unadjusted increased risk for middle SES group supplement users compared to other SES groups of supplement users is illustrated in the Kaplan Meier curve in Figure 22. Additionally, women who exercise vigorously had a higher risk of breast cancer if they took supplements than if they were not users (HR=1.16; 95% CI: 0.94 1.42). However women who did not exercise vigorously were at greater risk (HR=1.27; 95% CI: 0.99 1.62) if they were non-users than if they took supplement (HR=1.10; 95% CI: 0.87 1.39). However individual risk estimates did not reach statistical significance. Effects of HRT use, BMI, educational level and food group were not significantly modified by supplement use.

	Cases/ N	Cases/ Non-cases		HR (95% CI) <sup>a</sup>		
	Non-users	Users	Non-users	Users		
Menopausal statu	S	271/8213	1	0.88 (0.73, 1.07)		
Pre	124/5293	219/7803	0.98 (0.72, 1.33)	1.17 (0.89, 1.55)	0.04	
HRTUSE						
Never	182/6987	302/10661	1	1.07 (0.89, 1.29)		
Prior Current	27/775 89/2244	56/1603 132/3751	1.23 (0.82, 1.85) 1.43 (1.14, 1.97)	1.23 (0.90, 1.67) 1.32 (1.03, 1.68)	0.5	
Socio-economic s	tatus				010	
High Medium Low	109/4018 123/4250 66/1738	189/6351 230/6826 71/2838	1 1.08 (0.84, 1.40) 1.52 (1.11, 2.07)	1.09 (0.86, 1.38) 1.23 (0.98, 1.54) 0.97 (0.72, 1.31)	0.01	
Education level						
None O'level A'level Degree	50/1330 90/2894 69/2340 72/2710	75/2260 145/4741 120/3731 110/4082	1 0.93 (0.65, 1.32) 0.88 (0.60, 1.29) 0.83 (0.56, 1.22)	0.87 (0.61, 1.25) 0.91 (0.64, 1.24) 0.95 (0.67, 1.33) 0.83 (0.57, 1.19)	0.8	
BMI						
Low Normal Overweig Obese	2/97 166/6071 ht 92/2699 38/1139	5/232 311/10610 133/3810 41/1363	0.99 (0.19, 5.14) 1 0.94 (0.68, 1.31) 0.85 (0.53, 1.38)	0.80 (0.33, 1.95) 1.04 (0.86, 1.26) 1.19 (0.94, 1.50) 1.08 (0.76, 1.52)	0.9	
Exercise vigorous Yes No	ly 138/5515 158/4455	293/9812 193/6136	1 1.27 (0.99 1.62)	1.16 (0.94 1.42) 1.10 (0.87, 1.39)	0.05	
Food Group Meat Oily fish Fish Vegetaria	231/7444 4/163 17/743 n 46/1656	345/10667 12/488 49/1670 84/3190	1 0.80 (0.39, 2.15) 0.75 (0.46, 1.22) 0.94 (0.68, 1.30)	1.00 (0.85, 1.19) 0.79 (0.44, 1.42) 0.93 (0.68, 1.27) 0.88 (0.68, 1.13)	0.9	

**Table 35** Assessment of modifying effects of supplement use at baseline on breast cancer risk using interaction terms for joint effects of supplement use and other characteristics

<sup>a</sup>Adjusted for age, BMI (grouped), socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hrs exercise sweating per week, contraceptive pill use, HRT use, low alcohol, smoking status, food group, total calories.

<sup>b</sup>Test for interaction
Figure 22 Breast cancer time-to-event curve for supplement users split by socioeconomic status



## 7.5 Discussion

In line with previous research, descriptive results from these UKWCS baseline and phase 2 analyses support the inverse supplement hypothesis that supplement users lead a healthier lifestyle than non-users, as found in the UK<sup>25 28 29</sup> and elsewhere.<sup>30-35</sup> In particular, supplement users had on average a higher intake of fruit and vegetables, a lower intake of meat and spent more time exercising vigorously. Whilst UKWCS supplement users were generally more health conscious, they had spent less time on average breast feeding than non-users or never-users. Additionally, UKWCS users were less likely to be educated to degree level. However a previous analysis of an initial 13,822 respondents from the baseline UKWCS found no significant association for education, though this was after mutual adjustment for other variables,<sup>27</sup> and other studies have reported higher qualifications for supplement users.<sup>32,353</sup>

Furthermore, the descriptive results showed that consistent users (users at both baseline and phase 2) have healthier behaviours than inconsistent users, as well as never users. Only one other cohort has analysed the characteristics of consistent supplement users compared to non-users and inconsistent users;<sup>364</sup> women in this EPIC-Heidelberg cohort who were consistent users had the highest intake of dairy products, fish, fruit and vegetables and wine but the lowest intake of meat compared to inconsistent users as well as non-users. In this German cohort only inconsistent use was associated with self-reported cancers. In the UKWCS inconsistent or consistent users were not more likely to have a family history of breast cancer. Further analysis of the UKWCS using a multivariate logistic regression model mutually adjusting for variables is required to assess significant predictors of consistent supplement use in the UKWCS cohort, which remain significant over and above associations with other factors. In comparison to the German study which reported a low percentage of consistent women users (23%), a large percentage (54%) of women at phase 2 of the UKWCS were consistent users from baseline. Reasons for the differences may be as follows: baseline supplement use in the German study was much lower than in the UKWCS (39% vs. 62%); supplement recordings occurred over three follow-up time points in the German study compared to two in the UKWCS; and the German cohort may not have been as health conscious as the women in UKWCS which included a high percentage of vegetarians.

In the current analysis there was no evidence of significant associations between breast cancer risk and supplement use at baseline for total women and by menopausal status. This supports the results of both the Danish and US study reported<sup>257 259</sup> in the 2007 WCRF review which found no convincing evidence for beneficial effects of

supplements on breast cancer risk.<sup>7</sup> However the current results are in contrast to the Taiwanese study where general supplement use was associated with a reduced risk.<sup>258</sup> Taiwanese women may generally have lower intake levels of nutrients than western women and therefore may require supplementation to reduce their risk. In contrast, there was a weak increased risk for pre-menopausal supplement use at baseline in the UKWCS which was close to significance, but only in the unadjusted analysis. Given that users generally have a healthier lifestyle than non-users, lower breast cancer risks for users before adjustment for lifestyle factors had been expected.

Additionally, there was no evidence of significant associations between breast cancer risk and consistent supplement use between baseline and phase 2 for total women and by menopausal status. This is the first study to assess breast cancer risk and long-term supplement use. Only one previous study has analysed the consistency of general supplement use in relation to cancer, though this was for total cancer mortality.<sup>363</sup> This EPIC-Heidelberg analysis reported a weak increased cancer mortality risk for consistent users which was non-significant, but reported evidence of a significant moderate increased risk in new users which became non-significant after excluding women diagnosed between the baseline and follow-up surveys.<sup>363</sup> This supports research mentioned in the previous chapter 6 which reported cancer patients tend to take supplements after cancer diagnosis. A limitation in both studies is that it is unknown whether participants were taking the same types of supplements at the follow-up as at baseline: for how long, how frequently, how many and at what doses they were taking them is also unknown. Some breast cancer cases (129 of the total 414) occurred between baseline and phase 2 in the UKWCS, consequently the shorter duration of supplement use for these women was likely to have had less influence on cancer development than incidences occurring after phase 2. Additionally, it is unknown how many years supplements were used before baseline.

Additionally, the current analysis found that supplement use modified the effects of menopausal status, socio-economic status and vigorous exercise on breast cancer risk in UK health conscious women. For supplement users, differences in breast cancer risks between menopausal statuses could arise because pre-menopausal women may tend to take different supplements from post-menopausal women and these may have different effects on breast cancer risk. For instance, a small 2008 UK national survey reported cod liver oil was used by half of those aged 55 plus (who were likely to be post-menopausal) compared with a quarter (26%) of those aged 16-54, whereas multivitamins and vitamin C were more likely to be used by 16-54 year old supplement users than those aged 65 plus.<sup>26</sup> Another explanation for higher risk in pre-menopausal women at high risk

of genetically inherited breast cancer in this cohort, who in the hope of reducing their cancer risk may have been more likely to take supplements. Sensitivity analyses could be undertaken excluding women who had a family history of breast cancer, though these women were not associated with supplement use and the recording of this at baseline appeared incomplete, as mentioned in section 4.7.6.3.1. One reason why non-supplement users of lower SES were found to have significantly higher risks of breast cancer than low SES supplement users or non-supplement of other SES is perhaps they were more likely to be poorly nourished. In relation to exercise, one study reported vitamin C supplementation was detrimental to the beneficial effects of physical exercise on insulin resistance<sup>366</sup> which in turn may affect breast cancer risk, as mentioned in section 3.5.6 of chapter 3. Further sub-group analysis was not undertaken since it has been acknowledged that this should be discouraged because it may lead to spurious findings from multiple testing, especially when no overall significance was found prior to sub-grouping.<sup>367</sup> Conversely Hemila and Kaprio (2008) argue that given the investment of time and resources into the establishment of large databases it is morally right to sub-analyse data, especially when biological and social affects are complex and not well established.<sup>368</sup>

Although general supplement use in poorly nourished populations may be required for the body to function adequately against the development of cancer, any effects of general supplement use per se on breast cancer risk in well-nourished populations may be more difficult to explain biologically. The women in the UKWCS appear to be wellnourished, and high intakes of different nutrients from supplements are likely to have different effects on cancer risk, some may produce opposing effects. What is more, as seen in the previous chapter, 6, a large proportion of women take many different types of supplements. Alternatively differences in risk could arise due to lifestyles or characteristic differences between users and non-users that have not been adequately adjusted for. Another limitation is that UKWCS participants were more health conscious than the general population and therefore not representative of the UK population as a whole. Results were not weighted to account for the large proportion of vegetarians in the UKWCS; nevertheless the aim was to examine the effects on health conscious women who may be prepared to alter their behaviour in light of new evidence.

To conclude, there was no evidence of significant associations between general supplement use and breast cancer risk in this UK cohort which comprised of mainly health conscious women. However, the results suggest that UK women who take supplements may have different breast cancer risks from non-users depending on their menopausal status, their socio-economic status or level of physical activity.

165

## **CHAPTER 8**

## 8 Do women who take supplements containing vitamin C in the UK have an increased risk of breast cancer? Phase 2 analyses of the UKWCS.

## 8.1 Summary

*Objective:* To determine whether breast cancer incidence in the UKWCS is associated with the use of supplements containing vitamin C.

*Method:* 11,184 middle-aged women from phase 2 of the UK Women's Cohort Study (UKWCS) were followed up for a median of 7.4 years. Associations between 239 registered incident breast cancers and vitamin C contained in supplements recorded by 4-day diaries were analysed by Cox's regression models using four intake categories: no frequent use of supplements containing vitamin C; frequent intake up to and including EU recommended allowances (1-60mg/d); between 60mg and 500mg/d; and high intake (≥500mg/d). Additionally, the relationship between breast cancer risk and phase 2 questionnaire reports of daily multivitamin use, a common source of low dose vitamin C, was examined for 12,642 women, including 269 cases. Adjustments were made for relevant covariates.

*Results:* Compared to women who did not use supplements containing vitamin C, there was no evidence of significant associations between breast cancer incidence and frequent vitamin C supplementation in any intake category, or in post-menopausal sub-analyses. Additionally, no dose-response associations were found (HR=0.98 per 60mg/d; 95% CI: 0.94, 1.02; P<sub>trend</sub>=0.3). However pre-menopausal women in the lowest intake category (1-60mg/d) had statistically significantly increased risks (HR=2.56; 95% CI: 1.41, 4.66; p=0.002) compared to non-users of vitamin C. There was also a moderate increased risk for pre-menopausal daily multivitamin users, however, this was not significant (HR=1.51; 95% CI: 0.90, 2.54).

*Conclusion:* There was little evidence that supplementation with vitamin C per se was associated with breast cancer incidence in UK women, even at high doses. The increased breast cancer risk for pre-menopausal women consuming supplements containing vitamin C less than or equal to EU recommendations may be due to the effects of other ingredients in the supplements, though the multivitamin use results did not entirely support this. Alternatively the positive associations may be spurious.

## 8.2 Introduction

As mentioned in chapter 2, vitamin C is one of the most commonly used dietary supplements in the UK.<sup>26 352</sup> In a small 2008 nationally representative UK survey 24% of people reported its use and it was one of the top three reported supplements used along with multivitamins (36%) and cod liver oil (35%).<sup>26</sup> In the UKWCS 17.5% of women recorded they took vitamin C on the phase 2 questionnaire; other popular supplements were multivitamins, fish oils and evening primrose oil. In total, phase 2 questionnaire recordings showed that 35.5% of UKWCS women used vitamin C, multivitamins, or antioxidants; all likely to contain vitamin C (Figure 2, chapter 2). This was comparable to diary recordings at phase 2 which, as seen in chapter 6, showed that 34% of women frequently took supplements containing any dose of vitamin C. Additionally, diary recordings showed that 10% frequently took high doses of 500mg or more and 5% frequently took doses of 1000mg or above (chapter 6).<sup>1</sup>

In addition to associations with healthier lifestyle behaviours, the frequent use of supplements containing vitamin C by UKWCS participants was also associated with a family history of breast cancer, as reported in chapter 6.<sup>1</sup> It is unknown, however, whether these women took vitamin C supplements for cancer prevention, or took them to increase immune function or for general health and well-being, the latter being a popular reason for taking supplements in general in the 2008 UK survey.<sup>26</sup> The 2007 WCRF/AICR report states that it is unwise to recommend widespread supplement use for cancer prevention since effects cannot be confidently predicted in the general population.<sup>7</sup> Indeed there are mechanisms whereby vitamin C could potentially increase or decrease cancer, as discussed in chapter 2.<sup>77 81</sup> Whilst supplement use by UK women continues to increase,<sup>24 25</sup> evidence is lacking regarding the benefits of vitamin C, in relation to mortality,<sup>351</sup> cancer and breast cancer specifically.<sup>7 351</sup>

As seen in section 3.7.4 of the literature review chapter, results from prospective studies examining associations between vitamin C from supplements and breast cancer risk have been inconsistent. Only six cohort studies have assessed the relationship between breast cancer incidence and vitamin C supplement intake by dose, without the inclusion of dietary intake.<sup>242 245 247 267 270 273</sup> These are summarised together in Table 36, along with one RCT.<sup>14</sup> Each study has assessed different categories of intake, but no pattern of risk emerges.

Author,	Study	Design	Supplement	Follow up and	Menopause	Vitamin C supplement associations,
year			information requested	cancer incidence	status	compared to non-users: OR, RR, HR
Rohan et al. 1993 <sup>267</sup>	Canadian Breast screening Study	Prospective Nested case- control Age 50-59	Specific vitamins Multivitamins	1982-1987 519 cases 1184 controls	Total of pre & post	<250mg/d: 1.04 (0.78-1.39) >250mg/d: 1.46 (1.05-2.01)
Kushi et al. 1996 <sup>242</sup>	Iowa Women's Health Study, US	Prospective Cohort N=34,387 Age 55-69	A,C,E, Multivitamin brand name frequency	Av 7yrs: 1986-1992 879 cases	Post	<200mg/d: 0.91 (0.77-1.08) 200-500: 0.94 (0.73-1.21) 500-1000: 0.86 (0.65-1.14) >1000: 0.77 (0.50-1.17) p <sub>trend</sub> = 0.20
Zhang et al. 1999 <sup>247</sup>	Nurses Health Study US	Prospective Cohort N=77,925 Age 33-60	Brand, doses, duration of specific vitamins & multivitamins Ingredients obtained	14yrs: 1980-1994 2523 cases Pre-meno ~27%	Total of pre & post	<400mg/d: 0.96 (0.73-1.25) 400-700: 1.08 (0.93-1.27) 750-1250: 1.02 (0.85-1.23) >1300: 1.04 (0.77-1.42)
Nissen et al. 2003 <sup>245</sup>	Danish Diet & Cancer Cohort	Prospective Nested case- control Age 50-64	Brand, doses, frequency in last year Ingredients obtained	Mean 4.7yrs: 1993-1997 228 cases 246 controls	Post	Per additional 100mg/d 1.06 (1.01-1.13)
Cui et al. 2008 <sup>270</sup>	Women's Health Initiative US	Prospective Cohort N=84,805 Age 50-79	Multiple and single vitamins A,C, E estimated	Av 7.6yrs 2879 cases	Post	>0<61mg/d: 1.03 (0.91-1.16) 61<347: 0.99 (0.88-1.11) 347-711: 1.04 (0.92-1.16) >711: 1.16 (1.04-1.30) p <sub>trend</sub> = 0.029
Roswall et al 2010 <sup>273</sup>	Diet, Cancer & Health Cohort Denmark	Prospective Cohort N=26,224 1993-1997 Age 50-64	Vitamin C content by brand obtained	Median 10.6yrs 1072 cases	Post	> $0 \le 40 \text{ mg/d:} 0.95 (0.74-1.21)$ > $40 \le 64$ : 1.03 (0.80-1.33) > $64$ : 0.96 (0.77-1.21) $p_{\text{trend}} = 0.41$
Lin et al. 2009 <sup>14</sup>	Women's Antioxidant Cardio- vascular Study, US	RCT Factorial design N=7,627 Age >=40	500mg/d vitamin C 600 IU/d vitamin E 50/2d beta carotene	Av 9 yrs 135 cases	Post	Compared to placebo group 500mg/d: 1.11 (0.87–1.41)

 Table 36 Studies analysing breast cancer risk by vitamin C supplement dose

Author,	Exclusions	Hormonal factors	Dietary factors	Other factors controlled for
Rohan et al. 1993 <sup>267</sup>	History of breast cancer Had mammograms	Age at menarche Age at first live birth Surgical menopause	Energy intake	Age Years of education Family history of breast cancer History of benign breast disease
Kushi et al. 1996 <sup>242</sup>	Post menopausal Full or partial breast removal Cancer (other than skin) Missing >30 items on FFQ. Extreme total energy intake	Age at menarche Age at first live birth Parity Age at menopause	Alcohol intake (Energy intake in sensitivity analysis)	Age Educational attainment BMI @ BL & 18yrs Waist/hip ratio Breast cancer family history in first degree relatives History of benign breast disease
Zhang et al. 1999 <sup>247</sup>	Previous cancer diagnosis Implausible total energy intake <500>3500kcal/d Missing >10 FFQ blank	Age at menarche Age at first live birth Parity Age at menopause HRT use Menopausal status	Alcohol intake Energy intake	Length of follow up Age Height BMI & BMI @18 yr Weight change from 18 Breast cancer in mother or sister
Nissen et al. 2003 <sup>245</sup>	Previous cancer diagnosis Pre-menopausal Missing items on FFQ Missing items on lifestyle questionnaire regarded as potential confounders	Age at first birth Parity HRT duration	Intake of vitamin A & E Alcohol intake (Energy intake in sensitivity analysis)	School education History of benign breast disease BMI (Matched on age at baseline)
Cui et al. 2008 <sup>270</sup>	History of breast cancer Implausible total energy intake <600>5000kcal/d	Age at menarche Age at first live birth Parity Age at menopause Contraceptive use HRT use Hysterectomy Bilateral oophorectomy	Energy intake Alcohol intake Dietary folate intake Dietary vitamin C	Age Ethnicity Educational level Smoking Family history of breast cancer History of benign breast disease

 Table 37 Methods and factors controlled for in studies of breast cancer risk and vitamin C supplement use

_	rable continued - Methods and factors controlled for in studies of breast cancer risk and vitamin C supplement use								
	Roswall	Cancer diagnosis	HRT	Alcohol intake	Age (used as the underlying time scale for hazard				
	et al	No lifetime menstruation	Parity	Total intake of vitamin	ratio rather than adjusted for it as a covariate)				
	2010 <sup>273</sup>	Missing covariates and micronutrient	Age at first birth	E and β-carotene	BMI				
		intake	-	Dietary or supplement	Education				
				vitamin C intake					
				mutually adjusted					
	Lin	Self-reported cancer diagnosis (other	Randomised	Randomised	Randomised				
	et al.	than skin)							
	2009 <sup>14</sup>	Unwilling to avoid vitamin A,C or E use							

Table continued - Methods and factors controlled for in studies of breast cancer risk and vitamin C supplement use

Half of the cohorts in Table 36 show no evidence of associations compared to women with no intake. In the Nurses' Health Study, Zhang et al. (1999) reported no evidence of associations for intake categories of <400mg/d, 400-700mg/d, 750-1250, and >=1300mg/d supplement vitamin C for pre-menopausal, post-menopausal or total women.<sup>247</sup> Kushi et al. (1996) also found no significant associations with risk for intake categories of <200mg/d, 200-500mg/d, 500-1000 or >1000 or per 100mg/d increase intake for post-menopausal women in the Iowa Women's Health Study.<sup>242</sup> The recent analyses of the full Danish cohort by Roswall et al. (2010) using much lower categories of intake (<40, 40-64, >64mg/d) also reported no associations.<sup>273</sup> Similarly, in comparisons between total vitamin C users and non-vitamin C users, no associations with breast cancer risk were found in the Netherland Cohort Study or in the Nurses Health Study.<sup>246 247</sup> Duration of use was also not associated with risk.<sup>247</sup> Only one RCT has used vitamin C supplements to assess breast cancer risk, where in a factorial design 500mg/d was taken singly or with vitamin E or beta carotene by North American women at high risk for cardiovascular events. There was no evidence of a statistically significant effect on breast cancer risk for vitamin C (OR=1.11; 95% CI: 0.87, 1.41).<sup>14</sup>

Conversely, in the Danish nested case-control study of post-menopausal women, Nissen et al. (2003) provided weak evidence of an increased risk per 100mg/d increase intake (OR=1.06; 95% CI: 1.01, 1.13), and results were similar after excluding users who consumed above 300mg/d.<sup>245</sup> In another nested case-control study Rohan et al. (1993) reported evidence of a larger increase in breast cancer risk in the Canadian Breast Screening Study for women consuming more than 250mg/d (OR=1.45; 95% CI: 1.05, 2.01) compared to women with no vitamin C intake from supplements.<sup>267</sup> Results split by post-menopausal or other women were not statistically significantly different.<sup>267</sup> In addition, a modest increase in breast cancer incidence was reported by Cui et al. (2008) for post-menopausal women in the Women's Health Initiative Observational Study for intakes of 711mg/d and over (HR=1.16 95% CI: 1.0, 1.30).<sup>270</sup>

The current analysis is the first UK study to examine the relationship between breast cancer risk and vitamin C intake from supplements. The study is important since a variety of supplements containing vitamin C are commonly consumed in the UK, as single or multi-ingredient types, and the consequences of taking them are unclear at population level. In contrast to previous studies, the aim for the current study was to determine whether the frequent use of supplements containing vitamin C recorded by diary, and categorised in relation to the EU recommended daily allowance (60mg/d) and high dose intake, was associated with breast cancer risk in UK women. Evidence of a dose-response relationship was also assessed. The initial study hypothesis, based on antioxidant and pro-oxidant mechanisms discussed in chapter 2, was that women

taking supplements containing low to medium amounts of vitamin C (up to 60mg/d or between 60-500mg/d) may be at lower risk than non-users and also high users (over 500mg/d); i.e. there may be a U-shaped relationship with breast cancer incidence. Additionally, the relationship between breast cancer risk and questionnaire reports of daily multivitamin use, a common source of low dose vitamin C, was examined.

## 8.3 Methods

#### 8.3.1 Subjects

UKWCS baseline data was gathered between 1995-1998 from 35,367 women.<sup>279</sup> Further details are found in section 4.1.1.1. At baseline 62% of participants took some type of dietary supplement.

All the initial participants were re-contacted at phase 2 between 1999 and 2004, on average 4 years after recruitment; 12,453 (35%) completed a follow-up health and lifestyle questionnaire and a 4-day food diary. Of the 12,453 women who had completed both the phase 2 diary and questionnaire, 1011 women were excluded who had any registered malignant cancer prevalent at diary date (except for skin cancer). An extra 252 women who self-reported cancer and a further six were excluded because they developed breast cancer within six months of the diary date. This left 11,184 women for the vitamin C analyses, including 239 cases.

#### 8.3.2 Determining supplement use

#### 8.3.2.1 Vitamin C use from diary recordings

The diaries requested supplement brand, name, amount taken and dosage of any supplement taken for each of the four diary days as shown in section 4.5.3.1 of the methods chapter. As explained in section 4.5.3.2, the supplement data provided by the women were matched against a database of supplement descriptions and ingredient composition. The mean daily vitamin C intake contained in all supplement types was calculated across the total number of diary days vitamin C was taken; this meant the women were more likely to be categorised by doses actually taken than if intake had been averaged across the four diary days. (A focus towards doses actually taken was decided because averaging across the four diary days could not provide a meaningful mean vitamin C amount absorbed by the body since vitamin C is not stored in the body in large amounts. Excretion from the body in urine starts at intakes of about 100mg/day and tissue saturation is reached between 200-400 mg/day.<sup>65</sup> Doses above 400mg are likely to be excreted in healthy women.<sup>65</sup>) Frequent use of supplements containing vitamin C in this study was defined as taking these on at least three out of the four

diary days. Because excess vitamin C is excreted, the nutrient is most likely to have an effect on the body if taken daily, or almost daily. Women taking supplements containing vitamin C on only one or two diary days were categorised in the reference category as non-frequent users.

The study population was split into four intake categories: women not frequently taking supplements containing vitamin C; those frequently taking up to and including 60mg/d; frequently taking between 60mg and less than 500mg/d; frequently taking 500mg/d or more vitamin C. Substantial agreement with frequencies given in the phase 2 questionnaire, completed concurrently with the diary, was observed and reported in section 5.2 of chapter 5. Category cut-off points in the current analysis were chosen with the aim of determining whether intakes in relation to the EU recommended daily allowance (60mg/d) were beneficial or detrimental in relation to breast cancer risk. As described in section 4.5.3.2 of the methods chapter, multivitamins consumed in the UK often contain the EU recommended daily allowance (EU RDA 60mg/d),<sup>53</sup> and 'antioxidant' supplements usually include more than the EU RDA but less than 500mg/d. The highest intake category of 500mg/d or more was chosen because supplements containing 500mg/d and above are usually classed as high dose and these usually do not contain other micronutrients. Women's characteristics by these categories of consumption were described using means and percentages by intake category. Significant differences in characteristics between vitamin C intake categories were established using chi squared tests for trend over categorical characteristics or tests for linear trend over the continuous variables.

The breast cancer risk of users taking high doses of 1000mg/d was assessed separately. It was observed in chapter 6 that phase 2 women who had a family history of breast cancer were significantly more likely to frequently take supplements containing 1000mg/d or more of vitamin C. In line with this analysis and in contrast to the main analysis, these high dose users were compared to women who did not take 1000mg/d or more vitamin C, instead of being compared to women who did not take any vitamin C from supplements. These 502 users, however, comprised only 4.5% of the study population, providing considerably less numbers to power this secondary analysis compared to the main analysis where there were 9.5% of women in the highest intake category ( $\geq$ 500mg/d). This high dose of 1000mg/day has been recommended as the safe upper limit; intakes at this level and above have been linked to adverse effects, particularly gastrointestinal disturbance.<sup>52</sup> This level is more than 15 times the recommended daily allowance (EU RDA = 60mg/day<sup>53</sup>) normally found in multivitamins.

Note that dietary vitamin C could not be included in any analyses since this had not been captured electronically from the food diary details; however total fruit and vegetable intake, the main source of vitamin C, had been electronically captured from diary data, and thus was available for adjustment.

#### 8.3.2.2 Multivitamin use from questionnaire recordings

Since multivitamins are a common source of low-dose vitamin C (as seen in section 4.5.3.2) additional analyses were produced to explore whether multivitamin use as recorded on the phase 2 questionnaire was associated with breast cancer risk. The frequency use of multivitamins with minerals, and multivitamins without minerals had been recorded as: more than daily; daily; weekly; monthly; less than monthly (as seen in section 4.5.2 in the methods chapter). The study population was split into: women currently taking multivitamins or multivitamins with minerals either daily or more than daily; and those women not taking these supplements daily. In relation to frequent use, the reporting of daily multivitamin use on the questionnaires may be more representative than 4-d diary recordings. Of the 14172 women who had completed the phase 2 questionnaire, 78 were excluded because of a missing questionnaire date, 1154 women were excluded who had any registered malignant cancer prevalent at diary date (except for skin cancer). An extra 292 women who self-reported cancer and a further 6 were excluded because they developed breast cancer within six months of the diary date. This left 12,642 women for the analysis including 269 cases.

#### 8.3.3 Statistical analyses

Cox proportional hazard modelling was used to estimate breast cancer risks relating to vitamin C intake from supplements. Hazard ratios were adjusted for age, BMI (underweight; normal; overweight; obese), parity, estimated cumulative breast feeding (wks), age at menarche, estimated units of alcohol per week (based on a glass of wine = 2.5 unit; half pint of beer = 1 unit; glass of sherry =1 unit; spirit measure= 1.5 units), physical activity (none; light/moderate physical activity most weeks; vigorous activity for at least 20 minutes once or twice a week; vigorous activity at least 20 minutes three or more times a week), smoking status (never; past; current), highest education level (none; CSE/O' level; A' level/City & Guilds; HNC; Degree; other), current HRT use, ever used a contraceptive pill, menopausal status at phase 2 (post-menopausal; pre-menopausal) and family history of breast cancer. Vigorous activity was defined as activity causing shortness of breath, rapid heart rate and sweating. Unlike the majority of prior studies, further adjustment was made for total fruit and vegetable intake (gms/d) and intake of vitamin E from supplements (mg/d), both recorded by diary, as explained in the discussion. All variables were taken from responses to the phase 2

questionnaire except for education level which was derived from answers to the questionnaire used at baseline since this information was not provided at phase 2. Reasons for selecting confounders are discussed in section 4.7.5 of this thesis. Menopausal status at phase 2 was determined as explained in chapter 4. Red meat was adjusted for in an additional sensitivity analysis since intake was found to be significantly and inversely associated with vitamin C supplementation in chapter 6,<sup>1</sup> and was positively associated with breast cancer incidence in this dataset in unadjusted analysis. A further sensitivity analysis was performed to adjust for fish oil supplementation which had been inversely associated with breast cancer risk in a US cohort,<sup>369</sup> though not significantly associated in this cohort.

The length of follow-up was calculated for each individual in the study as the number of days since completion of the phase 2 diaries for the vitamin C analyses (or questionnaire date for the multivitamin analyses) until the date of breast cancer diagnosis or to 1<sup>st</sup> July 2008, whichever is the sooner. The median follow-up period was 7.4 years. The assumption of proportional hazards over follow-up time was met; the log-log curves for intake categories were more or less parallel for pre-menopausal, post-menopausal and total women. An increment of 60mg/day was used to test for linear trends over continuous intake.

Likelihood ratio tests were used to determine whether breast cancer risk differed by menopausal status and sub-group time-to-event analyses were produced by menopausal status. Likelihood ratio tests were also performed for socio-economic status (high: professional and managerial; intermediate; low: routine and manual) and exercise behaviour since these variables were found to have significant effect modifications on breast cancer risk with supplement use/none use in chapter 7 of this thesis.

#### 8.3.4 Excluding family history of breast cancer

Exclusions were made in two separate sensitivity analyses for women with a family history of breast cancer, and for women estimated to be at raised genetic risk of breast cancer. A family history of breast cancer was defined as having a blood relative diagnosed with breast cancer as reported by participants at phase 2 in the questionnaire; they were questioned about cancers in mothers, fathers, sisters, brothers, aunts, and uncles. This information was also used to identify a small percentage (2%) of phase 2 UKWCS women who may have been at raised risk of developing breast cancer based on guidelines from the NICE report on familiar breast cancer.<sup>319</sup> As explained in detail in section 4.7.6.3.2 and 0 of this thesis, being at raised risk depended on the age of diagnosis of their relatives and whether they were

1<sup>st</sup> or 2<sup>nd</sup> degree relatives. Unfortunately these NICE guidelines<sup>319</sup> could not be followed in some areas for this analysis; the UKWCS had no information on daughters or sons with breast cancer. NICE guidelines also stated that women with a first degree relative diagnosed with bilateral breast cancer before the age of 50 would also be at raised risk; however cases of bilateral breast cancer in the UKWCS could not be identified.

## 8.4 Results

In this dataset 34% of women frequently took supplements containing vitamin C, on at least three of the four diary days; 13% (1493) took doses less than or equal to 60mg/d; 11% (1235) took between 60mg and 500mg/d; 9.5% (1063) took 500mg/d or more (Table 38) and 4.5% (503) frequently took high doses of 1000mg or more.

There was a tendency for women who took supplements containing higher doses of vitamin C ( $\geq$ 500mg/d) to have a lower BMI, or to consume more fruit and vegetables or less red meat (Table 38). They also were more likely to take many types of other supplements, visit alternative practitioners, have a family history of cancer, or have fewer children. Women in the low or mid-range categories (1-60mg/d and >60<500mg/d) were less likely to be educated to A' level compared to women taking higher doses or not taking any. Although 49% of women not taking any supplements containing vitamin C at phase 2 did take some other types of supplements at baseline, the percentage taking other supplements at baseline was much higher in the other vitamin C intake categories (78-86%).

	Vitami	n C supplement	y days			
	None	1-60mg/d	>60<500mg/d	≥500mg/d	Total	P value <sup>a</sup>
	N=7393	N=1493	N=1235	N=1063	N=11184	
Age (years), mean (sd)	56.5 (9.1)	55.9 (8.7)	57.5 (8.8)	56.0 (8.7)	56.5 (9.0)	0.9
BMI (kg/m <sup>2</sup> ), mean (sd)	24.9 (4.5)	24.4 (4.2)	24.6 (4.2)	24.2 (4.1)	24.7 (4.4)	<0.001
Waist, mean (sd)	81.7 (11.4)	80.7 (10.8)	81.0 (11.0)	79.9 (10.7)	81.3 (11.3)	<0.001
Age at menarche, mean (sd)	12.8 (1.6)	12.8 (1.6)	12.7 (1.6)	12.8 (1.5)	12.8 (1.6)	0.2
Age at first birth, mean (sd)	26.1 (4.6)	25.7 (4.4)	25.6 (4.7)	26.0 (4.9)	26.0 (4.6)	0.006
Number of children, mean (sd)	1.9 (1.3)	1.8 (1.3)	1.8 (1.3)	1.7 (1.3)	1.8 (1.3)	<0.001
Est. cumulative breastfeeding weeks (sd)	27 (37)	26 (36)	23 (33)	24 (36)	26 (37)	<0.001
Portions red meat /week median (IQR)	2 (0-4)	1 (0-3)	1 (0-3)	0 (0-2)	1 (0-3)	<0.001
Portions white meat /week median (IQR)	2 (0-3)	1 (0-3)	1 (0-3)	1 (0-3)	2 (0-3)	<0.001
Portions fish meat /week median (IQR)	1 (1-2)	1 (0-2)	1 (0-2)	1 (0-2)	2 (0-3)	0.02
Portions fruit /week median (IQR)	12 (7-18)	14 (8-20)	14 (7-20)	14 (8-20)	14 (7-20)	<0.001
Portions vegetables /week median (IQR)	12 (7-15)	12 (7-17)	13 (7-18)	14 (8-18)	12 (7-16)	<0.001
Diary fruit intake (grams) mean (sd)	858 (628)	943 (658)	947 (647)	1026 (702)	895 (644)	<0.001
Diary veg. intake (grams) mean (sd)	827 (452)	865 (462)	922 (506)	960 (536)	855 (536)	<0.001
Vitamin E sup. intake (mg), mean (sd)	8 (39)	25 (69)	49 (82)	75 (130)	21 (67)	<0.001
Number of supplement types taken	1.2 (1.6)	2.7 (1.7)	3.7 (2.1)	4.5 (2.4)	2.0 (2.1)	<0.001
Professional or managerial n(%)	2981 (41)	592 (40)	510 (42)	455 (43)	4538 (41)	0.03
Educated to 'A'-level or above n(%)	4307 (60)	807 (55)	691 (57)	641 (61)	6446 (59)	0.7
No children n(%)	1492 (20)	307 (21)	290 (23)	300 (28)	2389 (21)	<0.001
Ever used contraceptive pill n(%)	4860 (68)	1030 (70)	832 (69)	752 (72)	7474 (68)	0.002
Current HRT use n(%)	1481(20)	343 (23)	300 (24)	254 (24)	2378 (21)	<0.001
Post-menopausal at phase 2	4952 (67)	973 (65)	894 (72)	705 (66)	7524 (67)	0.2
Ex-smoker n(%)	2151 (29)	438 (30)	398 (32)	377 (36)	3364 (30)	<0.001
Drinks alcohol less than once a week	2450 (33)	512 (34)	423 (34)	371 (35)	3756 (34)	0.2
Vigorous activity >=3/week n(%)	1241 (18)	294 (21)	275 (24)	236 (24)	2046 (19)	<0.001
Bowel movements more than once/day	1226 (17)	279 (19)	258 (22)	237 (23)	2000 (18)	<0.001
Takes any supplements at baseline n(%)	3312 (49)	1054 (78)	924 (82)	842 (86)	6132 (60)	<0.001
Takes more than 4-10 supplement types	560 (8)	324 (22)	479 (39)	476 (45)	1839 (16)	<0.001
Visits alternative practitioner >=5 /year	463 (6)	119 (8)	132 (11)	159 (15)	873 (8)	<0.001
Family history of breast cancer n(%)	1362 (18)	261 (17)	242 (20)	211 (20)	2076 (19)	0.2
Family history of any cancer n(%) <sup>b</sup>	4232 (57)	851 (57)	764 (62)	638 (60)	6485 (55)	0.006

 Table 38 Characteristics of women at phase 2 in the UKWCS by vitamin C supplement intake group

 $a^{a}$ p= test for trend over continuous variable, sd =standard deviation

# 8.4.1 Breast cancer risk and use of supplements containing vitamin C

Compared to women who did not take supplements containing vitamin C there was no evidence of an association between breast cancer risk and regular intake of supplements containing vitamin C for total women for any of the intake categories in the unadjusted and adjusted analyses (Table 39). Neither was there evidence of a dose-response association across these supplement users (HR=0.98 per 60mg/d; 95% CI: 0.94, 1.02;  $p_{trend}$ =0.3) or when non-users of these supplements were included. Excluding women with a family history of breast cancer or women at raised risk of breast cancer made little difference to the results (Table 40 and Table 41). In the tests for effect modification, menopausal status was found to significantly modify the effect of vitamin C supplement intake groups on breast cancer incidence (adjusted p=0.02). Socio-economic status was also found to modify the associations of risk with vitamin C intake (adjusted p=0.03) but exercise behaviour did not (adjusted p=0.7). Unfortunately the number of cases was considered too low to power sub-analyses by socio-economic status.

In the subgroup analysis by post-menopausal status no significant associations were found relating to dose-response relationships (HR=1.00 per 60 mg/d; 95% CI: 0.96, 1.05;  $p_{trend}$ =1.0) or intake category.

Dose-response relations for pre-menopausal women were also non-significant. However pre-menopausal women who frequently took supplements containing less than or equal to 60mg/d of vitamin C were over twice as likely to develop breast cancer (HR=2.37; 95% CI: 1.32, 4.27; p=0.004) compared to women not taking any supplements containing vitamin C, in the multivariate model after adjustment of the majority of potentially confounding factors. The odds increased further after adjustment by fruit and vegetable intake and vitamin E supplement intake recorded by diary (HR=2.56; 95% CI: 1.41, 4.66; p=0.002). Additional adjustments for red meat intake strengthened the risks slightly e.g. for all pre-menopausal women in the low dose group (HR=2.78; 95% CI: 1.52, 5.09; p=0.001), and adjusting for daily fish oil supplementation made little difference to hazard ratios (both not tabled). As observed in Table 40 and Table 41, a significant positive association for pre-menopausal women who frequently took supplements containing less than or equal to 60mg/d of vitamin C was also found in sensitivity analyses excluding women with a family history of breast cancer (HR=3.02; 95% CI: 1.53, 5.96; p=0.002) and when only women at raised risk of breast cancer (HR=2.78; 95% CI: 1.52, 5.09; p=0.001) were excluded.

	Cases/	Unadjusted	Multivariateb	Fully adjusted <sup>c</sup>
Supplement vitamin	Non-			
C intake <sup>a</sup>	cases	HR (95% CI)	HR (95% CI)	HR (95% CI)
Total women				
No vitamin C	156/7237	1	1	1
1-60mg/d	35/1458	1.10 (0.76, 1.58)	1.16 (0.77, 1.75)	1.15 (0.76,1.74)
>60<500mg/d	29/1206	1.10( 0.74, 1.64)	1.17 (0.75, 1.82)	1.15 (0.73,1.80)
≥500mg/d	19/1044	0.83 (0.51, 1.33)	0.77 (0.43, 1.37)	0.75 (0.41,1.35)
Per 60ma/d (users)	83/3708		0.98 (0.94 1.02)	0.98 (0.94 1.02)
n ei oomg/a (aseis)	03/3700	0.8	0.30 (0.34, 1.02)	0.30 (0.34, 1.02)
Ptrend Per 60ma/d (all)	230/100/5			0.5
n ci oomgia (anj	200/10040	0.5	0.00 (0.00, 1.02)	0.00 (0.00, 1.02)
Ptrend		0.5	0.4	0.4
Post-menopausal				
No vitamin C	112/4840	1	1	1
1-60mg/d	15/958	0.67 (0.39, 1.14)	0.67 (0.36, 1.26)	0.65 (0.34, 1.23)
>60<500mg/d	25/869	1.22 (0.79, 1.89)	1.25 (0.77, 2.05)	1.21 (0.74, 2.00)
≥500mg/d	14/691	0.86 (0.49, 1.50)	0.77 (0.38, 1.53)	0.70 (0.34, 1.43)
Per 60ma/d (users)	54/2518	1 02 (0 99 1 05)	1 00 (0 96 1 05)	1 00 (0 96 1 05)
Diment	0 1/2010	0.2	0.8	1.00 (0.00, 1.00)
Per 60ma/d (all)	166/7358	1 02 (0 99 1 04)	1 00 (0 96 1 03)	1 00 (0 96 1 03)
Dwand	100,1000	0.2	0.9	0.8
Pirena		0.2	0.0	010
Pre-menopausal				
No vitamin C	44/2397	1	1	1
1-60mg/d	20/500	2.14 (1.26, 3.63)	2.37 (1.32, 4.27)	2.56 (1.41, 4.66)
>60<500mg/d	4/337	0.64 (0.23, 1.79)	0.83 (0.29, 2.34)	1.01 (0.34, 2.99)
≥500mg/d	5/353	0.76 (0.30, 1.91)	0.77 (0.27, 2.20)	0.91 (0.31, 2.65)
Por 60ma/d (usors)	20/1100	0.02 (0.96, 1.01)	0.01 (0.91 1.01)	0.02 (0.82, 1.02)
n.	23/1130	0.00 (0.00, 1.01)		0.32 (0.02, 1.02)
Ptrend Per 60ma/d (all)	73/3587	0.00	0.07	
n.	10/0001	0.00 (0.00, 1.00)	0.00 (0.03, 1.03)	0.37 (0.30, 1.04)
Ptrend		0.7	0.2	0.0

**Table 39** Breast cancer risks of women supplementing with vitamin C recorded by diaries at phase 2 in the UKWCS

	Cases/	Unadjusted		Fully adjusted <sup>a</sup>	
Supplement vitamin C intake	Non-				
	cases	HR	(95% CI)	HR	(95% CI)
Total women					
No vitamin C taken 3-4 days	115/5916	1		1	
1-60mg/d	30/1202	1.26	(0.85, 1.89)	1.22	(0.77, 1.94)
>60<500mg/d	24/969	1.26	(0.81, 1.96)	1.20	(0.72, 2.00)
≥500mg/d	15/837	0.90	(0.53, 1.55)	0.75	(0.38, 1.48)
Per 60mg/d (users)	69/3008	0.98	(0.94, 1.02)	0.98	(0.94, 1.03)
Ptrend			0.4		0.4
Per 60mg/d (all)	184/8924	1.01	(0.98, 1.03)	0.99	(0.96, 1.03)
<i>p</i> <sub>trend</sub>			0.6		0.6
Post-monopousal					
No vitamin C taken 3-4 days	85/3025	1		1	
1-60mg/d	13/787	0.75	(0.42, 1.35)	0.67	(0.33, 1.36)
>60~500ma/d	20/69/	1 32	(0.42, 1.00)	1 20	(0.68, 2.14)
>500~000mg/d	11/553	0 90	(0.01, 2.14) (0.48, 1.69)	0.68	(0.00, 2.14) (0.30, 1.54)
Looong/d	11/000	0.00	(0.40, 1.00)	0.00	(0.00, 1.04)
Per 60mg/d (users)	44/2034	1.01	(0.97, 1.04)	1.01	(0.96, 1.06)
Ptrend			0.7		0.7
Per 60mg/d (all)	129/5959	1.02	(0.99, 1.05)	1.00	(0.96, 1.04)
$p_{trend}$			0.2		0.9
Pre-menonausal					
No vitamin C taken 3-4 days	30/1991	1		1	
1-60mg/d	17/415	2 67	(1 47 4 84)	3.02	(1.53, 5.96)
>60<500mg/d	4/275	0.95	(0.34, 2.70)	1 32	(0.43, 3.99)
>500mg/d	4/284	0.00	(0.32, 2.70)	1.02	(0.29, 3.47)
	7/207	0.01	(0.02, 2.00)	1.01	(0.20, 0.77)
Per 60mg/d (users)	25/974	0.91	(0.82, 1.01)	0.87	(0.75, 1.01)
Ptrend			0.07		0.07
Per 60mg/d (all)	55/2965	0.97	(0.90, 1.03)	0.96	(0.88, 1.05)
Ptrend			0.3		0.4

**Table 40** Breast cancer risks of women supplementing with vitamin C recorded by diaries in the UKWCS, excluding women with a family history of breast cancer

<sup>a</sup>Adjusted for age, BMI (grouped), age at menarche, parity, estimated alcohol units consumed per week, activity intensity and frequency, smoking status (never, ex, current), education level, current HRT use, ever used a contraceptive pill, estimated cumulative breast feeding, total fruit and vegetable intake recorded by diary (gms/d) and vitamin E supplements use (mg/d) recorded by diary

	Cases/	Unadjusted		Fully adjusted <sup>a</sup>	
Supplement vitamin C intake	Non-cases				
		HR	(95% CI)	HR	(95% CI)
Total women					
No vitamin C taken 3-4 days	147/7103	1		1	
1-60mg/d	35/1425	1.17	(0.81, 1.69)	1.24	(0.81, 1.88)
>60<500mg/d	28/1177	1.14	(0.76, 1.70)	1.19	(0.75, 1.90)
≥500mg/d	18/1023	0.83	(0.51, 1.36)	0.81	(0.45, 1.47)
Per 60mg/d (users)	81/3625	0.98	(0.95, 1.01)	0.98	(0.94, 1.02)
Ptrend			0.3		0.3
Per 60mg/d (all)	228/10728	1.00	(0.98, 1.02)	0.98	(0.96, 1.02)
<i>p</i> <sub>trend</sub>			0.9		0.5
Post-monopausal					
No vitamin C taken 3-4 days	106/4730	1		1	
1-60mg/d	15/938	0 70	(0.41 1.21)	0 70	(0.37, 1.33)
>60~500mg/d	24/851	1.24	(0.41, 1.21)	1 26	(0.37, 1.33)
>500~300mg/d	13/680	0.84	(0.00, 1.00) (0.47, 1.49)	0.77	(0.70, 2.11) (0.37, 1.57)
2300mg/d	13/000	0.04	(0.47, 1.43)	0.77	(0.07, 1.07)
Per 60mg/d (users)	52/2469	1.00	(0.96, 1.04)	1.00	(0.96, 1.05)
Ptrend			1.0		0.9
Per 60mg/d (all)	158/7208	1.01	(0.98, 1.04)	1.00	(0.96, 1.04)
P <sub>trend</sub>			0.5		0.9
Dro monopolical					
No vitamin C taken 3-4 days	11/2361	1		1	
1-60mg/d	20/487	2 3 2	(1 36 3 96)	2 78	(1.52, 5.09)
>60<500mg/d	20/407	0.70	(1.30, 3.30) (0.25, 1.96)	1.08	(1.32, 3.09)
>500<500mg/d	5/3/3	0.70	(0.23, 1.90)	1.00	(0.30, 3.19)
2500mg/u	5/343	0.62	(0.33, 2.08)	1.00	(0.34, 2.92)
Per 60mg/d (users)	29/1156	0.93	(0.86, 1.01)	0.92	(0.82, 1.02)
<b>P</b> trend			0.08		0.1
Per 60mg/d (all)	70/3520	0.98	(0.93, 1.03)	0.97	(0.91, 1.04)
Dtrend			0.5		0.4

 Table 41 Breast cancer risks of women supplementing with vitamin C recorded by diaries in the UKWCS, excluding women at raised risk of breast cancer

<sup>a</sup>Adjusted for age, BMI (grouped), age at menarche, parity, estimated alcohol units consumed per week, activity intensity and frequency, smoking status (never, ex, current), education level, current HRT use, ever used a contraceptive pill, estimated cumulative breast feeding, family history of breast cancer, total fruit and vegetable intake recorded by diary (gms/d), and vitamin E supplements use (mg/d) recorded by diary

No associations were found between breast cancer risk and frequent use of supplements containing 1000mg or more vitamin C compared with women not taking these amounts. Adjusted hazard ratios for pre-menopausal women were HR=0.37 (95% CI: 0.05, 2.70) and for post-menopausal women were HR=1.08 (95% CI: 0.47, 2.52); none were close to significance (Table 42). The number of cases in these analyses was extremely low.

**Table 42** Breast cancer risks of women supplementing with 1000mg/d or more vitamin C (N=502) compared to women not taking these doses at phase 2 in the UKWCS

	Cases	Unadjusted	Multivariate <sup>a</sup>	Fully adjusted <sup>⁵</sup>
Supplement vitamin C intake	/Non- cases	HR (95% CI)	HR (95% CI)	HR (95% CI)
Total women <1000mg/d <sup>c</sup>		1	1	1
1000mg or more (y/n)	10/492	0.92 (0.49, 1.73)	0.88 (0.41, 1.87)	0.85 (0.39, 1.84)
Post-menopausal <1000mg/d <sup>c</sup> 1000mg or more (y/n)	8/310	1 1.14 (0.56, 2.32)	1 1.17 (0.51, 2.66)	1 1.08 (0.47, 2.52)
Pre-menopausal <1000mg/d <sup>c</sup> 1000mg or more (y/n)	2/182	1 0.52 (0.13, 2.14)	1 0.32 (0.04, 2.34)	1 0.37 (0.05, 2.70)

There were no substantial differences in any of the above results if self-reported prevalent cancers were not excluded.

#### 8.4.2 Breast cancer risk and multivitamin use

From the questionnaire data on supplement use it was observed that compared to women who did not take multivitamin supplements every day there was no evidence of significant associations between breast cancer risk and daily use of multivitamin supplements for total women in the unadjusted and adjusted analyses (Table 43) or for post-menopausal women. However both the unadjusted and adjusted results for premenopausal women showed a moderate positive association and the unadjusted result was close to significance (HR=1.55; 95% CI: 0.98, 2.45 and HR=1.51; 95% CI: 0.90, 2.54 respectively). Results did not change substantially when women with a raised risk of breast cancer were excluded (Table 45). When women with a family history of breast cancer were excluded the risk for pre-menopausal multivitamin users was strengthened, the unadjusted results reached significance and the adjusted HR=1.79; 95% CI: 0.99, 3.22, Table 44).

Results did not alter substantially when women who recorded daily vitamin C supplement use on the phase 2 questionnaires were excluded (results not shown).

Daily multivitamin intake	Cases/	Una	Unadjusted		djusted <sup>a</sup>
	Non-cases	HR	(95% CI)	HR	(95% CI)
Total women					
Not taking daily multivitamins	198/9362	1		1	
Taking daily multivitamins	71/3011	1.09	(0.81, 1.43)	1.12	(0.81, 1.54)
Post-menopausal					
Not taking daily multivitamins	149/6396	1		1	
Taking daily multivitamins	42/1901	0.93	(0.66, 1.31)	0.95	(0.63, 1.42)
Pre-menopausal	/				
Not taking daily multivitamins	49/2966	1		1	
Taking daily multivitamins	29/1110	1.55	(0.98, 2.45)	1.51	(0.90, 2.54)

**Table 43** Breast cancer risks of women using multivitamins recorded by questionnaire at phase 2 in the UKWCS

<sup>a</sup>Adjusted for age, BMI (grouped), age at menarche, parity, estimated alcohol units consumed per week, activity intensity and frequency, smoking status (never, ex, current), education level, current HRT use, ever used a contraceptive pill, estimated cumulative breast feeding, family history of breast cancer and total fruit and vegetable intake recorded by diary (gms/d).

Daily multivitamin intake	Cases/	Unadiusted		Α	diusted <sup>a</sup>
Daily mainmainmaine	Non-cases	HR	(95% CI)	HR	(95% CI)
Total women					
Not taking daily multivitamins	151/7668	1		1	
Taking daily multivitamins	59/2447	1.20	(0.89, 1.62)	1.25	(0.88, 1.78)
Post-menopausal					
Not taking daily multivitamins	116/5204	1		1	
Taking daily multivitamins	35/1537	1.00	(0.69, 1.46)	1.05	(0.67, 1.65)
Pre-menopausal					
Not taking daily multivitamins	35/2464	1		1	
Taking daily multivitamins	24/910	1.81	(1.08, 3.05)	1.79	(0.99, 3.22)

**Table 44** Breast cancer risks of women using multivitamins recorded by questionnaire

 at phase 2 in the UKWCS, excluding women with a family history of breast cancer

<sup>a</sup>Adjusted for age, BMI (grouped), age at menarche, parity, estimated alcohol units consumed per week, activity intensity and frequency, smoking status (never, ex, current), education level, current HRT use, ever used a contraceptive pill, estimated cumulative breast feeding and total fruit and vegetable intake recorded by diary (gms/d).

**Table 45** Breast cancer risks of women using multivitamins recorded by questionnaire at phase 2 in the UKWCS, excluding women at raised risk of breast cancer

Daily multivitamin intake	Cases/	Unadjusted		Adjusted <sup>a</sup>	
	Non-cases	HR	(95% CI)	HR	(95% CI)
Total women					
Not taking daily multivitamins	190/9193	1		1	
Taking daily multivitamins	68/2938	1.09	(0.83, 1.44)	1.16	(0.84, 1.61)
Post-menopausal					
Not taking daily multivitamins	143/6271	1		1	
Taking daily multivitamins	40/1860	0.92	(0.65, 1.31)	0.97	(0.64, 1.48)
Pre-menopausal					
Not taking daily multivitamins	47/2922	1		1	
Taking daily multivitamins	28/1078	1.58	(0.99, 2.52)	1.60	(0.94, 2.70)

<sup>a</sup>Adjusted for age, BMI (grouped), age at menarche, parity, estimated alcohol units consumed per week, activity intensity and frequency, smoking status (never, ex, current), education level, current HRT use, ever used a contraceptive pill, estimated cumulative breast feeding and total fruit and vegetable intake recorded by diary (gms/d) and family history of breast cancer.

## 8.5 Discussion

This analysis of phase 2 of the UKWCS found little evidence of associations between breast cancer incidence and vitamin C supplementation per se, when compared to women not consuming supplements containing vitamin C, as recorded by food diaries. Specifically, there was no evidence of a dose-response relationship for total, pre- or post-menopausal women. Additionally there was no evidence of a U-shaped relationship that was initially hypothesised to occur between vitamin C intake and breast cancer risk. Conversely, the Danish nested-case control study found evidence of a slight dose-response relationship: a significant increase in risk per additional 100mg/d of vitamin C intake from supplements for postmenopausal women (HR=1.06; 95% CI: 1.01, 1.13) (Table 36).<sup>245</sup>

No significant associations were found for total women or for post-menopausal women in any of the intake categories. This is consistent with US results from the Nurses Health Study, the Iowa Women's Health Study and the RCT of the Women's Antioxidant Cardiovascular Study,<sup>14 242 247</sup> and also the most recent findings from a Danish cohort.<sup>273</sup> However, it is in contrast with another US study, the Women's Health Initiative, and also with the Canadian and a nested case-control analysis of the Danish study which did find increased risks for post-menopausal women taking vitamin C supplements.<sup>245</sup> <sup>267</sup> <sup>270</sup> The latter two studies used case-control methodology which unlike full cohort studies could have been affected by biased selection of controls.<sup>245 267</sup> The Women's Health Initiative had the highest average age of all the cohorts (64 years), involving women as old as 79,270 and their positive associations (shown in Table 36) could indicate that high doses of vitamin C may promote the progression of cancer in the later stages of the disease. Differences in results cannot be explained by the number of cases involved to power the analyses since the two largest studies, the Nurses Health Study and Women's Health Initiative, produced conflicting results.<sup>247 270</sup> The length of follow-up for all but one of the studies, including ours, was on average between four and nine years; the average follow-up for the Nurses Health Study which found no association was considerably longer: 14 years.<sup>247</sup> Longer follow-up studies may be less likely to find associations if supplement usage did not remain relatively constant. Another explanation for the null results of studies for total and postmenopausal women, is that vitamin C may not affect hormonal mechanisms of cancer initiation or development; as discussed in Chapter 3, levels of endogenous sex hormones are one of the main risk factors for post-menopausal breast cancer.<sup>147</sup>

In the current study, however, for pre-menopausal women an increased risk of breast cancer was associated with the lowest category of intake: below or equal to 60mg/d.

Similar to most studies, the classification of pre-menopausal women was at recording date rather than diagnosis or censor date, meaning that pre-menopausal women taking supplements in this category appear to have an increased risk of later developing breast cancer; whether this was when they were pre- or post-menopausal. Although vitamin C intake has been associated with increased breast density, an increased risk factor for breast cancer, this result for low dose vitamin C is difficult to explain biologically. The result may be due to other characteristics of these pre-menopausal women in the UKWCS. Indeed, in a previous UKWCS analysis of pre-menopausal women another result was difficult to account for; vegetarians were found to have a significantly higher breast cancer risk than low meat consumers.<sup>82</sup> The authors suggested a family history of breast cancer may have increased a tendency for vegetarianism as well as an increased risk of breast cancer.<sup>82</sup> However, the significant association in the current analysis is unlikely to be attributed to women genetically at high risk since the results were adjusted for women with a family history of breast cancer. Furthermore, excluding these women in sensitivity analyses increased the risk further. The level of significance was high therefore this result may not have occurred by chance despite the multiple testing through sub-analyses, although the number of cases was low and could have produced spurious results. Previously, the Canadian study reported a significant moderate increase in risk evident for intakes above 250mg/d compared to women with no intake for a subgroup combining pre- and perimenopausal women.<sup>267</sup> No previous study has analysed pre-menopausal women consuming intakes as low as 60mg/d vitamin C. In the UKWCS analysis there was no evidence of significant associations for pre-menopausal women supplementing with vitamin C above the EU recommended daily intake of 60mg/d. However, since there were less than 10 cases in these higher intake groups, the analysis is likely to have been underpowered to find an effect for pre-menopausal women. In chapter 7 of this thesis a non-significant increased risk of breast cancer was found for pre-menopausal women who took any type of supplement at baseline, compared to those who did not; the use of supplements containing low levels of vitamin C may have contributed to this result. The reason the results did not reach significance, may have been due to dilution by use of other supplement types which did not have the same effect.

Although the use of supplements containing 1000mg/d of vitamin C was associated with a family history of breast cancer in the UKWCS in chapter 6 of this thesis,<sup>1</sup> there was no evidence in the current analysis that women taking these high doses had a significantly different risk of breast cancer compared to women not taking them, even when adjusting for family history of breast cancer. Due to the very small number of women who developed breast cancer and who had taken doses more than or equal to 1000mg/d, the analysis is likely to have been underpowered to find an effect. Only one

previous study assessed intakes specifically above 1000mg/d, the Iowa Women's Health Study,<sup>247</sup> which showed a non-significant inverse association for postmenopausal women with wide confidence intervals (HR=0.77; 95% CI: 0.50, 1.17).

Women with prevalent self-reported cancers were excluded in the UKWCS analyses, which meant that some women who had non-malignant cancers may have been inadvertently excluded from the analyses. However sensitivity analyses which included prevalent self-reported cancers did not produce substantially different results. From the reports of previous studies it was not clear whether their exclusions for prevalent cancers were self-reported or not.

It is unlikely that differences in results between this and previous studies were due to variations in adjustments for potential confounding factors; the current analyses controlled for the majority of those used in previous studies (as shown in Table 37). However, unlike previous studies, no adjustment was made for energy intake in the UKWCS since this data was not electronically captured for all women. Nevertheless two studies reported that adjustments for energy intake in sensitivity analyses did not alter their results.<sup>242 245</sup> Contrary to previous studies no adjustment was made for age at first live birth in the UKWCS analyses due to large numbers of missing data; instead cumulative time of breast feeding and parity was used which have been found to be a good predictor of breast cancer risk.<sup>167</sup> Of prior studies that reported an increase in risk for post-menopausal women, the Women's Health Initiative adjusted for dietary vitamin C intake,<sup>270</sup> and the Danish study adjusted for vitamin A and E supplement intake;<sup>245</sup> this was in contrast to the other studies detailed in Table 37. Although dietary vitamin C and supplement vitamin A was not available in the UKWCS, further adjustments were made for vitamin E supplement intake and for dietary fruit and vegetable intake, which has been found to correlate well with plasma vitamin C;<sup>67</sup> these, however, did not substantially alter the results. Adjustment were made for these variables since vitamin C supplement use has been significantly associated with fruit and vegetable consumption<sup>1</sup> and due to vitamin C's role in the regeneration of vitamin  $E^{76}$  the latter could potentially modulate vitamin C's activity relating to cancer development. The majority of supplements containing vitamin C at medium to low doses such as those classed as multivitamins and antioxidants also contain vitamin E. However, as discussed in 4.7.6.2, vitamin E may be a mediator between vitamin C and breast cancer incident so adjusting for it could create bias, therefore this was only undertaken in the sensitivity analyses.

Fish oil supplements, which are taken by 26.5 % of UKWCS women (figure 2, chapter 2), have been inversely associated with breast cancer risk;<sup>369</sup> and in a recent UK survey half of women over 55 years of age reported taking cod-liver oil compared to a

quarter of younger women.<sup>26</sup> However, additional adjustment for daily use of fish oil supplements did not change the results for the breast cancer and vitamin C analysis in the UKWCS. Furthermore, although there was evidence of a reduced breast cancer risk from daily use of fish oil supplements in the UKWCS, it was not significant. Previous studies in Table 36 did not adjust for fish oil supplements.

Since the body has limited capacity to absorb and store this water soluble vitamin; <sup>65</sup> daily vitamin C supplement use is plausibly more likely to have an effect on the body than infrequent intake. Therefore only women recording at least 3 out of 4 days intake of supplements containing vitamin C were classed as frequent users in this analysis. Regardless of this, the diary recordings may represent short-term episodic intake, particularly because vitamin C is promoted for reducing seasonal common colds and no seasonal adjustments to the results have been made. Nevertheless, the majority of vitamin C users (82%) in the UKWCS reported taking a supplement on average four years earlier, however the type taken is unknown and respective seasons were not assessed.<sup>1</sup> Therefore due to its short-term nature, diary data may not be more superior than questionnaire data for elucidating the relationship between breast cancer risk and long-term use of supplements, unless repeated measures are taken.<sup>370</sup> Unfortunately baseline details of supplements used had not been electronically captured.

Vitamin C contained in a wide variety of supplement types recorded by diary was determined using a database of brand specific vitamin C content and extensive cleaning was done (as described in the methods chapter). Therefore intake assessed in relation to breast cancer risk in the UKWCS may be more accurate than other approaches used in some of the previous studies. Only the Nurses Health Study, which found no associations, and the two Danish studies, which found conflicting associations, appeared to have used a database of supplement ingredients.<sup>245 247 273</sup> In contrast some of the other studies appear have estimated or not accounted for the vitamin C content of many supplements used, and this may have reduced their ability to find a dose-response affect if one existed. Nevertheless, comparisons between UKWCS diary data and questionnaire data in section 5.2 of chapter 5 indicated that taking account of multivitamins and antioxidants, as well single supplement vitamin C, may pick up a large proportion of supplements containing vitamin C.

As seen in section 5.2.3 of chapter 5, the vitamin C responses on the phase 2 questionnaire had reasonable agreement with the recordings of supplements containing 500mg/d or more vitamin C from diaries. Therefore, it is possible that high dose users may be defined more accurately by combining recordings from both the phase 2 questionnaire and diaries. However, since some reporting was not consistent,

this would result in fewer high dose users in the analysis and therefore would lower the power to detect significant associations.

Previous chapters have provided evidence that women in the low intake category (less than or equal to 60mg/d of vitamin C) were likely to be taking multivitamins which contained a wide range of other micronutrients (as reported in section 4.5.3.2 and section 5.2.3.1, though the Kappa agreement was only weak to moderate K=0.41(0.40-42)). Conversely, women in the reference group did not take supplements containing any vitamin C and therefore were unlikely to be frequently taking multivitamins, most of which contain vitamin C. Furthermore, it was also observed from the questionnaire responses that 22% of women in the low intake category took four to ten types of supplements (though women in the highest vitamin C intake category were twice as likely to do this). Therefore, one or more, or an interaction of these micronutrients in multivitamins or from the variety of supplements taken, may have had an effect on breast cancer risk for premenopausal women rather than the low dose of vitamin C per se. For instance iron and vitamin C together may exert a pro-oxidant effect via the Fenton reaction,<sup>76</sup> or ingredients other than vitamin C may have an effect. A clearer way to assess the effects of vitamin C intake per se without confounding by other supplement ingredients would be by analysing the dietary vitamin C intake of nonsupplement users (as reported in chapter 9) or by implementing an RCT using a vitamin C supplement verse a placebo where the consumption of other supplements would be randomised between intervention groups.

A higher proportion of pre-menopausal compared to post-menopausal women took multivitamin supplements in the UKWCS (as seen in Table 43). Similarly, the 2008 UK survey found multivitamins were more likely to be used by 16-54 year old supplement users than those aged 65 plus.<sup>26</sup> The UKWCS adjusted analysis of multivitamin use recorded by questionnaire at phase 2 showed a moderate non-significant increased risk of breast cancer for pre-menopausal women. This strengthened and became close to significance when women with a family history of breast cancer were excluded. One explanation for the association with multivitamins being non-significant and weaker than the low dose vitamin C results may be because the former analysis did not exclude higher doses of vitamin C, whereas the latter did. This may indicate that taking multivitamins without also taking medium to high doses of vitamin C may increase breast cancer risk. A possible biological explanation is, perhaps, at low concentrations in the presence of certain nutrients vitamin C is a pro-oxidant but at high concentrations it is an antioxidant. However, the multivitamin use results did not alter substantially when women who recorded daily vitamin C supplement use on the phase 2 questionnaires were excluded, who were likely to be consuming high doses. Further

analyses could be undertaken to explore the risks for women specifically reporting multivitamin use in the diaries.

Results of a recent Swedish study show that multivitamin use was significantly associated with a 22% increased risk of breast cancer for all women using them for over 3 years.<sup>261</sup> The authors suggest that folic acid in the multivitamins may have promoted cancer, since there is no mandatory folic acid fortification of food in Sweden. Conversely, the US does fortify food with folic acid and several US studies, including the Nurses Health Study, have not found an increased risk of breast cancer with multivitamin use.<sup>247 371</sup> A recent meta-analysis found no evidence of a significant association between multivitamin use and breast cancer risk.<sup>260</sup> Since fortification is not mandatory in the UK, folic acid in multivitamins could explain the increased risk for UKWCS pre-menopausal women who took supplements containing low vitamin C doses, though the multivitamin results do not strongly support this. Despite many multivitamins containing folic acid, very few women reported on the UKWCS questionnaire that they took folic acid supplements; therefore the questionnaire data would be underpowered to find an association with breast cancer incidence. However, an analysis of associations between breast cancer risk and folic acid use by dose ascertained from the phase 2 diary supplement database may be able to illuminate this hypothesis further.

To conclude, this UK study found little evidence that supplementation with vitamin C per se was beneficial or detrimental in relation to breast cancer incidence in UK women, even at high doses. However the results indicate that frequent use of supplements containing less than or equal to 60mg/d vitamin C, or the frequent use of multivitamins, may increase breast cancer risk in pre-menopausal women. Although these results may be spurious, it would be prudent to conduct further research on supplement use in the UK, particularly relating to multivitamins, to identify whether some micronutrients or combination of micronutrients might produce an increase in breast cancer risk in pre-menopausal users.

## **CHAPTER 9**

# 9 Is breast cancer incidence associated with dietary vitamin C or fruit and vegetable intake? Does supplement use modify these relationships? A baseline analysis of the UKWCS

## 9.1 Summary

*Objectives:* To determine whether dietary vitamin C or fruit and vegetable intake at baseline in the UKWCS is associated with breast cancer risk. Since the use of supplements may have obscured associations in previous studies, sub-analyses by general supplement use were also undertaken.

*Method:* This time-to-event analysis of 33,520 middle-aged women at baseline in the UK Women's Cohort Study (UKWCS) calculated Hazard Ratios (HRs) over a median follow-up period of 11.2 years for 989 registered incident breast cancers in relation to fruit and vegetable and dietary vitamin C intake recorded by Food Frequency Questionnaire (FFQ) and cross-check questions. Sub-analyses by supplement use and menopausal status were produced. Additionally, non-linear relationships between intakes of 60mg-350mg/d vitamin C were modelled using restricted cubic splines.

*Results:* All hazard ratios were non-significant relating to intake and breast cancer risk. Furthermore, in tests for interactions, relationships were not significantly different between women who took any type of supplement and non-users for total fruit and vegetable servings (p =0.6) and for dietary vitamin C intake (p=0.2). Additionally, no significant interactions were found relating to menopausal status or family history of breast cancer. Although breast cancer risks appeared to increase with increasing dietary vitamin C intake for non-supplement users, and specifically post-menopausal women (HR per 60mg/d vitamin C HR=1.05; 95% CI: 0.94, 1.19), risks were non-significant. In the tabled analysis by fifths of intake, a U-shaped relationship was apparent for these women when vitamin C was expressed as nutrient density, but this was less clear on the graphed models.

*Conclusion:* There was no evidence of significant associations between breast cancer incidence and intake of fruit and vegetables or dietary vitamin C derived from FFQ in

the UKWCS. There was also no statistically significant evidence that supplement use modified the relationship between breast cancer risk and intake.

## 9.2 Introduction

Fruit and vegetables are the main dietary sources of vitamin C,<sup>67</sup> whose antioxidant properties are believed to protect against DNA damage and cancer development by decreasing reactive oxygen species (ROS) that may cause DNA damage.<sup>74</sup> Additionally, many other bioactive compounds such as phytochemicals in fruit and vegetables are thought to protect against cancer.<sup>7</sup> Although it is widely believed that high intakes of fruit and vegetables reduce cancer risk, there is no conclusive evidence for this or for breast cancer specifically.<sup>7 11 12</sup> The 2007 WCRF report stated there was probable rather than convincing evidence that they decreased the risk of many cancers such as mouth, oesophagus, stomach and colorectal cancer, but not breast cancer.<sup>7</sup> Furthermore, a recent EPIC analysis found only a very small inverse association between intake of total fruits and vegetables and total cancer risk.<sup>212</sup> Initial findings from retrospective case-control studies reported a reduction in breast cancer risk with increasing intake by as much as 25%.<sup>8 213</sup> However, results of a subsequent pooled analysis of cohort studies showed that fruit and vegetable consumption during adulthood was not significantly associated with breast cancer risk.<sup>12</sup>

Similarly, as stated in section 3.6.4.1 of the literature review, the meta-analyses in the WCRF 2007 report of 14 case-control studies of dietary vitamin C intake provides evidence of a statistically significant protective association for breast cancer (RR=0.88, 95% CI: 0.84-0.92, per 100mg/day); analyses comparing high versus low intake categories, plus pre-menopausal and post-menopausal analyses support this.<sup>7</sup> However, overall cohort results were inconclusive.<sup>239 240 242 243 245-247 264-267</sup> In the 2007 WCRF report no significant associations were produced for dietary vitamin C in the post-menopausal women meta-analyses of three cohort studies (HR=1.15 per 100mg/d, 95% CI: 0.92-1.43), <sup>7 245 246 265</sup> or in the high versus low intake analysis of two additional studies.<sup>242</sup> <sup>247</sup> Differences in findings between study types may be due to selection or recall bias in the retrospective case-control studies. Most recently no significant evidence was found between dietary intake of vitamin C and breast cancer risk for either pre- or post-menopausal women in the EPIC study, involving 7,502 breast cancer cases,<sup>244</sup> or for post-menopausal women in a Danish cohort analysed by Roswall et al. (2010).<sup>273</sup> A recent analysis of the Women's Health Initiative Observational Study found that dietary vitamin C was not associated with breast cancer risk, although weak positive associations were found for breast cancer for total vitamin C and supplemental vitamin C intake in the highest versus lowest intake analyses.<sup>270</sup>

In contrast to the overall WCRF meta-analysis results, Nissen et al. (2003) in an earlier nested case-control analysis of the Danish 'Diet, Cancer and Health' cohort (418 case and 394 controls) did find a significant doubling in risk per 100mg/d increase in intake for post-menopausal women (RR= 2.06, 95% CI: 1.45-2.91).<sup>245</sup> Unlike other studies mentioned, this dietary analysis was restricted to women who consumed vitamin C through both diet and supplements; women who did not consume vitamin C supplements were excluded.<sup>245</sup> This nested case-control study followed Danish, postmenopausal women for 4.7 years; a shorter period than the other studies. Unlike the other studies in the WCRF meta-analysis, adjustment was made for intake of the other antioxidant vitamins A and E.

As seen in previous chapters supplement users in general have healthier behaviours than non-users,<sup>27</sup> therefore excluding all types of supplement users from analyses will reduce potential confounding and provide a clearer picture of associations between dietary vitamin C intake and risk. There is a lack of evidence relating to nonsupplement users, other than non-significant findings reported above for the recent EPIC study, which did exclude all supplement users in their dietary vitamin C analysis.<sup>244</sup> When vitamin C supplement users were excluded in the Danish nested case-control study reported by Nissen et al. (2003), this reduced the association substantially from a doubling in risk to RR=1.54 (95% CI: 0.80-2.96 per 100mg/day), which also became non-significant.<sup>245</sup> No analysis, however, was produced excluding all supplement users. The more recent analysis of the Danish study by Roswall et al. (2010) also did not exclude supplement users.<sup>273</sup> An analysis of the Iowa Women's Health Study excluded women who took antioxidant supplements A,C and E and this increased the point estimate in the high verse low calculation from an non-significant decrease (HR=0.88 95% CI: 0.70-1.11) to a non-significant increase in risk (HR=1.06 95% CI: 0.77-1.47).

Furthermore, it is possible that the effect of dietary intake on cancer risk may be different for women genetically predisposed to developing breast cancer. In a test for interaction Zhang et al.  $(1999)^{247}$  found that family history of breast cancer significantly modified the relationship between dietary vitamin C and breast cancer risk (p<0.05). Moreover pre-menopausal women with a family history of breast cancer whose dietary vitamin C intake was in the highest fifth were found to have a significantly reduced risk of breast cancer compared to those in the lowest fifth; RR=0.37 (95% CI: 0.17, 0.80). However, no association was apparent when vitamin C from supplements was included; RR=0.97 (95% CI: 0.52, 1.79).

The current analyses examined the relationship between breast cancer incidence and fruit and vegetable intake, and dietary vitamin C intake at baseline in the UKWCS, which incorporates a wide range of intakes. Based on the hypothesis that a U-shaped relationship may exist, non-linear relationships between intake and breast cancer risk were explored, in addition to linear. Although the amount of vitamin C in supplements taken by users was unknown at baseline, women who reported supplement use of any type were excluded in sub-analyses to provide a clearer picture of the relationship between dietary vitamin C and breast cancer risk. The effects of family history and menopausal status were also examined.

## 9.3 Method

### 9.3.1 Study population

General information about the recruitment of the 35,372 women who completed the baseline questionnaire is provided in section 4.1.1.1 of this thesis. All analyses excluded 1735 women with any prevalent malignant cancers recorded in the cancer registry before the baseline questionnaire date and excluded a further 33 women who were diagnosed with breast cancer within 6 months after the questionnaire date. The analyses also excluded an additional 83 women with extremely low or high total energy intake (more than 6000kcal and less than 500kcal). During the median follow up period of 11.2 years, 989 women developed incident breast cancers; these were registered with cancer registries by the censor date 01/01/08.

Of the 33,520 women included in the analyses of dietary vitamin C and fruit and vegetable intake, 33,137 (99%) provided information on whether or not they took dietary supplements. Sixty-two percent (20,460 women) in the cohort took supplements. Ninety-four percent (31,559 women) also provided information on whether or not they had a family history of cancer, 8% (2,406) of whom had a family history of breast cancer and 39% (12,350) had a family history of any type of cancer.

#### 9.3.2 Exposure measurement

#### 9.3.2.1 Fruit and vegetable intake

Fruit and vegetable intake had been measured in grams and also by two methods of calculating total servings: from answers to the 217 item FFQ; also from answers to cross-check questions 7 and 11 of the baseline questionnaire. Fruit and vegetable intake in grams was derived from total servings from the FFQ.

The cross-check questions were:

Q7 How many servings of vegetables or vegetable containing dishes (excluding potatoes) do you usually eat each week?

Q11 How many servings of fruit or fruit containing dishes do you usually eat each week?

Five roughly equal groups were created from the cross-check question data for servings per day. Total servings derived from the answers to the FFQ, and also total grams of fruit and vegetable consumed per day derived from the FFQ, were divided into fifths. The range of fruit and vegetable servings in each group and the mean servings for each group are shown in Table 48. As reported in section 5.3 of the evaluation chapter, the agreement between the two methods of calculating servings was not good, and reporting misjudgements by participants are likely to have occurred with both methods. However, because there were substantial correlations between vitamin C derived from FFQ and fruit and vegetable servings or grams calculated from FFQ, analyses using FFQ derived fruit and vegetable servings and grams were not undertaken apart from for comparisons in the initial analysis in Table 48.

#### 9.3.2.2 Vitamin C intake

Dietary vitamin C consumed per day was derived from FFQs. As explained in section 4.5.1 in the methods chapter, the FFQ data had been entered into the DANTE program which used micronutrient content of foods as listed in The Royal Society of Chemistry Food tables, version 5 and supplements to estimate vitamin C consumed.<sup>287</sup>

Vitamin C nutrient density per mega joule was calculated for each participant as vitamin C/ MJ. The reason for undertaking additional analyses with nutrient densities is discussed in section 4.7.5.3 of the methods chapter.

Five equal groups were created from the dietary vitamin C as absolute values and also from vitamin C expressed as nutrient densities. The mean intake for each fifth of dietary vitamin C intake is shown in Table 48.

#### 9.3.2.3 Supplement use

Supplement users at baseline had been determined from yes/no answers to the question:

Q29 Do you take any vitamins, minerals, fish oils or other food supplements? Women who did not answer this question but recorded information about the type of supplements taken on the questionnaire were also classed as supplement users.

#### 9.3.3 Statistical analyses

Women's characteristics by category of consumption were described using means and percentages, and displayed in separate tables for absolute vitamin C intake and for fruit and vegetables intake recorded by the cross-check questions. Significant differences in characteristics in relation to intake were established from p-values using regression tests for trend for continuous variables and chi squared tests for trend for categorical variables.

Cox proportional hazard ratios were used to model the associations between breast cancer incidence and fifths of vitamin C and also fifths of fruit and vegetable intake. To test for linear trends, continuous intake variables were used per increment of one serving for fruit and vegetable intake and approximately one standard deviation of mean vitamin C intake (e.g. 60mg/d for absolute vitamin C intake and 8mg/MJ/day for vitamin C intake density). Tables of hazard ratios were adjusted for age, BMI (under; normal; overweight), parity, estimated cumulative breast feeding (weeks), age at menarche, hours exercise sweating per week, contraceptive pill use (never, past, current), HRT use (ever, past, current), low alcohol use, socio-economic class (professional and managerial; intermediate; routine and manual), smoking status (never, past current), total energy intake, and also supplement use (yes, no), and menopausal status (postmenopausal; premenopausal) where appropriate. Reasons for selecting confounders are discussed in section 4.7.5 of this thesis. In sensitivity analyses further adjustment was made for family history of cancer in first degree relatives (any, breast). As detailed in section 4.7.6.3 women were asked whether their mother, father, brothers or sisters ever suffered from cancer, and were asked to provide details. Family history may be a confounder since women in the UKWCS who were aware that they may be at increased risk of breast cancer may have consumed more fruit and vegetables and supplements in an effort to reduce the risk.

Both unadjusted and adjusted likelihood ratio test for overall interactions were used to test whether breast cancer risk differed between supplement groups, menopausal status or family history of breast cancer. Sub-group time-to-event analyses were produced by supplement use and for menopausal status.

Non-linear relationships between intake and breast cancer risk were modelled using restricted cubic splines consisting of four cubic polynomial segments ( $\alpha + \beta_1 x + \beta_2 x^2 + \beta_3 x^3$ ) separated by five knots. This produced a smooth curve for each model: the models show straight lines before the first and after the last knot and are continuous at knot boundaries.<sup>365</sup> Harrell (2001) recommended using five knots which are placed at

the following percentiles of the exposure variable: 5, 27.5, 50, 72.5, 95.<sup>372</sup> These models were adjusted using the means of the continuous covariates: age, BMI, total calories parity, estimated cumulative breast feeding (weeks), age at menarche, hours exercise sweating per week. This produced estimated breast cancer risks over a range of vitamin C intakes for a woman with these average UKWCS characteristics. No adjustments were made for categorical variables, since each of the categories would need to be adjusted for in separate analyses, thereby reducing the power of the analyses. In the model of total women (minus exclusions and outliers) knots occurred at 76.7, 120.3, 155.2, 197.7 and 288.5mg/d vitamin C. Separate knots and covariate means were calculated for menopausal and supplement status sub-groups.

Women with a family history of breast cancer were excluded from the restricted cubic spline models. The upper threshold for excluding outliers, i.e. 350mg/d of vitamin C, was determined as 1.5 times the inter-quartile range and comprised 3.6% of the total women.

25% percentile = 113.6mg/d vitamin C 75% percentile = 209.7mg/d Inter-quartile range = 96.1mg/d 1.5 \* range = 144.2mg/d Upper threshold = 209.7 + 144.2 = 353.9mg/d

Women with intakes below the RDA of 60mg/d were also excluded since they formed less than 3% of the total and appeared to disproportionately influence the extremities of the graphs. Seventy-one other data points were excluded using Mahalanobis distance calculations between vitamin C intake and the total fruit and vegetable intake from the cross-check question. This multivariate method of determining outliers was developed by Hadi and applied in Stata v10<sup>316,317</sup> (see section 4.7.6.1 of the methods chapter). The predicted hazard ratios shown for the graphs are relative to women taking 60mg/d vitamin C.

### 9.4 Results

The mean (sd) values for the lowest intake groups were 0.9 (0.4) and 4.4 (1.2) servings of fruit and vegetables per day recorded by cross-check question and by FFQ respectively, and 19mg per day of dietary vitamin C. Mean values for the highest intake groups were 5.1 (1.3) and 19.7 (6.6) servings per day for the cross-check question and FFQ respectively and 301mg (86) per day dietary vitamin C.
From the tests for trends across intakes, as observed in Table 46 and Table 47, there were significant relationships between the majority of the characteristics listed and increasing intakes of fruit and vegetables (as per the cross-check question) and also increasing intakes of dietary vitamin C. For instance, with increasing intake, there were obvious trends in relation to longer breast feeding, exercising vigorously for longer, higher energy intake, and also higher dietary vitamin A and E intake. Similarly, there were obvious trends towards having no children, taking supplements, and being a current smoker in the higher intake groups. Paradoxically, although lower meat consumption occurred with increased fruit and vegetables intake in Table 46, whereas, an increase in meat consumption occurred with increased for total alcohol (ethanol) intake, significant trends were seen in relation to drinking less than once a week; women who had low fruit and vegetable or vitamin C intake were also more likely to have a very low alcohol intake.

In the breast cancer time-to-event analyses in Table 48 there was no significant evidence of differences between the lowest intake group and other intake groups for any of the methods of recording intake before or after adjustment of potentially confounding covariates, or after further adjustment for family history of cancer. Furthermore, no linear trends in risks were apparent in this table, which was prior to testing for modification effects of supplements. Instead, the highest but weak and non-significant risks for breast cancer were found in the middle intake group of fruit and vegetables and also for absolute vitamin C values (HR=1.16; 95% CI: 0.93, 1.46; mean intake 156mg/d vitamin C). The lowest risks were found for the highest fruit and vegetable intake groups derived from the FFQ analysed by servings (HR=0.85; 95% CI: 0.66, 1.09) and by grams consumed (HR=0.89; 95% CI: 0.69, 1.15), though these were also non-significant. The restricted cubic spline model (Figure 23) also shows no clear association between breast cancer risk and dietary vitamin C, with little evidence of any significant non-linear relationships, as seen from the wide confidence intervals in the bottom graph.

		Total fruit an	d vegetable co	nsumption (as	per Q7 & Q11)	grouped	
	1	2	<u></u> 3	4	5	Total	P value <sup>a</sup>
Mean (sd) servings/day (sd)	0.9 (0.4)	1.9 (0.2)	2.5 (0.2)	3.3 (0.3)	5.1 (1.3)	2.8 (1.6)	
	N=6638	N=6235	N=6298	N=6675	N=7674	N=33520	
	(20%)	(19%)	(19%)	(20%)	(23%)	(100%)	
Age (years), mean (sd)	52.3 (9.5)	52.3 (9.9)	51.0 (9.1)	52.1 (9.3)	52.3 (9.0)	52.0 (9.3)	0.4
BMI (kg/m <sup>2</sup> ), mean(sd)	24.8 (4.7)	24.5 (4.3)	24.5 (4.2)	24.3 (4.2)	24.2 (4.4)	24.5 (4.4)	<0.001
Weight change since aged 20, mean(sd)	9.0 (10.4)	8.6 (9.8)	8.2 (9.7)	7.5 (9.7)	6.9 (10.2)	8.0 (10.0)	<0.001
Height (cm), mean (sd)	163.1 (7.1)	163.7 (6.9)	163.7 (6.7)	164.0 (6.7)	164.0 (6.7)	163.7 (6.9)	<0.001
Age at menarche, mean (sd)	12.93 (1.7)	12.86 (1.6)	12.81 (1.6)	12.76 (1.6)	12.72 (1.6)	12.8(1.6)	<0.001
Age at first birth, mean (sd)	25.2 (4.7)	25.9 (4.8)	25.9 (4.8)	26.1 (4.7)	26.0 (4.6)	25.8 (4.7)	<0.001
Parity, mean (sd)	1.9 (1.3)	1.9 (1.3)	1.9 (1.3)	1.9 (1.3)	1.9 (1.3)	1.9 (1.3)	0.006
Est. total breast feeding (wks), mean(sd)	19.0 (1.9)	21.2 (32.8)	24.3 (35.3)	25.1 (35.3)	27.6(38.4)	23.6 (35.0)	<0.001
Physical activity (min), mean (sd)	13 (28)	14 (28)	15 (29)	16 (30)	18 (30)	15 (29)	<0.001
Total meat (grams), mean (sd)	72 (63)	69 (63)	68 (62)	63 (61)	55 (59)	64 (62)	<0.001
Total fruit (grams), mean (sd)	224 (215)	261 (19)	273 (186)	340 (210)	444 (263)	314 (232)	<0.001
Total vegetables (grams), mean (sd)	250 (174)	285 (165)	297 (155)	333 (169)	396 (199)	315 (181)	<0.001
Total ethanol intake (grams), mean (sd)	8.2 (11.5)	8.4 (10.4)	9.6 (11.1)	8.8 (10.1)	9.5 (9.8)	8.7 (10.6)	0.1
Total energy intake (Kcal), mean (sd)	2212 (740)	2295 (726)	2338 (685)	2391 (697)	2473 (703)	2346 (716)	<0.001
Dietary vitamin A intake (µg), mean (sd)	1111 (622)	1187 (575)	1209 (532)	1288 (598)	1368 (613)	1238 (598)	<0.001
Dietary vitamin E intake (mg), mean (sd)	8.6 (4.1)	9.2 (4.1)	9.6 (4.1)	10.1 (4.2)	10.8 (4.4)	9.7 (4.3)	<0.001
Dietary vitamin C intake (mg), mean (sd)	139 (85)	155 (78)	160 (72)	181 (78)	213 (93)	171 (86)	<0.001
Vitamin C from fruit (mg), mean (sd)	43 (51)	51 (46)	54 (48)	66 (52)	88 (68)	62 (57)	<0.001
Vitamin C from veg (mg), mean (sd)	53 (42)	60 (40)	62 (37)	70 (40)	83 (49)	66 (43)	<0.001
Professional or managerial (%)	31	35	41	42	44	39	<0.001
Educated to 'A'-level or above (%)	38	46	55	58	60	52	<0.001
No children (%)	20	20	20	20	22	20	<0.001
Ever used contraceptive pill (%)	66	68	72	68	67	68	0.5
Ever used HRT (%)	29	28	27	27	28	28	0.07
Current smoker (%)	18	13	11	9	6	11	<0.001
Drinks alcohol less than once a week(%)	37	35	30	32	33	33	<0.001
Takes any supplements (%)	58	59	60	63	68	62	<0.001
Family history of breast cancer (%)	6.8	7.2	8.1	8.1	7.9	7.6	0.003
Family history of any cancer (%) <sup>b</sup>	38.4	39.7	38.7	38.9	40.0	39.2	0.2

Table 46 Characteristics of women at baseline in the UKWCS b	by category of fruit and vege	etable servings consumed (a	as per Q7 & Q11)

Family history of any cancer (%)<sup>b</sup> <sup>a</sup>p= test for trend, <sup>b</sup>family history in first degree relatives

<b>Table 47</b> Characteristics of women at baseline in the UKWCS by cate	gory o	of dietary	y vitamin (	C consumption
---	--------	------------	-------------	---------------

	Total dietary vitamin C consumption from FFQ by fifths						
	Lowest	2nd Fifth	3rd Fifth	4th Fifth	Highest	Total	P value <sup>a</sup>
Mean vitamin C mg/d (sd)	79 (18)	122 (10)	156 (10)	197 (15)	301 (87)	171 (86)	
	N=6704	N=6704	N=6704	N=6704	N=6704	N=33520	
Age (years), mean (sd)	51.2 (9.3)	51.6 (9.3)	52.1 (9.3)	52.6 (9.2)	52.7 (9.2)	52.0 (9.3)	<0.001
BMI (kg/m <sup>2</sup> ), mean (sd)	24.6 (4.6)	24.4 (4.3)	24.4 (4.2)	24.4 (4.2)	24.4 (4.5)	24.5 (4.4)	0.2
Weight change since aged 20 mean,(sd)	8.4 (10.5)	8.0 (9.9)	7.8 (9.6)	7.9 (9.7)	7.8 (10.4)	8.0 (10.0)	0.05
Height (cm), mean (sd)	163.3 (6.9)	163.7 (6.9)	163.9 (7.1)	164.0 (6.8)	163.9 (6.6)	163.7 (6.9)	<0.001
Age at menarche, mean (sd)	12.85 (1.6)	12.86 (1.6)	12.79 (1.6)	12.79 (1.6)	12.78 (1.6)	12.8(1.6)	0.02
Age at first birth, mean (sd)	25.6 (4.8)	26.0 (4.8)	26.0 (4.8)	25.9 (4.6)	25.7 (4.6)	25.8 (4.7)	0.1
Parity, mean (sd)	1.7 (1.3)	1.8 (1.3)	1.9 (1.3)	1.9 (1.3)	2.0 (1.3)	1.9 (1.3)	<0.001
Est. total breast feeding (wks), mean(sd)	18.6 (31.4)	22.6 (33.8)	24.6 (35.3)	25.4 (36.3)	26.7 (37.7)	23.6 (35.0)	<0.001
Physical activity (min), mean (sd)	12 (26)	13 (28)	14 (27)	16 (28)	20 (35)	15 (29)	<0.001
Total meat (grams), mean (sd)	61 (54)	64 (58)	65 (61)	67 (64)	66 (71)	64 (62)	<0.001
Total fruit (grams), mean (sd)	145 (93)	224 (116)	288 (138)	358 (161)	553 (329)	313 (231)	<0.001
Total vegetables (grams), mean (sd)	168 (71)	241 (84)	298 (103)	361 (124)	509 (244)	315 (181)	<0.001
Total ethanol intake (grams), mean (sd)	8.4 (11.3)	8.9 (10.9)	8.9 (10.7)	8.7 (9.8)	8.5 (10.1)	8.7 (10.6)	0.2
Total energy intake (Kcal), mean (sd)	1904 (570)	2157 (580)	2322 (596)	2501 (655)	2849 (796)	2346 (716)	<0.001
Dietary vitamin A intake (µg), mean (sd)	895 (440)	1089 (464)	1222 (509)	1361 (556)	1620 (712)	1237 (598)	<0.001
Dietary vitamin E intake (mg), mean (sd)	7.1 (3.3)	8.6 (3.6)	9.6 (3.7)	10.5 (3.8)	12.7 (4.8)	9.7 (4.3)	<0.001
Vitamin C from fruit (mg), mean (sd)	24 (16)	40 (21)	53 (26)	70 (34)	120 (90)	62 (57)	<0.001
Vitamin C from veg (mg), mean (sd)	31 (14)	48 (19)	62 (23)	78 (29)	112 (60)	66 (43)	<0.001
Professional or managerial (%)	36	39	40	41	39	39	<0.001
Educated to 'A'-level or above (%)	47	53	54	54	53	52	<0.001
No children (%)	24	21	19	19	18	20	<0.001
Ever used contraceptive pill (%)	69	70	68	67	66	68	<0.001
Ever used HRT (%)	26	26	28	29	30	28	<0.001
Current smoker (%)	16	12	10	8	9	11	<0.001
Drinks alcohol less than once a week(%)	38	33	31	30	34	33	<0.001
Takes any supplements (%)	57	58	62	64	68	62	<0.001
Family history of breast cancer (%) <sup>b</sup>	7.3	7.7	7.7	7.8	7.6	7.6	0.5
Family history of any cancer (%) <sup>b</sup>	39.0	38.7	38.3	39.9	39.8	39.2	0.2

<sup>a</sup>p= test for trend. <sup>b</sup>In first degree relatives sd =standard deviation

	Mean (SD)	Cases/	Unadjusted HR	Adjusted HR	Adjusted for	
		non-cases <sup>a</sup>	(95% CI)	(95% CI) <sup>b</sup>	family history	
					(95% CI) <sup>c</sup>	
Fruit & vegetable servin	gs/d, from Q7 & Q	11				
1 (0-1.4)	0.9 (0.4)	188/6450	1	1	1	
2 (1.6-2)	1.9 (0.2)	176/6059	0.98 (0.80, 1.21)	1.09 (0.86, 1.38)	1.08 (0.85, 1.37)	
3 (2.1-2.9)	2.5 (0.2)	191/6107	1.06 (0.87, 1.30)	1.12 (0.89, 1.42)	1.13 (0.88, 1.41)	
4 (3-3.9)	3.3 (0.3)	200/6475	1.04 (0.85, 1.27)	1.05 (0.83, 1.33)	1.04 (0.82, 1.33)	
5 (4 & above)	5.1 (1.3)	234/7440	1.05 (0.87, 1.28)	1.04 (0.83, 1.31)	1.04 (0.82, 1.31)	
Per serving/d			1.01 (0.97, 1.05)	1.00 (0.95, 1.04)	1.00 (0.95, 1.05)	
<i>p</i> <sub>trend</sub>			0.6	0.9	1.0	
Fruit & vegetable servings/d. EEO derived						
1 (0-6 1)	4 4 (1 2)	188/6515	1	1	1	
2 (6 1-8 4)	7 2 (0 7)	202/6499	1 06 (0 87 1 30)	0 99 (0 79 1 24)	0 99 (0 79 1 25)	
3 (8 4-10 7)	9.5 (0.7)	217/6489	1 13 (0 93 1 38)	1 04 (0 83 1 30)	0.99 (0.79, 1.25)	
4 (10 7-14 2)	12 3 (0.9)	208/6498	1.08 (0.88, 1.31)	0.94 (0.75, 1.19)	0.93 (0.74, 1.18)	
5 (14.2 & above)	19.7 (6.6)	174/6530	0.89 (0.72, 1.09)	0.85 (0.66, 1.09)	0.84 (0.65, 1.08)	
Per serving/d	(0.0)		0.97 (0.91, 1.03)	0.97 (0.90, 1.05)	0.96 (0.88, 1.05)	
P <sub>trend</sub>			0.3	0.5	0.4	
	ris co al					
1 lowost fifth		100/0504	4	4	4	
	200y/u (72) 122a/d (20)	100/0524				
2	433g/d (39)	196/6506	1.09 (0.89, 1.33)	0.99 (0.79, 1.24)	0.97 (0.77, 1.23)	
3	566g/d (40)	220/6484	1.21 (0.99, 1.47)	1.16 (0.93, 1.45)	1.12 (0.89, 1.40)	
4 5 bisboot fifth	730g/d (58)	212/0492	1.15 (0.95, 1.41)	1.01(0.80, 1.28)	0.99 (0.78, 1.26)	
	1149g/a (360)	179/6525	0.96(0.78, 1.18)	0.89 (0.69, 1.15)	0.87 (0.67, 1.13)	
Per 80g/a			0.99 (0.97, 1.00)	0.99 (0.88, 1.01)	0.99 (0.97, 1.01)	
$p_{trend}$			0.2	0.2	0.2	

 Table 48 Breast cancer risks by fruit and vegetable intake and dietary vitamin C intake at baseline in the UKWCS

	Mean (SD)	Cases/ non-cases <sup>a</sup>	Unadjusted HR (95% CI)	Adjusted HR (95% CI) <sup>b</sup>	Adjusted for family history (95% CI) <sup>c</sup>
Dietary vitamin C mg	g/d, FFQ derived				
1 lowest fifth	79mg/d (18)	182/6522	1	1	1
2	122mg/d (10)	176/6528	0.96 (0.78, 1.18)	0.96 (0.76,1.21)	0.94 (0.74, 1.20)
3	156mg/d (10)	218/6486	1.19 (0.98, 1.45)	1.16 (0.93, 1.46)	1.15 (0.91, 1.45)
4	197mg/d (15)	219/6485	1.20 (0.98, 1.45)	1.12 (0.88, 1.41)	1.08 (0.85, 1.37)
5 highest fifth	301mg/d (87)	194/6512	1.05 (0.86, 1.28)	1.05 (0.81, 1.35)	1.03 (0.79, 1.33)
Per 60mg/d			1.00 (0.96, 1.05)	1.00 (0.94, 1.06)	0.94 (0.93, 1.05)
$p_{trend}$			1.0	0.9	0.7
Dietary vitamin C de	nsity, FFQ derived				
1 lowest fifth	9.1/d (1.9)	178/6526	1	1	1
2	13.3/d (0.9)	190/6514	1.06 (0.86, 1.30)	1.04 (0.83, 1.31)	1.07 (0.84, 1.35)
3	16.6/d (1.0)	204/6500	1.14 (0.93, 1.39)	1.07 (0.85, 1.34)	1.09 (0.87, 1.38)
4	20.4/d (1.4)	206/6498	1.14 (0.93, 1.39)	1.08 (0.86, 1.35)	1.06 (0.84,1.33)
5 highest fifth	29.7/d (7.2)	211/6493	1.16 (0.95, 1.42)	1.09 (0.87, 1.37)	1.07 (0.85, 1.35)
Per 8mg/MJ/d			1.02 (0.96, 1.09)	1.00 (0.93, 1.08)	0.99 (0.92, 1.07)
<i>P</i> <sub>trend</sub>			0.5	0.9	0.9

Table continued: Breast cancer risks by fruit and vegetable intake and dietary vitamin C intake at baseline in the UKWCS

<sup>a</sup>Numbers of cases for the unadjusted analyses

<sup>b</sup>Adjusted for age, BMI, socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hours exercise sweating per week, contraceptive pill use, HRT use, low alcohol use, smoking status, supplement use, total calories (other than nutrient density).

<sup>c</sup>Adjusted for the above, plus family history of cancer in first degree relatives (any cancer, breast cancer).

There were no significant modifying effects of menopausal status, supplement use, or family history of breast cancer on the relationship between dietary intake and breast cancer risk (Table 49). Although the interactions were non-significant in the test between supplement use and fruit and vegetable intake or dietary vitamin C intake (p=0.6, 0.2, 0.3), breast cancer time-to-event analyses were stratified by supplement use status (Table 51). Similarly, stratified analyses were tabled by menopausal status (Table 50) and by supplement use for post-menopausal (Table 52) and premenopausal women (Table 53).

Although different patterns of risk were seen in sub-analyses, there was little evidence of non-linear relationships, and indeed no evidence of any significant associations between breast cancer incidence and dietary intake after stratifying by supplement use or by menopausal status. Additionally, there were no differences in patterns of hazard ratios in sensitivity analyses after adjusting for family history of breast cancer and any cancer (not shown). Patterns of risk appeared different for pre- and post-menopausal women as seen in Table 50, but these differences were not statistically significant. In the restricted cubic-spline models (Figure 24), compared to risks at 60mg/d vitamin C, deviations of risk from unity for intakes between 60mg/d to 350mg/d were weak and also non-significant; confidence intervals, which were not shown, were wide. If anything, the spline model for pre-menopausal women indicated risks increased slightly, but non-significantly, with increasing vitamin C intake, though the results in Table 50 did not reflect this.

For non-supplement users, there appeared to be non-significant trends towards higher risks with increased dietary vitamin C intake, with moderate but non-significant increases in risk in the high verse low risk comparison (HR=1.38; 95% CI: 0.92, 2.06; Table 51), and similarly for post-menopausal non-supplement users (HR=1.52; 95% CI: 0.89, 2.58; Table 52). Likewise, in the predicted spline model (Figure 25) breast cancer risk tended to increase in non-supplement users with increasing dietary vitamin C intake, with the predicted risk at 350mg/d roughly 2.3 compared to 60mg/d intake, but confidence intervals were wide. Similarly, a predicted hazard ratios of 3 for postmenopausal non-supplement users was seen at this intake in spline models (not shown). The results, however, were not statistically significant since the upper confidence limits were very wide.

The hazard ratios for the nutrient density analyses for non-supplement users in Table 51 and Table 52, however, did indicate a U-shaped relationship, with women in the 2<sup>nd</sup> group having the lowest, though non-significant risks (HR=0.68; 95% CI: 0.40, 1.14; Table 52 of post-menopausal women). However, the restricted cubic spline model

showed that the relationship was reasonably flat (Figure 27), and non-significant (confidence intervals not shown).

For supplement users the relationships were largely flat (as observed in Table 51 to Table 52). This was also apparent in the spline model for total supplement users (Figure 26), and also separately for post-menopausal and pre-menopausal women (not shown). Any deviations occurring in the middle intake groups in the tables and graphs were weak and also non-significant as confidence intervals were wide. The shape of the graphs did not change when women with the top and bottom 5% or 10% of intakes were excluded (not shown).

**Table 49** Assessment of modification by menopausal status, supplement use and family history of breast cancer on fruit and vegetable and dietary vitamin C intake and associations with breast cancer risk in the UKWCS

Tests for modification by	P values	P values
	Unadjusted	Adjusted
1) Menopausal status		
Fruit & veg servings, Q7 & Q11 * Menopausal status	0.6	0.8
Dietary vitamin C mg/d * Menopausal status	0.7	0.6
Dietary vitamin C (density) * Menopausal status	0.8	0.8
2) Supplement use status		
Fruit & veg servings, Q7 & Q11 * supplement use	0.7	0.6
Dietary vitamin C mg/d * supplement use	0.1	0.2
Dietary vitamin C (density) * supplement use	0.3	0.3
Supplement use status: pre-menopausal women		
Fruit & veg servings, Q7 & Q11 * supplement use	0.6	0.6
Dietary vitamin C mg/d * supplement use	0.7	0.9
Dietary vitamin C (density) * supplement use	0.8	0.8
Supplement use status: post-menopausal women		
Fruit & veg servings, Q7 & Q11 * supplement use	0.4	0.9
Dietary vitamin C mg/d * supplement use	0.1	0.2
Dietary vitamin C (density) * supplement use	0.5	0.3
3) Family history of breast cancer status		
Fruit & veg servings, Q7 & Q11 * family history	0.9	0.9
Dietary vitamin C mg/d * family history	0.9	1.0
Dietary vitamin C (density) * family history	0.9	1.0
Family history of breast cancer: pre-menopausal women		
Fruit & veg servings, Q7 & Q11 * family history	0.8	0.5
Dietary vitamin C mg/d * family history	0.6	0.6
Dietary vitamin C (density) * family history	0.9	0.5
Family history of breast cancer: post-menopausal women		
Fruit & veg servings, Q7 & Q11 * family history	0.9	0.7
Dietary vitamin C mg/d * family history	1.0	0.9
Dietary vitamin C (density) * family history	0.7	0.5

<sup>a</sup>Adjusted for age, BMI, socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hrs exercise sweating per week, contraceptive pill use, HRT use, low alcohol use, smoking status, supplement use, total calories (other than nutrient density).

rotal women split by Fre-	Pre-menopausal		Post-menopausal		
cases/		cases/			
non-	HR (95% CI) <sup>b</sup>	non-	HR (95% CI) <sup>b</sup>		
cases <sup>a</sup>		cases <sup>a</sup>			
Fruit & vegetable servings/day, Q7	7 & Q11				
1 lowest 57/2376	1	77/2368	1		
2 70/2362	1.19 (0.84, 1.69)	79/2380	1.00 (0.73, 1.37)		
3 76/2753	1.12 (0.79, 1.58)	83/2274	1.10 (0.80, 1.50)		
4 71/2667	1.04 (0.74, 1.49)	89/2652	1.03 (0.76, 1.41)		
5 highest 69/2937	0.92 (0.64, 1.32)	117/3252	1.10 (0.82, 1.48)		
Per serving/d	0.95 (0.89, 1.03)		1.02 (0.96, 1.08)		
<i>P</i> <sub>trend</sub>	0.2		0.5		
	_				
Dietary vitamin C mg/d, FFQ deriv	ed				
1 lowest fifth 70/2832	1	74/2279	1		
2 74/2797	1.04 (0.75, 1.45)	70/2484	0.88 (0.63, 1.22)		
3 78/2625	1.13 (0.81, 1.58)	97/2591	1.17 (0.85, 1.59)		
4 62/2459	0.96 (0.67, 1.38)	109/2797	1.21 (0.89, 1.65)		
5 highest fifth 59/2382	0.94 (0.64, 1.38)	95/2775	1.10 (0.79, 1.54)		
Per 60mg/d	0.96 (0.87, 1.05)		1.01 (0.94, 1.09)		
P <sub>trend</sub>	0.4		0.7		
Distancy itamin C density FEO de	rived				
1 Jowest fifth 71/2068	nveu 4	70/0040	4		
1 IOWEST IIITI 7 1/2900		12/2242			
2 70/2700	1.10(0.79, 1.52)	01/2010	0.96(0.71, 1.33)		
3 72/2020 4 60/2406	1.07 (0.77, 1.49)	05/2001	1.04 (0.76, 1.42)		
4 09/2490	1.07 (0.70, 1.49)	90/2700	1.03(0.77, 1.44)		
5 highest hith 55/2250	0.30 (0.00, 1.30)	100/2009	1.10 (0.00, 1.00)		
Per 8ma/M.I/d	0 03 (0 83 1 05)		1 04 (0 05 1 15)		
	0.33 (0.03, 1.03)		(0.33, 1.13) $\cap A$		
<b>₽</b> trena	0.0		0.4		

**Table 50** Breast cancer risks by fruit and vegetable intake and dietary vitamin C intake at baseline in the UKWCS by menopausal status

<sup>b</sup>Adjusted for age, BMI, socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hours exercise sweating per week, contraceptive pill use, HRT use, low alcohol use, smoking status, supplement use, total calories (other than nutrient density).

Total women split by	Non-users		Supplement users		
	cases/		cases/		
	non-	HR (95% CI) <sup>b</sup>	non-	HR (95% CI) <sup>b</sup>	
	cases <sup>a</sup>		cases <sup>a</sup>		
Fruit & vegetable serv	/ings/day, Q	7 & Q11			
1 lowest	55/2026	1	79/2718	1	
2	65/1956	1.20 (0.84,1.72)	84/2786	1.01 (0.74, 1.37)	
3	56/2029	1.04 (0.71,1.51)	103/2998	1.16 (0.86, 1.55)	
4	64/1977	1.20 (0.83, 1.72)	96/3342	0.96 (0.71, 1.30)	
5 highest	58/2018	1.04 (0.71, 1.52)	128/4171	1.03 (0.79, 1.37)	
Per serving/d		0.97 (0.90, 1.05)		1.01 (0.95, 1.06)	
$p_{trend}$		0.5		0.8	
Dietary vitamin C mg/	d, FFQ deriv	ved			
1 lowest fifth	58/2234	1	86/2877	1	
2	63/2208	1.09 (0.76, 1.57)	81/3073	0.87 (0.64, 1.18)	
3	56/2001	1.07 (0.74, 1.57)	119/3215	1.20 (0.90, 1.59)	
4	61/1883	1.22 (0.84, 1.79)	110/3373	1.04 (0.77, 1.40)	
5 highest fifth	60/1680	1.38 (0.92, 2.06)	94/3477	0.88 (0.64, 1.22)	
-					
Per 60mg/d		1.05 (0.96, 1.16)		0.96 (0.89, 1.04)	
$p_{trend}$		0.3		0.3	
Dietary vitamin C den	sity, FFQ de	erived			
1 lowest fifth	64/2266	1	79/2944	1	
2	52/2164	0.80 (0.55,1.16)	105/3107	1.21 (0.91 1.63)	
3	60/2009	0.98 (0.69, 1.40)	101/3218	1.13 (0.84, 1.51)	
4	62/1848	1.11 (0.78, 1.58)	102/3356	1.07 (0.80, 1.44)	
5 highest fifth	60/1719	1.12 (0.79, 1.60)	103/3390	1.07 (0.80, 1.44)	
Per 8mg/MJ/d		1.07 (0.95, 1.20)		0.96 (0.88, 1.06)	
$p_{trend}$		0.3		0.4	

**Table 51** Breast cancer risks by fruit and vegetable intake and dietary vitamin C intake at baseline in the UKWCS by supplement use

<sup>b</sup>Adjusted for age, BMI, socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hours exercise sweating per week, contraceptive pill use, HRT use, low alcohol use, smoking status, total calories (other than nutrient density).

Post-menopausal	Non-users		Supplement users	
	cases/		cases/	
	non-	HR (95% CI) <sup>b</sup>	non-	HR (95% CI) <sup>b</sup>
	cases <sup>a</sup>		cases <sup>a</sup>	
Fruit & vegetable ser	rvings/day, C	7 & Q11		
1 lowest	34/927	1	43/1441	1
2	33/934	0.98 (0.61,1.59)	46/1446	1.01 (0.66, 1.53)
3	31/869	1.00 (0.61, 1.63)	52/1405	1.16 (0.77, 1.74)
4	37/956	1.13 (0.70,1.82)	52/1696	0.97 (0.65, 1.46)
5 highest	39/1027	1.09 (0.68, 1.76)	78/2225	1.11 (0.76, 1.62)
Per servings/d		0.99 (0.89, 1.10)		1.03 (0.96, 1.11)
$p_{trend}$		0.9		0.4
Dietary vitamin C mg	/d, FFQ deri	ved		
1 lowest fifth	29/951	1	45/1328	1
2	34/976	1.13 (0.68, 1.87)	36/1508	0.70 (0.45, 1.09)
3	32/932	1.11 (0.66, 1.87)	65/1659	1.14 (0.77, 1.69)
4	39/964	1.35 (0.81, 2.23)	70/1833	1.09 (0.74, 1.62)
5 highest fifth	40/890	1.52 (0.89, 2.58)	55/1885	0.87 (0.56, 1.35)
Per 60mg/d		1.05 (0.94, 1.19)		0.99 (0.90, 1.09)
$ ho_{trend}$		0.3		0.8
Dietary vitamin C de	nsity, FFQ de	erived		
1 lowest fifth	33/913	1	39/1329	1
2	25/972	0.68 (0.40, 1.14)	56/1544	1.21 (0.80, 1.82)
3	37/958	1.02 (0.64 1.64)	52/1643	1.04 (0.68, 1.57)
4	39/911	1.14 (0.71, 1.82)	56/1797	1.01 (0.67,1.53)
5 highest fifth	40/959	1.12 (0.70, 1.78)	68/1900	1.16 (0.78, 1.72)
Per 8mg/MJ/d		1.08 (0.93, 1.26)		1.02 (0.90, 1.15)
$p_{trend}$		0.3		0.8

**Table 52** Breast cancer risks by fruit and vegetable intake and dietary vitamin C intake at baseline in the UKWCS by supplement use in post-menopausal women

<sup>b</sup>Adjusted for age, BMI, socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hours exercise sweating per week, contraceptive pill use, HRT use, low alcohol use, smoking status, total calories (other than nutrient density).

Pre-menopausal	Non-users		Supplement users	
	cases/		cases/	
	non-	HR (95% CI) <sup>b</sup>	non-	HR (95% CI) <sup>♭</sup>
	cases <sup>a</sup>		cases <sup>a</sup>	
Fruit & vegetable se	rvings/day, Q	7 & Q11		
1 lowest	21/1099	1	36/1277	1
2	32/1022	1.50 (0.86, 2.60)	38/1340	1.00 (0.63, 1.58)
3	25/1160	1.05 (0.59, 1.89)	51/1593	1.11 (0.73, 1.72)
4	27/1021	1.27 (0.71, 2.27)	44/1646	0.91 (0.59, 1.44)
5 highest	19/991	0.89 (0.49, 1.69)	50/1946	0.90 (0.58, 1.40)
Per servings/d		0.93 (0.82, 1.06)		0.96 (0.88, 1.05)
$p_{trend}$		0.3		0.4
Dietary vitamin C m	g/d, FFQ deri	ved		
1 lowest fifth	29/1283	1	41/1549	1
2	29/1232	1.05 (0.62, 1.77)	45/1565	1.04 (0.68,1.60)
3	24/1069	1.02 (0.59, 1.78)	54/1556	1.22 (0.80, 1.86)
4	22/919	1.05 (0.59, 1.89)	40/1540	0.91 (0.58, 1.45)
5 highest fifth	20/790	1.13 (0.60, 2.13)	39/1592	0.86 (0.53, 1.41)
Per 60mg/d		1.03 (0.88, 1.20)		0.92 (0.82, 1.04)
$p_{trend}$		0.7		0.2
Dietary vitamin C de	ensity, FFQ de	erived		
1 lowest fifth	31/1353	1	40/1615	1
2	27/1192	0.94 (0.56, 1.59)	49/1563	1.20 (0.79, 1.83)
3	23/1051	0.89 (0.52, 1.54)	49/1575	1.19 (0.78, 1.82)
4	23/937	1.00 (0.58, 1.72)	46/1559	1.12 (0.73, 1.72)
5 highest fifth	20/760	1.07 (0.60, 1.89)	35/1490	0.91 (0.58, 1.45)
Per 8mg/MJ/d		1.02 (0.85, 1.25)		0.89 (0.76, 1.03)
$p_{trend}$		0.8		0.1

**Table 53** Breast cancer risks by fruit and vegetable intake and dietary vitamin C intake at baseline in the UKWCS by supplement use in pre-menopausal women

<sup>b</sup>Adjusted for age, BMI, socio-economic status, parity, estimated cumulative breast feeding, age at menarche, hours exercise sweating per week, contraceptive pill use, HRT use, low alcohol use, smoking status, total calories (other than nutrient density).

Figure 23 Adjusted restricted cubic spline models of breast cancer risks by dietary vitamin C intake before sub-grouping (confidence intervals shown in bottom graph)



(confidence intervals shown in bottom graph)

Adjusted covariates set to mean: age = 51.8, total calories = 2341, BMI = 24.4, cumulative breast feeding wks = 24.7, age at menarche = 12.8, vigorous exercise hr/wk = 0.25, parity = 1.86

210

Figure 24 Adjusted restricted cubic spline models of breast cancer risks by vitamin C intake for pre-menopausal and post-menopausal women (without confidence intervals)



Adjusted covariates set to mean: age = 44.6, total calories = 2360, BMI = 23.9, cumulative breast feeding wks = 29.2, age at menarche = 12.8, vigorous exercise hr/wk =0.27, parity = 1.67



Breast cancer risk for post-menopausal women

Adjusted covariates set to mean: age = 58.8, total calories = 2323, BMI = 25.0, cumulative breast feeding wks = 20.2, age at menarche = 12.9, vigorous exercise hr/wk = 0.24, parity = 2.05

Figure 25 Adjusted restricted cubic spline models of breast cancer risks by vitamin C intake for non-supplement users (confidence intervals shown in bottom graph)



(Confidence intervals shown in bottom graph)

Adjusted covariates set to mean: age = 51.2, total calories = 2333, BMI = 24.8, cumulative breast feeding wks = 26.6, age at menarche = 12.8, vigorous exercising hr/wk = 0.24, parity = 1.93

**Figure 26** Adjusted restricted cubic spline models of breast cancer risks by vitamin C intake for supplement users (confidence intervals shown in bottom graph)



(Confidence intervals shown in bottom graph)

Adjusted covariates set to mean: age = 52.2, total calories = 2346, BMI = 24.2, cumulative breast feeding wks = 23.5, age at menarche = 12.8, vigorous exercising hr/wk = 0.26, parity 1.82

**Figure 27** Adjusted restricted cubic spline models of breast cancer risks by nutrient density for non-supplement users and supplement users (without confidence intervals)



Adjusted covariates set to mean: age = 51.2, total calories = 2327, BMI = 24.8, cumulative breast feeding wks = 26.4, age at menarche = 12.8, vigorous exercise hr/wk = 0.24, parity = 1.92



Adjusted covariates set to mean: age = 51.2, total calories = 2363, BMI = 24.2, cumulative breast feeding wks = 23.5, age at menarche = 12.8, vigorous exercise hr/wk = 0.26, parity = 1.82

### 9.5 Discussion

The results of this UKWCS baseline analysis showed that neither total fruit and vegetable intake nor dietary vitamin C intake were significantly associated with breast cancer incidence. This is inline with previous research.<sup>7 11 12 244 270 273</sup> The 2007 WCRF research review and the 2008 WCRF continuous update reported no associations between breast cancer risk and fruit and vegetable or vitamin C intake.<sup>7 169</sup>

Although the amount of vitamin C in supplements taken by users was unknown at baseline in the UKWCS, meaning relationships with total vitamin C intake could not be assessed, the sub-analysis of women who did not take any supplements provided a clearer picture of the relationship between dietary vitamin C intake and breast cancer risk. A very weak positive dose-response relationship was noticeable for UKWCS nonsupplement users in the vitamin C intake models, but this was not statistically significant. Nissen et al. (2003) reported significant evidence of a doubling in breast cancer risk with dietary vitamin C intake per 100mg/d increases in post-menopausal women in a small Danish nested case-control study, however vitamin C supplement users were included in this analysis.<sup>245</sup> When vitamin C supplement users were excluded in this Danish study, the risk was reduced to a moderate association and this also became non-significant.<sup>245</sup> Unlike our analysis other supplement users were not excluded. The recent large European wide EPIC analysis did exclude all supplement users and found no evidence of associations between dietary vitamin C intake and breast cancer risk for all pre- or post-menopausal women.<sup>244</sup> A significant trend of reduced risk with increasing vitamin C intake, however, was apparent for postmenopausal women using exogenous hormones.<sup>244</sup> These relationships should be examined in other cohorts; however, there may be insufficient power to undertake HRT sub-analyses in the UKWCS.

For supplement users in the UKWCS, the non-significant decreased risks evident at higher vitamin C intakes could be attributable to other micronutrients in the supplements which have protective effects; and, as seen in chapter 8, these women were likely to have used a variety of supplements. Alternatively, the results could be explained by residual confounding; reduced risk in supplement users may be due to other health related behaviours associated with supplement use despite attempts to adjust for them. However, since the results were non-significant they were likely to have occurred by chance. Apart from the Nissen et al. study (2003),<sup>245</sup> the current results support the non-significant results for dietary vitamin C analyses reported in the 2007 WCRF review,<sup>7</sup> and results of more recent analyses which did not exclude supplement users<sup>270 273</sup> (as seen in Table 2 and Table 3 in section 3.7.2). Only the lowa

Women's Heath Study has reported dietary vitamin C analyses by supplement users only, i.e. excluding all non-users, which reported no evidence of associations.<sup>242</sup>

There is no simple biological explanation for these non-significant patterns observed in these UKWCS analyses. As explained in chapter 2, it is thought that via its antioxidant properties, vitamin C reduces the risk of cancer initiation by reducing DNA damage from ROS, particularly in individuals with high levels of ROS.<sup>77</sup> A biological explanation for increased risk with increased intake, which was the most pronounced for UKWCS post-menopausal women non-supplement users, is that antioxidant properties of vitamin C may decrease beneficial apoptosis, the ROS induced programmed death of damaged cells.<sup>81</sup> Such a reduction in programmed cell death could lead to the progression of cancer particularly in individuals with age-related damage.<sup>77</sup> Additionally, natural increases in iron stores during menopause may cause vitamin C to act as a pro-oxidant creating highly reactive and damaging hydroxyl radicals via the Fenton reaction.<sup>76</sup> A U-shaped relationship, which could be hypothesised from the above, was observed for vitamin C dietary intake expressed as nutrient density for nonsupplement users. However, since high intakes of this water-soluble vitamin are excreted,<sup>65</sup> risks were expected to flatten out at high intakes, rather than increase or decrease, as seen in some of the graphed models. No previous study of vitamin C and breast cancer risk reported results analysed by nutrient density.

Given the antioxidant hypothesis, at very low vitamin C intakes, decreased risks with increasing intake up to 60mg/d would be expected. However, from FFQ recordings, only 3% of women in this predominately health conscious cohort were calculated to have vitamin C dietary intakes below the EU recommended daily allowance of 60mg/d. Indeed, the mean intake was 79mg/d (sd 18) in the reference category, the lowest fifth of intake. Furthermore, the graphed models excluded these women since they may have disproportionately influenced the extremities of the graphs. Thus, the analyses apparently could not assess the relationship between vitamin C deficiency and breast cancer risk. Nevertheless, as discussed in chapter 5, the evaluation chapter, the use of FFQs tends to overestimate intake, therefore this cohort may contain a larger percentage of women with intakes below the EU RDA than calculated.

Contrary to previous research,<sup>247</sup> the effect of dietary vitamin C on breast cancer risk was not modified by family history of breast cancer in first degree relatives, even after stratifying the interaction test by menopausal status. Therefore the current results do not support the hypothesis that mechanisms of breast carcinogenesis in relation to dietary intake may vary according to genetic predispositions or intergeneration transmission of other factors that may affect risk. Sensitivity analysis using breast cancer family history data from France, Spain, Cambridge and Norway centres in the

EPIC study also found no substantial differences.<sup>244</sup> The current results, however, may have been under powered to find an effect. It was expected that the UKWCS would have more women with a genetic predisposition than the national average since they were recruited via the WCRF. Previous analysis of pre-menopausal women at baseline found that vegetarian women had a higher risk than low meat consumers, which the authors suggested may relate to a family history of breast cancer.<sup>82</sup> Although, initial adjustments for family histories of cancers did not alter results in the current analyses, further sensitivity analyses could be undertaken to exclude women with a family history of breast cancer. However, misclassification of breast cancer family history may have occurred at baseline when participants did not specify the type of cancer their relatives developed.

A limitation of the study is that patterns of risk varied depending upon the tools and methods used to assess intake. Vitamin C intake was calculated from the list of fruit and vegetables ticked on the FFQ using food table nutrient compositions. As seen in section 5.3 of the thesis the agreement between servings of fruit and vegetables measured by the cross-check questions, Q7 & Q11, and that measured by the FFQ, was not good. The mean servings per day from the FFQ were considerably higher than those calculated from the cross-check questions. Indeed 11% of women (731) in group 1 of the fruit and vegetable cross-check question had not answered either the fruit or vegetable serving question; some participants may have missed these cross-check questions. Others may have misread the servings per week as servings per day and therefore under-reported their intake. Additionally, there may be a tendency for participants to underestimate in their answers to cross-check questions. Conversely, FFQs tended to encourage over-reporting of fruit and vegetables consumed, therefore the derived vitamin C amount is also likely to be overestimated especially in the high intake group. As explained in section 5.3 overestimation of fruit and vegetables by FFQs is a known problem which has been assessed in the UKWCS.<sup>288</sup> The authors suggest errors can be attenuated by applying a weighting factor for each subject. However, it was decided not to apply this weighting for this analysis, since it was unclear whether the cross-check answers reflected true intake or at least true ranking of fruit and vegetable servings any better than the FFQ in the UKWCS.

Using vitamin C density, i.e. dividing vitamin C intake by total energy intake, may have reduced the effect of over-reporting on the FFQ. Furthermore, Willet (1998) explains that using nutrient densities has a tendency to produce inverse association with disease if total energy intake is associated with the disease, even when the nutrient is not independently associated with the disease.<sup>301</sup> Willet suggests adjusting for total energy in nutrient density analyses to reduce this potential problem which could be

done in sensitivity analyses; this additional energy adjustment has been used in the next chapter on total vitamin C intake.<sup>301</sup>

A strength of the restricted cubic splines, over the tables, was that the splines estimated breast cancer risks at all levels of intake within a wide intake range (60-350mg/d, in comparison to 60mg/d intake), allowing any potential non-linear relationships to become apparent. Conversely, risk estimates shown in the tables were based on average intakes for each of the five intake groups so any nonlinear relationships from these would have been less clear. Also the small numbers of women with extreme intakes, who could have influenced the extremities of the graphs, were excluded. Taking account of potential confounders in the cubic splines, however, was much more limited compared to the adjustments made in the tables. The graphs showed estimated risks for only the average woman in the cohort or in the subgroups with respect to the confounding continuous variables; the mean values for these continuous variables, such as age and BMI, were noted below the graphs. Graphs for each category of the confounding categorical variables were not produced since subgrouping the women further would have substantially reduced the power of the cubic spline analyses. Nevertheless, the conclusions from the results of the tables and from the graphs were the same; that there was no evidence of any significant linear or non-linear associations between breast cancer risk and vitamin C intake.

To explore whether breast cancer risk is influenced by nutrients in fruit and vegetables which are highly correlated with vitamin C rather than vitamin C per se, risks could be sub-analysed by vitamin C in fruit and by vitamin C in vegetables, though there may be insufficient power for further sub-analyses in the UKWCS. If different statistically significant patterns are found between fruit and vegetables with equivalent vitamin C content this would indicate that other bioactive compounds in fruit and vegetables may have more influence on breast cancer than vitamin C.

In summary the results observed from this baseline analysis of the UKWCS indicated that fruit and vegetable intake and dietary vitamin C intake are not associated with breast cancer risk.

# **CHAPTER 10**

# 10 Is total vitamin C intake associated with breast cancer incidence? An analysis of diary data from the UK Dietary Cohort Consortium nested case-control study

## 10.1 Summary

*Objectives*: To determine whether total vitamin C intake is associated with breast cancer risk as few studies have included intake from both supplements and diet in their analyses. Furthermore, Food Frequency Questionnaires used in prospective studies can over-report fruit and vegetable intake, the main source of vitamin C. This is the first study to investigate breast cancer risk and vitamin C intake using food diaries.<sup>2</sup>

*Methods*: Estimated dietary vitamin C intake was derived from four to seven day food diaries pooled from five prospective studies in the UK Dietary Cohort Consortium. This nested case-control study of 851 incident breast cancer cases and 2727 matched controls examined breast cancer risk in relation to dietary vitamin C intake using conditional logistic regression adjusting for relevant covariates. Additionally, total vitamin C intake from supplements and diet was analysed in three cohorts.

*Results*: No evidence of associations were observed between breast cancer risk and vitamin C intake analysed for dietary vitamin C intake (OR = 1.00 per 60mg/d, 95%CI: 0.91 to 1.09,  $P_{trend} = 1.0$ ) or total vitamin C intake (OR = 1.01 per 60mg/d, 95%CI: 1.00 to 1.03,  $P_{trend} = 0.1$ ). Additionally, there was no significant association for post-menopausal women (OR = 1.02 per 60mg/d, 95%CI: 1.00 to 1.05,  $P_{trend} = 0.06$ ). In the supplement only analysis, there was evidence of significant associations between vitamin C intake and incidence of breast cancer for pre-menopausal women (OR=2.51; 95%CI: 1.22 to 5.15; p=0.01) and total women (OR=1.39; 95%CI: 1.05 to 1.82; p=0.02).

*Conclusions*: This pooled analysis of individual UK women found no evidence of significant associations between breast cancer incidence and dietary or total vitamin C intake derived uniquely from detailed diary recordings. However, there was some evidence that supplements containing low doses of vitamin C were associated with an increased risk of breast cancer.

# **10.2 Introduction**

In the previous two chapters the relationship between breast cancer risk and vitamin C intake has been examined for supplement intake only, recorded by diary at phase 2 (Chapter 8) and for dietary intake only, recorded by FFQ at baseline (Chapter 9). These analyses have produced little evidence that vitamin C intake per se, or fruit and vegetable intake were associated with breast cancer incidence in UK women. Total vitamin C intake, however, could not be assessed for the full UKWCS baseline or phase 2 datasets since both dietary and supplement vitamin C intake were not electronically captured at both collection points for the full cohort.

In summary, initial findings from retrospective case-control studies have shown that fruit and vegetable intake, the main source of vitamin C, and also vitamin C intake were inversely associated with breast cancer risk. <sup>7 8 213</sup> However, no conclusive evidence of a protective effect from fruit and vegetables has been produced prospectively from cohort studies.<sup>7 10-13</sup> Similarly, the meta-analyses of prospective cohorts using FFQs in the 2007 WCRF report showed no significant associations with dietary or supplement vitamin C intake, nor in subgroup analyses by menopausal status.<sup>7</sup> Only four prospective studies in this report included vitamin C from supplements as well as diet,<sup>239 242 245 247</sup> one of which showed an increased risk with increased total vitamin C intake and breast cancer risk,<sup>270 273</sup> one of which found a weak positive association.<sup>270</sup>

FFQs tend to encourage the over-reporting of fruit and vegetable consumption,<sup>288 335 347</sup> leading to the over-estimation of vitamin C intake,<sup>347</sup> Alternatively, diaries may more accurately record numbers of fruit and vegetable portions consumed individually or in mixed dishes,<sup>347</sup> over a period of days, though they are limited by their short-term nature.

The current analysis is one of a small number of prospective studies assessing the relationship of breast cancer risk with total vitamin C intake, which includes intake from supplements as well as from diet. Pooling of the phase 2 UKWCS nested case-control data with those from four other UK cohorts was necessary to power the analysis. This pooled analysis of the UK Dietary Cohort Consortium is the first study to investigate the relationship between breast cancer risk and vitamin C intake using food diaries, an alternative tool to FFQs used in previous analyses. (Note that this uses an additional 583 controls and 144 matched cases than the UK Dietary Cohort Consortium analysis accepted for publication.<sup>2</sup> These were omitted from the dataset analysed for publication due to potential selection bias, as discussed later.)

# 10.3 Methods

#### 10.3.1 Subjects

Individual participant data were pooled from five established cohort studies within the UK Dietary Cohort Consortium: EPIC-Norfolk;<sup>283</sup> the UK Women's Cohort Study (UKWCS);<sup>279</sup> EPIC-Oxford;<sup>281</sup> Whitehall II;<sup>284</sup> and the MRC National Survey of Health and Development (NSHD).<sup>285</sup> Additional information on these cohorts is provided in section 4.1.2 of the methods chapter. Methods used were similar to those previously described for colorectal case-control analyses nested within this UK consortium.<sup>293</sup>

#### 10.3.2 Case ascertainment and matching

Incident cases of breast cancer were identified from data provided by UK cancer registries based on the International Classification of Diseases (ICD) version 9 (174) or 10 (C50). Diagnoses within six months of food diary completion were excluded to ensure that latent disease without formal diagnosis was not present; otherwise disease suspected by participants could have influenced their dietary habits. Across the cohorts 851 incident cases and 2727 controls were used in the dietary vitamin C analysis. Only three cohorts (EPIC-Oxford, EPIC-Cambridge and UKWCS) were used in the total vitamin C analysis which involved 745 incident cases and 2308 controls (85% of the consortium participants); the remaining two cohorts did not have adequate supplement use data to determine the vitamin C content of supplements consumed at diary date. Within each cohort, each case was matched to randomly selected controls based on age at recruitment (± 3 years) and date of diary completion (± 3 months or as close as possible). The number of controls matched to cases was four for EPIC-Norfolk, Whitehall and NSHD, and up to five for UKWCS. In EPIC-Oxford one control was matched to each case, to within six months of case diary completion. Controls had no registry-reported cancer diagnosis at recruitment (except non-melanoma skin cancer) and were free from breast cancer at the end of the follow-up period. The mean length of follow-up for cases in the cohorts ranged from 2.4 years to 10.8 years as detailed in Table 54; these were not adjusted for in the analyses.

222	

Table 54 Characteristics of the five cohorts from the UK Dietary Cohort Consortium included in the analyses of vitamin C and breast cancer risk

Cohort	Participants	Diary days	Years when food diary completed	Last follow up date	Mean time to diagnosis	Cases	Controls	Mean(sd) dietary vit C	Mean(sd) total vit C
					of cases			intake	intake
EPIC-Norfolk	General population in Norfolk	7 days	1993-1998	31.12.2006	6.0 yrs	365	1329	91 (50)	118 (167)
EPIC-Oxford	General population and vegetarians in the UK	7 days	1993-1998	31.12.2004	3.5 yrs	194	194	111 (61)	233 (436)
UK Women's Cohort Study (UKWCS)	Middle aged women in the UK	4 days	1999-2003	31.12.2006	2.4 yrs	186	785	117 (61)	239 (361)
Whitehall II	Civil servants in the UK	7 days	1991-1993	30.09.2005	7.8 yrs	70	275	101 (51)	a —
National Survey of Health and Development (NSHD)	Nationally reprehensive cohort of women who were born in one week in March 1946 in England, Wales and Scotland.	5 days	1989	31.12.2006	10.8 yrs	36	144	66 (37)	a 

<sup>a</sup>Whitehall and NSHD did not have detailed diary data of vitamin C intake from supplements

#### 10.3.3 Dietary methods

All cohorts collected dietary information using semi-weighed food diaries or included photographs to aid the estimation of portion size. The number of days' intake recorded for each cohort is shown in Table 54; this varied from four to seven day of recording. Food diary details were input by trained food diary analysts; the majority (88%) were entered into Data into Nutrients for Epidemiological Research (DINER) developed at Cambridge. The details were checked and nutrient data was derived using a linked system (DINERMO).<sup>373</sup> Diaries from 42 cases and 201 controls from the UKWCS were entered using an in-house Microsoft Access-based dietary analysis program (DANTE), which had previously been validated against DINER on a subsample of 100 randomly selected diaries, with acceptable agreement.<sup>293</sup> Diaries from the NSHD were entered into DIDO<sup>295</sup> which, after validation, proved to use portion sizes and recipes that were more concurrent with the time of NSHD diary completion. Owing to time and cost it was not possible for all diet diaries to be coded using a single data entry program. All estimated dietary vitamin C intake was based on standard tables of food composition and daily averages were calculated.<sup>287</sup> The underlying food tables used were the same for all coding systems. Furthermore, since cases were matched to controls within the same cohort, the different coding systems were not expected to have had any adverse effect on results.

In three cohorts (EPIC-Norfolk, EPIC-Oxford and UKWCS), in separate sections of the diaries, participants were asked to record supplement brand, name and amount per day for any supplement taken. Databases had been created to match this information against manufacturers' information for EPIC-Norfolk:<sup>38</sup> and UKWCS<sup>1</sup>. The UKWCS ingredient supplement database is described in section 4.5.3.2; this database was also used to allocate vitamin C amounts to supplements taken by EPIC-Oxford women. The two ingredient databases used included supplement descriptions and ingredient composition from product labels directly obtained from manufacturers or the participants' descriptions and/or labels. Where participants were unclear in their description, a weighted average of vitamin C from similar supplements was calculated from the database and applied.<sup>38</sup> For instance, separate generic averages were calculated for multivitamins, antioxidant ACE supplements and high dose vitamin C supplements. For each participant the average daily vitamin C amount consumed from all supplement types was calculated. No supplement data was analysed for Whitehall II and NSHD. For the Whitehall II study supplement data had been collected via a questionnaire but was not detailed enough to determine the vitamin C content of all supplements consumed. Since the NSHD cohort had collected supplement data at later follow-up phase, this was not included.

223

#### **10.3.4 Statistical analyses**

#### 10.3.4.1 Dietary and total vitamin C analyses

Separate quintile cut points were determined for dietary intake (mg per day), dietary vitamin C intake density (mg per megajoule per day) and total vitamin C intake including supplements (mg per day). Dietary vitamin C intake density was analysed as an alternative method of controlling for potential confounding by total energy intake. Conditional logistic regression was used to model the associations between fifths of vitamin C intake and breast cancer incidence. To test for linear trends continuous intake variables per increment of approximately one standard deviation of mean intake (being 60mg/day for dietary intake and 8mg/MJ/day for intake density) were used. No supplement intakes were implausible.

Owing to the process of matching cases and controls the conditional logistic regression model automatically adjusted for date of diary completion, age (in years) and cohort. The multivariate model adjusted for exact age, parity (0, 1, 2, 3, 4+, missing), hormone replacement therapy use (current, non-current, missing), alcohol intake, total energy intake, weight (<60kg, 60-, 66-, >72kg, missing), height (<158cm, 158-, 163-, >168cm, missing), physical activity (low, low-medium, medium-high, high, missing), and menopausal status (pre, peri or post-menopausal, missing). The level of missing data ranged from 0% for alcohol and total energy intake, to 0.3% for parity to 4.2% for physical activity. Alcohol and total energy intake were ascertained from the diaries. All other covariates were collected by standard questionnaires, either self-administered or by trained researchers at or close to the time of diary completion. Sensitivity analysis was performed to adjust for variables which have weaker associations with breast cancer risk (smoking status and level of education), and also to adjust for important risk variables which had moderate levels of missing data (age at menarche (13%) and cumulative duration of breastfeeding (weeks) (15%)). This restricted the sensitivity analysis to 2811 participants. Additional sensitivity analyses also controlled for dietary vitamin E and iron which affect vitamin C bioavailability. No allowance was made for the different number of days reporting for the diaries in the different cohorts. However, the assumption of no heterogeneity across the different cohorts was formally tested by including an exposure by centre interaction term in the models. Analyses were carried out using Stata version 10 and results were based on a significance level of p<0.05. Note that peri-menopausal women were included in the analyses of total women but excluded from menopausal sub-analyses.

#### 10.3.4.2 Supplement only vitamin C analyses

To test the results reported in chapter 8, which found pre-menopausal women from phase 2 of UKWCS who took supplements containing low dose vitamin C (1-60mg/d) had a significantly increased risk of breast cancer, compared to women who did not use supplement containing any vitamin C, a similar analysis was undertaken with the pooled nested case-control data from the three cohorts: EPIC-Norfolk, EPIC-Oxford and UKWCS. These were adjusted for covariates described above.

## 10.4 Results

#### 10.4.1 Dietary vitamin C intake

On average the total women (2851) in the five cohorts were 56 years old and consumed 366g/d fruit and vegetables; 65% were post-menopausal, 60% had never smoked, 23% were educated to degree, HNC or HND level, and only 20% took HRT at the date of diary completion.

As observed in Table 55, total cases (851) had similar characteristics to the 2727 controls and their mean (sd) dietary vitamin C intakes were 99mg/d (55) and 101mg/d (50) respectively. Women with a higher dietary vitamin C intake tended to have a higher energy intake, consume more alcohol, dietary vitamin E and iron as well as more fruit and vegetables. Additionally, they were more active, had attained higher levels of education, or were more likely to be of higher socio-economic status or to have never smoked (Table 55).

		Breast	cancer	Dietary vitamin C intake (diary fifths)					
Covariates (at diary date)		Cases	Controls	1	2	3	4	5	p <sup>a</sup>
Cases/controls		851	2727	171/544	161/555	167/548	179/537	173/543	
Dietary vitamin C intake (mg/day)	mean (SD)	99 (55)	101 (50)	38.6 (10.1)	65.0 (6.7)	88.8 (7.4)	119.1 (10.3)	186.4(47.6)	
Fruit Intake (g/d)	mean (SD)	196 (140)	199 (143)	90 (78)	142 (89)	193 (103)	235 (121)	325 (167)	<0.001
Vegetable intake (g/d)	mean (SD)	167 (88)	175 (95)	107 (52)	146 (65)	168 (69)	196 (92)	228 (108)	<0.001
Age at diary completion (yr)	mean (SD)	56.5 (9.4)	56.0 (9.7)	55.4 (9.7)	55.9 (9.6)	56.9 (9.1)	57.1 (9.1)	56.6 (9.2)	0.03
Height (cm)	mean (SD)	162 (7)	163 (7)	161.0 (6.4)	162.0 (6.5)	162.1 (6.8)	162.8 (6.4)	163.6 (6.4)	<0.001
Weight (kg)	mean (SD)	67.2 (12.2)	68.1 (12.2)	67.7 (11.5)	67.9 (13.1)	67.6 (13.1)	67.0 (11.5)	66.8 (11.4)	0.02
Energy intake (diary, MJ/day)	mean (SD)	7.5 (1.7)	7.6 (1.7)	6.8 (1.7)	7.3 (1.6)	7.5 (1.7)	7.7 (1.7)	8.1 (1.7)	<0.001
Alcohol intake (diary, g/day)	mean (SD)	9.5 (13.8)	10.6 (13.9)	8.4 (13)	9.8 (14)	9.9 (13)	9.9 (13)	10.7 (15)	0.009
Total fat (g/d)	mean (SD)	67.5 (22.6)	69.3 (22.0)	64.5 (21.9)	68.3 (21.5)	67.3 (22.5)	69.3 (22.9)	69.7 (23.1)	<0.001
Dietary vitamin E (mg/d)	mean (SD)	9.5 (4.0)	10.0 (4.1)	8.2 (3.9)	9.2 (3.6)	9.5 (3.7)	10.2 (4.6)	11.0 (4.3)	<0.001
Dietary Iron (mg/d)	mean (SD)	11.5 (3.5)	11.9 (3.5)	9.8 (3.1)	11.1 (3.0)	11.8 (3.5)	12.2 (3.4)	13.3 (3.6)	<0.001
Parity (number of children)	mean (SD)	2.0 (4.9)	2.3 (6.0)	2.5 (6.8)	2.6 (8.2)	1.9 (1.3)	2.0 (3.9)	2.1 (6.4)	0.1
Exercise (medium - high)	n (%)	297 (37)	1021 (39)	206 (30)	246 (36)	261 (38)	297 (43)	308 (45)	<0.001
HRT use (current user)	n (%)	157 (19)	524 (20)	125 (18)	142 (20)	139 (20)	154 (22)	121 (17)	0.9
Menopausal status (post-menopausal)	n (%)	519 (63)	1801 (67)	439 (63)	460 (66)	465 (66)	488 (69)	468 (66)	0.3
Never smoked	n (%)	499 (59)	1613 (60)	341 (49)	405 (57)	445 (63)	433 (61)	488 (69)	<0.001
Education level (degree)	n (%)	202 (26)	568 (22)	75 (11)	116 (17 )	142 (21)	203 (30)	234 (35)	<0.001
Social class (professional or intermediate)	n (%)	331 (51)	1272 (51)	257 (40)	304 (48)	321 (51)	356 (57)	365 (61)	<0.001

Table 55 Characteristics of women by fifth of dietary vitamin C intake derived from food diaries in the UK Dietary Cohort Consortium

<sup>a</sup>p is p<sub>trend</sub> over continuous exposure for continuous variables, and trend over ordered exposure groups for  $\chi 2$  tests for categorical variables

The odds ratios for breast cancer according to dietary intake of vitamin C in the five cohorts are shown in Table 56 for the unadjusted and multivariate model. There was no evidence of any significant association between dietary vitamin C intake and incidence of breast cancer for total women in the five cohorts. In the adjusted analysis for total women the odds ratio of breast cancer per 60mg/day increments was 1.00 (95%CI: 0.91 to 1.09,  $P_{trend} = 1.0$ ) Similarly, there was no evidence of any linear trends or significant associations between dietary vitamin C intake groups and incidence of breast cancer in the sub-analysis by post-menopausal status (OR=1.00 per 60mg/day, 95%CI: 0.89 to 1.13,  $P_{trend} = 1.0$ ) or by pre-menopausal status (OR=0.98 per 60mg/day, 95%CI: 0.77 to 1.24,  $P_{trend} = 0.8$ ). The results remained non-significant in sensitivity analyses after further adjustment for smoking status, age at menarche, cumulative duration of breastfeeding (weeks), and level of education. Odds ratios did not alter substantially. There was no evidence of any linear trends or significant associations between the incidence of breast cancer and dietary vitamin C expressed as intake density (Table 57).

In tests for heterogeneity there was evidence of differences between the five study centres when a study centre by dietary vitamin C intake group interaction term was included (p=0.03 total women; p=0.07 post-menopausal; pre-menopausal p=0.3), but no evidence when testing only the three largest study centres (p=0.5 total women; p=0.3 post-menopausal). When only UKWCS, EPIC-Norfolk and EPIC-Oxford, the three largest cohorts, were pooled (data not shown) there was no substantial difference in odds ratios for dietary vitamin C compared to the results for all five cohorts presented in Table 56. As observed in Table 54, the mean (sd) dietary intakes for NSHD (66mg/d (37)) was lower than EPIC-Norfolk, Whitehall II, EPIC-Oxford, and UKWCS (91mg/d (50), 101mg/d (51), 111mg/d (61) and 117mg/d (61) respectively). The lower intake for the younger, nationally representative NSHD women (mean age 43 vs. 50s in other cohorts) reflected previous findings from households with similar aged adults.<sup>374</sup>

Dietary vitamin C intake	Cases/	Unadjusted <sup>a</sup>	Adjusted	Sensitivity <sup>c</sup>				
Fifths: mean mg/day (SD	) Controls	OR (95% CI)	OR (95% CI)	OR (95% CI)				
Total women								
1 (lowest): 38.6 (10.1	) 171/544	1.10 (0.85, 1.42)	1.14 (0.88, 1.47)	1.09 (0.80,1.50)				
2 65.0 (6.7)	161/555	1	1	1				
3 88.8 (7.4)	167/548	1.02 (0.80, 1.32)	1.02 (0.79,1.31)	1.01 (0.75, 1.37)				
4 119.1 (10.3	3) 179/537	1.12 (0.87, 1.44)	1.07 (0.82, 1.38)	1.18 (0.86, 1.60)				
5 (highest): 186.4 (47.0	6) 173/543	1.04 (0.81, 1.35)	0.99 (0.76, 1.29)	1.10 (0.80, 1.52)				
P trend per 60mg/d		0.5	1.0	0.5				
Estimate per 60mg/d		1.02 (0.93, 1.12)	1.00 (0.91, 1.09)	1.04 (0.93, 1.16)				
Post-menopausal								
1 (lowest) 38.5 (9.9	) 100/339	1.08 (0.77, 1.52)	1.16 (0.82, 1.64)	1.17 (0.79, 1.73)				
2 65.3 (6.6	) 99/361	1	1	1				
3 88.8 (7.3	) 106/359	1.12 (0.81, 1.55)	1.16 (0.83, 1.62)	1.29 (0.88, 1.88)				
4 119.1 (10.2	2) 112/376	1.12 (0.81, 1.56)	1.10 (0.78, 1.53)	1.26 (0.86, 1.84)				
5 (highest) 186.3 (47.	5) 102/366	0.99 (0.71, 1.38)	0.96 (0.67, 1.36)	1.03 (0.69, 1.53)				
P trend per 60mg/d		0.4	1.0	0.7				
Estimate per 60mg/d		1.03 (0.91, 1.15)	1.00 (0.89,1.13)	1.03 (0.90, 1.18)				
Pre-menopausal								
1 (lowest) 38.5 (10.6	6) 41/136	1.32 (0.76, 2.27)	1.39 (0.76, 2.53)	1.39 (0.57, 3.37)				
2 64.6 (6.9	) 36/127	1	1	1				
3 89.1 (7.9	) 30/120	0.98 (0.54, 1.78)	1.01 (0.54, 1.90)	0.83 (0.32, 2.15)				
4 119.3 (10.0	6) 37/101	1.25 (0.71, 2.20)	1.11 (0.59, 2.07)	1.46 (0.58, 3.69)				
5 (highest) 182.1 (41.4	4) 37/110	1.29 (0.71, 2.28)	1.17 (0.63, 2.19)	1.24 (0.49, 3.15)				
P trend per 60mg/d		0.8	0.8	0.6				
Estimate per 60mg/d		1.03 (0.83, 1.28)	0.98 (0.77, 1.24)	1.09 (0.77, 1.54)				

**Table 56** Breast cancer risks by dietary vitamin C intake recorded by diaries in the UK

 Dietary Cohort Consortium

<sup>a</sup>Conditional logistic regression on cases and controls matched by centre, age, date of diary completion and length of follow-up

<sup>b</sup>As for the unadjusted model with additional adjustment for exact age, height (<158cm, 158-, 163-, 168+), weight (<60kg, 60-, 66-, 72+), physical activity, parity (0,1,2,3,4+), current HRT use, menopausal status, diary-derived alcohol consumption and total energy intake. Missing data added as a category.

<sup>c</sup>Sensitivity analysis, as for the adjusted model with additional adjustment for age at menarche, cumulative breast feeding, educational level and smoking status

vitamin C nutrient density	Cases/	Unadjusted <sup>a</sup>	Adjusted <sup>b</sup>	Sensitivity <sup>c</sup>
Fifths:mean mg/MJ/d (SD)	Controls	OR (95% CI)	OR (95% CI)	OR (95% CI)
Total women				
1 (lowest): 5.4 (1.3)	175/540	1.04 (0.81, 1.32)	1.03 (0.80, 1.33)	1.15 (0.85,1.57)
2 8.8 (0.9)	170/546	1	1	1
3 12.0 (1.0)	166/549	0.94 (0.73, 1.20)	0.93 (0.72, 1.19)	1.05 (0.78, 1.42)
4 16.2 (1.4)	177/539	1.03 (0.80, 1.32)	1.02 (0.79, 1.32)	1.13 (0.84, 1.53)
5 (highest): 25.6 (7.0)	163/553	0.89 (0.69, 1.15)	0.90 (0.69, 1.17)	1.08 (0.78, 1.48)
P trend per 8 mg/MJ/d		0.7	0.8	0.6
Estimate per 8 units		0.98 (0.90, 1.07)	0.99 (0.90, 1.08)	1.03 (0.92,1.14)
Post-menopausal				
1 (lowest): 5.5 (1.3)	99/315	1.04 (0.74, 1.46)	1.03 (0.73, 1.46)	1.15 (0.77, 1.70)
2 8.9 (0.9)	95/344	1	1	1
3 12.0 (1.0)	104/371	0.94 (0.67, 1.31)	0.93 (0.66, 1.32)	1.02 (0.69, 1.50)
4 16.2 (1.4)	120/385	1.05 (0.75, 1.46)	1.07 (0.77, 1.51)	1.13 (0.78, 1.65)
5 (highest): 25.8 (7.1)	101/386	0.87 (0.62, 1.23)	0.90 (0.63, 1.29)	1.01 (0.67, 1.51)
P trend per 8 mg/MJ/d		0.8	0.9	0.7
Estimate/ 8 units		0.98 (0.88, 1.10)	1.00 (0.89,1.12)	1.03 (0.90, 1.17)
Pre-menopausal				
1 (lowest): 5.4 (1.4)	46/149	1.30 (0.76, 2.24)	1.32 (0.74, 2.34)	1.33 (0.57, 3.10)
2 8.8 (0.9)	42/136	1	1	1
3 12.0 (1.0)	31/107	1.02 (0.57, 1.84)	0.95 (0.52, 1.75)	0.89 (0.37, 2.18)
4 16.2 (1.4)	35/98	1.14 (0.63, 2.03)	1.04 (0.56, 1.93)	1.46 (0.64, 3.30)
5 (highest): 24.8 (6.0)	27/104	0.95 (0.52, 1.74)	0.90 (0.47, 1.74)	1.13 (0.45, 2.80)
P trend per 8 mg/MJ/d		0.7	0.7	0.6
Estimate/ 8 units		0.96 (0.78, 1.19)	0.94 (0.75, 1.20)	1.11 (0.79, 1.56)

**Table 57** Breast cancer risks by dietary vitamin C density intake recorded by diaries in the UK Dietary Cohort Consortium

<sup>a</sup>Conditional logistic regression on cases and controls matched by centre, age, date of diary completion and length of follow-up.

<sup>b</sup>As for the unadjusted model with additional adjustment for exact age, height (<158cm, 158-, 163-, 168+), weight (<60kg, 60-, 66-, 72+), physical activity, parity (0,1,2,3,4+), current HRT use, menopausal status, alcohol consumption and total energy intake. Missing data added as a category.

<sup>c</sup>Sensitivity analysis, as for the adjusted model with additional adjustment for age at menarche, cumulative breast feeding, educational level and smoking status

#### 10.4.2 Total vitamin C intake

In the analyses of total vitamin C, cases had a somewhat higher total vitamin C intake than controls 191mg/d (sd 400) vs. 165mg/d (sd 246). As observed in Table 54 the mean vitamin C supplement intake per day for EPIC-Norfolk was significantly less than those for UKWCS or EPIC-Oxford (means mg/d (sd) were 28 (160), 122 (529) and 123 (695) respectively); likewise for total intake (means mg/d (sd) were 119 (168), 239 (361) and 234 (436) respectively). Based on diary completion date, mean total intake in autumn and winter compared to spring and summer were not significantly different (173.3 (sd 327) vs. 169.9 (sd 254) mg/d); comprising respectively of 47.1% and 52.9% of these women. The relationships between total vitamin C intake split by fifths and lifestyle characteristics are shown in Table 58 and were similar to those for dietary only intake (Table 55). The highest intake group had the highest vitamin C intake from both diet and supplements (mean (sd) 159 (76) mg/d) and 326 (561) mg/d respectively); 72% took supplements containing vitamin C and 84% of these women took them every day.

In pooling the three cohorts which recorded vitamin C intake from supplements there was also no evidence of any significant associations between total vitamin C intake and incidence of breast cancer in the adjusted analysis for the continuous estimate for all women (OR = 1.01 per 60mg/d, 95%CI: 1.00 to 1.03,  $P_{trend} = 0.1$ ), or for post-menopausal women (OR = 1.02 per 60mg/d, 95%CI: 1.00 to 1.05,  $P_{trend} = 0.06$ ) or for post-menopausal women (OR = 1.01 per 60mg/d, 95%CI: 0.94 to 1.09,  $P_{trend} = 0.8$ ) or by fifths of total vitamin C intake (Table 59). The results remained non-significant in sensitivity analyses after further adjustment for smoking status, age at menarche, cumulative duration of breastfeeding, and level of education. There was no evidence of significant differences between the three study centres when formally tested using a study centre by fifths of total vitamin C intake interaction term, for total, post- or premenopausal women (p=0.7, p=0.9 and 0.5 respectively).

For both dietary and total intake no substantial differences in the estimates were found in sensitivity analyses controlling for dietary vitamin E and iron.

				Total vitamin C intake (diary fifths)					
Covariates (at diary date)		Cases	Controls	1	2	3	4	5	p <sup>a</sup>
Cases/controls		745	2308	132/478	145/466	163/447	153/458	152/459	
Total vitamin C intake (mg/day)	mean (SD)	190.8 (399.6)	165.3 (245.8)	42.4 (11.8)	74.7 (8.4)	105.2 (9.9)	149.3 (17.5)	485.6 (541.8)	
Dietary vitamin C intake (mg/day)	mean (SD)	104.4 (59.7)	100.5 (55.5)	42.0 (11.9)	73.1 (10.5)	99.3 (18.5)	134.0 (31.6)	158.9 (75.5)	<0.001
Supplement vitamin C intake (mg/day)	mean (SD)	86.4 (391.7)	64.8 (236.2)	0.4 (3.3)	1.6 (7.3)	6.0 (16.1)	15.3 (29.9)	326 (561)	<0.001
Days/week vitamin C supplement taken	mean (SD	1.4 (2.5)	1.1 (2.6)	0.1 (0.8)	0.3 (1.2)	0.7 (2.0)	1.3 (2.4 )	3.6 (2.7)	<0.001
Age at diary completion (yr)	mean (SD)	57.1 (9.6)	58.0 (9.1)	57.9 (9.2)	58.5 (9.5)	58.4 (9.1)	57.1 (8.8)	57.3 (9.5)	0.004
Height (cm)	mean (SD)	163.3 (6.5)	162.0 (6.5)	161.1 (6.1)	161.9 (6.6)	162.3 (6.8)	162.9 (6.4)	163.4 (6.5)	<0.001
Weight (kg)	mean (SD)	68.4 (12.1)	67.4 (12.1)	68.5 (11.5)	68.8 (13.3)	67.6 (12.8)	67.8 (11.8)	65.6 (10.8)	0.2
Energy intake (diary, MJ/day)	mean (SD)	7.6 (1.7)	7.4 (1.7)	6.8 (1.6)	7.3 (1.7)	7.6 (1.7 )	7.8 (1.6)	7.9 (1.7)	<0.001
Alcohol intake (diary, g/day)	mean (SD)	10.5 (13.3)	9.2 (13.4)	8.0 (12.9)	9.0 (12.5)	9.7 (12.3)	10.1 (13.3)	10.7 (14.8)	0.1
Total fat (g/d)	mean (SD)	68.4 (21.9)	66.5 (22.1)	63.4 (19.9)	66.6 (22.2)	68.0 (22.4)	67.8 (20.9)	69.1 (23.7)	0.006
Non starch Polysaccharide (g/d)	mean (SD)	15.6 (5.7)	15.2 (5.6)	11.6 (4.2)	14.2 (4.7)	15.6 (5.1)	16.8 (5.3)	18.0 (6.2)	<0.001
Parity (number of children)	mean (SD)	1.8 (1.2)	2.0 (1.3)	2.1 (1.2)	2.0 (1.2)	2.0 (1.2)	2.0 (1.3)	1.8 (1.3)	<0.001
Taking supplements with vitamin C	n (%)	193 (26)	541 (23)	9 (1)	33 (5)	93 (15)	159 (26)	440 (72)	<0.001
Takes any supplement	n (%)	387 (52)	1175 (51)	179(29)	239 (39)	290 (48)	345 (56)	509 (83)	<0.001
Exercise (moderate - high)	n (%)	254 (37)	866 (39)	176 (30)	219 (37)	240 (41)	232 (40)	253 (44)	<0.001
HRT use (current user)	n (%)	144 (20)	471 (21)	113 (19)	126 (21)	124 (21)	120 (20)	132 (22)	0.4
Menopausal status (post-menopausal)	n (%)	489 (67)	1682 (74)	447 (74)	456 (76)	444 (74)	415 (69)	409 (68)	0.001
Never smoked	n (%)	440 (60)	1398 (61)	313 (52)	361 (60)	376 (62)	381 (63)	407 (67)	<0.001
Education level (degree)	n (%)	191 (27)	506 (23)	61 (10)	103 (17)	145 (25)	181 (31)	207 (36)	<0.001
Social class (professional or intermediate)	n (%)	271 (50)	1033 (50)	198 (37)	256 (47)	269 (52)	274 (54)	307 (62)	<0.001

Table 58 Characteristics by case/control status & total vitamin C intake of women in EPIC-Oxford, EPIC-Norfolk and UKWCS nested case-control cohorts

<sup>a</sup>p is p<sub>trend</sub> over continuous exposure for continuous variables, and trend over ordered exposure groups for  $\chi 2$  tests for categorical variables

$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Total vitamin C inta	ke Cases/	Unadjusted <sup>a</sup>	Adjusted <sup>b</sup>	Sensitivity <sup>c</sup>
Total women1 (lowest): $42.4$ (11.8) $132/478$ $0.86$ ( $0.66$ , $1.14$ ) $0.88$ ( $0.66$ , $1.16$ ) $0.96$ ( $0.71$ , $1.32$ )2 $74.7$ ( $8.4$ ) $145/466$ 1113 $105.2$ ( $9.9$ ) $163/447$ $1.15$ ( $0.88$ , $1.50$ ) $1.13$ ( $0.86$ , $1.49$ ) $1.14$ ( $0.85$ , $1.54$ )4 $149.3$ ( $17.5$ ) $153/458$ $1.00$ ( $0.76$ , $1.31$ ) $0.96$ ( $0.72$ , $1.26$ ) $1.07$ ( $0.78$ , $1.46$ )5 (highest): $485.6$ ( $542$ ) $152/459$ $1.01$ ( $0.76$ , $1.35$ ) $0.99$ ( $0.74$ , $1.32$ ) $0.98$ ( $0.70$ , $1.35$ )P for trend per 60mg/d $0.2$ $0.1$ $0.3$ Estimate per 60mg/d $1.01$ ( $0.99, 1.03$ ) $1.01$ ( $1.00, 1.03$ ) $1.01$ ( $0.99, 1.03$ )Post-menopausal1 (lowest) $42.7$ ( $11.7$ ) $95/352$ $1.01$ ( $0.72, 1.41$ ) $1.07$ ( $0.75, 1.51$ ) $1.18$ ( $0.81, 1.72$ )2 $74.3$ ( $8.4$ ) $99/357$ 11113 $105.4$ ( $9.9$ ) $112/322$ $1.23$ ( $0.88, 1.71$ ) $1.29$ ( $0.92, 1.82$ ) $1.39$ ( $0.96, 2.01$ )4 $149.1$ ( $17.8$ ) $91/324$ $1.00$ ( $0.70, 1.41$ ) $0.97$ ( $0.67, 1.39$ ) $1.09$ ( $0.73, 1.61$ )5 (highest) $463.3$ ( $492$ ) $92/317$ $1.12$ ( $0.78, 1.59$ ) $1.11$ ( $0.77, 1.74$ )Pre-menopausal1 (lowest) $41.1$ ( $12.5$ ) $17/74$ $1.04$ ( $0.45, 2.37$ ) $0.79$ ( $0.31, 2.02$ ) $1.02$ ( $0.33, 3.14$ )2 $75.3$ ( $8.1$ ) $22/64$ 111 </td <td>Fifths: mean mg/da</td> <td>y (SD) Controls</td> <td>OR (95% CI)</td> <td>OR (95% CI)</td> <td>OR (95% CI)</td>	Fifths: mean mg/da	y (SD) Controls	OR (95% CI)	OR (95% CI)	OR (95% CI)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Total women				
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1 (lowest): 42.4	(11.8) 132/478	0.86 (0.66, 1.14)	0.88 (0.66, 1.16)	0.96 (0.71, 1.32)
3       105.2 (9.9)       163/447       1.15 (0.88, 1.50)       1.13 (0.86, 1.49)       1.14 (0.85, 1.54)         4       149.3 (17.5)       153/458       1.00 (0.76, 1.31)       0.96 (0.72, 1.26)       1.07 (0.78, 1.46)         5 (highest): 485.6 (542)       152/459       1.01 (0.76, 1.35)       0.99 (0.74, 1.32)       0.98 (0.70, 1.35)         P for trend per 60mg/d       0.2       0.1       0.3         Estimate per 60mg/d       1.01 (0.99, 1.03)       1.01 (1.00, 1.03)       1.01 (0.99, 1.03)         Post-menopausal       1.01 (0.72, 1.41)       1.07 (0.75, 1.51)       1.18 (0.81, 1.72)         2       74.3 (8.4)       99/357       1       1       1         3       105.4 (9.9)       112/332       1.23 (0.88, 1.71)       1.29 (0.92, 1.82)       1.39 (0.96, 2.01)         4       149.1 (17.8)       91/324       1.00 (0.70, 1.41)       0.97 (0.67, 1.39)       1.09 (0.73, 1.61)         5 (highest) 463.3 (492)       92/317       1.12 (0.78, 1.59)       1.11 (0.77, 1.59)       1.16 (0.77, 1.74)         P for trend per 60mg/d       0.09       0.06       0.2         Estimate per 60mg/d       1.02 (1.00, 1.04)       1.02 (1.00, 1.05)       1.02 (0.33, 3.14)         2       75.3 (8.1)       22/64       1       1	2 74.7	(8.4) 145/466	1	1	1
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	3 105.2	2 (9.9) 163/447	1.15 (0.88, 1.50)	1.13 (0.86, 1.49)	1.14 (0.85, 1.54)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	4 149.3	8 (17.5) 153/458	1.00 (0.76, 1.31)	0.96 (0.72, 1.26)	1.07 (0.78, 1.46)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	5 (highest): 485.6	6 (542) 152/459	1.01 (0.76, 1.35)	0.99 (0.74, 1.32)	0.98 (0.70, 1.35)
Estimate per 60mg/d1.01 (0.99, 1.03)1.01 (1.00, 1.03)1.01 (0.99, 1.03)Post-menopausal1 (lowest) $42.7 (11.7)$ $95/352$ $1.01 (0.72, 1.41)$ $1.07 (0.75, 1.51)$ $1.18 (0.81, 1.72)$ 2 $74.3 (8.4)$ $99/357$ 1113 $105.4 (9.9)$ $112/332$ $1.23 (0.88, 1.71)$ $1.29 (0.92, 1.82)$ $1.39 (0.96, 2.01)$ 4 $149.1 (17.8)$ $91/324$ $1.00 (0.70, 1.41)$ $0.97 (0.67, 1.39)$ $1.09 (0.73, 1.61)$ 5 (highest) $463.3 (492)$ $92/317$ $1.12 (0.78, 1.59)$ $1.11 (0.77, 1.59)$ $1.16 (0.77, 1.74)$ P for trend per $60mg/d$ $0.09$ $0.06$ $0.2$ Estimate per $60mg/d$ $1.02 (1.00, 1.04)$ $1.02 (1.00, 1.05)$ I (lowest) $41.1 (12.5)$ 17/74 $1.04 (0.45, 2.37)$ $0.79 (0.31, 2.02)$ $1.02 (0.33, 3.14)$ 2 $75.3 (8.1)$ $22/64$ 113 $104.0 (9.3)$ $24/72$ $1.28 (0.56, 2.92)$ $1.27 (0.51, 3.11)$ $1.34 (0.44, 4.08)$ 4 $149.8 (17.4)$ $32/78$ $1.26 (0.59, 2.66)$ $1.13 (0.49, 2.58)$ $1.61 (0.56, 4.66)$ 5 (highest) $457.5 (349)$ $26/85$ $1.46 (0.65, 3.29)$ $1.33 (0.55, 3.23)$ $1.06 (0.35, 3.20)$	P for trend per 60m	g/d	0.2	0.1	0.3
Post-menopausal1 (lowest) $42.7 (11.7)$ $95/352$ $1.01 (0.72, 1.41)$ $1.07 (0.75, 1.51)$ $1.18 (0.81, 1.72)$ 2 $74.3 (8.4)$ $99/357$ 1113 $105.4 (9.9)$ $112/332$ $1.23 (0.88, 1.71)$ $1.29 (0.92, 1.82)$ $1.39 (0.96, 2.01)$ 4 $149.1 (17.8)$ $91/324$ $1.00 (0.70, 1.41)$ $0.97 (0.67, 1.39)$ $1.09 (0.73, 1.61)$ 5 (highest) $463.3 (492)$ $92/317$ $1.12 (0.78, 1.59)$ $1.11 (0.77, 1.59)$ $1.16 (0.77, 1.74)$ P for trend per 60mg/d $0.09$ $0.06$ $0.2$ Estimate per $60mg/d$ $1.02 (1.00, 1.04)$ $1.02 (1.00, 1.05)$ $1.02 (0.99, 1.05)$ Pre-menopausalI (lowest) $41.1 (12.5)$ $17/74$ $1.04 (0.45, 2.37)$ $0.79 (0.31, 2.02)$ $1.02 (0.33, 3.14)$ 2 $75.3 (8.1)$ $22/64$ 1113 $104.0 (9.3)$ $24/72$ $1.28 (0.56, 2.92)$ $1.27 (0.51, 3.11)$ $1.34 (0.44, 4.08)$ 4 $149.8 (17.4)$ $32/78$ $1.26 (0.59, 2.66)$ $1.13 (0.49, 2.58)$ $1.61 (0.56, 4.66)$ 5 (highest) $457.5 (349)$ $26/85$ $1.46 (0.65, 3.29)$ $1.33 (0.55, 3.23)$ $1.06 (0.35, 3.20)$	Estimate per 60mg	/d	1.01 (0.99, 1.03)	1.01 (1.00, 1.03)	1.01 (0.99, 1.03)
Post-menopausal1 (lowest) $42.7 (11.7)$ $95/352$ $1.01 (0.72, 1.41)$ $1.07 (0.75, 1.51)$ $1.18 (0.81, 1.72)$ 2 $74.3 (8.4)$ $99/357$ 1113 $105.4 (9.9)$ $112/332$ $1.23 (0.88, 1.71)$ $1.29 (0.92, 1.82)$ $1.39 (0.96, 2.01)$ 4 $149.1 (17.8)$ $91/324$ $1.00 (0.70, 1.41)$ $0.97 (0.67, 1.39)$ $1.09 (0.73, 1.61)$ 5 (highest) $463.3 (492)$ $92/317$ $1.12 (0.78, 1.59)$ $1.11 (0.77, 1.59)$ $1.16 (0.77, 1.74)$ P for trend per $60mg/d$ $0.09$ $0.06$ $0.2$ Estimate per $60mg/d$ $1.02 (1.00, 1.04)$ $1.02 (1.00, 1.05)$ $1.02 (0.99, 1.05)$ Pre-menopausal1 (lowest) $41.1 (12.5)$ $17/74$ $1.04 (0.45, 2.37)$ $0.79 (0.31, 2.02)$ $1.02 (0.33, 3.14)$ 2 $75.3 (8.1)$ $22/64$ 1113 $104.0 (9.3)$ $24/72$ $1.28 (0.56, 2.92)$ $1.27 (0.51, 3.11)$ $1.34 (0.44, 4.08)$ 4 $149.8 (17.4)$ $32/78$ $1.26 (0.59, 2.66)$ $1.13 (0.49, 2.58)$ $1.61 (0.56, 4.66)$ 5 (highest) $457.5 (349)$ $26/85$ $1.46 (0.65, 3.29)$ $1.33 (0.55, 3.23)$ $1.06 (0.35, 3.20)$					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Post-menopausal				
274.3 (8.4)99/3571113105.4 (9.9)112/3321.23 (0.88, 1.71)1.29 (0.92, 1.82)1.39 (0.96, 2.01)4149.1 (17.8)91/3241.00 (0.70, 1.41)0.97 (0.67, 1.39)1.09 (0.73, 1.61)5 (highest) 463.3 (492)92/3171.12 (0.78, 1.59)1.11 (0.77, 1.59)1.16 (0.77, 1.74)P for trend per 60mg/d0.090.060.2Estimate per 60mg/d1.02 (1.00, 1.04)1.02 (1.00, 1.05)1.02 (0.99, 1.05)Pre-menopausal1 (lowest)41.1 (12.5)17/741.04 (0.45, 2.37)0.79 (0.31, 2.02)1.02 (0.33, 3.14)275.3 (8.1)22/641113104.0 (9.3)24/721.28 (0.56, 2.92)1.27 (0.51, 3.11)1.34 (0.44, 4.08)4149.8 (17.4)32/781.26 (0.59, 2.66)1.13 (0.49, 2.58)1.61 (0.56, 4.66)5 (highest) 457.5 (349)26/851.46 (0.65, 3.29)1.33 (0.55, 3.23)1.06 (0.35, 3.20)	1 (lowest) 42.7	(11.7) 95/352	1.01 (0.72, 1.41)	1.07 (0.75, 1.51)	1.18 (0.81, 1.72)
3       105.4 (9.9)       112/332       1.23 (0.88, 1.71)       1.29 (0.92, 1.82)       1.39 (0.96, 2.01)         4       149.1 (17.8)       91/324       1.00 (0.70, 1.41)       0.97 (0.67, 1.39)       1.09 (0.73, 1.61)         5 (highest) 463.3 (492)       92/317       1.12 (0.78, 1.59)       1.11 (0.77, 1.59)       1.16 (0.77, 1.74)         P for trend per 60mg/d       0.09       0.06       0.2         Estimate per 60mg/d       1.02 (1.00, 1.04)       1.02 (1.00, 1.05)       1.02 (0.99, 1.05)         Pre-menopausal       1       1.04 (0.45, 2.37)       0.79 (0.31, 2.02)       1.02 (0.33, 3.14)         2       75.3 (8.1)       22/64       1       1       1         3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest) 457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)	2 74.	3 (8.4) 99/357	1	1	1
4149.1 (17.8)91/3241.00 (0.70, 1.41)0.97 (0.67, 1.39)1.09 (0.73, 1.61)5 (highest) 463.3 (492)92/3171.12 (0.78, 1.59)1.11 (0.77, 1.59)1.16 (0.77, 1.74)P for trend per 60mg/d0.090.060.2Estimate per 60mg/d1.02 (1.00, 1.04)1.02 (1.00, 1.05)1.02 (0.99, 1.05)Pre-menopausal1 (lowest)41.1 (12.5)17/741.04 (0.45, 2.37)0.79 (0.31, 2.02)1.02 (0.33, 3.14)275.3 (8.1)22/641113104.0 (9.3)24/721.28 (0.56, 2.92)1.27 (0.51, 3.11)1.34 (0.44, 4.08)4149.8 (17.4)32/781.26 (0.59, 2.66)1.13 (0.49, 2.58)1.61 (0.56, 4.66)5 (highest) 457.5 (349)26/851.46 (0.65, 3.29)1.33 (0.55, 3.23)1.06 (0.35, 3.20)	3 105.	4 (9.9) 112/332	1.23 (0.88, 1.71)	1.29 (0.92, 1.82)	1.39 (0.96, 2.01)
5 (highest) 463.3 (492)       92/317       1.12 (0.78, 1.59)       1.11 (0.77, 1.59)       1.16 (0.77, 1.74)         P for trend per 60mg/d       0.09       0.06       0.2         Estimate per 60mg/d       1.02 (1.00, 1.04)       1.02 (1.00, 1.05)       1.02 (0.99, 1.05)         Pre-menopausal       1       1.04 (0.45, 2.37)       0.79 (0.31, 2.02)       1.02 (0.33, 3.14)         2       75.3 (8.1)       22/64       1       1       1         3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest) 457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)	4 149.1	(17.8) 91/324	1.00 (0.70, 1.41)	0.97 (0.67, 1.39)	1.09 (0.73, 1.61)
P for trend per 60mg/d       0.09       0.06       0.2         Estimate per 60mg/d       1.02 (1.00, 1.04)       1.02 (1.00, 1.05)       1.02 (0.99, 1.05)         Pre-menopausal       1       1       1       1         2       75.3 (8.1)       22/64       1       1       1       1         3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest)       457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)	5 (highest) 463.3 (492) 92/317		1.12 (0.78, 1.59)	1.11 (0.77, 1.59)	1.16 (0.77, 1.74)
Estimate per 60mg/d       1.02 (1.00, 1.04)       1.02 (1.00, 1.05)       1.02 (0.99, 1.05)         Pre-menopausal         1 (lowest)       41.1 (12.5)       17/74       1.04 (0.45, 2.37)       0.79 (0.31, 2.02)       1.02 (0.33, 3.14)         2       75.3 (8.1)       22/64       1       1       1         3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest)       457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)	P for trend per 60m	g/d	0.09	0.06	0.2
Pre-menopausal       1 (lowest)       41.1 (12.5)       17/74       1.04 (0.45, 2.37)       0.79 (0.31, 2.02)       1.02 (0.33, 3.14)         2       75.3 (8.1)       22/64       1       1       1         3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest)       457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)	Estimate per 60mg	/d	1.02 (1.00, 1.04)	1.02 (1.00,1.05)	1.02 (0.99, 1.05)
Pre-menopausal         1 (lowest)       41.1 (12.5)       17/74       1.04 (0.45, 2.37)       0.79 (0.31, 2.02)       1.02 (0.33, 3.14)         2       75.3 (8.1)       22/64       1       1       1         3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest)       457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)					
1 (lowest)       41.1 (12.5)       17/74       1.04 (0.45, 2.37)       0.79 (0.31, 2.02)       1.02 (0.33, 3.14)         2       75.3 (8.1)       22/64       1       1       1         3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest)       457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)	Pre-menopausal				
2       75.3 (8.1)       22/64       1       1       1         3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest)       457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)	1 (lowest) 41.1	(12.5) 17/74	1.04 (0.45, 2.37)	0.79 (0.31, 2.02)	1.02 (0.33, 3.14)
3       104.0 (9.3)       24/72       1.28 (0.56, 2.92)       1.27 (0.51, 3.11)       1.34 (0.44, 4.08)         4       149.8 (17.4)       32/78       1.26 (0.59, 2.66)       1.13 (0.49, 2.58)       1.61 (0.56, 4.66)         5 (highest)       457.5 (349)       26/85       1.46 (0.65, 3.29)       1.33 (0.55, 3.23)       1.06 (0.35, 3.20)	2 75.	3 (8.1) 22/64	1	1	1
4149.8 (17.4)32/781.26 (0.59, 2.66)1.13 (0.49, 2.58)1.61 (0.56, 4.66)5 (highest)457.5 (349)26/851.46 (0.65, 3.29)1.33 (0.55, 3.23)1.06 (0.35, 3.20)	3 104	.0 (9.3) 24/72	1.28 (0.56, 2.92)	1.27 (0.51, 3.11)	1.34 (0.44, 4.08)
5 (highest) 457.5 (349) 26/85 1.46 (0.65, 3.29) 1.33 (0.55, 3.23) 1.06 (0.35, 3.20)	4 149.8	3 (17.4) 32/78	1.26 (0.59, 2.66)	1.13 (0.49, 2.58)	1.61 (0.56, 4.66)
	5 (highest) 457.5 (349) 26/85		1.46 (0.65, 3.29)	1.33 (0.55, 3.23)	1.06 (0.35, 3.20)
<i>P</i> for trend per 60mg/d 0.7 0.8 0.9	P for trend per 60m	g/d	0.7	0.8	0.9
Estimate per 60mg/d 1.01 (0.95, 1.09) 1.01 (0.94, 1.09) 0.99 (0.91, 1.09)	Estimate per 60mg	/d	1.01 (0.95, 1.09)	1.01 (0.94, 1.09)	0.99 (0.91, 1.09)

**Table 59** Breast cancer risks by total vitamin C intake recorded by diaries in EPIC 

 Oxford, EPIC-Norfolk and UKWCS cohorts of the UK Dietary Cohort Consortium

<sup>a</sup>Conditional logistic regression on cases and controls matched by centre, age, date of diary completion and length of follow-up

<sup>b</sup>As for the unadjusted model with additional adjustment for height (<158cm, 158-, 163-, 168+), weight (<60kg, 60-, 66-, 72+), physical activity, parity (0,1,2,3,4+), current HRT use, menopausal status, diary-derived alcohol consumption and total energy intake

<sup>c</sup>Senstivity analysis as for the adjusted model with additional adjustment for age at menarche, cumulative breast feeding, educational level and smoking status

#### 10.4.3 Supplement-only vitamin C intake

In the supplement-only analysis there was evidence of significant associations between vitamin C intake and incidence of breast cancer for total women and pre-menopausal women who took supplements containing low doses of vitamin C (1-60mg/d). As seen in Table 60, the odds ratio was higher for pre-menopausal women (OR=2.51; 95%CI: 1.22 to 5.15; p=0.01) than for total women (OR=1.39; 95%CI: 1.05 to 1.82; p=0.02). A moderate increased risk was also seen for post-menopausal women in this intake range but this was not significant (OR=1.24; 95%CI: 0.84 to 1.80). No significant associations were found in the other intake ranges (>60<500mg/d and  $\geq$ 500mg/d).

Similar results were produced in sensitivity analyses after further adjustment for smoking status, age at menarche, cumulative duration of breastfeeding (weeks), and level of education. Odds ratios increased for pre-menopausal women taking supplements containing low doses of vitamin C (OR=3.34; 95%CI: 1.37, 8.17; p=0.008).

As observed in Table 61, the majority (66%) of the pre-menopausal women in this supplement intake category were from the UKWCS. Results of separate analyses for the UKWCS nest case-control for pre-menopausal women using supplements containing low dose of vitamin C were much higher than the analysis which combined the other two cohorts and the latter was not significant ((OR=4.54; 95%CI: 5.3, 13.6; p=0.007) vs. (OR=1.62; 95%CI: 0.54, 4.91; p=0.4) not tabled).
Supplement vit C	Cases/	Unadjusted <sup>a</sup>	Adjusted <sup>b</sup>	Sensitivity <sup>c</sup>
Intake mean mg/day	Controls	OR (95% CI)	OR (95% CI)	OR (95% CI)
Total women				
No vitamin C	552/1767	1	1	1
1-60mg/d	93/221	1.35 (1.03, 1.78)	1.39 (1.05, 1.83)	1.39 (1.03, 1.90)
>60<500mg/d	59/199	0.89 (0.64, 1.24)	0.89 (0.64, 1.24)	0.83 (0.56, 1.22)
≥500mg/d	41/121	0.99 (0.66, 1.48)	0.98 (0.65, 1.48)	0.90 (0.56, 1.44)
Post menopausal				
No vitamin C	383/1312	1	1	1
1-60mg/d	48/153	1.16 (0.80, 1.67)	1.24 (0.84, 1.80)	1.34 (0.89, 2.02)
>60<500mg/d	35/139	0.91 (0.60, 1.38)	0.90 (0.58, 1.37)	0.89 (0.55, 1.45)
≥500mg/d	23/78	1.04 (0.60, 1.80)	1.01 (0.58, 1.78)	0.89 (0.47, 1.68)
Pre- menopausal				
No vitamin C	81/270	1	1	1
1-60mg/d	22/39	2.34 (1.21, 4.51)	2.51 (1.22, 5.15)	3.34 (1.37, 8.17)
>60<500mg/d	8/39	0.92 (0.36, 2.35)	0.88 (0.30, 2.58)	0.90 (0.25, 3.27)
≥500mg/d	10/25	1.34 (0.51, 3.54)	1.28 (0.45, 3.67)	0.92 (0.27, 3.18)

**Table 60** Breast cancer risks according to vitamin C content of supplements recorded by diaries in EPIC-Oxford, EPIC-Norfolk UK and UKWCS cohorts of the UK Dietary Cohort Consortium

<sup>a</sup>Conditional logistic regression on cases and controls matched by cohort, age and date of diary completion

<sup>b</sup>As for the unadjusted model with additional adjustment for exact age, height (<158cm, 158-, 163-, 168+), weight (<60kg, 60-, 66-, 72+), physical activity, parity (0,1,2,3,4+), current HRT use, menopausal status, diary-derived alcohol consumption total energy intake, and dietary vitamin C. Missing data added as a category.

<sup>c</sup>Senstivity analysis as for the adjusted model with additional adjustment for age at menarche cumulative breast feeding, educational level and smoking status.

Numbers (%)	Epic-Norfolk	UKWCS	Epic-Oxford	Total
Total				
No vitamin C	1433 (62%)	618 (26%)	268(12%)	2319 (100%)
1-60mg/d	141 (45%)	133 (42%)	40 (13%)	314 (100%)
>60<500mg/d	92 (36%)	120 (46%)	46 (18%)	258 (100%)
≥500mg/d	28 (17%)	100 (62%)	34 (21%)	162 (100%)
Total	1694 (55%)	971 (32%)	388 (13%)	3053 (100%)
Post-				
No vitamin C	1183 (70%)	389 (23%)	123 (7%)	1695 (100%)
1 60mg/d	114 (57%)	76 (38%)	11 (5%)	201 (100%)
1-001119/0	73 (42%)	84 (48%)	17 (10%)	174 (100%)
>500<300mg/d	24 (24%)	66 (65%)	11 (11%)	101 (100%)
Total	1394 (64%)	615 (28%)	162 (8%)	2171 (100%)
Pre-menopausal				
No vitamin C	171 (49%)	155 (44%)	25 (7%)	351 (100%)
1-60mg/d	19 (31%)	40 (66%)	2 (3%)	61 (100%)
- >60<500mg/d	13 (28%)	27 (57%)	7 (15%)	47 (100%)
≥500mg/d	4 (11%)	25 (72%)	6 (17%)	35 (100%)
Total	207 (42%)	247 (50%)	40 (8%)	494 (100%)

 Table 61
 The percentage of women in each supplement category from the three cohorts of the UK Dietary Cohort Consortium in the vitamin C supplement analyses

## **10.5 Discussion**

This pooled analysis of individual participant data from five UK cohorts found no evidence of an association between incidence of breast cancer and dietary vitamin C intake recorded by food diaries. Neither was there any evidence of an association with total vitamin C intake when vitamin C from supplements was included for three of the cohorts. The non-significant results relating to dietary vitamin C intake support null results found in chapter 9 which used data from the whole of the UKWCS baseline. The current results for post-menopausal women also support the results of the 2007 WCRF meta-analyses of three cohort studies (RR=1.15 per 100mg/d, 95% CI: 0.92-1.43),<sup>7 245 246 265</sup> also the high versus low intake results of two US studies,<sup>242 247</sup> and the recent European Prospective Investigation into Cancer (EPIC) analysis involving the pooling of data from 10 European countries (highest vs. lowest quintile HR = 0.98, 95% CI: 0.87–1.11);<sup>244</sup> all of which used FFQs. The current results for dietary vitamin C are in conflict with significant evidence of a 12-14% reduced risk found in the meta-analysis of retrospective case-control studies which,<sup>7</sup> unlike our study, are prone to recall bias.

In contrast to the results from the current analysis and other studies,<sup>239 242 247 273</sup> the large Women's Health Initiative study<sup>270</sup> found significant but weak evidence of increased breast cancer risk for total intake. The advanced age of the participants in this cohort (average 64 years) might suggest that high vitamin C intake may promote the progression of cancer in older people or at later stages of the disease. Similarly positive associations with post-menopausal breast cancer for both dietary and total vitamin C intake (OR= 2.06 per 100mg/d, 95% CI: 1.45-2.91; and OR=1.08 per 100mg/d, 95% CI: 1.02-0.1.15 respectively) were found in a small Danish nested case-control study,<sup>245</sup> but not in the recent full analysis of this Danish cohort;<sup>273</sup> selection bias of controls or restricting to women who consumed vitamins from both dietary and supplement sources may have possibly influenced the earlier results. Total intake could not be assessed for the full UKWCS baseline or phase 2 datasets, since both dietary and supplement vitamin C intake was not captured at both collection points.

The doubling of breast cancer risk for pre-menopausal women consuming supplements containing low dose of vitamin C (1-60mg/d) in the supplement-only analyses supports earlier results found in chapter 8 which analysed data from the full phase 2 UKWCS dataset. This was despite a difference in calculating the average intake: the average supplement vitamin C intake used in this nested case-control was calculated over the total diary days, whereas the average used in the full dataset was calculated over the number of days taken. Possible reasons for the similar results were because diary recordings were used in both, and because a large proportion of these women in the

pooled nested case-control dataset were from the UKWCS (66%). Indeed, a significant association was found in the analysis using only UKWCS data, but not in the analyses combining EPIC-Norfolk and EPIC Oxford, though the latter may be due to low numbers involved. One reason why the result for total women consuming supplements containing low dose vitamin C in this analysis, and not in the previous analysis in chapter 8, is that pre-menopausal women made up a larger proportion of the total in this dataset than the full UKWCS phase 2 dataset.

Pooling of the cohorts was necessary since none of the separate nested case-control studies was large enough to power the analysis to detect a significant moderate effect for total vitamin C intake (about 650 cases with five matched controls were needed to detect an OR of 1.4, as calculated in the methods chapter, section 4.7.7). Pooling individual participant data in this consortium had three other advantages. Firstly, it ensured that vitamin C intake from the whole consortium could be categorised into fifths; secondly, the variations in intake across the cohorts increase the power to detect smaller effect sizes,<sup>375</sup> i.e. many women in EPIC-Oxford and UKWCS were vegetarians and/ or consumed supplements containing vitamin C compared to the other cohorts; thirdly, analysis and adjustment by covariates could be done in a uniform way.

The study had a few caveats. Whilst the use of missing covariate categories may have grouped dissimilar individuals and introduced some bias, its effect on the adjusted results may be considered acceptable since the level of missing data was small, and confounding was judged to be weak. To account for the possible modulation of vitamin C on cancer development due to its the role in the regeneration of vitamin E, in the absorption of iron and in the Fenton reaction,<sup>76</sup> sensitivity analysis adjustments were made for these dietary nutrients. Supplement intake data for these nutrients, however, was not available. The Danish studies, one of which found a positive association, controlled for both dietary and supplement intake of vitamin A and E.<sup>245 273</sup> In the current study data were unavailable to adjust for family history of breast cancer which has been associated with high-dose vitamin C supplement use in the UK <sup>1</sup>. Data were unavailable from all cohorts to exclude general supplement users from the dietary analysis; the different health behaviours of users may have influenced the results.<sup>27</sup> There was inadequate power to sub-analyse by HRT users, oestrogen receptor-negative or pre-menopausal breast cancers to find moderate effects.

Unfortunately, it was considered that the selection of some of the UKWCS controls was biased, in that they may have not been truly representative of the cohort, the population from which the cases were drawn, with respect to vitamin C exposure. This selection bias could have been minimised by selecting controls at random from the

whole cohort. However, the controls from the UKWCS which were coded using DINER, a system developed at Cambridge, were considered not to be a random selection since it was later found that they were more likely to be meat-eaters and less likely to be vegetarians than the cohort as a whole. It was assumed that, on average, the meateaters would have consumed less vitamin C than vegetarians, and therefore controls were likely to have had lower vitamin C intake than the cohort on average. An artificially inflated difference in vitamin C intake between cases and controls could have attenuated any potential associations between breast cancer risk and vitamin C intake in the results. Therefore, for the manuscript accepted for publication which used the same dietary and total analysis methods used in this chapter,<sup>2</sup> these 583 controls and the 144 matched cases were omitted from the dataset due to potential selection bias. However, their omission made little difference to the results, giving further support to the conclusion that vitamin C intake is not associated with breast cancer risk. Additionally, further inspection indicated that the vitamin C intake of these controls was, in fact, unlikely to be much different from the cohort as a whole; as seen in chapter 5, section 5.3.3, the vegetarians in the UKWCS on average consumed only 3% more vitamin C than non-vegetarians, the majority of whom were meat eaters. Given these facts, it was decided to include these cases and controls in the analyses for this chapter.

This is the first time the relationship between breast cancer risk and vitamin C intake has been analysed using prospective data from food diaries. Diaries can capture detailed and accurate intake over a narrow period of days due to their open format, whereas FFQs aim to reflect intake over a much longer period, normally an estimated average of the previous 12 months. Repeated diary data collections may reduce their short-term limitations but were not undertaken for the whole consortium due to expense and time taken to administer, complete and analyse. The required commitment and awareness of intake may have also influenced participants' consumption during diary recording. As discussed in section 5.4.4 of the evaluation chapter, when compared to FFQs, food diaries have shown stronger correlations with plasma vitamin C biomarkers in validity tests when collected in close temporal proximity. However this may reflect the short-term nature of both plasma vitamin C and diary data, particularly since correlations with biomarker levels re-measured several years later were similar for diaries and FFQs.<sup>346 347 370</sup> Furthermore, other UK validation studies have shown similar associations between biomarkers and vitamin C estimated from FFQs and diaries. <sup>294 338</sup> Overall correlations between biomarkers and FFQs or diaries are generally weak to moderate.<sup>335 343</sup> Since the absorption and storage of vitamin C is limited, particularly above 400mg/d,<sup>65</sup> biomarkers are unlikely to reflect dietary vitamin C intake well. Therefore it is difficult to assess objectively whether diaries or FFQs can rank individual intake sufficiently well in order to find associations between vitamin C and cancer risk. Given the limitations, results of vitamin C analyses from either FFQs and diaries need to be treated with some caution. Additional analysis could be undertaken using dietary vitamin C FFQ data from this UK Dietary Cohort Consortium dataset, to compare them with the diary dietary vitamin C results. Nevertheless, the dietary results using diary data in this chapter supports those reported in chapter 9 using FFQ recordings.

To conclude, the supplement-only results support those found in chapter 8, that use of supplements containing low-dose vitamin C is associated with increased breast cancer risk, though as explained in chapter 8, the results may be due to the effects of other ingredients in supplements, or may be spurious. In relation to dietary or total vitamin C intake, the evidence to date from this and other prospective studies does not indicate either a beneficial or a detrimental effect of vitamin C intake on breast cancer risk.

# **CHAPTER 11**

## **11 Summary discussion**

The analyses in this thesis have used data from the UKWCS and other cohorts in the UK Dietary Cohort Consortium. These are some of the largest population-based prospective studies in the UK which were designed to assess associations between diet and chronic diseases. The majority of the analyses used previously unexploited supplement data to explore, for the first time, breast cancer risk in UK women in relation to:

- any supplement use, including use of any supplements over two recording periods, roughly 4 years apart (chapter 7)
- vitamin C contained in supplements, examining dose-response relationships, as well as the risk of taking supplements containing high, medium and low vitamin C content (chapter 8)
- dietary vitamin C intake derived from FFQs for supplement users and nonusers; the latter sub-analysis provided a clearer picture of the relationship between breast cancer risk and dietary vitamin C intake without residual confounding by factors relating to supplement use. The use of restricted cubic spline models enabled non-linear relationships between vitamin C intake and breast cancer risk to be explored for the first time (chapter 9)
- total vitamin C intake from both diet and supplements; this was examined prospectively for the first time using food diary records instead of FFQs (chapter 10)

The combination of these objectives ensured that vitamin C intake in relation to breast cancer risk in UK women had been thoroughly examined in this thesis. The first three objectives used data from the baseline or phase 2 of the UKWCS only, whilst the fourth objective used nested case-control data from the UKWCS and other UK studies, which were part of the UK Dietary Cohort Consortium. Data from the extensive supplement ingredient database and participant supplement use database at phase 2 of the UKWCS was used for the second objective. Additionally, a large number of vegetarian women consuming high levels of dietary vitamin C from fruit and vegetables were included in the baseline analysis. This provided a wide range of vitamin C intakes for analysis. Furthermore, women from baseline and phase 2 in the UKWCS were followed-up for cancer incidences for over 11 and 7 years respectively, which was

comparable to or better than most studies detailed in the literature review chapter in section 3.7.2.

A major strength of these studies is that their prospective nature minimised reverse causality, recall bias and responder bias which can affect results of retrospective casecontrol studies. Selection bias, however, may have been present in the nested casecontrol analysis of total vitamin C, though this may have been minimal. Their large size increased the power of the analyses, particularly in the baseline and nested casecontrol analyses which involved substantial numbers of cases. Additionally, the study questionnaires captured many important health and lifestyle factors which were potential confounders and were adjusted for in the analyses.

The UKWCS, instigated through the World Cancer Research Fund, in general comprised of health conscious women. The results are therefore directly applicable to similar women, who would be interested to know how lifestyles choices could affect their breast cancer risk, and who may be prepared to alter their behaviours. The UK Dietary Cohort Consortium data produced results slightly more generalizable to the UK population as a whole, in addition to providing an opportunity to examine breast cancer risk in relation to total vitamin C intake recorded by diary data.

In addition to the objectives listed above, the phase 2 UKWCS data provided an opportunity to determine, for the first time, whether UK women who have a history of breast cancer were more likely to use high dose vitamin C supplements. The results showed women who had a family history of breast cancer, or who self-reported having had breast cancer, were respectively 26% and 70% more likely to use vitamin C supplements of 1000mg/d or above, (chapter 6).<sup>1</sup> Since these women were likely to have different health behaviours than others, women with a family history of breast cancer risk sensitivity analyses in this thesis.

In relation to risk, the results showed there was no evidence of statistically significant associations between general supplement use, fruit and vegetable intake and vitamin C intakes per se and breast cancer incidence in the UK in any of the analyses reported in this thesis. Similarly, no associations were found after excluding all supplement users in the dietary vitamin C analysis; previously non-supplement users had been examined in only one other cohort.<sup>273</sup> Moreover, the unique use of diary recorded data in dietary and total vitamin C analyses produced no evidence of associations with breast cancer risk. These results are inline with the majority of previous prospective studies from other countries reported in the 2007 WCRF review and the 2008 WCRF continuous update.<sup>7 169</sup> As discussed in section 3.7 of the literature review chapter, the

inclusion of the results from more recently published studies in future meta-analyses are unlikely to produce a different conclusion from the 2007 WCRF review,<sup>14 244 270 273</sup> despite a significantly increased breast cancer risk with high vitamin C intake from supplements reported in 2008 by a large US study.<sup>270</sup> Meta-analyses including these recent studies have not been done for this thesis since these are likely to be undertaken by the WCRF in their future continuous updates. Regardless, there was insufficient evidence from the UKWCS and other studies to support a U–shaped relationship between vitamin C intake and breast cancer risk, which at the start of this thesis had been hypothesised due to the potential of vitamin C to act as both an antioxidant and a pro-oxidant.

The current study did, however, find a more than doubled increased risk for premenopausal women taking supplements containing low dose vitamin C, up to and including 60mg/d (the EU RDA), compared to women not taking supplements containing vitamin C. Since there appears to be no biological rationale that this increased risk is specifically due to low vitamin C content, it was thought to be due to other ingredients in the supplements taken. However, in exploring the relationship with multivitamin supplements, which usually contain EU RDA amounts of vitamin C, no significant associations with breast cancer in the UKWCS were observed (although a non-significant increased risk of 50% for pre-menopausal women was found for daily users). Additionally, as seen in chapter 7, general supplement use was associated with a weak, but non-significant increased risk for pre-menopausal women. The significant finding may be spurious due to residual confounding, or possible inappropriate use of covariates or due to low numbers of pre-menopausal women involved. Alternatively multiple testing of sub-groups may have produced spurious significant associations by chance, though the pattern seen for pre-menopausal women does not support this. To avoid chance findings, lower p values for testing levels of statistical significance could have been used;<sup>298</sup> the p-values, nevertheless, for the doubling in risk were very low (e.g. 0.004). Instead, the significant results may be specific to this group of cases in the UKWCS. Indeed, in a previous UKWCS analysis of pre-menopausal women another result was difficult to explain; vegetarians were found to have a significantly higher breast cancer risk than low meat consumers.<sup>82</sup> Since few studies have examined use of supplements containing vitamin C in well-nourished pre-menopausal women more research is needed. The two studies which have assessed low level intakes of vitamin C from supplements included post-menopausal women only.<sup>270 273</sup>

Although the UK cohorts were of high quality and large size, the possibility that null results have been found when a true association exists should not be ruled out entirely. Various reasons for null results have been suggested by Willet, which can be related to

the current findings.<sup>301</sup> From the life-course perspective, vitamin C intake prior to breast tissue differentiation before first pregnancy may have a greater influence on breast cancer risk than intake in later life. Nevertheless, it is difficult to argue that midlife is not a critical period for susceptibility to the effects of vitamin C intake in relation to cancer development; midlife in general being the age when the women in these cohorts recorded intake. Considering the antioxidant theory, and given the potential accumulation of DNA damage caused by free radicals as we age, midlife appears to be a time when the antioxidant effects of vitamin C would be most beneficial in preventing cancer initiation. On the other hand, non-invasive cancer cells may be already present in many women at midlife, and these may be susceptible to further cancer progression by potentially detrimental effects of vitamin C discussed in chapter 2. Since vitamin C plays many roles in the body and interacts with other nutrients, any weak to moderate positive and negative effects these create in relation to cancer may balance out to produce a null effect overall. An additional reason for null results is vitamin C may only affect cancer risk in certain types of women and there may have been insufficient cases to power some of the sub-analyses undertaken to find statistically significant effects. Different mechanisms in the body are likely to dominate in certain sub-groups of pre- or post-menopausal women, for instance it is possible that vitamin C may have a differential effect with women who are overweight compared to those who are not; however this was not analysed in this thesis. Since subgroup analysis can produce significant associations by chance, they should be stated a priori or, as mentioned above, at least smaller p values should be used for statistical significance levels.

Although there was considerable variation in vitamin C intake across the UKWCS, the women were well-nourished, meaning there may have been insufficient women with low vitamin C status to reveal a raised risk of breast cancer in such women, if one existed. Only 3% of women in the baseline analyses had dietary intakes below the EU RDA (60mg/d) and less than 1% had intakes below the UK RNI (40%). Furthermore, simple linear dose-response relationships may not have been apparent when calculating continuous estimates over a wide intake range, due to the limits of absorption, storage, etc of vitamin C at high intake levels. Nevertheless, the restricted cubic spline graphs did not show any significant relationships, despite excluding outliers that may have undue influence. Another reason for finding a null result is that, as discussed in chapters 5 and 10, the FFQ, and 4-d and 7-d diary methods in general may be too imprecise to measure vitamin C accurately. Although Bingham et al. (1987) suggested that 36 days of weighed food records are required to correctly classify vitamin C intake at an individual level,<sup>376</sup> this is not practical, and besides participants are likely to change their eating habits to reduce the burden of recording. Repeat

measures over a period of years are needed to examine the effects of longer term intake, or fluctuations in intake. The largest fluctuations in nutrient intakes are likely to be due to supplements, but as seen in chapters 3 and 8, little research on this has been reported, and further research is needed. Finally, another limitation common to all the UKWCS results is that, since the cohort includes a high proportion of vegetarians, the results are not generalizable to the UK population, therefore absolute risk rates for the UK as a whole could not be calculated from the results. Sensitivity analyses could be undertaken to reweight the cohort by the percentage of vegetarians in the general population. In the 2004 NDNS survey 7% of women (and about 5% over the age of 35 years) stated they were vegetarian,<sup>24</sup> whereas at baseline in the UKWCS 28% stated they were vegetarian. However, a more accurate but pragmatic definition classified 18% as vegetarian (those who ate meat or fish less than once a week).<sup>279</sup> Previous analyses of the UKWCS, however, have found no substantial differences before or after weighting for vegetarians.

Although adjustments were made for many confounding factors, one potential limitation was that no adjustment could be made for breast screening, since this information had not been gathered. Breast screening, which has been provided to over 50 year olds since the start of the UKWCS, could have been a confounding factor, particularly in the post-menopausal analyses. This was because breast screening was likely to pick up cases sooner and because health conscious women were more likely to attend screening as well as tending to have a higher vitamin C intake. Additionally, these women may have been more likely to have had a family history of breast cancer, and potentially had a higher risk of breast cancer themselves. Future analyses of other cohorts which have gathered screening information could control for this, though prior studies however, have not done so. It is possible that given the long length of follow-up of this cohort, and many prior studies, screening may not have detected substantially more cases.

### 11.1 Ideas for future research

#### 11.1.1 UKWCS

Further sensitivity analyses of the current data could be undertaken to observe the effects of excluding self-reported cancers in the baseline analyses, or the effects of excluding women with a family history of cancer, where this has not already been done. Additionally, sensitivity analyses could be undertaken to exclude women who developed breast cancer within two years, since some of these women may have suspected they had cancer and altered their diet before their diagnosis. Furthermore,

the interaction, seen in chapter 7, between general supplement use and socioeconomic status in relation to breast cancer could be explored further.

Since total cancer incidence is available for the UKWCS, associations between this risk and vitamin C intake or multivitamin use could be examined. Although, there are too few women (861) who had prevalent breast cancer at baseline to examine risk of breast cancer reoccurrence, there is scope to examine changes in diet from baseline to phase 2 for the ~130 women who developed cancer between these data collections.

Within phase 2 of the UKWCS there is a range of possibilities for examining risk of cancers or other diseases in relation to other supplements, whether this is from recordings by diary or questionnaire. Supplement types recorded by questionnaire, however, are limited to those displayed in Figure 2 in chapter 2, with fish oil (26.5%), evening primrose oil/ starflower oil (18.7%) and calcium (14.2%) being used by a relatively high percentage of women. Unfortunately, vitamin D supplement use, which has been the focus of more recent attention,<sup>377</sup> has not been recorded by questionnaire in the UKWCS. Additionally, folic acid supplement use, which as discussed in chapter 8 has been linked to an increased risk of cancer, was only recorded by 2.2% of women on the questionnaire which would result in insufficient cases to power analyses. Nevertheless, folic acid is present in many supplements, particularly multivitamins; therefore the amount of this or other nutrients such as vitamin D in supplements recorded by diary could be determined from the UKWCS participant supplement database. Further cleaning of this database would be needed, and ideally, these analyses should be delayed to ensure sufficient cancer cases have accumulated to power the analyses.

Furthermore, breast cancer risks for UKWCS women who followed most of the eight WCRF cancer prevention guidelines, summarised at the front of the 2007 WCRF report,<sup>7</sup> could be compared to those who did not. Similar research has been undertaken in a US cohort where a score of 0-2 was given for adherence to each of the four American Cancer Society prevention guidelines, and subjects with total scores of 7-8 were compared to those with total scores of 0-2.<sup>378</sup>

Although women at phase 2 who were at raised risk or high risk of breast cancer, according to the NICE guidelines, were excluded from some of the analyses in this thesis, they comprised such a small percentage of UKWCS (2.1% and 0.8% respectively) that breast cancer risk analyses for these women were not feasible. Also, there are too few of these women to undertake a nested case-control study.

245

#### 11.1.2 Other

As previously discussed, there is a need for more accurate methods of measuring dietary intake in order to examine relationships between diet and cancer risks. There is potential for cohort dietary data to be collected using innovative technologies in the future; for instance using on-line 24 hour recall questionnaires and FFQs, or using handheld devices such as the Smartphone with the 'My Meal Mate' application that is being trialled at the University of Leeds.<sup>379</sup> These have facilities for participants to select foods and have the potential for data to be downloaded to an automatic coding system, which would reduce data collection and coding time, as well as coding errors. Additionally, the portability and convenience of the Smartphone has an added advantage that recordings can be done in real-time and not be reliant on memory.<sup>379</sup> Further work, however, may be needed on the application to ensure portion sizes are correctly recorded.<sup>379</sup> It is expected to be less burdensome than paper diary methods, particularly for the younger generation, and therefore participants may be less likely to alter their normal intake and may be willing to produce many days of reporting necessary to correctly classify their nutrient intake, and to repeat these at future dates. In order to minimise costs, the recruitment of sufficient participants with their own personal Smartphones for a large cohort appears possible in the future since these phones are becoming widely used. Although this device may not be suitable to gather follow-up data for the women of the UKWCS, three quarters of whom will now be over the age of 60, it may be suitable for collecting data from their offspring, if required.

Future research on other cohorts needs to take account of supplements, whether this is through sub-analysis by supplement users and non-users or, for nutrient analyses, by including nutrient intake from supplements. As discussed above more research is needed relating to supplement use in pre-menopausal women, and in particular use of supplements containing low dose vitamin C such as multivitamins. Vitamin C intake sub-analyses by HRT or by BMI should be undertaken in large cohorts to test for significant results found in the pooled EPIC study and the RCT previously mentioned.<sup>14</sup> <sup>255</sup> Additionally, very few studies have sub-analysed by hormone receptor status of breast cancers,<sup>270 273</sup> since these data have not been widely available, or there has been insufficient cases to power the analyses. In general, the prevalence of oestrogen receptor negative breast cancers may be less than 20%, and for triple negative breast cancers may be about 15%.<sup>380</sup> Many women who develop the latter are carriers of BRCA1 gene faults,<sup>380</sup> and little research has been done to identify whether lifestyle factors including diet could modify risks for these women, or others who are highly susceptible to breast cancer. Due to the small percentage of these women, nested case-control studies would be needed. A Korean case-control study reported breast cancer risk was higher in women who were carriers of ATM (Ataxia telangiectasia mutated cells) and had low intakes of vitamin C or other antioxidants compared to those with high intake.<sup>272</sup> Since intake in this study was measured after diagnosis of cancer, temporal bias cannot be ruled out.

Genetic susceptibility to having low blood vitamin C levels could be researched with respect to breast cancer risk. A genetic variation (at the *SLC23A1* gene locus) in coding for a sodium-dependent vitamin C active transporter which facilitates ascorbic acid transport across intestinal cell membranes has been associated with low plasma ascorbic acid levels.<sup>381</sup> This genetic variant could be used as an instrumental variable within observational studies as a proxy for plasma vitamin C levels to determine the causal effect of lifetime variation in vitamin C concentrations on the risk of cancer and other chronic diseases.<sup>381</sup> This 'Mendelian Randomisation' method has the added benefit of being free from confounding influences since potential confounders are naturally randomised between carriers and non-carriers of the genetic variant.<sup>382</sup> However, this method is limited by the need for very large sample sizes.<sup>382</sup>

Nutrigenomics is a relatively new research discipline involving the study of genetic variations and their interactions with nutritional factors and subsequent associations with disease. Not only can genetic variation affect an individual's processing of food, but constituents of food can also change the expression of genes, i.e. the phenotype, which may then impact on the development of disease. It may be some considerable time, however, before research evidence from nutrigenomics can be employed at an individual and commercial level due to the cost and time involved in the genotype-phenotype-biomarker testing of individuals that would be necessary in order to provide tailored dietary advice.

More research using larger cohorts, nested case-control studies or RCTs is needed in relation to vitamin C intake or supplement use and risk of cancer reoccurrence, especially since cancer survivors are more likely to use supplements, as seen in chapter 6. There has been controversy surrounding evidence of benefits of high dose vitamin C supplementation for prolonged cancer survival,<sup>87-89</sup> indeed there is some evidence that antioxidants can potentially reduce the effectiveness of anti-cancer drugs.<sup>77 356</sup> Patients, therefore, should be encouraged to discuss their supplement use with their doctors in order to avoid contraindications. More recently clinical trials using intravenous vitamin C, with or without chemotherapy have been undertaken (e.g. trial reference NCT00441207 and NCT01050621). Phase I results of a recent Canadian trial failed to demonstrate an effect on patients with advanced cancer.<sup>383</sup>

### **11.2 Public health issues**

The potential for antioxidants, contained in fruit and vegetables, to reduce the risk of cancers, was one of the original promoted benefits of the 5-A-day fruit and vegetable campaign. Despite the evidence in chapter 9 and in the 2007 second WCRF review,<sup>7</sup> that neither the increased intake of fruit and vegetables, nor dietary intake of vitamin C appear to reduce the risk of breast cancer, the public health message to consume high intakes of fruit and vegetables need not be changed since there is no convincing evidence of detrimental effects. Moreover, fruit and vegetable intake should be encouraged since there is evidence they reduce the risk of other cancers,<sup>7</sup> as well as heart disease.<sup>384</sup> Whether it is antioxidants and specifically vitamin C, or other bioactive chemicals in the fruit and vegetables that are beneficial in these cases is still unclear.

The current results also suggest, along with some but not all previous studies reported in chapter 8, that there are no detrimental effects in relation to breast cancer risk from using high dose vitamin C supplements, meaning the current results do not support a need to restrict their sale to the general public. Furthermore, after reviewing evidence on any aspect of health, both the European Food Safety Authority and the UK Expert Group on Vitamins and Minerals currently believe there is insufficient data to set an upper limit for vitamin C.<sup>22 52</sup> One reason why vitamin C may not have an effect on humans is that it is water-soluble and is readily excreted above levels of about 400mg/d.65 Being water-soluble, though, does not preclude micronutrients from being toxic as there have been recommendations that other soluble micronutrients, such as folic acid, should be re-categorised as over-the-counter medicine because of concerns over toxicity.<sup>385</sup> Vitamins are currently sold as dietary supplements where the onus in the US has been on the food authorities to prove supplements are unsafe, rather than manufacturers needing to provide evidence of their benefits and needing to obtain licences to sell them like drugs.<sup>385</sup> In addition however, in Europe the EFSA has recently implemented regulations for approving or rejecting general health claims made on foods and supplements based on whether these claims are independently verified by research studies.<sup>21</sup>

Indeed, as seen in chapters 8 and 10, no protective effects of taking high dose vitamin C, or any dose of vitamin C, were found in terms of breast cancer risk, which in general supports previous research from cohort studies discussed in chapter 8. Results, therefore, do not support the use of vitamin C supplements for reducing breast cancer risk. As reported in chapter 2 and chapter 6, as many as a third of UKWCS women used supplements containing vitamin C, and about 10% of women frequently took high dose vitamin C. Furthermore, chapter 6 reported that women with a family history of

breast cancer or personal history of breast cancer in the UK were more likely to frequently take high doses of 1000mg/d or above.<sup>1</sup> Although the reasons these women took high dose supplements is unknown, it is possible they hoped they may reduce breast cancer risk. It appears, therefore, some women may be spending money and consuming supplements unnecessarily. Indeed, antioxidant supplements are sold in a multimillion pound worldwide industry. Although there are regulations to prevent specific health claims being stated explicitly, antioxidants are sold on the basis they reduce free radicals, which is widely believed, by the general public, to play a role in preventing chronic diseases including cancer. However, some researchers doubt antioxidants can provide this role and believe claims relating to them may be overstated.<sup>386</sup> Indeed, the EFSA has recently reported that no evidence has been provided to them to establish that having antioxidant activity/content and/or antioxidant properties exerts beneficial physiological effects on humans.<sup>46</sup> Any future restrictions on EU selling, however, would not prevent supplements being purchased outside the EU by UK residents via mail order.

The 2007 WCRF report, nevertheless, judged there was convincing or probable evidence from prospective studies that some antioxidants are associated with some cancers (Appendix B).<sup>7</sup> For instance, the antioxidant selenium was associated with a reduced risk of lung, stomach and prostate cancer,<sup>7</sup> although more recent analyses from the SELECT trial do not support the latter.<sup>237 387</sup> Furthermore, supplementing with low doses of antioxidants appears to reduce cancer risk or cancer mortality in people who may have low antioxidant status,<sup>17</sup> for instance men,<sup>256</sup> and people researched in remote parts of China who may consume insufficient nutrients via their diet.<sup>388</sup> Importantly, however, there has been convincing evidence that supplementation with the antioxidant  $\beta$ -carotene increases the risk of lung cancer in smokers.<sup>238</sup>

As Bjekelakovic has pointed out, dietary supplementation has moved from preventing deficiencies to being used in attempts to promote wellness and prevent disease. He maintains consumers are being taking advantage of and are not getting value for money since supplementation cannot be recommended for preventative measures in well-nourished populations.<sup>389</sup> Using a peer-reviewed Cochrane protocol, a systematic review and meta-analysis by Bjelakovic et al. (2007) of low-bias, primary and secondary prevention randomised trials of antioxidant supplements (published to October 2005) concluded that antioxidant supplements carotene and vitamin E, and also vitamin A may increase all-cause mortality.<sup>351</sup> Vitamin C and selenium were found to have no effect on mortality. Several researchers, nevertheless, have questioned the Bjelakovic et al. (2007) results, for instance, because 405 out of 815 trials reviewed were excluded since no deaths occurred.<sup>390</sup>

249

The question remains: Should vitamins be considered as drugs?<sup>385</sup> The evidence from this thesis for vitamin C does not support this consideration; however, the results indicate that more research evidence is needed on pre-menopausal supplement users to help answer this question. The current evidence, nevertheless, does not justify the widespread use of vitamin C supplements to reduce breast cancer risk. Indeed, the WCRF does not recommend supplements to prevent cancer and in 2009 issued a press release warning on high strength supplements in general.<sup>362</sup> Whether the advice has been heard and acted upon is, as yet, unknown.

The WCRF reported that a 2009 YouGov survey found media reporting of conflicting studies appears to have caused many people to ignore dietary advice relating to cancer risks, with 60% of people over the age of 55 years being cynical about dietary advice.<sup>15</sup> Consequently, media reporting of more important results from reviews, metaanalysis and large pooling studies are likely to be overlooked by the general public, in the midst of reports of many smaller less important results. Therefore, there may be a need for greater control over the types and amount of reporting by journalists, who may need some education in epidemiology. Additionally, academics should be encouraged to disseminate important work in popular press. Given the often inappropriate affect of the media, more government controls may be needed relating to the pricing, availability and production of food and supplements, in order to reduce the risk of cancer in general. Policy recommendations such as these have been detailed by the WCRF in their policy report.<sup>391</sup>

## **12 References**

- 1. Hutchinson J, Burley VJ, Greenwood DC, Thomas JD, Cade JE. High-dose vitamin C supplement use is associated with self-reported histories of breast cancer and other illnesses in the UK Women's Cohort Study. *Public Health Nutr* 2011;14(05):768-777.
- Hutchinson J, Lentjes M, Greenwood D, Burley V, Cade J, Cleghorn C, et al. Vitamin C intake from diary recordings and risk of breast cancer in the UK Dietary Cohort Consortium. *Eur J Clin Nutr* 2012;66(5):561-568.
- 3. Ferlay J, Parkin DM, Steliarova-Foucher E. Estimates of cancer incidence and mortality in Europe in 2008. *Eur J Cancer* 2010;46(4):765-781.
- Ferlay J, Shin H-R, Bray F, Forman D, Mathers C, Parkin DM. Estimates of worldwide burden of cancer in 2008: GLOBOCAN 2008. Int J Cancer 2010;127(12):2893-2917.
- 5. Office for National Statistics. Registrations of cancer diagnosed in 1994-1997, England & Wales. *Health Stat Quart* 2000;07:71-82.
- 6. Cancer Research UK. Breast cancer UK incidence statistics [Internet]. London: Cancer Research UK:[cited 2011 June]. Available from: http://info.cancerresearchuk.org/cancerstats/types/breast/incidence/#source5.
- 7. WCRF/AICR. Food, nutrition, physical activity, and the prevention of cancer: a global perspective. Washington, DC: AICR, 2007.
- 8. WCRF/AICR. Food, nutrition and the prevention of cancer: a global perspective. Washington, DC: AICR, 1997.
- World Health Organization. Diet, nutrition and the prevention of chronic diseases: report of a joint WHO/FOA expert consultation. *Technical report series No 916*: WHO, 2003.
- 10. Key TJ. Fruit and vegetables and cancer risk. Br J Cancer 2010;104(1):6-11.
- 11. Michels KB, Mohllajee AP, Roset-Bahmanyar E, Beehler GP, Moysich KB. Diet and breast cancer. *Cancer* 2007;109(S12):2712-2749.
- 12. Smith-Warner SA, Spiegelman D, Yaun S-S, Adami H-O, Beeson WL, van den Brandt PA, et al. Intake of fruits and vegetables and risk of breast cancer: a pooled analysis of cohort studies. *JAMA*. 2001;285(6):769-776.
- 13. van Gils CH, Peeters PHM, Bueno-de-Mesquita HB, Boshuizen HC, Lahmann PH, Clavel-Chapelon F, et al. Consumption of vegetables and fruits and risk of breast cancer. *JAMA* 2005;293(2):183-193.
- 14. Lin J, Cook NR, Albert C, Zaharris E, Gaziano JM, Van Denburgh M, et al. Vitamins C and E and beta carotene supplementation and cancer risk: a randomized controlled trial. *J Natl Cancer Inst* 2009;101(1):14-23.
- 15. WCRF. Scientists "always changing their minds" on cancer [internet]. London: WCRF:[Updated 2009 May 25; cited 2009 May]. Available from: http://www.wcrf-uk.org/audience/media/press\_release.php?recid=63.
- 16. Fisher B, Dowding D, Pickett KE, Fylan F. Health promotion at NHS breast cancer screening clinics in the UK. *Health Promot Int* 2007;22(2):137-145.
- 17. Hercberg S, Czernichow S, Galan P. Antioxidant vitamins and minerals in prevention of cancers: lessons from the SU.VI.MAX study. *Br J Nutr* 2006(96):S28-S30.
- 18. DaVanzo JE, Heath S, El-Gamil A, Dobson A. The economic contribution of the dietary supplement industry. Analyses of the economic benefits to the US economy. Washington: Natural Products foundation, 2009.
- 19. Food and Drug Administration. Overview of Dietary Supplements [Internet]. Silver Spring: FDA:[Updated 2009 Oct 19; cited 2011 Aug]. Available from

http://www.fda.gov/Food/DietarySupplements/ConsumerInformation/ucm110417.htm

- 20. European Parliament. Directive 2002/46/EC of the European Parliament relating to food supplements. : European Commission, 2002.
- 21. European Food Safety Authority. "General function" health claims under Article 13 [Internet]: EFSA:[updated 2011 Oct 31; cited 2011 Nov]. Available from http://www.efsa.europa.eu/en/ndaclaims/ndaclaims13.htm.
- 22. Scientific Committee on Food. Tolerable upper intake levels for vitamins and minerals. Parma: EFSA, 2006.
- 23. Mintel. Vitamins and Supplements UK May 2009 [Internet]. London: Mintel Group Ltd:[cited 2011 Aug]. Available from http://oxygen.mintel.com/sinatra/oxygen/display/id=393960.
- 24. Hoare J, Henderson L, Bates C, Prentice A, Birch M, Swan G, et al. The National Diet & Nutrition Survey: adults aged 19 to 64 years. Summary report. London: HMSO, 2004.
- 25. Bates B, Lennox A, Swan G. The National Diet & Nutrition Survey: Headline results from Year 1 of the Rolling Programme (2008/2009). London: FSA, 2009
- 26. GfK Social Research. Consumer consumption of vitamin and mineral food supplements: UK Random Location Omnibus Survey 2008, 2009.
- 27. Kirk SFL, Cade JE, Barrett JH, Conner M. Diet and lifestyle characteristics associated with dietary supplement use in women. *Public Health Nutr* 1999;2(01):69-73.
- 28. Harrison RA, Holt D, Pattison DJ, Elton PJ. Are those in need taking dietary supplements? A survey of 21,923 adults. *Br J Nutr* 2004;91(04):617-623.
- 29. McNaughton SA, Mishra GD, Paul AA, Prynne CJ, Wadsworth MEJ. Supplement use is associated with health status and health-related behaviors in the 1946 British birth cohort. *J Nutr* 2005;135(7):1782-1789.
- 30. Brownie S. Characteristics of older dietary supplement users: review of the literature. *Australas J Ageing* 2005;24(2):77-87.
- 31. Frank E, Bendich A, Denniston M. Use of vitamin-mineral supplements by female physicians in the United States. *Am J Clin Nutr* 2000;72(4):969-975.
- Lyle BJ, Mares-Perlman JA, Klein BEK, Klein R, Greger JL. Supplement users differ from nonusers in demographic, lifestyle, dietary and health characteristics. *J Nutr* 1998;128(12):2355-2362.
- 33. Patterson RE, Neuhouser ML, White E, Hunt JR, Kristal AR. Cancer-related behavior of vitamin supplement users. *Cancer Epidemiol Biomarkers Prev* 1998;7(1):79-81.
- 34. Reinert A, Rohrmann S, Becker N, Linseisen J. Lifestyle and diet in people using dietary supplements. A German cohort study. *Eur J Nutr* 2007;46(3):165-173.
- 35. Shikany JM, Patterson RE, Agurs-Collins T, Anderson G. Antioxidant supplement use in Women's Health Initiative participants. *Prev Med* 2003;36(3):379-387.
- 36. Ajzen I. The Theory of Planned Behaviour. Organ Behav Hum Dec 1991;50:179-211.
- Conner M, Kirk SFL, Cade JE, Barrett JH. Why do women use dietary supplements? The use of the Theory of Planned Behaviour to explore beliefs about their use. Soc Sci Med 2001;52(4):621-633.
- 38. Lentjes MA, Bhaniani A, Mulligan AA, Khaw K-T, Welch AA. Developing a database of vitamin and mineral supplements (ViMiS) for the Norfolk arm of the European Prospective Investigation into Cancer (EPIC-Norfolk) *Public Health Nutr* 2011;14(3):459-471.
- 39. Schlueter AK, Johnston CS. Vitamin C: overview and update. *Journal of Evidence-Based Complementary & Alternative Medicine* 2011;16(1):49-57.

- 40. Davies M, Partridge D, Austin J. Vitamin C Its Chemistry and Biochemistry. Cambridge: The Royal Society of Chemistry, 1991.
- 41. Bender D. *Nutritional Biochemistry of the vitamins* 2nd ed. Cambridge: University Press, 2003.
- 42. Naidu KA. Vitamin C in human health and disease is still a mystery? An overview. *Nutr J* 2003;2:7.
- 43. Sato K, Niki E, Shimasaki H. Free radical-mediated chain oxidation of low density lipoprotein and its synergistic inhibition by vitamin E and vitamin C. *Arch Biochem Biophys* 1990;279(2):402-405.
- 44. Bendich A, Cohen M. Ascorbic acid safety: analysis of factors affecting iron absorption. *Toxicol Lett* 1990;51(2):189-201.
- 45. Hemilä H, Chalker E, Douglas B. Vitamin C for preventing and treating the common cold. *Vitamin C for preventing and treating the common cold. Cochrane Database of Systematic Reviews: Reviews* 2007(3).
- 46. EFSA Panel on Dietetic Products Nutrition and Allergies (NDA). Scientific opinion on the substantiation of health claims in relation to various food(s)/food constituent(s) and protection of cells from premature ageing, antioxidant activity, antioxidant content and antioxidant properties, protection of DNA, proteins and lipids from oxidative damage, and bioavailability of anthocyanins in black currants *EFSA J* 2010;8(10):1752.
- 47. Food and Nutrition Board and Panel on Dietary Antioxidants and Related Compounds. Dietary reference intakes for vitamin C, vitamin E, selenium, and carotenoids. Washington, DC: National Academy Press, 2000:95-185.
- 48. Brubacher, Moser, Jordan. Vitamin C concentrations in plasma as a function of intake: a meta-analysis. *Int J Vitam Nutr Res* 2000;70(5):226-237.
- 49. Carr AC, Frei B. Toward a new recommended dietary allowance for vitamin C based on antioxidant and health effects in humans. *Am J Clin Nutr* 1999;69:1086 1107.
- 50. Biesalski HK, Bohles H, Esterbauer H, Furst P, Gey F, Hundsdorfer G, et al. Antioxidant vitamins in prevention. *Clin Nutr* 1997;16(3):151-155.
- 51. COMA Committee on Medical Aspects of Food and Nutrition Policy. Dietary Reference Values for food energy and nutrients for the United Kingdom. Report of the panel on Dietary Reference Values London: HMSO, 1991.
- 52. Expert Group on Vitamins and Minerals. Safe upper levels for vitamins and minerals [Internet], 2003:[cited 2009 August]. Available from: http://cot.food.gov.uk/pdfs/vitmin2003.pdf
- 53. The European Food Information Council. Vitamins: what they do and where to find them [Internet]. Brussels: EUFIC:[Updated 2011 Nov 17; cited 2011 Nov]. Available from: http://www.eufic.org/article/en/page/MARCHIVE/expid/miniguidevitamins/#10
- 54. Sauberlich H. Bioavailability of vitamins. Prog Food Nutr Sci 1985;9:1-33.
- 55. Hathcock JN, Azzi A, Blumberg J, Bray T, Dickinson A, Frei B, et al. Vitamins E and C are safe across a broad range of intakes. *Am J Clin Nutr* 2005;81(4):736-745.
- 56. Alhadeff LC, Gualtieri T, Upton M. Toxic Effects of Water-Soluble Vitamins. *Nutr Rev* 1984;42(2):33-40.
- 57. Johnston CS. Biomarkers for Establishing a Tolerable Upper Intake Level for Vitamin C. *Nutr Rev* 1999;57(3):71-77.
- 58. EFSA. Opinion of the Scientific Panel on Dietetic Products, Nutrition and Allergies on a request from the Commission related to the Tolerable Upper Intake Level

of Vitamin C (L-Ascorbic acid, its calcium, potassium and sodium salts and L-ascorbyl-6-palmitate). *EFSA J* 2004;59:1-21.

- 59. Mangels AR, Block G, Frey CM, Patterson BH, Taylor PR, Norkus EP, et al. The bioavailability to humans of ascorbic acid from oranges, orange juice and cooked broccoli is similar to that of synthetic ascorbic acid. *J Nutr* 1993;123(6):1054-1061.
- 60. Rumsey SC, Levine M. Absorption, transport, and disposition of ascorbic acid in humans. *J Nutr Biochem* 1998;9(3):116-130.
- Levine M, Conry-Cantilena C, Wang Y, Welch RW, Washko PW, Dhariwal KR, et al. Vitamin C pharmacokinetics in healthy volunteers: evidence for a recommended dietary allowance. *Proc Natl Acad Sci USA* 1996;93(8):3704-3709.
- 62. Kallner A, Hartmann D, Hornig D. Steady-state turnover and body pool of ascorbic acid in man. *Am J Clin Nutr* 1979;32(3):530-539.
- 63. Rivers JM. Safety of High-level Vitamin C Ingestion. *Ann NY Acad Sci* 1987;498(Third Conference on Vitamin C):445-454.
- 64. Hornig D. Distribution of ascorbic acid, metabolites and analogues in man and animals *Ann NY Acad Sci* 1975;258(Second Conference on Vitamin C):103-118.
- 65. Levine M, Wang Y, Padayatty SJ, Morrow J. A new recommended dietary allowance of vitamin C for healthy young women. *Proc Natl Acad Sci USA* 2001;98(17):9842-9846.
- 66. Bates CJ, Thurnham DI, Bingham SA, Margetts BM, Nelson M. *Biochemical* markers of nutrient intake. In: Margetts BM, Nelson M (eds). Design Concepts in Nutritional Epidemiology 2nd ed. Oxford: Oxford University Press, 1997.
- 67. Khaw K-T, Bingham S, Welch A, Luben R, Wareham N, Oakes S, et al. Relation between plasma ascorbic acid and mortality in men and women in EPIC-Norfolk prospective study: a prospective population study. *Lancet* 2001;357(9257):657-663.
- 68. Canoy D, Wareham N, Welch A, Bingham S, Luben R, Day N, et al. Plasma ascorbic acid concentrations and fat distribution in 19 068 British men and women in the European Prospective Investigation into Cancer and Nutrition Norfolk cohort study. *Am J Clin Nutr* 2005;82(6):1203-1209.
- 69. Sargeant LA, Wareham NJ, Bingham S, Day NE, Luben RN, Oakes S, et al. Vitamin C and hyperglycemia in the European Prospective Investigation into Cancer--Norfolk (EPIC-Norfolk) study: a population-based study. *Diabetes Care* 2000;23(6):726-732.
- 70. Wilson JX. Regulation of vitamin C transport. Ann Rev Nutr 2005;25(1):105-125.
- 71. Wannamethee SG, Lowe GDO, Rumley A, Bruckdorfer KR, Whincup PH. Associations of vitamin C status, fruit and vegetable intakes, and markers of inflammation and hemostasis. *Am J Clin Nutr* 2006;83(3):567-574.
- 72. Jialal I, Singh U. Is vitamin C an antiinflammatory agent? *Am J Clin Nutr* 2006;83(3):525-526.
- 73. Lawlor DA, Smith GD, Bruckdorfer KR, Kundu D, Ebrahim S. Those confounded vitamins: what can we learn from the differences between observational versus randomised trial evidence? *Lancet* 2004;363(9422):1724-1727.
- 74. Willcox JK, Ash SL, Catignani GL. Antioxidants and prevention of chronic disease. *Crit Rev Food Sci Nutr* 2004;44(4):275-295.
- 75. Albright CD, Salganik RI, Craciunescu CN, Mar M-H, Zeisel SH. Mitochondrial and microsomal derived reactive oxygen species mediate apoptosis induced by transforming growth factor-beta1 in immortalized rat hepatocytes. *J Cell Biochem* 2003;89(2):254-261.

- 76. Valko M, Rhodes CJ, Moncol J, Izakovic M, Mazur M. Free radicals, metals and antioxidants in oxidative stress-induced cancer. *Chemico-Biol Interact* 2006;160(1):1-40.
- 77. Salganik RI. The benefits and hazards of antioxidants: Controlling apoptosis and other protective mechanisms in cancer patients and the human population. *J Am Coll Nutr* 2001;20(90005):464S-476S.
- 78. Halliwell B. Vitamin C and genomic stability. *Mutat Res-Fund Mol M* 2001;475(1-2):29-35.
- 79. Huang H-Y, Helzlsouer KJ, Appel LJ. The effects of vitamin C and vitamin E on oxidative DNA damage: results from a randomized controlled trial. *Cancer Epidemiol Biomarkers Prev* 2000;9(7):647-652.
- 80. Shen J, Gammon MD, Terry MB, Wang Q, Bradshaw P, Teitelbaum SL, et al. Telomere length, oxidative damage, antioxidants and breast cancer risk. *Int J Cancer* 2009;124(7):1637-43.
- 81. Salganik RI, Albright CD, Rodgers J, Kim J, Zeisel SH, Sivashinskiy MS, et al. Dietary antioxidant depletion: enhancement of tumor apoptosis and inhibition of brain tumor growth in transgenic mice. *Carcinogenesis* 2000;21(5):909-914.
- Taylor EF, Burley VJ, Greenwood DC, Cade JE. Meat consumption and risk of breast cancer in the UK Women's Cohort Study. *Br J Cancer* 2007;96(7):1139-1146.
- 83. Albanes D, Heinonen OP, Taylor PR, Virtamo J, Edwards BK, Rautalahti M, et al. Alpha-tocopherol and beta-carotene supplements and lung cancer incidence in the alpha-tocopherol, beta-carotene cancer prevention study: effects of base-line characteristics and study compliance. J Natl Cancer Inst 1996;88(21):1560-1570.
- 84. Li Y, Schellhorn HE. New developments and novel therapeutic perspectives for vitamin C. *J Nutr* 2007;137(10):2171-2184.
- 85. Catani MV, Costanzo A, Savini I, Levrero M, de Laurenzi V, Wang JYJ, et al. Ascorbate up-regulates MLH1 (Mut L homologue-1) and p73: implications for the cellular response to DNA damage. *Biochem J* 2002;364(2):441-447.
- 86. Block KI, Koch AC, Mead MN, Tothy PK, Newman RA, Gyllenhaal C. Impact of antioxidant supplementation on chemotherapeutic efficacy: a systematic review of the evidence from randomized controlled trials. *Cancer Treat Rev* 2007;33(5):407-418.
- 87. Cameron E, Pauling L. Supplemental ascorbate in the supportive treatment of cancer: Reevaluation of prolongation of survival times in terminal human cancer. *Proc Natl Acad Sci USA* 1978;75:4538-4542.
- 88. Creagan ET, Moertel CG, O'Fallon JR, Schutt AJ, O'Connell MJ, Rubin J, et al. Failure of high-dose vitamin C (ascorbic acid) therapy to benefit patients with advanced cancer. A controlled trial. *N Engl J Med* 1979;301(13):687-690.
- 89. Moertel CG, Fleming TR, Creagan ET, Rubin J, O'Connell MJ, Ames MM. Highdose vitamin C versus placebo in the treatment of patients with advanced cancer who have had no prior chemotherapy. A randomized double-blind comparison. *N Engl J Med* 1985;312(3):137-141.
- 90. Tappel A. Heme of consumed red meat can act as a catalyst of oxidative damage and could initiate colon, breast and prostate cancers, heart disease and other diseases. *Medical Hypotheses* 2007;68(3):562-564.
- 91. Lee D-H, Jacobs DR. Interaction among heme iron, zinc, and supplemental vitamin C intake on the risk of lung cancer: Iowa Women's Health Study. *Nutr Cancer* 2005;52(2):130 - 137.
- 92. Kabat G, Rohan T. Does excess iron play a role in breast carcinogenesis? an unresolved hypothesis. *Cancer Causes Control* 2007;18(10):1047-1053.

- 93. Bhat SH, Azmi AS, Hanif S, Hadi SM. Ascorbic acid mobilizes endogenous copper in human peripheral lymphocytes leading to oxidative DNA breakage: A putative mechanism for anticancer properties. *Int J Biochem Cell B* 2006;38(12):2074-2081.
- 94. Singh VN, Gaby SK. Premalignant lesions: role of antioxidant vitamins and betacarotene in risk reduction and prevention of malignant transformation. *Am J Clin Nutr* 1991;53(1):386S-390.
- 95. Jenab M, Riboli E, Ferrari P, Sabate J, Slimani N, Norat T, et al. Plasma and dietary vitamin C levels and risk of gastric cancer in the European Prospective Investigation into Cancer and Nutrition (EPIC-EURGAST). *Carcinogenesis* 2006;27(11):2250-2257.
- Zhang ZW, Abdullahi M, Farthing MJG. Effect of physiological concentrations of vitamin C on gastric cancer cells and Helicobacter pylori. *Gut* 2002;50(2):165-169.
- 97. Bjelakovic G, Nikolova D, Simonetti Rosa G, Gluud C. Antioxidant supplements for preventing gastrointestinal cancers. *Cochrane Database of Systematic Reviews: Reviews* 2008(3).
- Potischman N, Troisi R, Vatten L. A Life Course Approach to Chronic Disease Epidemiology. In: Kuh D, Ben-Shlomo Y, editors. 2nd ed. Oxford: Oxford University Press, 2004.
- 99. Hilakivi-Clarke L, de Assis S. Fetal origins of breast cancer. *Trends Endocrin Met* 2006;17(9):340-348.
- 100. Painter RC, De Rooij SR, Bossuyt PMM, Osmond C, Barker DJP, Bleker OP, et al. A possible link between prenatal exposure to famine and breast cancer: A preliminary study. *Am J Hum Biol* 2006;18(6):853-856.
- 101. Finaud J, Lac G, Filaire E. Oxidative stress: relationship with exercise and training. *Sports Med* 2006;36(4):327-358.
- 102. Rolo AP, Palmeira CM. Diabetes and mitochondrial function: Role of hyperglycemia and oxidative stress. *Toxicol Appl Pharmacol* 2006;212(2):167-178.
- 103. Herceg Z. Epigenetics and cancer: towards an evaluation of the impact of environmental and dietary factors. *Mutagenesis* 2007;22(2):91-103.
- 104. Fraga MF, Ballestar E, Paz MF, Ropero S, Setien F, Ballestar ML, et al. Epigenetic differences arise during the lifetime of monozygotic twins. *Proc Natl Acad of Sci USA* 2005;102(30):10604-10609.
- 105. Dunn GA, Bale TL. Maternal high-fat diet promotes body length increases and insulin insensitivity in second-generation mice. *Endocrinol* 2009;150(11):4999-5009.
- 106. Lakhani SR, Audretsch W, Cleton-Jensen A-M, Cutuli B, Ellis I, Eusebi V, et al. The management of lobular carcinoma in situ (LCIS). Is LCIS the same as ductal carcinoma in situ (DCIS)? *Eur J Cancer* 2006;42(14):2205-2211.
- 107. Cancer Research UK. Invasive ductal breast cancer [Internet]. London: Cancer Research UK:[cited 2011 Jun]. Available from: http://cancerhelp.cancerresearchuk.org/type/breast-cancer/about/types/invasiveductal-breast-cancer
- 108. Singletary SE, Allred C, Ashley P, Bassett LW, Berry D, Bland KI, et al. Revision of the American Joint Committee on Cancer Staging System for Breast Cancer. *J Clin Oncol* 2002;20(17):3628-3636.
- 109. Cancer Research UK. Cancer Statistics breast cancer [Internet]. London: Cancer Research UK:[cited 2011 Jun]. Available from: http://info.cancerresearchuk.org/cancerstats/types/breast/

- 110. Brewster DH, Sharpe KH, Clark DI, Collins J. Declining breast cancer incidence and decreased HRT use. *Lancet* 2009;373(9662):459-460.
- 111. Beral V, Banks E, Reeves G. Evidence from randomised trials on the long-term effects of hormone replacement therapy. *Lancet* 2002;360(9337):942-944.
- 112. Coleman MP, Quaresma M, Berrino F, Lutz J-M, De Angelis R, Capocaccia R, et al. Cancer survival in five continents: a worldwide population-based study (CONCORD). *Lancet Oncol* 2008;9(8):730-756.
- 113. Sant M, Allemani C, Capocaccia R, Hakulinen T, Aareleid T, Coebergh JW, et al. Stage at diagnosis is a key explanation of differences in breast cancer survival across Europe. *Int J Cancer* 2003;106(3):416-422.
- 114. Collaborative Group on Hormonal Factors in Breast Cancer. Familial breast cancer: collaborative reanalysis of individual data from 52 epidemiological studies including 58 209 women with breast cancer and 101 986 women without the disease. *Lancet* 2001;358(9291):1389-1399.
- 115. Stratton MR, Rahman N. The emerging landscape of breast cancer susceptibility. *Nat Genet* 2008;40(1):17-22.
- 116. Risch HA, McLaughlin JR, Cole DEC, Rosen B, Bradley L, Fan I, et al. Population BRCA1 and BRCA2 mutation frequencies and cancer penetrances: A kin-cohort study in Ontario, Canada. *J Natl Cancer Inst* 2006;98(23):1694-1706.
- 117. McCormack VA, dos Santos Silva I. Breast density and parenchymal patterns as markers of breast cancer Rrsk: A meta-analysis. *Cancer Epidemiol Biomarkers Prev* 2006;15(6):1159-1169.
- 118. Boyd NF, Lockwood GA, Byng JW, Tritchler DL, Yaffe MJ. Mammographic densities and breast cancer risk. *Cancer Epidemiol Biomarkers Prev* 1998;7(12):1133-1144.
- 119. Vachon CM, Kushi LH, Cerhan JR, Kuni CC, Sellers TA. Association of diet and mammographic breast density in the Minnesota Breast Cancer Family Cohort. *Cancer Epidemiol Biomarkers Prev* 2000;9(2):151-160.
- 120. Masala G, Ambrogetti D, Assedi M, Giorgi D, Del Turco MR, Palli D. Dietary and lifestyle determinants of mammographic breast density. A longitudinal study in a Mediterranean population. *Int J Cancer* 2006;118(7):1782-1789.
- 121. Martin LJ, Boyd NF. Mammographic density. Potential mechanisms of breast cancer risk associated with mammographic density: hypotheses based on epidemiological evidence. *Breast Cancer Res* 2008;10(1):201.
- 122. Mitchell G, Antoniou AC, Warren R, Peock S, Brown J, Davies R, et al. Mammographic density and breast cancer risk in BRCA1 and BRCA2 mutation carriers. *Cancer Res* 2006;66(3):1866-1872.
- 123. Vachon C, van Gils C, Sellers T, Ghosh K, Pruthi S, Brandt K, et al. Mammographic density, breast cancer risk and risk prediction. *Breast Cancer Res* 2007;9(6):217.
- 124. Hartmann LC, Sellers TA, Frost MH, Lingle WL, Degnim AC, Ghosh K, et al. Benign breast disease and the risk of breast cancer. *N Engl J Med* 2005;353(3):229-237.
- 125. Robinson D, Holmberg L, Moller H. The occurrence of invasive cancers following a diagnosis of breast carcinoma in situ. *Br J Cancer* 2008;99(4):611-615.
- 126. Rubino C, Arriagada R, Delaloge S, M.G. L. Relation of risk of contralateral breast cancer to the interval since the first primary tumour. *Br J Cancer* 2010;102 (1):213–219.
- 127. Henderson T, Amsterdam A, Bhatia S, Hudson MM, Meadows AT, Neglia JP, et al. Surveillance for breast cancer in women treated with chest radiation for a childhood, adolescent or young adult cancer: a report from the Children's Oncology Group. *Ann Intern Med* 2010;152(7):444.

- 128. Lundell M, Mattsson A, Karlsson P, Holmberg E, Gustafsson A, Holm L-E. Breast cancer risk after radiotherapy in infancy: a pooled analysis of two Swedish cohorts of 17,202 infants. *Radiation Res* 1999;151(5):626-632.
- 129. Jacobi C, de Bock G, Siegerink B, van Asperen C. Differences and similarities in breast cancer risk assessment models in clinical practice: which model to choose? *Breast Cancer Res Treat* 2009;115(2):381-390.
- 130. Kelsey JL, Gammon MD, John EM. Reproductive factors and breast cancer. *Epidemiologic Reviews* 1993;15(1):36-47.
- 131. Key TJ, Verkasalo PK, Banks E. Epidemiology of breast cancer. *Lancet Oncol* 2001;2(3):133-140.
- 132. Kaaks R, Rinaldi S, Key TJ, Berrino F, Peeters PHM, Biessy C, et al. Postmenopausal serum androgens, oestrogens and breast cancer risk: the European prospective investigation into cancer and nutrition. *Endocr Relat Cancer* 2005;12(4):1071-1082.
- 133. Berkey CS, Gardner JD, Lindsay Frazier A, Colditz GA. Relation of childhood diet and body size to menarche and adolescent growth in girls. *Am J Epidemiol* 2000;152(5):446-452.
- 134. Merzenich H, Boeing H, Wahrendorf Jr. Dietary fat and sports activity as determinants for age at menarche. *Am J Epidemiol* 1993;138(4):217-224.
- 135. Collaborative Group on Hormonal Factors in Breast Cancer. Breast cancer and hormone replacement therapy: collaborative reanalysis of data from 51 epidemiological studies of 52 705 women with breast cancer and 108 411 women without breast cancer. *Lancet* 1997;350(9084):1047-1059.
- 136. Million Women Study Collaborators. Breast cancer and hormone-replacement therapy in the Million Women Study. *Lancet* 2003;362(9382):419-427.
- 137. Schairer C, Lubin J, Troisi R, Sturgeon S, Brinton L, Hoover R. Menopausal estrogen and estrogen-progestin replacement therapy and breast cancer risk. *JAMA* 2000;283(4):485-491.
- 138. Conner P, Svane G, Azavedo E, Soderqvist G, Carlstrom K, Graser T, et al. Mammographic breast density, hormones, and growth factors during continuous combined hormone therapy. *Fertil Steril* 2004;81(6):1617-1623.
- 139. von Schoultz B. Androgens and the breast. Maturitas 2007;57(1):47-49.
- 140. Dimitrakakis C, Zhou J, Wang J, Belanger A, LaBrie F, Cheng C, et al. A physiologic role for testosterone in limiting estrogenic stimulation of the breast. *Menopause* 2003;10(4):292-298.
- 141. Dimitrakakis C, Jones RA, Liu A, Bondy CA. Breast cancer incidence in postmenopausal women using testosterone in addition to usual hormone therapy. *Menopause* 2004;11(5):531-535.
- 142. Tamimi RM, Hankinson SE, Chen WY, Rosner B, Colditz GA. Combined estrogen and testosterone use and risk of breast cancer in postmenopausal women. *Arch Intern Med* 2006;166(14):1483-1489.
- 143. Collaborative Group on Hormonal Factors in Breast Cancer. Breast cancer and hormonal contraceptives: collaborative reanalysis of individual data on 53 297 women with breast cancer and 100 239 women without breast cancer from 54 epidemiological studies. *Lancet* 1996;347(9017):1713-1727.
- 144. Kahlenborn C, Modugno F, Potter DM, Severs WB. Oral contraceptive use as a risk factor for premenopausal breast cancer: a meta-analysis. *Mayo Clin Proc* 2006;81(10):1290-1302.
- 145. Marchbanks PA, McDonald JA, Wilson HG, Folger SG, Mandel MG, Daling JR, et al. Oral contraceptives and the risk of breast cancer. *N Eng J Med* 2002;346(26):2025-2032.

- 146. Key TJ. Endogenous oestrogens and breast cancer risk in premenopausal and postmenopausal women. *Steroids* 2011;76(8):812-815.
- 147. Endogenous Hormones Breast Cancer Collaborative Group, Key TJ, Appleby PN, Barnes I, Reeves GK. Endogenous sex hormones and breast cancer in postmenopausal women: Reanalysis of nine prospective studies. J Natl Cancer Inst 2002;94(8):606-616.
- 148. Baglietto L, Severi G, English DR, Krishnan K, Hopper JL, McLean C, et al. Circulating steroid hormone levels and risk of breast cancer for postmenopausal women. *Cancer Epidemiol Biomarkers Prev* 2010;19(2):492-502.
- 149. Missmer SA, Eliassen AH, Barbieri RL, Hankinson SE. Endogenous estrogen, androgen, and progesterone concentrations and breast cancer risk among postmenopausal women. *J Natl Cancer Inst* 2004;96(24):1856-1865.
- 150. Yager JD, Davidson NE. Estrogen carcinogenesis in breast cancer. *N Engl J Med* 2006;354(3):270-282.
- 151. Zeleniuch-Jacquotte A, Toniolo P, Levitz M, Shore RE, Koenig KL, Banerjee S, et al. Endogenous estrogens and risk of breast cancer by estrogen receptor status: a prospective study in postmenopausal women. *Cancer Epidemiol Biomarkers Prev* 1995;4(8):857-860.
- 152. Sieri S, Krogh V, Bolelli G, Abagnato CA, Grioni S, Pala V, et al. Sex hormone levels, breast cancer risk, and cancer receptor status in postmenopausal women: the ORDET Cohort. *Cancer Epidemiol Biomarkers Prev* 2009;18(1):169-176.
- 153. Diaz-Chico BN, Rodriguez FG, Gonzalez A, Ramirez R, Bilbao C, De Leon AC, et al. Androgens and androgen receptors in breast cancer. *J Steroid Biochem Mol Biol* 2007;105(1-5):1-15.
- 154. Dimitrakakis C, Zhou J, Bondy CA. Androgens and mammary growth and neoplasia. *Fertil Steril* 2002;77:26-33.
- 155. Liao DJ, Dickson RB. Roles of androgens in the development, growth, and carcinogenesis of the mammary gland. *J Steroid Biochem Mol Biol* 2002;80(2):175-189.
- 156. Secreto G, Venturelli E, Meneghini E, Greco M, Ferraris C, Gion M, et al. Testosterone and biological characteristics of breast cancers in postmenopausal women. *Cancer Epidemiol Biomarkers Prev* 2009;18(11):2942-2948.
- 157. Zeleniuch-Jacquotte A, Shore RE, Koenig KL, Akhmedkhanov A, Afanasyeva Y, Kato I, et al. Postmenopausal levels of oestrogen, androgen, and SHBG and breast cancer: long-term results of a prospective study. *Br J Cancer* 2004;90(1):153-159.
- 158. Lorincz AM, Sukumar S. Molecular links between obesity and breast cancer. *Endocr Relat Cancer* 2006;13(2):279-292.
- 159. Tworoger SS, Eliassen AH, Sluss P, Hankinson SE. A prospective study of plasma prolactin concentrations and risk of premenopausal and postmenopausal breast cancer. *J Clin Oncol* 2007;25(12):1482-1488.
- 160. Endogenous Hormones and Breast Cancer Collaborative Group, Key TJ, Appleby PN, Reeves GK, Roddam AW. Insulin-like growth factor 1 (IGF1), IGF binding protein 3 (IGFBP3), and breast cancer risk: pooled individual data analysis of 17 prospective studies *Lancet Oncol* 2010;11:530–542.
- 161. Renehan AG, Harvie M, Howell A. Insulin-like growth factor (IGF)-I, IGF binding protein-3, and breast cancer risk: eight years on. *Endocrine-Related Cancer* 2006;13(2):273-278.
- 162. Crowe FL, Key TJ, Allen NE, Appleby PN, Roddam A, Overvad K, et al. The association between diet and serum concentrations of IGF-I, IGFBP-1, IGFBP-2, and IGFBP-3 in the European Prospective Investigation into Cancer and Nutrition. *Cancer Epidemiol Biomarker Prev* 2009;18(5):1333-1340.

- 163. Larsson SC, Mantzoros CS, Wolk A. Diabetes mellitus and risk of breast cancer: a meta-analysis. *Int J Cancer* 2007;121(4):856-862.
- 164. Neilson HK, Friedenreich CM, Brockton NT, Millikan RC. Physical activity and postmenopausal breast cancer: proposed biologic mechanisms and areas for future research. *Cancer Epidemiol Biomarkers Prev* 2009;18(1):11-27.
- 165. Lawlor DA, Smith GD, Ebrahim S. Hyperinsulinaemia and increased risk of breast cancer: findings from the British Women's Heart and Health Study. *Cancer Causes Control* 2004;15(3):267-275.
- 166. Megdal SP, Kroenke CH, Laden F, Pukkala E, Schernhammer ES. Night work and breast cancer risk: a systematic review and meta-analysis. *Eur J Cancer* 2005;41(13):2023-2032.
- 167. Collaborative Group on Hormonal Factors in Breast Cancer. Breast cancer and breastfeeding: collaborative reanalysis of individual data from 47 epidemiological studies in 30 countries, including 50,302 women with breast cancer and 96, 973 women without the disease. *Lancet* 2002;360(9328):187-195.
- 168. Lord SJ, Bernstein L, Johnson KA, Malone KE, McDonald JA, Marchbanks PA, et al. Breast cancer risk and hormone receptor status in older women by parity, age of first birth, and breastfeeding: a case-control study. *Cancer Epidemiol Biomarkers Prev* 2008;17(7):1723-1730.
- 169. WCRF/AICR. Systematic literature review continuous update report. The associations between food, nutrition and physical activity and the risk of breast cancer: WCRF/AICR, 2008.
- 170. Berclaz G, Li S, Price KN, Coates AS, Castiglione-Gertsch M, Rudenstam CM, et al. Body mass index as a prognostic feature in operable breast cancer: the International Breast Cancer Study Group experience. *Ann Oncol* 2004;15(6):875-884.
- 171. Calle EE, Rodriguez C, Walker-Thurmond K, Thun MJ. Overweight, obesity, and mortality from cancer in a prospectively studied cohort of US adults. *N Engl J Med* 2003;348(17):1625-1638.
- 172. Christou NV, Lieberman M, Sampalis F, Sampalis JS. Bariatric surgery reduces cancer risk in morbidly obese patients. *Surg Obes Relat Dis* 2008;4(6):691-695.
- 173. Carpenter CL, Ross RK, Paganini-Hill A, Bernstein L. Effect of family history, obesity and exercise on breast cancer risk among postmenopausal women. *Int J Cancer* 2003;106(1):96-102.
- 174. Endogenous Hormones Breast Cancer Collaborative Group. Body mass index, serum sex hormones, and breast cancer risk in postmenopausal women. *J Natl Cancer Inst* 2003;95(16):1218-1226.
- 175. Stephenson GD, Rose DP. Breast cancer and obesity: an update. *Nutr Cancer* 2003;45(1):1 16.
- 176. Ahn J, Schatzkin A, Lacey JV, Jr., Albanes D, Ballard-Barbash R, Adams KF, et al. Adiposity, adult weight change, and postmenopausal breast cancer risk. *Arch Intern Med* 2007;167(19):2091-2102.
- 177. Tehard B, Clavel-Chapelon F. Several anthropometric measurements and breast cancer risk: results of the E3N cohort study. *Int J Obes Relat Metab Disord* 2005;30(1):156-163.
- 178. Gibson L, Lawrence D, Dawson C, Bliss J. Aromatase inhibitors for treatment of advanced breast cancer in postmenopausal women. *Cochrane Database of Systematic Reviews: Reviews*, 2009.
- 179. Kaaks R. Nutrition, hormones, and breast cancer: is insulin the missing link? *Cancer Causes Control* 1996;7(6):605-625.

- 180. Pugeat M, Crave JC, Elmidani M, Nicolas MH, Garoscio-Cholet M, Lejeune H, et al. Pathophysiology of sex hormone binding globulin (SHBG): Relation to insulin. J Steroid Biochem Mol Biol 1991;40(4-6):841-849.
- 181. Allen NE, Appleby PN, Kaaks R, Rinaldi S, Davey GK, Key TJ. Lifestyle determinants of serum insulin-like growth-factor-I (IGF-I), C-peptide and hormone binding protein levels in British women. *Cancer Causes Control* 2003;14(1):65-74.
- 182. Aggarwal BB, Shishodia S, Sandur SK, Pandey MK, Sethi G. Inflammation and cancer: How hot is the link? *Biochem Pharmacol* 2006;72(11):1605-1621.
- 183. Barb D, Williams CJ, Neuwirth AK, Mantzoros CS. Adiponectin in relation to malignancies: a review of existing basic research and clinical evidence. Am J Clin Nutr 2007;86(3):858S-866S.
- 184. Pérez-Martin A, Raynaud E, Mercier J. Insulin resistance and associated metabolic abnormalities in muscle: effects of exercise. *Obes Rev* 2001;2(1):47-59.
- 185. Michels KB, Xue F. Role of birthweight in the etiology of breast cancer. *Int J Cancer* 2006;119(9):2007-2025.
- 186. Trichopoulos D. Hypothesis: does breast cancer originate in utero? *Lancet* 1990;335(8695):939-940.
- 187. Kaijser M, Granath F, Jacobsen G, Cnattingius s, Ekbom A. Maternal pregnancy estriol levels in relation to anamnestic and fetal anthropometric data. *Epidemiol* 2000;11:315-9.
- 188. Petridou E, Panagiotopoulou K, Katsouyanni K, Spanos E, Trichopoulos D. Tobacco smoking, pregnancy estrogens and birth weight. *Epidemiol* 1990;1:247–50.
- 189. Boyne MS, Thame M, Bennett FI, Osmond C, Miell JP, Forrester TE. The relationship among circulating Insulin-Like Growth Factor (IGF)-I, IGF-Binding proteins-1 and -2, and birth anthropometry: a prospective study. *J Clin Endocrinol Metab* 2003;88(4):1687-1691.
- 190. Troisi R, Potischman N, Hoover RN. Exploring the underlying hormonal mechanisms of prenatal risk factors for breast cancer: a review and commentary. *Cancer Epidemiol Biomarkers Prev* 2007;16(9):1700-1712.
- 191. Ruder E, Dorgan J, Kranz S, Kris-Etherton P, Hartman T. Examining breast cancer growth and lifestyle risk factors: early life, childhood, and adolescence. *Clin Breast Cancer* 2008;8(4):334-342.
- 192. van den Brandt PA, Spiegelman D, Yaun S-S, Adami H-O, Beeson L, Folsom AR, et al. Pooled analysis of prospective cohort studies on height, weight, and breast cancer risk. *Am J Epidemiol* 2000;152(6):514-527.
- 193. Adair LS. Size at birth predicts age at menarche. *Pediatrics* 2001;107(4):e59.
- 194. Baer HJ, Tworoger SS, Hankinson SE, Willett WC. Body fatness at young ages and risk of breast cancer throughout life. *Am J Epidemiol* 2010;171(11):1183-1194.
- 195. Hilakivi-Clarke L, Forsen T, Eriksson JG, Luoto R, Tuomilehto J, Osmond C, et al. Tallness and overweight during childhood have opposing effects on breast cancer risk. *Br J Cancer* 2001;85(11):1680-1684.
- 196. Ziegler RG, Hoover RN, Pike MC, Hildesheim A, Nomura AMY, West DW, et al. Migration patterns and breast cancer risk in Asian-American women. *J Natl Cancer Inst* 1993;85(22):1819-1827.
- 197. Murtaugh MA, Sweeney C, Giuliano AR, Herrick JS, Hines L, Byers T, et al. Diet patterns and breast cancer risk in Hispanic and non-Hispanic white women: the Four-Corners Breast Cancer Study. *Am J Clin Nutr* 2008;87(4):978-984.

- 198. Cade JE, Taylor EF, Burley VJ, Greenwood DC. Does the Mediterranean dietary pattern or the Healthy Diet Index influence the risk of breast cancer in a large British cohort of women? *Eur J Clin Nutr* 2011;65(8):920-928.
- 199. Edefonti V, Randi G, La Vecchia C, Ferraroni M, Decarli A. Dietary patterns and breast cancer: a review with focus on methodological issues. *Nutr Rev* 2009;67(6):297-314.
- 200. Fung TT, Hu FB, McCullough ML, Newby PK, Willett WC, Holmes MD. Diet quality is associated with the risk of estrogen receptor-negative breast cancer in postmenopausal women. *J Nutr* 2006;136(2):466-72.
- 201. Nkondjock A, Ghadirian P. Diet quality and BRCA-associated breast cancer risk. *Breast Cancer Res Treat* 2007;103(3):361-9.
- 202. Brennan SF, Cantwell MM, Cardwell CR, Velentzis LS. Dietary patterns and breast cancer risk: a systematic review and meta-analysis. *Am J Clin Nutr* 2010;91(5):1294-302.
- 203. Key TJ, Allen NE, Spencer EA, Travis RC. Nutrition and breast cancer. *Breast* 2003;12(6):412-416.
- 204. Sugimura T, Wakabayashi K, Nakagama H, Nagao M. Heterocyclic amines: Mutagens/carcinogens produced during cooking of meat and fish. *Cancer Sci* 2004;95(4):290-299.
- 205. Santamaria P. Nitrate in vegetables: toxicity, content, intake and EC regulation. J Sci Food Agric 2006;86(1):10-17.
- 206. Galland L. Diet and inflammation. Nutr Clin Pract 2010;25(6):634-640.
- 207. Sant M, Allemani C, Sieri S, Krogh V, Menard S, Tagliabue E, et al. Salad vegetables dietary pattern protects against HER-2-positive breast cancer: a prospective Italian study. *Int J Cancer* 2007;121(4):911-914.
- 208. Key TJ, Appleby PN, Spencer EA, Travis RC, Roddam AW, Allen NE. Cancer incidence in vegetarians: results from the European Prospective Investigation into Cancer and Nutrition (EPIC-Oxford). *Am J Clin Nutr* 2009;89(5):1620S-1626S.
- 209. Cade JE, Taylor EF, Burley VJ, Greenwood DC. Common dietary patterns and risk of breast cancer: analysis from the United Kingdom Women's Cohort Study. *Nutr Cancer* 2010;62(3):300-6.
- 210. Gann PH, Chatterton RT, Gapstur SM, Liu K, Garside D, Giovanazzi S, et al. The effects of a low-fat/high-fiber diet on sex hormone levels and menstrual cycling in premenopausal women. *Cancer* 2003;98(9):1870-1879.
- 211. Gartner C, Stahl W, Sies H. Lycopene is more bioavailable from tomato paste than from fresh tomatoes. *Am J Clin Nutr* 1997;66(1):116-122.
- 212. Boffetta P, Couto E, Wichmann J, Ferrari P, Trichopoulos D, Bueno-de-Mesquita HB, et al. Fruit and vegetable intake and overall cancer risk in the European Prospective Investigation Into Cancer and Nutrition (EPIC). *J Natl Cancer Inst* 2010;102(8):529-537.
- 213. Gandini S, Merzenich H, Robertson C, Boyle P. Meta-analysis of studies on breast cancer risk and diet: the role of fruit and vegetable consumption and the intake of associated micronutrients. *Eur J Cancer* 2000;36(5):636-646.
- 214. Higdon JV, Delage B, Williams DE, Dashwood RH. Cruciferous vegetables and human cancer risk: epidemiologic evidence and mechanistic basis. *Pharmacol Res* 2007;55(3):224-236.
- 215. Monroe KR, Murphy SP, Kolonel LN, Pike MC. Prospective study of grapefruit intake and risk of breast cancer in postmenopausal women: the Multiethnic Cohort Study. *Br J Cancer* 2007;97(3):440-445.

- 216. Kim EH, Hankinson SE, Eliassen AH, Willett WC. A prospective study of grapefruit and grapefruit juice intake and breast cancer risk. *Br J Cancer* 2008;98(1):240-1.
- 217. Monroe KR, Murphy SP, Henderson BE, Kolonel LN, Stanczyk FZ, Adlercreutz H, et al. Dietary fiber intake and endogenous serum hormone levels in naturally postmenopausal Mexican American women: The Multiethnic Cohort Study. *Nutr Cancer* 2007;58(2):127-135.
- 218. Olsen A, Tjonneland A, Thomsen BL, Loft S, Stripp C, Overvad K, et al. Fruits and vegetables intake differentially affects estrogen receptor negative and positive breast cancer incidence rates. *J Nutr* 2003;133(7):2342-2347.
- 219. Agurs-Collins T, Rosenberg L, Makambi K, Palmer JR, Adams-Campbell L. Dietary patterns and breast cancer risk in women participating in the Black Women's Health Study. *Am J Clin Nutr* 2009;90(3):621-8.
- 220. Boggs DA, Palmer JR, Wise LA, Spiegelman D, Stampfer MJ, Adams-Campbell LL, et al. Fruit and vegetable intake in relation to risk of breast cancer in the Black Women's Health Study. *Am J Epidemiol* 2010.
- 221. Rogers I, Emmett P, Gunnell D, Dunger D, Holly J. Milk as a food for growth? The insulin-like growth factors link. *Public Health Nutr* 2006;9(03):359-368.
- 222. Sieri S, Krogh V, Ferrari P, Berrino F, Pala V, Thiebaut ACM, et al. Dietary fat and breast cancer risk in the European Prospective Investigation into Cancer and Nutrition. *Am J Clin Nutr* 2008;88(5):1304-1312.
- 223. Bingham SA, Luben R, Welch A, Wareham N, Khaw K-T, Day N. Are imprecise methods obscuring a relation between fat and breast cancer? *Lancet* 2003;362(9379):212-214.
- 224. Freedman LS, Potischman N, Kipnis V, Midthune D, Schatzkin A, Thompson FE, et al. A comparison of two dietary instruments for evaluating the fat-breast cancer relationship. *Int J Epidemiol* 2006;35(4):1011-21.
- 225. Key TJ, Appleby PN, Cairns BJ, Luben R, Dahm CC, Akbaraly T, et al. Dietary fat and breast cancer: comparison of results from food diaries and food-frequency questionnaires in the UK Dietary Cohort Consortium. *Am J Clin Nutr* 2011;94(4):1043-1052.
- 226. Prentice RL, Caan B, Chlebowski RT, Patterson R, Kuller LH, Ockene JK, et al. Low-fat dietary pattern and risk of invasive breast cancer: the Women's Health Initiative Randomized Controlled Dietary Modification Trial. JAMA 2006;295(6):629-42.
- 227. Byers T, Sedjo RL. Does intentional weight loss reduce cancer risk? *Diabetes Obes Metab* 2011;13(12):1063-1072.
- 228. Sieri S, Pala V, Brighenti F, Pellegrini N, Muti P, Micheli A, et al. Dietary glycemic index, glycemic load, and the risk of breast cancer in an Italian prospective cohort study. *Am J Clin Nutr* 2007;86(4):1160-1166.
- 229. McCann SE, McCann WE, Hong CC, Marshall JR, Edge SB, Trevisan M, et al. Dietary patterns related to glycemic index and load and risk of premenopausal and postmenopausal breast cancer in the Western New York Exposure and Breast Cancer Study. *Am J Clin Nutr* 2007;86(2):465-471.
- 230. Mulholland HG, Murray LJ, Cardwell CR, Cantwell MM. Dietary glycaemic index, glycaemic load and breast cancer risk: a systematic review and meta-analysis. *Br J Cancer* 2008;99(7):1170-1175.
- 231. Cade JE, Burley VJ, Greenwood DC, The UKWCS Steering Group. Dietary fibre and risk of breast cancer in the UK Women's Cohort Study. *Int J Epidemiol* 2007;36(2):431-438.

- 232. Dong J-Y, He K, Wang P, Qin L-Q. Dietary fiber intake and risk of breast cancer: a meta-analysis of prospective cohort studies. *Am J Clin Nutr* 2011;94(3):900-905.
- 233. Lew JQ, Freedman ND, Leitzmann MF, Brinton LA, Hoover RN, Hollenbeck AR, et al. Alcohol and risk of breast cancer by histologic type and hormone receptor status in postmenopausal women: the NIH-AARP Diet and Health Study. Am J Epidemiol 2009;170(3):308-17.
- 234. Suzuki R, Orsini N, Mignone L, Saji S, Wolk A, Suzuki R, et al. Alcohol intake and risk of breast cancer defined by estrogen and progesterone receptor status: a meta-analysis of epidemiological studies. *Int J Cancer* 2008;122(8):1832-41.
- 235. Zhang SMM, Lee IM, Manson JE, Cook NR, Willett WC, Buring JE. Alcohol consumption and breast cancer risk in the Women's Health Study. *Am J Epidemiol* 2007;165(6):667-676.
- 236. Larsson SC, Giovannucci E, Wolk A. Folate and risk of breast cancer: a metaanalysis. *J Natl Cancer Inst* 2007;99(1):64-76.
- 237. Lippman SM, Klein EA, Goodman PJ, Lucia MS, Thompson IM, Ford LG, et al. Effect of selenium and vitamin E on risk of prostate cancer and other cancers: The Selenium and Vitamin E Cancer Prevention Trial (SELECT). *JAMA* 2009;301(1):39-51.
- 238. Virtamo J, Pietinen P, Huttunen JK, et al. Incidence of cancer and mortality following alpha-tocopherol and beta-carotene supplementation: a postintervention follow-up. *JAMA* 2003;290(4):476-485.
- 239. Cho E, Spiegelman D, Hunter DJ, Chen WY, Zhang SM, Colditz GA, et al. Premenopausal intakes of vitamins A, C, and E, folate, and carotenoids, and risk of breast cancer. *Cancer Epidemiol Biomarkers Prev* 2003;12(8):713-720.
- 240. Horn-Ross PL, Hoggatt KJ, West DW, Krone MR, Stewart SL, Anton-Culver H, et al. Recent diet and breast cancer risk: the California Teachers Study (USA). *Cancer Causes Control* 2002;13(5):407-415.
- 241. Hunter DJ, Manson JE, Colditz GA, Stampfer MJ, Rosner B, Hennekens CH, et al. A prospective study of the intake of vitamins C, E, and A and the risk of breast cancer. *N Engl J Med* 1993;329(4):234-240.
- 242. Kushi LH, Fee RM, Sellers TA, Zheng W, Folsom AR. Intake of vitamins A, C, and E and postmenopausal breast cancer: The Iowa Women's Health Study. *Am J Epidemiol* 1996;144(2):165-174.
- 243. Michels KB, Holmberg L, Bergkvist L, Ljung H, Bruce Å, Wolk A. Dietary antioxidant vitamins, retinol, and breast cancer incidence in a cohort of Swedish women. *Int J Cancer* 2001;91(4):563-567.
- 244. Nagel G, Linseisen J, van Gils C, Peeters P, Boutron-Ruault M, Clavel-Chapelon F, et al. Dietary β-carotene, vitamin C and E intake and breast cancer risk in the European Prospective Investigation into Cancer and Nutrition (EPIC). *Breast Cancer Res Treat* 2010;119(3):753-765.
- 245. Nissen SB, Tjønneland A, Stripp C, Olsen A, Christensen J, Overvad K, et al. Intake of vitamins A, C, and E from diet and supplements and breast cancer in postmenopausal women. *Cancer Causes Control* 2003;14(8):695-704.
- 246. Verhoeven DTH, Assen N, Goldbohm RA, Dorant E, van 't Veer P, Sturmans F, et al. Vitamins C and E, retinol, beta-carotene and dietary fibre in relation to breast cancer risk: a prospective cohort study. *Br J Cancer* 1997;75(1):149-155.
- 247. Zhang S, Hunter DJ, Forman MR, Rosner BA, Speizer FE, Colditz GA, et al. Dietary carotenoids and vitamins A, C, and E and risk of breast cancer. *J Natl Cancer Inst* 1999;91(6):547-556.

- 248. Chlebowski RT, Johnson KC, Kooperberg C, Pettinger M, Wactawski-Wende J, Rohan T, et al. Calcium plus vitamin D supplementation and the risk of breast cancer. *J Natl Cancer Inst* 2008;100(22):1581-1591.
- 249. Cole BF, Baron JA, Sandler RS, Haile RW, Ahnen DJ, Bresalier RS, et al. Folic acid for the prevention of colorectal adenomas: a Randomized Clinical Trial. *JAMA* 2007;297(21):2351-2359.
- 250. Ishitani K, Lin J, Manson JE, Buring JE, Zhang SM. A prospective study of multivitamin supplement use and risk of breast cancer. *Am J Epidemiol* 2008;167(10):1197-1206.
- 251. Neuhouser ML, Wassertheil-Smoller S, Thomson C, Aragaki A, Anderson GL, Manson JE, et al. Multivitamin use and risk of cancer and cardiovascular disease in the Women's Health Initiative Cohorts. *Arch Intern Med* 2009;169(3):294-304.
- 252. Satia JA, Littman A, Slatore CG, Galanko JA, White E. Long-term use of beta-Carotene, retinol, lycopene, and lutein Supplements and lung cancer risk: results from the VITamins And Lifestyle (VITAL) Study. *Am J Epidemiol* 2009;169(7):815-828.
- 253. Slatore CG, Littman AJ, Au DH, Satia JA, White E. Long-term use of supplemental multivitamins, vitamin C, vitamin E, and folate does not reduce the risk of lung cancer. *Am J Respir Crit Care Med* 2008;177(5):524-530.
- 254. Zhang SM, Cook NR, Albert CM, Gaziano JM, Buring JE, Manson JE. Effect of combined folic acid, vitamin B6, and vitamin B12 on cancer risk in Women: a randomized trial. *JAMA* 2008;300(17):2012-2021.
- 255. Myung SK, Kim Y, Ju W, Choi HJ, Bae WK. Effects of antioxidant supplements on cancer prevention: meta-analysis of randomized controlled trials. *Ann Oncol* 2010;21(1):166-179.
- 256. Hercberg S, Galan P, Preziosi P, Bertrais S, Mennen L, Malvy D, et al. The SU.VI.MAX Study: a randomized, placebo-controlled trial of the health effects of antioxidant vitamins and minerals. *Arch Intern Med* 2004;164(21):2335-2342.
- 257. Ewertz M, Gill C. Dietary factors and breast cancer risk in Denmark. *Int J Cancer* 1990;46(5):779-784.
- 258. Lee MM, Chang IYH, Horng CF, Chang JS, Cheng SH, Huang A. Breast cancer and dietary factors in Taiwanese women. *Cancer Causes Control* 2005;16(8):929-937.
- 259. Moorman PG, Ricciuti MF, Millikan RC, Newman B. Vitamin supplement use and breast cancer in a North Carolina population. *Public Health Nutr* 2001;4(3):821-827.
- 260. Chan ALF, Leung HWC, Wang SF. Multivitamin supplement use and risk of breast cancer: a meta-analysis. *Ann Pharmacother* 2011;45(4):476-484.
- 261. Larsson SC, Akesson A, Bergkvist L, Wolk A. Multivitamin use and breast cancer incidence in a prospective cohort of Swedish women. *Am J Clin Nutr* 2010;91(5):1268-72.
- 262. Maruti SS, Ulrich CM, White E. Folate and one-carbon metabolism nutrients from supplements and diet in relation to breast cancer risk. *Am J Clini Nutr* 2009;89(2):624-633.
- 263. Pocobelli G, Peters U, Kristal AR, White E. Use of supplements of multivitamins, vitamin C, and vitamin E in relation to mortality. *Am J Epidemiol* 2009;170(4):472-483.
- 264. Giovannucci E, stampfer MJ, Colditz GA, Manson JE, Rosner BA, Longnecker M, et al. A comparison of prospective and retrospective assessments of diet in the study of breast cancer. *Am J Epidemiol* 1993;137(5):502-511.

- 265. Graham S, Zielezny M, Marshall J, Priore R, Freudenheim J, Brasure J, et al. Diet in the epidemiology of postmenopausal breast cancer in the New York State Cohort. *Am J Epidemiol* 1992;136(11):1327-1337.
- 266. Li W, Ray RM, Lampe JW, Lin M-G, Gao DL, Wu C, et al. Dietary and other risk factors in women having fibrocystic breast conditions with and without concurrent breast cancer: A nested case-control study in Shanghai, China. *Int J Cancer* 2005;115(6):981-993.
- 267. Rohan TE, Howe GR, Friedenreich CM, Jain M, Miller AB. Dietary fiber, vitamins A, C, and E, and risk of breast cancer: a cohort study. *Cancer Causes Control* 1993;4(1):29-37.
- 268. Wu K, Helzlsouer KJ, Alberg AJ, Comstock GW, Norkus EP, Hoffman SC. A prospective study of plasma ascorbic acid concentrations and breast cancer (United States). *Cancer Causes Control* 2000;11(3):270-283.
- 269. BMJ. Default search strategies used for BMJ Clinical Evidence [Internet]:[cited 2011 Jun]. Available from: http://clinicalevidence.bmj.com/ceweb/about/search filters.jsp.
- 270. Cui Y, Shikany JM, Liu S, Shagufta Y, Rohan TE. Selected antioxidants and risk of hormone receptor-defined invasive breast cancers among postmenopausal women in the Women's Health Initiative Observational Study. *Am J Clin Nutr* 2008;87(4):1009-1018.
- 271. Dorjgochoo T, Shrubsole M, Shu X, Lu W, Ruan Z, Zheng Y, et al. Vitamin supplement use and risk for breast cancer: the Shanghai Breast Cancer Study. *Breast Cancer ResTreat* 2008;111(2):269-278.
- 272. Lee SA, Lee KM, Lee SJ, Yoo KY, Park SK, Noh DY, et al. Antioxidant vitamins intake, ataxia telangiectasia mutated (ATM) genetic polymorphisms, and breast cancer risk. *Nutr Cancer* 2010;62(8):1087-94.
- 273. Roswall N, Olsen A, Christensen J, Dragsted LO, Overvad K, Tjonneland A. Micronutrient intake and breast cancer characteristics among postmenopausal women. *Eur J Cancer Prev* 2010;19(5):360-5.
- 274. Wang C, Baumgartner RN, Yang D, Slattery ML, Murtaugh MA, Byers T, et al. No evidence of association between breast cancer risk and dietary carotenoids, retinols, vitamin C and tocopherols in Southwestern Hispanic and non-Hispanic White women. *Breast Cancer Res Treat* 2009;114(1):137-45.
- 275. Zhang CX, Ho SC, Chen YM, Fu JH, Cheng SZ, Lin FY. Greater vegetable and fruit intake is associated with a lower risk of breast cancer among Chinese women. *Int J Cancer* 2009;125(1):181-8.
- 276. Zhu K, Hunter S, Payne-Wilks K, Sutcliffe C, Bentley C, Roland CL, et al. Potential differences in breast cancer risk factors based on CYP1A1 MspI and African-American-specific genotypes. *Ethnicity & Disease* 2006;16(1):207-15.
- 277. Fulan H, Changxing J, Baina W, Wencui Z, Chunqing L, Fan W, et al. Retinol, vitamins A, C, and E and breast cancer risk: a meta-analysis and meta-regression. *Cancer Causes Control* 2011:1-14.
- 278. Frazier AL, Ryan C, Rockett H, Willett W, Colditz G. Adolescent diet and risk of breast cancer. *Breast Cancer Res* 2003;5(3):R59 R64.
- 279. Cade JE, Burley VJ, Greenwood DC, UK Women's Cohort Study Steering Group. The UK Women's Cohort Study: comparison of vegetarians, fish-eaters and meat-eaters. *Public Health Nutr* 2004;7(07):871-878.
- 280. Greenwood DC, Cade JE, White K, V. B, Schorah C. The impact of high nonstarch polysaccharide intake on serum micronutrient concentrations in a cohort of women. *Public Health Nutr* 2003;7(4):543-548.
- 281. Davey GK, Spencer EA, Appleby PN, Allen NE, Knox KH, Key TJ. EPIC-Oxford:lifestyle characteristics and nutrient intakes in a cohort of 33 883 meat-

eaters and 31 546 non meat-eaters in the UK. *Public Health Nutr* 2003;6(03):259-268.

- 282. Appleby PN, Thorogood M, Mann JI, Key TJA. The Oxford Vegetarian Study: an overview. *Am J Clin Nutr* 1999;70(3):525S-531.
- 283. Bingham SA, Welch AA, McTaggart A, Mulligan AA, Runswick SA, Luben R, et al. Nutritional methods in the European Prospective Investigation of Cancer in Norfolk. *Public Health Nutr* 2001;4(03):847-858.
- 284. Marmot M, Brunner E. Cohort Profile: The Whitehall II study. *Int J Epidemiol* 2005;34(2):251-256.
- 285. Wadsworth M, Kuh D, Richards M, Hardy R. Cohort Profile: The 1946 National Birth Cohort (MRC National Survey of Health and Development). *Int J Epidemiol* 2006;35(1):49-54.
- 286. Dickinson HO, Salotti JA, Birch PJ, Reid MM, Malcolm A, Parker L. How complete and accurate are cancer registrations notified by the National Health Service Central Register for England and Wales? J Epidemiol Community Health 2001;55(6):414-422.
- 287. Holland B, Welch AA, Unwin ID, Buss DH, Paul AA, Southgate DAT. *McCance and Widdowson's the Composition of Foods*. 5th ed. Cambridge, UK: Royal Society of Chemisty, 1991.
- 288. Calvert C, Cade J, Barrett JH, Woodhouse A, UKWCS Steering Group. Using cross-check questions to address the problem of mis-reporting of specific food groups on Food Frequency Questionnaires. *Eur J Clin Nutr* 1997;51(10):708-712.
- 289. Riboli E. Nutrition and cancer: Background and rationale of the European Prospective Investigation into Cancer and Nutrition (EPIC). *Ann Oncol* 1992;3(10):783-791.
- 290. Riboli E, Kaaks R. The EPIC Project: rationale and study design. European Prospective Investigation into Cancer and Nutrition. *Int J Epidemiol* 1997;26(suppl\_1):S6-14.
- 291. National Audit Office. Statistical and Technical Team. A Practical Guide to Sampling [Internet]. London: NAO, 2001:[Updated 2001 Jun; cited 2009 sept]. Available from: http://www.nao.org.uk/publications/0001/sampling\_guide.aspx.
- 292. Marshall JR, Chen Z. Diet and health risk: risk patterns and disease-specific associations. *Am J Clin Nutr* 1999;69(6):1351S-1356.
- 293. Dahm CC, Keogh RH, Spencer EA, Greenwood DC, Key TJ, Fentiman IS, et al. Dietary fiber and colorectal cancer risk: A nested case control study using food diaries. *J Natl Cancer Inst* 2010;102(9):614-626.
- 294. Brunner E, Stallone D, Juneja M, Bingham S, Marmot M. Dietary assessment in Whitehall II: comparison of 7 d diet diary and food-frequency questionnaire and validity against biomarkers. *Br J Nutr* 2001;86:405-414.
- 295. Price GM, Paul AA, Key FB, Harter AC, Cole TJ, Day KC, et al. Measurement of diet in a large national survey: comparison of computerized and manual coding of records in household measures. *J Hum Nutr Diet* 1995;8(6):417-428.
- 296. Kirkwood BR, Stern JAC. *Essential medical statistics*. 2nd ed. Oxford: Blackwell Science, 2003.
- 297. Kleinbaum D, Klein M. Survival analysis: a self-learning text. 2nd ed. New York: Springer Science+Business Inc, 2005.
- 298. Bland JM. An introduction to medical statistics. 3rd ed. Oxford: Oxford University Press, 2000.
- 299. Wacholder S, Silverman DT, McLaughlin JK, Mandel JS. Selection of controls in case-control studies: III. design options. *Am J Epidemiol* 1992;135(9):1042-1050.

- 300. Farquhar CMS, Lynn. Harvey, Sally A. Stewart, Alistair W. The association of hysterectomy and menopause: a prospective cohort study. *Int J Obstet Gynaecol* 2005;112(7):956-962.
- 301. Willet W. *Nutritional Epidemiology*. 2nd ed. New York: Oxford university Press, 1998.
- 302. Greenland S, Brumback B. An overview of relations among causal modelling methods. *Int J Epidemiol* 2002;31(5):1030-1037.
- 303. Greenland S, Pearl J, Robins JM. Causal diagrams for epidemiologic research. *Epidemiol* 1999;10(1): 37-48.
- 304. Ambrosone CB, Kropp S, Yang J, Yao S, Shields PG, Chang-Claude J. Cigarette Smoking, N-Acetyltransferase 2 Genotypes, and Breast Cancer Risk: Pooled Analysis and Meta-analysis. *Cancer Epidemiol Biomarkers Prev* 2008;17(1):15-26.
- 305. Stringhini S, Dugravot A, Shipley M, Goldberg M, Zins M, Kivimaki M, et al. Health behaviours, socioeconomic status, and mortality: further analyses of the British Whitehall II and the French GAZEL prospective cohorts. *PLoS Med* 2011;8(2):e1000419.
- 306. Schernhammer ES, Hankinson SE, Rosner B, Kroenke CH, Willett WC, Colditz GA, et al. Job stress and breast cancer risk: The Nurses' Health Study. Am J Epidemiol 2004;160(11):1079-1086.
- 307. Macleod J, Davey Smith G. Re: 'Job Stress and breast cancer risk: the Nurses Health Study' *Am J Epidemiol* 2005;162(11):1133-1134.
- 308. MacInnis RJ, English DR, Gertig DM, Hopper JL, Giles GG. Body size and composition and risk of postmenopausal breast cancer. *Cancer Epidemiol Biomarkers Prev* 2004;13(12):2117-2125.
- 309. Costa PT, Jr, McCrae RR. Revised NEO personality inventory and NEO five factor inventory. *Psychol Assessment*: Odessa, FL, 1992.
- 310. Bogg T, Roberts BW. Conscientiousness and health behaviors: a meta-analysis. *Psychol Bull* 2004;130:887–919.
- 311. Hagger-Johnson GE, Whiteman MC. Conscientiousness facets and health behaviors: A latent variable modeling approach. *Pers Indiv Differ* 2007;43(5):1235-1245.
- 312. Costa PT, Jr., McCrae RR. Neuroticism, Somatic Complaints, and Disease: Is the Bark Worse than the Bite? *J Pers* 1987;55(2):299-316.
- 313. Laurikkala J, Juhola M, Kentala E. Informal Identification of Outliers in Medical Data. *Fifth International Workshop on Intelligent Data Analysis in Medicine and Pharmacology.* Berlin, 2000:20-24.
- 314. Hodge V, Austin J. A Survey of Outlier Detection Methodologies. *Artif Intell Rev* 2004;22(2):85-126.
- 315. Grubbs FE. Procedures for Detecting Outlying Observations in Samples. *Technometrics* 1969;11(1):1-21.
- 316. Hadi AS. Identifying multiple outliers in multivariate data. *J Roy Stat Soc B Met* 1992;54(3):761-771.
- 317. Hadi AS. A modification of a method for the detection of outliers in multivariate samples. *J Roy Stat Soc B Met* 1994;56(2):393-396.
- 318. Murff HJ, Spigel DR, Syngal S. Does this patient have a family history of cancer? An evidence-based analysis of the accuracy of family cancer history. *JAMA* 2004;292(12):1480-1489.
- 319. National Institute for Health and Clinical Excellence. Familial breast cancer. The classification and care of women at risk of familial breast cancer in primary, secondary and tertiary care. London: NICE, 2006.

- 320. Cohen J. A coefficient of agreement for nominal scales. *Educ Psychol Meas* 1960;20:37-46.
- 321. Landis JR, Koch GG. The measurement of observer agreement for categorical data. *Biometrics* 1977;33(1):159-174.
- 322. Murphy SP, Wilkens LR, Hankin JH, Foote JA, Monroe KR, Henderson BE, et al. Comparison of two instruments for quantifying intake of vitamin and mineral supplements: a brief questionnaire versus three 24-hour recalls. *Am J Epidemiol* 2002;156(7):669-675.
- 323. Patterson RE, Kristal AR, Levy L, McLerran D, White E. Validity of methods used to assess vitamin and mineral supplement use. *Am J Epidemiol* 1998;148(7):643-649.
- 324. Willett WC, Hu FB. Not the time to abandon the food frequency questionnaire: point. *Cancer Epidemiol Biomarkers Prev* 2006;15(10):1757-1758.
- 325. Amanatidis S, Mackerras D, Simpson JM. Comparison of two frequency questionnaires for quantifying fruit and vegetable intake. *Public Health Nutr* 2001;4:233-239.
- 326. Block G, Woods M, Potosky A, Clifford C. Validation of a self-administered diet history questionnaire using multiple diet records. *J Clin Epidemiol* 1990;43(12):1327-1335.
- 327. Subar AF, Heimendinger J, Patterson BH, Krebs-Smith SM, Pivonka E, Kessler R. Fruit and vegetable intake in the United States: the baseline survey of the Five A Day for Better Health Program. *Am J Health Promot* 1995;9(5):352-60.
- 328. Bland MJ, Altman DG. Statistical methods for assessing agreement between two methods of clinical measurement. *Lancet* 1986;327(8476):307-310.
- 329. Bland JM, Altman DG. Measuring agreement in method comparison studies. *Stat Methods Med Res* 1999;8(2):135-160.
- 330. Pollard J, Greenwood D, Kirk S, Cade J. Lifestyle factors affecting fruit and vegetable consumption in the UK Women's Cohort Study. *Appetite* 2001;37(1):71-79.
- 331. Pollard J, Wild CP, White KLM, Greenwood DC, Cade JE, Kirk SFL. Comparison of plasma biomarkers with dietary assessment methods for fruit and vegetable intake. *Eur J Clin Nutr* 2003;57(8):988-998.
- 332. Molag ML, de Vries JHM, Ocke MC, Dagnelie PC, van den Brandt PA, Jansen MCJF, et al. Design characteristics of Food Frequency Questionnaires in relation to their validity. *Am J Epidemiol* 2007;166(12):1468-1478.
- 333. Miller T, Abdel-Maksoud M, Crane L, Marcus A, Byers T. Effects of social approval bias on self-reported fruit and vegetable consumption: a randomized controlled trial. *Nutr J* 2008;7(1):18.
- 334. Krebs-Smith SM, Heimendinger J, Subar AF, Patterson BH, Pivonka E. Using food frequency questionnaires to estimate fruit and vegetable intake: association between the number of questions and total intakes. *J Nutr Educ* 1995;27(2):80-85.
- 335. Cade J, Thompson R, Burley V, Warm D. Development, validation and utilisation of food-frequency questionnaires a review. *Public Health Nutr* 2002;5(04):567-587.
- 336. Ministry of Agriculture Fisheries and Food. National Food Survey 1998. London: The Stationery Office, 1999.
- 337. Cade JE, Burley VJ, Warm DL, Thompson RL, Margetts BM. Food-frequency questionnaires: a review of their design, validation and utilisation. *Nutrition Res Rev* 2004 17:5-22.
- 338. Michels KB, Welch AA, Luben R, Bingham SA, Day NE. Measurement of fruit and vegetable consumption with diet questionnaires and implications for analyses and interpretation. *Am J Epidemiol* 2005;161(10):987-994.
- 339. Gerber M, Richardson S, Salkeld R, Chappuis P. Antioxidants in female breast cancer patients. *Cancer Invest* 1991;9(4):421-428.
- 340. Hietanene E, Bartsch H, Béréziat JC, Camus AM, McClinton S, Eremin O, et al. Diet and oxidative stress in breast, colon and prostate cancer patients: a casecontrol study. *Eur J Clin Nutr* 1994;48(8):575-86.
- 341. Ramaswamy G, Krishnamoorthy L. Serum carotene, vitamin A, and vitamin C levels in breast cancer and cancer of the uterine cervix. *Nutr Cancer* 1996;25(2):173 - 177.
- 342. Ness A, Khaw K, Bingham S, Day NE. Plasma vitamin C: what does it measure? . *Public Health Nutr* 1999;2:51-54.
- 343. Henríquez-Sánchez P, Sánchez-Villegas A, Doreste-Alonso J, Ortiz-Andrellucchi A, Pfrimer K, Serra-Majem L. Dietary assessment methods for micronutrient intake: a systematic review on vitamins. Br J Nutr 2009;102:S10-S37
- 344. Sobala GM, Pignatelli B, Schorah CJ, Bartsch H, Sanderson M, Dixon MF, et al. Levels of nitrite, nitrate, N-nitroso compounds, ascorbic acid and total bile acids in gastric juice of patients with and without precancerous conditions of the stomach. *Carcinogenesis* 1991;12(2):193-198.
- 345. Kaaks R, Slimani N, Riboli E. Pilot phase studies on the accuracy of dietary intake measurements in the EPIC project: overall evaluation of results. European Prospective Investigation into Cancer and Nutrition. *Int J Epidemiol* 1997;26(suppl 1):S26.
- 346. Bingham S, Luben R, Welch A, Low YL, Khaw KT, Wareham N, et al. Associations between dietary methods and biomarkers, and between fruits and vegetables and risk of ischaemic heart disease, in the EPIC Norfolk Cohort Study. *Int J Epidemiol* 2008;37(5):978-987.
- 347. Bingham SA, Gill C, Welch A, Cassidy A, Runswick SA, Oakes S, et al. Validation of dietary assessment methods in the UK arm of EPIC using weighed records, and 24-hour urinary nitrogen and potassium and serum vitamin C and carotenoids as biomarkers. *Int J Epidemiol* 1997;26(suppl 1):S137.
- 348. Boucher BA. Dietary supplements, quality scores and missing data in the review of validation studies *Br J Nutr* 2010;104:1878-1879.
- 349. Willett WC, Sampson L, Stampfer MJ, Rosner B, Bain C, Witschi J, et al. Reproducibility and validity of a semiquantitative food frequency questionnaire. *Am J Epidemiol* 1985;122(1):51-65.
- 350. Marshall JR, Chen Z. Diet and health risk: risk patterns and disease-specific associations. *Am J Clin Nutr* 1999;69(6):1351S-1356S.
- 351. Bjelakovic G, Nikolova D, Gluud LL, Simonetti RG, Gluud C. Mortality in randomized trials of antioxidant supplements for primary and secondary prevention: systematic review and meta-analysis. *JAMA* 2007;297(8):842-857.
- 352. Skeie G, Braaten T, Hjartaker A, Lentjes M, Amiano P, Jakszyn P, et al. Use of dietary supplements in the European Prospective Investigation into Cancer and Nutrition calibration study. *Eur J Clin Nutr* 2009;63(S4):S226-S238.
- 353. Satia-Abouta J, Kristal AR, Patterson RE, Littman AJ, Stratton KL, White E. Dietary supplement use and medical conditions: The VITAL study. *Am J Prev Med* 2003;24(1):43-51.
- 354. Velicer CM, Ulrich CM. Vitamin and mineral supplement use among US adults after cancer diagnosis: a systematic review. *J Clin Oncol* 2008;26(4):665-673.
- 355. Expert Group on Vitamins and Minerals. Risk assessment: Vitamin C, 2003.

- 356. Heaney ML, Gardner JR, Karasavvas N, Golde DW, Scheinberg DA, Smith EA, et al. Vitamin C antagonizes the cytotoxic effects of antineoplastic drugs. *Cancer Res* 2008;68(19):8031-8038.
- 357. Patterson RE, Neuhouser ML, Hedderson MM, Schwartz SM, Standish LJ, Bowen DJ. Changes in diet, physical activity, and supplement use among adults diagnosed with cancer. *J Am Diet Ass* 2003;103(3):323-328.
- 358. McDavid K, Breslow RA, Radimer K. Vitamin/mineral supplementation among cancer survivors:1987 and 1992 National Health interview surveys. *Nutr Cancer* 2001;41(1):29 32.
- 359. Alamian A, Rouleau I, Simard J, Dorval M. Use of dietary supplements among women at high risk of hereditary breast and ovarian cancer (HBOC) tested for cancer susceptibility. *Nutr Cancer* 2006;54(2):157 165.
- 360. Seema P, Paul B, Paolo B. Meta-analysis of social inequality and the risk of cervical cancer. *Int J Cancer* 2003;105(5):687-691.
- 361. Rock CL, Newman VA, Neuhouser ML, Major J, Barnett MJ. Antioxidant Supplement Use in Cancer Survivors and the General Population. *J Nutr* 2004;134(11):3194S-3195.
- 362. WCRF. Warning on high-strength vitamin supplements [internet]. London: WCRF:[Updated 2009 Oct 19; cited 2009 Nov]. Available from: http://www.wcrf-uk.org/audience/media/press\_release.php?recid=77.
- 363. Li K, Kaaks R, Linseisen J, Rohrmann S. Vitamin/mineral supplementation and cancer, cardiovascular, and all-cause mortality in a German prospective cohort (EPIC-Heidelberg). *Eur J Nutr*:doi:10.1007/s00394-011-0224-1. Epub 2011.
- 364. Li K, Kaaks R, Linseisen J, Rohrmann S. Consistency of vitamin and/or mineral supplement use and demographic, lifestyle and heath-status predictors: findings from the European Prospective Investigation into Cancer and Nutrition (EPIC)-Heidelberg cohort. Br J Nutr 2010;104(7):1058-1064.
- 365. Dupont W. Statistical modeling for biomedical researchers: a simple introduction to the analysis of complex data. 2nd ed. Cambridge: Cambridge University Press, 2009.
- 366. Ristow M, Zarse K, Oberbach A, Kloting N, Birringer M, Kiehntopf M, et al. Antioxidants prevent health-promoting effects of physical exercise in humans. *Proc Natl Acad Sci USA* 2009;106(21):8665-8670.
- 367. Assmann SF, Pocock SJ, Enos LE, Kasten LE. Subgroup analysis and other (mis)uses of baseline data in clinical trials. *Lancet* 2000;355(9209):1064-1069.
- 368. Hemila H, Kaprio J. Modification of the effect of vitamin E supplementation on the mortality of male smokers by age and dietary vitamin C. *Am J Epidemiol* 2009;169(8):946-953.
- 369. Brasky TM, Lampe JW, Potter JD, Patterson RE, White E. Specialty supplements and breast cancer risk in the VITamins And Lifestyle (VITAL) cohort. *Cancer Epidemiol Biomarkers Prev* 2010;19(7):1696-1708.
- 370. Willett WC. Commentary: Flawed study designs are not salvaged by large samples. *Int J Epidemiol* 2008;37(5):987-989.
- 371. Park S-Y, Murphy SP, Wilkens LR, Henderson BE, Kolonel LN. Multivitamin use and the risk of mortality and cancer incidence. *Am J Epidemiol* 2011;173:906-914.
- 372. Harrell F. Regression Modeling Strategies: With Applications to Linear Models, Logistic Regession, and Survival Analysis. New York: Springer-Verlag, 2001.
- 373. Welch AA, McTaggart A, Mulligan AA, Luben R, Walker N, Khaw KT, et al. DINER (Data Into Nutrients for Epidemiological Research) - a new data-entry program for nutritional analysis in the EPIC-Norfolk cohort and the 7-day diary method. *Public Health Nutr* 2001;4(06):1253-1265.

- 374. Defra. Family Food. A Report on the 2002-2003 Expenditure and Food Survey. London: The Stationary Office, 2004.
- 375. Schatzkin A, Subar AF, Thompson FE, Harlan LC, Tangrea J, Hollenbeck AR, et al. Design and serendipity in establishing a large cohort with wide dietary intake distributions. *Am J Epidemiol* 2001;154(12):1119-1125.
- 376. Bingham S. The dietary assessment of individuals; methods, accuracy, new techniques and recommendations. *Nutr Abstr Rev* 1987;57(10):705-74.
- 377. Byers T. Anticancer vitamins du jour The ABCED's so far. Am J Epidemiol 2011;172(1):1-3.
- 378. McCullough ML, Patel AV, Kushi LH, Patel R, Willett WC, Doyle C, et al. Following cancer prevention guidelines reduces risk of cancer, cardiovascular disease, and all-cause mortality. *Cancer Epidemiol Biomarkers Prev* 2011;20(6):1089-1097.
- 379. Carter M, Burley V, Nykjaer C, Cade J. "My Meal Mate" (MMM): Validation of the diet measures captured on a smartphone application to facilitate weight loss. *Br J Nutr* 2012; In Press.
- 380. Cancer Research UK. Triple negative breast cancer [Internet]:[cited 2011 June]. Available from: http://cancerhelp.cancerresearchuk.org/about-cancer/cancerquestions/triple-negative-breast-cancer#triple
- 381. Timpson NJ, Forouhi NG, Brion M-J, Harbord RM, Cook DG, Johnson P, et al. Genetic variation at the SLC23A1 locus is associated with circulating concentrations of L-ascorbic acid (vitamin C): evidence from 5 independent studies with >15,000 participants. Am J Clin Nutr 2010;92(2):375-382.
- 382. Lawlor DA, Harbord RM, Sterne JAC, Timpson N, Davey Smith G. Mendelian randomization: Using genes as instruments for making causal inferences in epidemiology. *Stat Med* 2008;27(8):1133-1163.
- 383. Hoffer LJ, Levine M, Assouline S, Melnychuk D, Padayatty SJ, Rosadiuk K, et al. Phase I clinical trial of i.v. ascorbic acid in advanced malignancy. Ann Oncol 2008;19(11):1969-1974.
- 384. Crowe FL, Roddam AW, Key TJ, Appleby PN, Overvad K, Jakobsen MU, et al. Fruit and vegetable intake and mortality from ischaemic heart disease: results from the European Prospective Investigation into Cancer and Nutrition (EPIC)-Heart study. *Eur Heart J* 2011;32(10):1235-1243.
- 385. Rogovik AL, Vohra S, Goldman RD. Safety considerations and potential interactions of vitamins: should vitamins be considered drugs? *Ann Pharmacother* 2010;44(2):311-324.
- 386. Howes RM. Mythology of antioxidant vitamins? *Journal of Evidence-Based Complementary & Alternative Medicine JEBCAM* 2011;16(2):149-159.
- 387. Klein EA, Thompson IM, Tangen CM, Crowley JJ, Lucia MS, Goodman PJ, et al. Vitamin E and the risk of prostate cancer. *JAMA* 2011;306(14):1549-1556.
- 388. Qiao Y-L, Dawsey SM, Kamangar F, Fan J-H, Abnet CC, Sun X-D, et al. Total and cancer mortality after supplementation with vitamins and minerals: followup of the Linxian General Population Nutrition Intervention Trial. *J Natl Cancer Inst* 2009;101(7):507-518.
- 389. Bjelakovic G, Gluud C. Vitamin and mineral supplement use in relation to allcause mortality in the Iowa Women's Health Study: Comment on "Dietary supplements and mortality rate in older women". *Arch Intern Med* 2011;171(18):1633-1634.
- 390. Albanes D. Antioxidant supplements and mortality. JAMA 2007;298(4):400.
- 391. WCRF/AICR. Policy and Action for Cancer Prevention. Washington, DC: AICR, 2009.

#### Appendix A.

## Summary of research findings relating to breast cancer risks from the 2007 WCRF report on food nutrition, physical activity and cancer prevention

FOOD, NUTR AND CANCE	ITION, PHYSICAL A R OF THE BREAST (	ACTIVITY, PREMENOPAUSE)	FOOD, NUTR CANCER OF	RITION, PHYSICAL A THE BREAST (POST	CTIVITY, AND MENOPAUSE)
In the judgement of cancer of the brea to the strength of	of the Panel, the factors liste st (premenopause). Judgem the evidence.	ed below modify the risk of ents are graded according	In the judgement cancer of the brea to the strength of	of the Panel, the factors liste ist (postmenopause). Judger the evidence.	ed below modify the risk of nents are graded according
	DECREASES RISK	INCREASES RISK		DECREASES RISK	INCREASES RISK
Convincing	Lactation	Alcoholic drinks	Convincing	Lactation	Alcoholic drinks
Probable	Body fatness	Adult attained height <sup>1</sup> Greater birth weight			Body fatness Adult attained height <sup>1</sup>
Limited — suggestive	Physical activity <sup>2</sup>		Probable	Physical activity <sup>2</sup>	Abdominal fatness Adult weight gain
Limited — no conclusion	Cereals (grains) and thei potatoes; vegetables; fru	r products; dietary fibre; uits; pulses (legumes); soya	Limited — suggestive		Total fat
	and soya products; meat and dairy products; fats. vegetable fat; fatty acid acids; cholesterol; sugar sugary foods and drinks; starch; glycaemic index; p vitamin B6; folate; vitam D; vitamin B6; folate; vitam D; vitamin E; calcium; iro isoflavones; dichlorodipl dichlorodiphenyltrichlor hexachlorobenzene; hex <i>trans</i> -nonachlor; polychl patterns; culturally defin gain; energy intake; beir	; poultry; fish; eggs; milk and oils; total fat; composition, <i>trans</i> -fatty (sucrose); other sugars; coffee; tea; carbohydrate; rotetin; vitamin A; riboflavin; in B12; vitamin C; vitamin n; selenium; carotenoids; nenyldichloroethylene; oethane; dieldrin; cachlorocyclohexane; orinated biphenyls; dietary ned diets; adult weight ng breastfed	Limited — no conclusion	Cereals (grains) and thei potatoes; vegetables an soya and soya products; milk and dairy products; fatty acid composition; ; sugary foods and drinks; starch; glycaemic index; p vitamin B6; folate; vitam D; vitamin B; folate; vitam D; vitamin E; calcium; irc isoflavones; dichlorodip dichlorodiphenytrichlon hexachlorobenzene; hey nonachlor; polychlorina patterns; culturally defin bitch horache anorqui int	r products; dietary fibre; d fruits; pulses (legumes); meat; poultry; fish; eggs; fats and oils; vegetable fat; coffee; tea; carbohydrate; croftei; vitamin A; riboflavin; ini B12; vitamin A; riboflavin; no; selenium; carotenoids; henyldichloroethylene; coethane; dieldrin; cachlorocyclohexane; trans- ted biphenyls; dietary hed diets; birth weight; vise bing heartfod
Substantial effect on risk unlikely	None ic	dentified	Substantial effect on risk unlikely	None id	dentified
<ol> <li>Adult attained h marker for gene affecting growth linear growth (s</li> <li>Physical activity recreational.</li> </ol>	eight is unlikely directly to mo tic, environmental, hormonal, n during the period from prece e chapter 6.2.1.3). of all types: occupational, hou	wify the risk of cancer. It is a and also nutritional factors onception to completion of sehold, transport, and	<ol> <li>Adult attained I marker for gene affecting growth linear growth (s</li> <li>Physical activity recreational.</li> </ol>	height is unlikely directly to mo etic, environmental, hormonal, h during the period from prec ee chapte 6.2.1.3). of all types: occupational, hou	odify the risk of cancer. It is a and also nutritional factors onception to completion of sehold, transport, and
For an explanation of please see chapter 3 and the glossary.	or all the terms used in the mains. 5.5.1, the text of this section,	Weil American Casee Recently Fand	For an explanation please see chapter 3 and the glossary.	of all the terms used in the ma 3.5.1, the text of this section,	trix, Generation Rowenth final

World Cancer Research Fund /American Institute for Cancer Research. Food, Nutrition, Physical Activity, and the Prevention of Cancer: a Global Perspective. Washington, DC: AICR, 2007. P289

#### Appendix B.

Summary of research findings relating to supplement use from the 2007 WCRF report on food nutrition, physical activity and cancer prevention

DIETARY CO	NSTITUENTS AND	SUPPLEMENTS, AND TH	E RISK OF CANCER	
In the judgement of	the Panel, the factors listed	below modify the risk of cancer. Judgen	nents are graded according to the str	ength of the evidence.
	DECF	REASES RISK	INCREASE	S RISK
	Exposure	Cancer site	Exposure	Cancer site
Convincing			Beta-carotene supplements <sup>1</sup>	Lung
Probable	Calcium² Selenium³	Colorectum Prostate		
Limited — suggestive	Retinol <sup>4</sup> Alpha-tocopherol <sup>2</sup> Selenium <sup>3</sup>	Skin <sup>5</sup> Prostate Lung <sup>3</sup> Colorectum <sup>6</sup>	Retinol supplements <sup>1</sup> Selenium supplements <sup>2</sup>	Lung Skin
Substantial effect on risk unlikely		Beta-carotene <sup>7</sup> : prostat	te; skin (non-melanoma)	
The evidence is de     Applies only to sqi     The evidence is de     The evidence is de     beta-carotene: see     For an explanation of	rived from studies using high- drived from studies using supple rived from studies using supple rived from studies using supple rived from studies using supple rived from studies using supple a chapter 4.2. all the terms used in the matrix	tose supplements (20 mg/day for beta-carof iments at a dose of 200 µg/day. iments at 200 µg/day. Selenium is toxic at h iments at a dose of 25 000 international un iments at a dose of 200 µg/day. Selenium is iments (at doses of 20, 30, 50 mg for prosta s, please see chapter 3.5.1, the text of this s	tene; 25 000 international units/day for igh doses. its/day. toxic at high doses. te, and doses of 30, 50 mg/day for skin), ection, and the glossary.	and foods containing

World Cancer Research Fund /American Institute for Cancer Research. Food, Nutrition, Physical Activity, and the Prevention of Cancer: a Global Perspective. Washington, DC: AICR, 2007. P289

#### Appendix C.

#### Search strategy for vitamin C and breast cancer

#### OVID MEDLINE (Jan 2006 - June 2011)

- 1. (vitamin\$ adj5 c).ti,ab,sh.
- 2. Ascorbate.ti,ab,sh.
- 3. ascorbic acid.ti,ab,sh.
- 4. exp Ascorbic acid/
- 5. or/1-4
- 6. antioxidant\*.ti,ab,sh.
- 7. anti-oxidant\*.ti,ab,sh.
- 8. antioxidants/
- 9. or/6-8
- 10. vitamin\*.ti,ab.
- 11. vitamins/
- 12. 10 or 11
- 13. 9 and 12
- 14. (micronutrient\$ or nutrient\$).ti,ab,sh.
- 15. fruit\$.ti,ab,sh.
- 16. vegetable\$.ti,ab,sh.
- 17. 14 and 15 and 16
- 18. 5 or 13 or 17
- 19. exp breast neoplasms/
- 20. exp neoplasms, hormone-dependent/
- 21. (breast adj25 neoplasm\$).ti,ab.
- 22. (breast adj25 cancer\$).ti,ab.
- 23. (breast adj25 tumo\$).ti,ab.
- 24. (breast adj25 malign\$).ti,ab.
- 25. (breast adj25 carcinoma\$).ti,ab.
- 26. (breast adj25 adenocarcinoma\$).ti,ab.
- 27. or/19-26
- 28. 18 and 27
- 29. Food/
- 30. Diet/
- 31. Nutrition Surveys/
- 32. Dietary Supplements/
- 33. diet records/
- 34. (diet\$ or intake\$ or nutri\$ or food\$ or supplement\$).ti,ab,sh.

- 35. or/29-34
- 36. exp cohort studies/
- 37. exp prospective studies/
- 38. exp follow-up studies/
- 39. exp longitudinal studies/
- 40. exp diet surveys/
- 41. cohort\$.ti,ab,sh.
- 42. longitudinal\$.ti,ab,sh.
- 43. prospective\$.ti,ab,sh.
- 44. controlled clinical trial.pt.
- 45. epidemiologic methods/
- 46. limit 45 to yr=1966-1989
- 47. exp case-control studies/
- 48. (case\$ and control\$).tw.
- 49. or/36-44,46-48
- 50. "randomized controlled trial".pt.
- 51. (random\$ or placebo\$ or single blind\$ or double blind\$ or triple blind\$).ti,ab.
- 52. (retraction of publication or retracted publication).pt.
- 53. or/50-52
- 54. (animals not humans).sh.

55. ((comment or editorial or meta-analysis or practice-guideline or review or letter or journal correspondence) not "randomized controlled trial").pt.

56. (random sampl\$ or random digit\$ or random effect\$ or random survey or random regression).ti,ab. not "randomized controlled trial".pt.

- 57. 53 not (54 or 55 or 56)
- 58. (review or review, tutorial or review, academic).pt.
- 59. (pooling or pooled or mantel haenszel).tw,sh.
- 60. meta-analysis.pt.
- 61. meta-analysis.sh.
- 62. (meta-analys\$ or meta analys\$ or metaanalys\$).tw,sh.
- 63. (systematic\$ adj5 review\$).tw,sh.
- 64. (systematic\$ adj5 overview\$).tw,sh.
- 65. (quantitativ\$ adj5 review\$).tw,sh.
- 66. (quantitativ\$ adj5 overview\$).tw,sh.
- 67. (quantitativ\$ adj5 synthesis\$).tw,sh.
- 68. (methodologic\$ adj5 review\$).tw,sh.
- 69. (methodologic\$ adj5 overview\$).tw,sh.
- 70. (integrative research review\$ or research integration).tw.

71. or/58-7072. 49 or 57 or 7173. 28 and 35 and 72

Note relating to search: it was possible that articles may have been missed which reported the analysis of vitamin C and breast cancer risk within the main body of the article but did not include the key search terms in the title, subject heading and abstract. To check this, a search was undertaken restricting key words to those indicating only cohorts (i.e. *cohort, prospective, follow-up, longitudinal*), dietary studies and breast cancer. Titles and abstracts for 303 articles were found and examined. Only four of these, over and above those found in the initial search appear to be potentially relevant; the full articles were retrieved but these did not meet the full inclusion criteria.

#### Appendix D.

#### NICE guidelines used to identify women in the UKWCS who may be at raised risk or high risk of developing breast cancer

The following NICE guidelines <sup>319</sup> were used to identify women in the UKWCS who may be at raised risk as those who met any of the following criteria:

- One 1st degree relative and one 2nd degree relative diagnosed with breast cancer before average age 50
- Two 1st degree relatives diagnosed with breast cancer before average age 50
- Three or more 1st or 2nd degree relatives diagnosed with breast cancer at any age
- One 1st degree male relative diagnosed with breast cancer at any age
- One 1st or 2nd degree relative with ovarian cancer at any age and one 1st or 2nd degree relative with breast cancer at any age (one should be a 1st degree relative)

Note: First-degree relatives: mother; father; daughter; son; sister; brother. Seconddegree relatives: grandparent; grandchild; aunt; uncle; niece and nephew; half sister and half brother.

The following NICE guidelines<sup>319</sup> were used to identify women in the UKWCS who may be at high risk as those who met any of the following criteria:

- Two 1st or 2nd degree relatives diagnosed with breast cancer before average age 50
- Three 1st or 2nd degree relatives diagnosed before average age 60
- Four relatives including one 1<sup>st</sup> degree relative diagnosed at any age

Or one relative diagnosed with ovarian cancer at any age and on the same side of the family there is

- One 1st (including relative with ovarian cancer) or one 2nd degree relative diagnosed with breast cancer before age 50
- One additional relative diagnosed with ovarian cancer at any age
- Two 1st or 2nd degree relatives diagnosed with breast cancer before average age 60

Or a relative with bilateral breast cancer as follows

- One 1st degree relative with cancer diagnosed in both breasts before average age 50
- One 1st or 2nd degree relative diagnosed with bilateral breast cancer *and* one 1st or 2nd degree relative diagnosed with breast cancer before average age 60

Or one male breast cancer at any age and on the same side of the family there is

•

• Two 1st or 2nd degree relatives diagnosed with breast cancer before average age 60

Unfortunately the NICE guidelines could not be followed in some areas; therefore the categories are only estimates. From the UKWCS data it was impossible to determine whether aunts were from the mothers' or the fathers' side of the family. Additionally, there was no information on daughters or sons with breast cancer or any relatives with bilateral breast cancer in the UKWCS.

#### Appendix E.

#### Poster and oral conference presentations by Jayne Hutchinson

## Symposium: University of Leeds LIGHT's Postgraduate Symposium. 24 January 2012

**Oral presentation:** Vitamin C intake from diary recordings and risk of breast cancer in the UK Dietary Cohort Consortium (based on results from chapter 10)

### Conference: IEA World Conference of Epidemiology. Edinburgh, 7-10 August 2011

**Poster:** Use of Supplements containing vitamin C and breast cancer risk in the UK Women's Cohort Study (based on results from chapter 8)

**Poster:** Vitamin C intake from diary recordings and risk of breast cancer in the UK Dietary Cohort Consortium (based on results from chapter 10)

Abstracts published in the Journal of Epidemiology and Community Health, August 2011, Vol 65, supplement 1

# Conference: World Cancer Research Fund: Nutrition, Physical Activity and Cancer Prevention: Current Challenges, New Horizons. London 12-13 September 2010

**Poster:** High dose vitamin C supplement use is associated with a personal and family history of cancer (based on chapter 6)

## Conference: University of Leeds Faculty of Medicine and Health Conference Day 2009: Identifying the Importance and Impact of Your Research. Weetwood Hall, Leeds 4 November 2009

**Poster:** Do women who take supplements have a greater risk of cancer? (awarded poster prize)

	Lt	isted below are food items divide o indicate how often, on average nonths.	ed into sectio , you have e	ons acc aten th	ording e spec	to food fied an	i type. nount c	Pleason Pleaso	e put a	a tick ( during	✓ ) in the last	the box ist 12
	E	Example: White bread, so if you e lay".	eat 4 or 5 slid	ces a d	ay, you	should	l put a	tick in	the co	olumn	heade	d "4-5
TY Ox		FOODS AND AMOUNTS	HOW OFT	EN HAVE	E YOU E	TEN TH	ESE FOO	DDS IN	THE LA	ST 12 N	IONTHS	2
			NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
TIN CO S		BREAD White slices or rolls	o	1	2	3	4	5	6	7	8√	9
		Example: For seasonal fruit such season you should put a tick in t	as strawber	ries, if eaded	you ea "once a	strawt week	berries	about	once	a week	k wher	
		FOODS AND AMOUNTS	HOW OF	TEN HAV	E YOU E	ATEN TH	IESE FC	ODS W	HEN IN	SEASO	N?	0.02590
The UK Women's Nutrition & Lifestyle Survey		6	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
		FRUIT Strawberries	0	1	2	3	4	S	6	7	8	9
		If you make a mistake and put a another tick in the correct box. Example: If you eat apples twice through as shown, and tick in the	tick in the wi a week, but 2-4 per we	ticked	the "2- x instea	ross th 3 times ad.	rough daily"	the tic	k as s istead,	hown t	oelow, ross th	and put
This questionnaire is mostly about your usual food intake over the last year. It is		FOODS AND AMOUNTS	HOW OF	TEN HA	VE YOU I	EATEN T	HESE FO	N SOOC	N THE L	AST 12	MONTH	IS?
designed for both vegetarians and non-vegetarians, so some questions may not seem relevant to you. There are also some questions about other topics such as smoking and exercise.			NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
Please answer every question. If you are uncertain about how to answer a question then do the best you can, but please do not leave a question blank. Don't be put off once you've started! It may be quite lengthy but it is straightforward and quick to work your way through.		FRUIT Apples	0	1	2	3	4	5	6	X	8	9
We want to find out about the relationship of nutrition with the occurrence of certain diseases. Please complete this questionnaire and return it in the pre-paid envelope as soon as possible. Your answers will be treated as strictly confidential and will be used only for medical research.												

# **Baseline combined Food** Lifestyle Questionnaire Frequency Questionnaire and Health and

Appendix F.

FOOD INTAKE:

CONFIDENTIAL

FOODS AND AMOUNTS	HOW OF	TEN HAV	E YOU E	ATEN TH	IESE FC	ODS IN	THE LA	ST 12	NONTH	IS?
	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ pe da
BREAD/SAVOURY BISCUITS				131				26.2		
White bread & rolls	0	ĭ .	2	3	4	5	6	7	8	9
Brown bread & rolls	0	1	2	3	4	5	6	7	8	9
Wholemeal bread & rolls	0	1	2	3	4	5	6	7	8	9
Chapatis, Nan, Paratha	0	1	2	3	4	5	6	7	8	9
Papadums	0	1	2	3	4	5	6	7	8	9
Tortillas	0	1	2	3	4	5	6	7	8	9
Pitta Bread	0	1	2	3	4	5	6	7	8	9
Crispbread e.g. Ryvita	0	1	2	3	4	5	6	7	8	9
Cream crackers, cheese biscuits	0	1	2	3	4	5	6	7	8	9
BREAKFAST CEREALS										
Porridge, Readybrek	0	1	2	3	4	5	6	7	8	9
Sugar coated cereals e.g. Sugar Puffs	o	1	2	3	4	5	6	7	8	9
Non-sugar coated cereals e.g. Cornflakes, Rice Krispies	0	1	2	3	4	5	6	7	8	9
Muesli	0	1	2	3	4	5	6	7	8	9
All Bran, Bran Flakes	0	1	2	3	4	5	6	7	8	g
Weetabix, Shredded Wheat	0	1	2	3	4	5	6	7	8	9
POTATOES, RICE & PASTA										
Potatoes e.g. boiled, mashed	0	1	2	3	4	5	6	7	8	9
Chips	0	1	2	3	4	5	6	7	8	9
Jacket Potato	0	1	2	3	4	5	6	7	8	9
Roast Potatoes	0	10020	2	3	4	5	8	7	8	9
Potato Salad	0	1	2	3	4	5	6	7	8	9
White Pasta e.g. Spaghetti, Green Pasta, Red Pasta, Noodles	0	1	2	3	4	5	6	7	8	9
Wholemeal Pasta, Brown Spaghetti	0	1	2	3	4	5	6	7	8	9
White Rice	0	1	2	3	4	5	6	7	8	9
Brown Rice	0	1	2	3	4	5	6	7	8	9
Wild Rice	0	1	2	3	4	5	6	7	8	9
Macaroni Cheese	0	1	2	3	4	5	6	7	8	9

OODS AND AMOUNTS	HOW OF	TEN HAV	E YOU E	ATEN TH	HESE FC	ODS IN	THE LA	ST 12 J	MONTH	HS?
	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
DAIRY & NON-DAIRY PRODUCTS	6			-		300			1	
Thick & Creamy Yoghurt (125g carton)	0	1	2	3	4	5	6	7	8	9
ow fat Yoghurt (125g carton)	0	1	2	3	4	5	6	7	8	9
Diet Yoghurt (125g carton)	0	1	2	3	4	5	6	7	8	9
Greek Yoghurt (125g carton)	0	1	2	3	4	5	6	7	8	9
Fromage Frais/Creme Fraiche 125g -carton)	0	1	2	3	4	5	6	7	8	9
Dairy Desserts (125g carton)	0	1	2	3	4	5	6	7	8	9
Single/Sour Cream (tablespoon)	0	1	2	3	4	5	6	7	8	9
Double/Clotted Cream (tablespoon)	0	1	2	3	4	5	6	7	8	9
cecream	0	1	2	3	4	5	6	7	8	9
Vilk Puddings	0	1	2	3	4	5	6	7	8	9
.ow-fat Cheese	0	1	2	3	4	5	6	7	8	9
Cheese e.g. Cheddar, Brie, Edam	0	1	2	3	4	5	6	7	8	9
Cottage Cheese	0	1	2	3	4	6	6	7	8	9
Cheese and Onion Pastie	0	1	2	3	4	5	6	7	8	9
Soya Cheese	0	1	2	3	4	5	6	7	8	9
Soya Yoghurt	0	1	2	3	4	5	6	7	8	9
MARGARINES/BUTTERS & SPREADS										
Butter (enough for 1 slice of bread)	0	1	2	3	4	5	6	7	8	9
Block Margarine e.g. Stork, Krona, NOT in tub (enough for 1 slice of bread)	0	1	2	3	4	5	6	7	8	9
Polyunsaturated Margarine e.g. Flora, Sunflower, Granose, in tub (enough for 1 slice of bread)	0	1	2	3	4	5	6	7	8	9
Other soft Margarine, Dairy spreads e.g. Blue Band, Clover, in tub (enough for 1 slice of bread)	0	a	2	3	4	5	6	7	8	9
Low fat spread e.g. Outline, Gold, Flora Lite, in tub (enough for 1 slice of bread)	0	1	2	3	4	5	6	Ŷ	8	9
Very low fat spread, in tuble.g. St Ivel Lowest Fat Spread (enough for 1 slice of bread)	0	1	2	3	4	5	6	7	8	9
Monounsaturated Margarine eg Mono,	0	1	2	3	4	5	6	7	8	9

FOODS AND AMOUNTS	HOW OF	TEN HAV	E YOU E	ATEN T	HESE FO	DODS IN	THE LA	ST 12	MONTH	HS?
	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
SPREADS										
Marmite/Bovril/Vegemite	0	1	2	3	4	5	6	7	8	9
Peanut Butter	0	1	2	3	4	5	6	7	8	9
Chocolate/Chocolate & Nut Spread	0	1	2	3	4	5	6	7	8	9
Jam/Marmalade	0	1	2	3	4	5	6	7	8	9
Honey	0	1	2	3	4	5	6	7	8	9
Vegetable pâté	0	1	2	3	4	5	6	7	8	9
Nut Pàté	0	1	2	3	4	5	6	7	8	9
SAUCES & SOUPS			-		100		. 4	1	- 10- 5	
Low Calorie Salad Cream (tablespoon)	0	1	2	3	4	5	6	7	8	9
Mayonnaise, Salad Cream Type Dressing (tablespoon)	0	.1	2	3	4	5	6	7	8	9
French Type Dressing (tablespoon)	0	1	2	3	4	5	6	7	8	9
Sauces e.g. white/cheese/'Cook In'/curry	0	1	2	3	4	5	6	7	8	9
Tomato Ketchup (tablespoon)	a	1	2	3	4	5	6	7	8	9
Pickles/Chutney/Pesto sauce	0	1	2	3	4	5	6	7	8	9
Packet Soups - Meat & Veg (Bowl)	0	1	2	3	4	5	6	7	8	9
Other - Vegetable Soups (Bowl)	0	1	2	3	4	5	6	7	8	9
Other - Meat Soups (Bowl)	0	1	2	3	4	5	6	7	8	9
Low Calorie Soups (Bowl)	0	1	2	3	4	5	6	7	8	9
GRAINS (Medium serving)										
Barley	0	1	2	3	4	5	6	7	8	9
Oats	0	1	2	3	4	5	6	7	8	9
Bulgar Wheat	0	1	2	3	4	5	6	7	8	9
Wheat Germ (tablespoon)	0	1	2	з	4	5	6	7	8	9
Cous-cous	0	1	2	3	4	5	6	7	8	9
White Rice	0	1	2	3	4	5	6	7	8	9
Brown Rice	0	1	2	3	4	5	6	7	8	9
NUTS & SEEDS		a solar		1			8			
Peanuts/Pistachio Nuts	0	1	2	3	4	5	6	7	8	9
Cashew Nuts & Almonds	0	1	2	3	4	5	6	7	8	9
Pecan Nuts/Walnuts	0	1	2	3	4	5	6	7	8	9
Sunflower Seeds/ Sesame Seeds	0	1	2	3	4	5	6	7	A	0

FOODS AND AMOUNTS	HOW OF	TEN HAV	E YOU EA	TEN THE	ESE FO	DDS IN	THE LAS	T 12 M	ONTH	s?
	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
PULSES (Include when used in recipes)									N.	
Lentils, dals	0	1	2	3	4	5	6	7	8	9
Chick Peas, Chanas	0	1	2	3	4	5	6	7	8	9
Hummus	0	1	2	3	4	5	6	7	8	9
Baked beans	0	1	2	3	4	5	6	7	8	9
Mung Beans & Red Kidney Beans	0	1	2	3	4	5	6	7	8	9
Bean Sprouts	0	1	2	3	4	5	6	7	8	9
Black Eyed Beans	0	1	2	3	4	5	6	7	8	9
Butter Beans/Broad Beans	0	1	2	3	4	5	6	7	8	9
EGGS/EGG DISHES										
Boiled/ Poached egg	0	1	2	3	4	5	6	7	8	9
Omelette, Scrambled egg	0	1	2	3	4	5	6	7	8	9
Fried egg	0	1	2	3	4	5	6	7	8	9
Quiche	0	1	2	3	4	5	6	7	8	9
VEGETABLE DISHES										
Quorn	0	1	2	3	4	5	6	7	8	9
Textured vegetable protein/ Sosmix/burger mix/soya sausages	0	1	2	3	4	5	6	7	8	9
Vegetarian Chilli/Vegetable Curry	0	1	2	3	4	5	6	7	8	9
Mixed Bean Casserole/Ratatouille	0	1	2	з	4	5	6	7	8	9
Stir-fry vegetables	0	4	2	3	4	5	6		8	9
Vegetable - Lasagne/Moussaka/Ravioli/ filled pasta with sauce	0	1	2	3	4	5	6	7	8	9
Vegetable Pizza	0	1	2	3	4	5	6	7	8	9
MEAT										
Beef e.g. roast, steak	0	1	2	3	4	5	6	7	8	9
Beef Stew/Casserole/Mince/Curry	0	t	2	3	4	5	6	7	8	9
Beefburger/Hamburger	0	1	2	3	4	5	6	7	8	9
Pork e.g. Roast, Chops, Slices	0	1	2	3	4	5	6	7	8	9
Pork Stew/Casserole	Ō	1	2	3	4	5	6	7	8	9
Lamb e.g. Roast, Chops	0	1	2	3	4	5	6	7	8	9
Lamb Stew/Casserole	0	1	2	3	4	5	6	7	8	9

FOODS AND AMOUNTS	HOW OF	TEN HAV	E YOU EA	TEN TH	ESE FOO	DS IN T	THE LAS	T 12 M	ONTHS	3?
	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
OTHER MEATS										
Chicken/Turkey roast, slices	0	1	2	3	4	5	6	7	8	9
Breadcrumbed e.g.chicken nuggets/kievs	0	1	2	3	4	5	6	7	8	9
Chicken/Turkey in creamy sauce, curry	0	i	2	3	4	5	6	7	8	9
Bacon	0	1	2	3	4	5	6	7	8	9
Ham	0	1	2	3	4	5	6	7	8	9
Corned Beef, Spam, Luncheon Meats	0	1	2	3	4	5	6	7	8	9
Sausages e.g. Beef Pork	Ö	1	2	3	4	5	6	7	8	9
Pies/Pasties/Sausage Rolls	0	1	2	3	4	5	6	7	8	9
Offal e.g. Liver, Kidney	0	1	2	3	4	5	6	7	8	9
Liver Pâté/Sausage, Salami	0	1	2	3	4	5	6	7	8	9
Meat - Lasagne/Moussaka/Ravioli/ filled pasta with sauce	0	1	2	3	4	5	6	7	8	9
Meat Pizza	0	1	2	3	4	5	6	7	8	9
FISH										
Fish fingers/cakes	0	1	2	3	4	5	6	7	8	9
Fried fish in batter (as in fish and chips)	0	1	2	3	4	5	6	7	8	9
White fish e.g. Cod, Haddock, Plaice, Sole, Halibut (fresh or frozen)	0	1	2	3	4	5	6	7	8	9
Oily fish e.g. Mackerel, Kippers, Tuna, Salmon, Sardines, Herring	0	1	2	3	4	5	6	7	8	9
Shellfish e.g. Crab, Prawns, Mussels	0	1	2	3	4	5	6	7	8	9
Fish Roe, Taramasalata	0	1	2	3	4	5	6	7	8	9
Fish Pie/Fish Lasagne	0	1	2	3	4	5	6	7	8	9
VEGETABLES										
Beetroot	0	1	2	3	4	5	6	7	8	9
Broccoli, Spring Greens, Kale	0	1	2	3	4	5	6	7	8	9
Brussel Sprouts	0	1	2	3	4	5	6	7	8	9
Cabbage	0	1	2	3	4	5	6	7	8	9
Carrots	0	1	2	3	4	5	6	7	8	9
Cauliflower	0	1	2	3	4	5	6	7	8	9
Celery	0 .	1	2	3	4	5	6	7	8	9

5

Please estimate how often you eat the following foods, and please answer every question.

#### PLEASE PUT A TICK(✓) ON EVERY LINE

	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
VEGETABLES (continued)								-		
Colesiaw	0	1	2	3	4	5	6	7	8	9
Low-calorie Coleslaw	0	1	2	3	4	5	8	7	8	9
Courgettes, Marrow, Squash	0	1	2	3	4	5	6	7	8	9
Cucumber	0	1	2	3	4	5	6	7	8	9
Garlic	0	1	2	3	4	5	6	7	8	9
Green Beans, Runner Beans	0	1	2	3	4	5	6	7	8	9
Leeks	0	1	2	3	4	5	6	7	8	9
Lettuce	0	1	2	3	4	5	6	7	8	9
Mushrooms	0	1	2	3	4	5	6	7	8	9
Aubergine, Okra/Ladies Finger	0	1	2	3	4	5	6	7	8	9
Olives	0	1	2	3	4	5	6	7	8	g
Parsnips	0	1	2	3	4	5	6	7	8	9
Peas, Mushy peas, Mange-tout	0	1	2	3	4	5	6	7	8	9
Peppers - Red, Green, Yellow, Black etc	0	1	2	3	4	5	6	7	8	9
Swedes	0	1	2	3	4	5	6	7	8	9
Sweetcorn	0	1	2	3	4	5	6	7	8	9
Tomatoes - raw/canned/sauce	0	1	2	3	4	5	6	7	8	9
Turnip	0	1	2	3	4	5	6	7	8	9
Watercress, Mustard & Cress	0	1	2	3	4	5	6	7	8	9
FRUIT										
Apples	0	1	2	3	4	5	6	7	8	9
Avocado	0	1	2	3	4	5	6	7	8	9
Bananas	0	1	2	3	4	5	6	7	8	9
Grapes	0	1	2	3	4	5	6	7	8	9
Kiwi Fruit	0	1	2	3	4	5	6	7	8	ę
Mangoes	0	1	2	3	4	5	6	7	8	4
Oranges, Satsumas, Grapefruit, etc	0	1	2	3	4	5	6	7	8	5
Papaya	0	1	2	3	4	5	6	7	8	3
Pears	0	1	2	3	4	5	6	7	8	1
Pineapple	0	1	2	3	4	5	6	7	8	

OODS AND AMOUNTS	HOW OF	TEN HAN	E YOU E	ATEN TH	HESE FC	ODS IN 1	THE LA	ST 12	MONT	HS?
	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ per day
SEASONAL FRUIT How often have you eaten these fruits, when they are in season?										
Apricots	0	1	2	3	4	5	6	7	8	9
Melon	0	1	2	3	4	5	6	7	8	9
Nectarines	0	1	2	3	4	5	6	7	8	9
Peaches	0	1	2	3	4	5	6	7	8	9
Plums	0	1	2	3	4	5	6	7	8	9
Raspberries	0	1	2	3	4	5	6	7	8	9
Red currants/Black currants	0	1	2	3	4	5	6	7	8	9
Rhubarb	0	1	2	3	4	5	6	7	8	9
Strawberries	0	1	2	3	4	5	6	7	8	9
DRIED FRUIT										
Dates	0	1	2	3	4	5	6	7	8	9
Figs	0	1	2	3	4	5	6	7	8	9
Prunes	0	1	2	3	4	5	6	7	8	9
Mixed Dried Fruit e.g. Apricots, Apples, Pears, Mangoes	0	1	2	3	4	5	6	7	8	9
Currants, Raisins, Sultanas	0	1	2	3	4	5	6	7	8	9
SWEET SNACKS										
Cereal Bars/Flapjacks (one)	0	1	2	3	4	5	6	7	8	9
Fruit bars (one) eg Apricot, Date	0	1	2	3	4	5	6	7	8	9
Chocolate Snack Bars e.g. Mars, Crunchie (1 bar)	0	1	2	3	4	5	6	7	8	9
Mini chocolate snack bars, Chocolates - singles or squares (1)	0	1	2	з	4	5	6	7	8	9
Boiled Sweets, Toffees, Mints	0	1	2	3	4	5	6	7	8	9
SAVOURY SNACKS										
Crisps (1 bag)	0	1	2	3	4	5	6	7	8	9
Other fried snacks e.g. Wotsits (1 bag)	0	1	2	3	4	5	6	7	8	9
Low fat or baked snacks e.g. Low-fat Crisps (1 bag)	0	1	2	3	4	5	6	7	8	9
Bombay Mix (small handful)	0	1	2	3	4	5	6	7	8	9
Peanuts/Pistachio Nuts (small handful)	0	1	2	3	4	5	6	7	8	9
Mixed Nuts and Raisins (small handful)	0	1	2	3	4	5	6	7	8	9

DODS AND AMOUNTS	HOW OF	TEN HAV	E YOU EA	TEN THE	ESE FOC	DS IN T	HE LAS	T 12 M	DNTHS	?
	NEVER	Less than once a month	1-3 per month	once a week	2-4 per week	5-6 per week	once per day	2-3 per day	4-5 per day	6+ pe da
EVERAGES					11		1	-		
ea (cup)	0	1	2	3	4	5	6	7	8	9
lerbal Tea (cup)	0	1	2	3	4	5	6	7	8	9
Coffee - instant/ground (cup)	0	1	2	3	4	5	6	7	8	9
Coffee - decaffeinated (cup)	0	1	2	3	4	5	6	7	8	9
Coffee substitute e.g Caro/Bambu (cup)	0	1	2	3	4	5	6	7	8	9
Coffee whitener (teaspoon)	0	1	2	3	4	5	6	7	8	9
Cocoa, Hot Chocolate (cup)	0	1	2	3	4	5	6	7	8	9
Horlicks, Ovaltine (cup)	0	1	2	3	4	5	6	7	8	9
ow Calorie/Low-fat Horlicks, Ovaltine, Hot Chocolate, (cup)	0	1	2	3	4	5	6	7	8	
Orange Juice (Pure Fruit) (glass)	0	1	2	-3	4	5	6	7	8	-
Other -(100%) Pure fruit juices (glass)	0	1	2	3	4	5	6	7	8	
Fruit Squash/Cordial - diluted (glass)	0	1	2	3	4	5	6	7	8	
Fizzy soft drinks e.g. Coke, Lemonade (glass/can)	0	1	2	3	4	5	6	7	8	
Low Calorie/Diet Soft Drinks (glass/can)	0	1	2	3	4	5	6	7	8	
ALCOHOLIC BEVERAGES		-				4				1
Wines (wineglassful)	0	1	2	3	4	5	6	7	8	-
Beer, Lager (half pint)	0	1	2	3	4	5	6	7	8	_
Cider (half pint)	0	1	2	З	4	5	6	7	8	
Port, Sherry, Liqueurs (glass)	0	1	2	3	4	5	6	7	8	_
Spirits e.g. Whisky, Gin, Vodka, Brandy (single/1 measure)	0	1	2	3	4	5	6	7	8	
BISCUITS, SWEETS & PUDDINGS					-		-	-		-
Plain Biscuits e.g. Marie, Nice, Digestive (one)	0	1	2	3	4	5	6	7	8	
Chocolate Biscuits (one)	0	1	2	3	4	5	6	7	8	
Sandwich/Cream Biscuits (one)	0	1	2	3	4	5	6	7	8	
Fruitcake (1 slice)	0	1	2	3	4	5	6	7	8	
Sponge cakes (1 slice)	0	1	2	3	4	5	6	7	8	
Buns/Pastries e.g. Croissants, Doughnuts, Tray Bakes, (one)	0	1	2	3	4	5	6	7	8	
Scones/Pancakes/Muffins/Crumpets (1)	0	1	2	3	4	5	6	7	8	1
Fruit Pies, Tarts, Crumbles, (1 slice)	0	1	2	3	4	5	6	7	8	
Sponge Puddings (1 serving)	0	1	2	3	4	5	6	.7	8	_

Coner Poods						VEGETABLES: 7: How many servi	ngs of vegetabl	es or vegeta	ble containi	ng dishes,		
Are there any other fo	ods which you eat m	ore than once a we	ek?	Yes 1	No 2	(excluding potatoes)	do vou usually	eat each we	eek?			
If yes, please list below	w			TOO Inmand			autorea hauren	uph of the vis	a atablas vo	u oat ara Er	ach Erozar	Canned or Drie
Food	Usua	l serving size	Number o	of times eaten ea	ach week	8: Call you please i	priste hoves e	a <sup>1</sup> / <sub>4</sub> Dried <sup>3</sup>	Byetables yo	u cataro i i	0.011, 110201	i, cuintea er erte
						Flease lick the applo	priate boxes, e.	g. /•Dricu,	1.1102011.			
								10	roportion	31	All	
				and the second second			Never	/4	/2	-/4	All	
						Fresh	1	2	2	4	5	
Would you describe yo	ourself as a vegetaria	n?		Vec 1		Frozen	1	2	3	4	5	
If yes, how long have	vou been vegetarian	vears			100	Dried	1	2	3	4	5	
	,					Canned						
Would you describe y	ourself as a vegan?			Yes 1	No 2	9: Do you ever eat	raw vegetables	apart from	salad vegeta	ables?		Yes 1 N
If yes, how long have	you been vegan?	years			1713	10: How do you usu	ally cook your y	egetables 7	(Excluding n	otatoes). Tir	sk more tha	in one box if nece
						to. There do you usu	any cook your v	- Jorgio ant				Detto
: Do you use herbs and	spices at least once j	per week when coo	king food?	Yes 1	No 2							Bouing
Which fresh herbs and	d spices would you u	se at least once a w	veek? Please lis	sthere						a		Steaming
	in the second second				the second second	Grilli	ng/Barbecuing/I	Jaking/Roas	sting (Cooke	a ary or usi	ng a small	amount of on)
Which dried herbs and	d spices would you u	se at least once a w	veek? Please lis	st here							Sur Frying	g/Frying/Saule
										Deve	- in inclu	which down a which
										Deep II	ying - mon	uning in batter
										Cas	carolina//R	aking in cause
ORTION SIZE:										Cas	seroling//B	aking in sauce
PORTION SIZE: Compared to other per	ople would you descr	ibe your typical ave	erage portion s	size of foods as?						Cas	seroling//B	aking in sauce Other
ORTION SIZE: : Compared to other peo	ople would you descr Small? 1	ibe your typical ave Medium?	erage portion s	size of foods as? Large? 3		Please describe				Cas	seroling//B	aking in sauce Other
ORTION SIZE: : Compared to other per	ople would you descr Small? 1	ibe your typical ave Medium?	erage portion s	size of foods as? Large? 3		Please describe				Cas	seroling//B	aking in sauce Other
ORTION SIZE: Compared to other per ULSES:	ople would you descr Small? <sup>1</sup>	ibe your typical ave Medium?	erage portion s	size of foods as? Large? 3		Please describe				Cas	seroling//B	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g.	ople would you descr Smail? <sup>1</sup> beans, peas, lentils e	ibe your typical av Medium? ttc. Yes	erage portion s	size of foods as? Large? 3 No 2		Please describe FRUIT: 11: How many serv	ings of fruit or f	ruit containi	ing dishes d	Cas o you usual	seroling//B	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qui	ople would you descr Small? <sup>1</sup> beans, peas, lentils e estion 7.	ibe your typical av Medium? ttc. Yes	erage portion s	size of foods as? Large? 3 No 2		Please describe FRUIT: 11: How many serv Can you please indic Please tot the appro	ings of fruit or f	ruit containi of the fruit y	ing dishes d /ou eat is Fro	Cas o you usual osh, Cannec	seroling//B ly eat each l, Dried or !	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qui	ople would you descr Small? <sup>1</sup> beans, peas, lentils e estion 7.	ibe your typical ave Medium? itc. Yes	erage portion s	size of foods as? Large? 3 No 2		Please describe FRUIT: 11: How many serv Can you please indic Please tick the appro	ings of fruit or f cate how much opriate boxes e.	ruit containi of the fruit y g. ¼ Fresh,	ing dishes d /ou eat is Fro <sup>3</sup> /4 Canned Proportion	Cas: o you usual osh, Cannec	seroling//B ly eat each l, Dried or 1	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qui Can you please indicat	ople would you descr Small? 1 beans, peas, lentils of estion 7. te how much of the P	ibe your typical av Medium? itc. Yes ulses you eat are F	erage portion s 2 1 resh, Frozen, (	size of foods as? Large? <sup>3</sup> No <sup>2</sup> Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the appro	ings of fruit or f cate how much opriate boxes e. Never	ruit containi of the fruit y g. ¼ Fresh, 1/4	ing dishes d rou eat is Fro <sup>3</sup> /4 Canned Proportion <sup>1</sup> /2	Cas: o you usual osh, Cannec <sup>3</sup> /4	seroling//B ly eat each l, Dried or S All	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. <b>If no,</b> please go to qui Can you please indicat ck the appropriate boxes,	ople would you descr Small? <sup>1</sup> beans, peas, lentils e estion 7. te how much of the P , e.g. <sup>1</sup> / <sub>4</sub> Dried, <sup>3</sup> / <sub>4</sub> Froz	ibe your typical av Medium? ttc. Yes ulses you eat are F en.	erage portion s	Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the appro	ings of fruit or f cate how much ppriate boxes e. Never	ruit containi of the fruit y g. ¼ Fresh, 4 1/4 2	ing dishes d you eat is Fro <sup>3</sup> /4 Canned Proportion <sup>1</sup> /2 3	Cas: o you usual osh, Cannec <sup>3</sup> /4 *	Iy eat each J, Dried or : All	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qu Can you please indicat ck the appropriate boxes,	ople would you descr Small? 1 beans, peas, lentils e estion 7. te how much of the P , e.g. <sup>1</sup> / <sub>4</sub> Dried, <sup>3</sup> / <sub>4</sub> Froz	ibe your typical av Medium? ttc. Yes ulses you eat are F en. Proportion	erage portion s	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the appro Fresh Stawed	ings of fruit or f cate how much opriate boxes e. Never	ruit containi of the fruit y g. ¼ Fresh, s ¼ 1/s 2 2	ing dishes d rou eat is Fro 3/4 Canned Proportion 1/2 3 3	Casi o you usual issh, Cannec <sup>3</sup> /4 4 4	lly eat each d, Dried or S All 5 5	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qui Can you please indicat ck the appropriate boxes,	ople would you descr Small? 1 beans, peas, lentils e estion 7. te how much of the P , e.g. ¼ Dried, ¾ Froz	ibe your typical av Medium? ttc. Yes ulses you eat are F en. Proportion 1/2 3/4	erage portion s	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the approv Fresh Stewed Dried	ings of fruit or f cate how much opriate boxes e. Never	ruit contain of the fruit y g. <sup>1</sup> /4 Fresh, <sup>1</sup> /4 2 2 2 2	ing dishes d rou eat is Fro 3/4 Canned Proportion 1/2 3 3 3 3 3	Casi o you usual o sish, Cannec <sup>3</sup> /4 4 4 4	Iy eat each I, Dried or : All 5 6 5	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qui Can you please indicat ck the appropriate boxes, Fresh	ople would you descr Small? 1 beans, peas, lentils e estion 7. te how much of the P , e.g. ¼ Dried, ¾ Froz Never 1/4 1 2	ibe your typical av Medium? htc. Yes ulses you eat are F en. Proportion 1/2 3/4 3	erage portion s 2 1 resh, Frozen, C All	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indis Please tick the appro Fresh Stewed Dried Canned	ings of fruit or f pate how much opriate boxes e. Never	ruit contain of the fruit y 1/4 2 2 2 2 2	ing dishes d rou eat is Fro 3/4 Canned Proportion 1/2 3 3 3 3 3 3	Cas o you usual ish, Cannec <sup>3</sup> /4 4 4 4 4 4	ly eat each b, Dried or : All 5 6 5 5 5 5	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qu Can you please indicat sk the appropriate boxes, Fresh Frozen	ople would you descr Small? 1 beans, peas, lentils of estion 7. te how much of the P , e.g. ¼ Dried, ¾ Froz Never 1/4 1 2 1 2 1 2	ibe your typical ava Medium? htc. Yes ulses you eat are F en. Proportion 1/2 3/4 3 3	erage portion s 2 1 resh, Frozen, C All 4 5 4 5	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indis Please tick the appro Fresh Stewed Dried Canned	ings of fruit or f cate how much opriate boxes e. Never	ruit contain of the fruit y g. <sup>7</sup> /4Fresh, 1/4 2 2 2 2 2 2 2	ing dishes d rou eat is Fro 3/4 Canned Proportion 1/2 3 3 3 3 3 3 3 3 3 3	Cas o you usual issh, Cannec <sup>3</sup> /4 4 4 4 4 4 4	ly eat each b, Dried or 3 All 5 5 5 5	aking in sauce Other
DRTION SIZE: Compared to other per JLSES: Do you eat pulses e.g. If no, please go to qu Can you please indicat sk the appropriate boxes, Fresh Frozen Dried	ople would you descr Small? 1 beans, peas, lentils of estion 7. te how much of the P , e.g. ¼ Dried, ¾ Froz Never 1/4 1 2 1 2 1 2 1 2	ibe your typical ava Medium? etc. Yes ulses you eat are F en. Proportion 1/2 3/4 3 3 3	erage portion s 2 1 resh, Frozen, C All 4 5 4 5 4 5 4 5 4 5 7 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8 7 8	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indis Please tick the approv Fresh Stewed Dried Canned	ings of fruit or f cate how much opriate boxes e. Never 1 1 1 1 1	ruit contain of the fruit y g. <sup>7</sup> / <sub>4</sub> Fresh, <sup>1</sup> / <sub>8</sub> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u>	ing dishes d /ou eat is Fro /4 Canned Proportion 1/2 3 3 3 3 2 3 3 2 3 3 2 3 3 3 3 3 3 3 3 3 3 3 3 3	Cas o you usual ssh, Cannec <sup>3</sup> /4 4 4 4 4 4 4 4 4 4 8	Ily eat each b, Dried or 3 All 5 5 5 5	aking in sauce Other
DRTION SIZE: Compared to other per JLSES: Do you eat pulses e.g. If no, please go to qu Can you please indicat sk the appropriate boxes, Fresh Frozen Dried Canned	ople would you descr Small? 1 beans, peas, lentils of estion 7. te how much of the P , e.g. <sup>1</sup> /4 Dried, <sup>3</sup> /4 Froz Never <sup>1</sup> /4 1 2 1 2 1 2 1 2	ibe your typical ava Medium? etc. Yes ulses you eat are F en. Proportion 1/2 3/4 3 3 3 3	erage portion s 2 1 resh, Frozen, C All 4 5 4 5 4 5 4 5	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the approvement Fresh Stewed Dried Canned 12: Do you ever com	ings of fruit or f cate how much opriate boxes e. Never T T T T T T T	ruit containi of the fruit y g. <sup>7</sup> / <sub>4</sub> Fresh, <sup>1</sup> / <sub>4</sub> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u>	ing dishes d rou eat is Fro /4 Canned Proportion 1/2 3 3 3 3 3 3 4 3 3 3 4 2 3 3 4 3 4 3 5 7 8 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7	Case o you usual rish, Cannec <sup>3</sup> /4 4 4 4 4 4 8 9 9 9 9 0 12	Ily eat each b, Dried or 1 All 5 5 5 5 5	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qu Can you please indicat ck the appropriate boxes, Fresh Frozen Dried Canned	ople would you descr Small? 1 beans, peas, lentils of estion 7. te how much of the P , e.g. <sup>1</sup> /4 Proz Never <sup>1</sup> /4 <u>1 2</u> <u>1 2</u> <u>1 2</u> <u>1 2</u>	ibe your typical ave Medium? etc. Yes ulses you eat are F en. Proportion 1/2 3/4 3 3 3 3	erage portion s 2 1 resh, Frozen, C All 4 5 4 5 4 5 4 5 4 5	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the appro Fresh Stewed Dried Canned 12: Do you ever cou 13: If so, how do you	ings of fruit or f cate how much opriate boxes e. Never T T T T T T Sok the fruit you bu usually cook	ruit containi of the fruit y g, <sup>7</sup> /sFresh, <sup>1</sup> /s 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ing dishes d rou eat is Fro 3/4 Canned Proportion 1/2 3 3 3 3 3 Yes 1 1	Casi o you usual osh, Cannec <sup>3</sup> /4 4 4 4 4 8 9 9 9 0 0 2	Ily eat each b, Dried or the All 5 5 5 5	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. <b>If no</b> , please go to qu Can you please indicat ck the appropriate boxes, Fresh Frozen Dried Canned ow do you usually cook p	ople would you descr Small? 1 beans, peas, lentils of estion 7. te how much of the P , e.g. <sup>1</sup> /4 Dried, <sup>3</sup> /4 Froz Never 1/4 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2	ibe your typical av Medium? etc. Yes ulses you eat are F en. Proportion 1/2 3/4 3 3 3 3	erage portion s 2 3 resh, Frozen, C 4 5 4 5 4 5 4 5 4 5 4 5	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the approver Fresh Stewed Dried Canned 12: Do you ever cou 13: If so, how do you Stewing	Never	ruit containi of the fruit y g, <sup>1</sup> / <sub>4</sub> Fresh, <u>1</u> / <sub>8</sub> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u> <u>2</u>	ing dishes d rou eat is Fro 3/4 Canned Proportion 1/2 3 3 3 3 3 Yes 1 1 Poaching/	Cas o you usual osh, Cannec <sup>3</sup> /4 4 4 4 4 4 8 No 2 2 Steaming	Ily eat each I, Dried or S All 5 5 5 5 5	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qu Can you please indication ck the appropriate boxes, Fresh Frozen Dried Canned ow do you usually cook p Steaming/Boiling/Pres	ople would you descr Small? 1 beans, peas, lentils e estion 7. te how much of the P , e.g. <sup>1</sup> /4 Dried, <sup>3</sup> /4 Froz Never 1/4 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2	ibe your typical av Medium? ttc. Yes ulses you eet are F en. Proportion 1/2 3/4 3 3 3 able. 1 Stewing	erage portion s 2 1 resh, Frozen, C All 4 5 4 5 4 5 4 5 9/Casseroling/I	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the approver Fresh Stewed Dried Canned 12: Do you ever cou 13: If so, how do you Stewing	Ings of fruit or f cate how much opriate boxes e. Never	ruit contain of the fruit y g. <sup>1</sup> / <sub>4</sub> Fresh, <u>1</u> / <sub>4</sub> 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ing dishes d rou eat is Fro 3/4 Canned Proportion 1/2 3 3 3 3 3 Yes 1 1 Poaching/	Cas o you usual ssh, Cannec <sup>3</sup> /4 4 4 4 4 8 No 2 2 Steaming	seroling//B	aking in sauce Other
PORTION SIZE: Compared to other per PULSES: Do you eat pulses e.g. If no, please go to qu Can you please indica- ick the appropriate boxes Fresh Frozen Dried Canned Steaming/Boiling/Pres Microwaving	ople would you descr Small? 1 beans, peas, lentils e estion 7. te how much of the P , e.g. <sup>1</sup> /4 Dried, <sup>3</sup> /4 Froz Never 1/4 1 2 1 2 1 2 1 2 1 2 1 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ibe your typical av Medium? ttc. Yes ulses you eat are F an. Proportion 1/2 3/4 3 3 able. 1 Stewing 2 Stir Fryi	erage portion s 2 1 resh, Frozen, C All 4 5 4 5 4 5 4 5 9/Casseroling/R ing/Frying	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the approv Fresh Stewed Dried Canned 12: Do you ever coo 13: If so, how do yo Stewing Baking	ings of fruit or f cate how much ppriate boxes e. Never 1 1 1 1 1 0 bk the fruit you pu usually cook	ruit contain of the fruit y g. <sup>1</sup> / <sub>4</sub> Fresh, <sup>1</sup> / <sub>4</sub> 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ing dishes d /ou eat is Fro /a Canned Proportion 1/2 3 3 3 Yes 1 Poaching/ Microway	Casi o you usual ish, Cannec <sup>3</sup> /4 4 4 4 4 8 1 9 No 2 2 Steaming ing	Iv eat each J, Dried or 3 All 5 5 5 5 2 1 2 2	aking in sauce Other
ORTION SIZE: Compared to other per ULSES: Do you eat pulses e.g. If no, please go to qu Can you please indication ck the appropriate boxes Fresh Frozen Dried Canned Steaming/Boiling/Prest Microwaving Brasting	ople would you descr Small? 1 beans, peas, lentils e estion 7. te how much of the P , e.g. <sup>1</sup> /4 Dried, <sup>3</sup> /4 Froz Never 1/4 <u>1 2</u> <u>1 2</u> <u>1 2</u> <u>1 2</u> oulses? Tick all applic ssure Cooking	ibe your typical av Medium? ttc. Yes ulses you eat are F en. Proportion 1/2 3/4 3 3 3 able. 1 Stewing 2 Stir Fry 3 Paurico	erage portion s 2 1 resh, Frozen, C All 4 5 4 5 4 5 9/Casseroling/I ing/Frying abod/Bau corr	size of foods as? Large? 3 No 2 Canned or Dried.	. Please	Please describe FRUIT: 11: How many serv Can you please indic Please tick the approvement Fresh Stewed Dried Canned 12: Do you ever coor 13: If so, how do you Stewing Baking Other	ings of fruit or f cate how much opriate boxes e. Never	ruit contain of the fruit y g. <sup>1</sup> / <sub>4</sub> Fresh, <sup>1</sup> / <sub>4</sub> 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	ing dishes d /ou eat is Fro /a Canned Proportion 1/2 3 3 3 Yes 1 1 Poaching/ Microwav	Casi o you usual ish, Cannec <sup>3</sup> /4 4 4 4 4 4 8 5 1 Steaming ing	Ily eat each J, Dried or S All 5 5 5 5 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	aking in sauce Other

MEAT: (If you <b>never</b> eat meat please go to question 16)	BREAKFAST:
14: How many servings of meat or meat containing dishes do you usually eat each week?	normally eat that were not mentioned earlier? Yes 1 No 2
What do you do with the visible fat on your meat? Eat all/most of the fat	If yes, which brand and type of breakfast cereal, do you usually eat?
Eat some of the fat	List the types most often used
Eat as little as possible/none	Brand
15. How do you usually cook meat? Tick more than one boy if necessary	
Grilling/Barbecuing/Baking/Roasting (Cooked dry or using a small amount of oil)	
Stir Frving/Frving	2
Microwaving	
Deep frying - including in batter	20: Do you usually take sugar on your breakfast cereary fes in the interview of the second se
Casseroling/Baking in sauce	5 If yes, how many teaspoons?
Other	6 21: Do you usually take sugar/honey in tea, herbal tea,
Please describe	coffee or coffee substitute? Yes 1 No 2
	If Yes, please write the number of teaspoons per cup.
16: How many servings of fish or fish containing dishes do you usually eat each week?	Sugar/honey in tea
How do you usually cook fish. Tick more than one box if necessary.	Sugar/honey in herbal tea
Steaming	2 Sugar/honey in coffee
Grilling/Barbecuing/Baking/Roasting (Cooked dry or using a small amount of oil)	<sup>3</sup> Sugar/honey in coffee substitute teaspoons
Stir Frying/Frying	<sup>4</sup> Do you use sweeteners instead of sugar or honey, Yes 1 No 2
Microwaving	5 Which brand of augustance do you upo placed should y
Deep frying - including in batter	e Which brand of sweetener do you use, please specify
Casseroling//Baking in sauce	If yes how many tablets per day, or how many teaspoons of powder per day?
Other	22. On days when you get bread, how many elices of bread or rolls do you get ?
Please describe	
	USE OF FATS:
MILK:	23: Do you usually spread butter/margarine on your bread?
17. What type of milk do you use most often? Select one only	Yes 1 No 2 Sometimes 3
Full cream (Silver Top)	How many slices of bread/rolls/crackers do you have with spread each day?
Skimmed/fat free 3 Channel Islands (Gold Top)	Uset a corrange/highy enread
Dried Milk 5 Sova 6	How much spread do you use? Sust a scrape/unity spread a
	Thickly spread
Other <sup>9</sup> Specify	24: What kind of fat do you most often use for frying, roasting, grilling etc?
If you used soya milk, please describe brand and type	lick more than one if applicable Butter
	Solid White Vegetable Fat
18: How much milk do you drink each day, including milk with tea, coffee, milky drinks, cereals etc?	Margarine
	None 6
1/2 Pint 3 3/4 Pint 4	
1 Pint 5 More than 1 Pint 6	If you used vegetable oil, or margarine, please give type e.g. corn, sunflower
11	
	51

25: What kind of fat do you most often use for baking cakes etc. ( lick more than one if applicable Butter 2 Solid White Vegetable Eat	II) Describe below how your diet has changed
Lard/Dripping 🔲 ° Margarine 🛄 °	
Vegetable Oil 5 None 6	Do you currently follow any of these diets? Tick more than one box if neccessary.
	Low fat 1 Low salt 2 Diabetic 3
If you use margarine, please give Brand e.g. Flora, Stork	Slimming 4 Gluten free 5 High fibre 6
USE OF SALT:	
26. Here after do you add calt to food while cooking?	Other Please give details
26: How often do you and sait to food while cooking!	CONSUMPTION OF ALCOHOL.
Sometimes 3 Barely 4	31: How often, if ever do you drink alcohol?
Never 5	
27: How often do you add salt to any food at the table?	Less than once a week
Always 1 Usually 2	32: In a typical week, how much do you drink?
Sometimes 3 Rarely 4	Beer or cider pints each week
Never 5	Wine glasses each week
28: Do you regularly use a salt substitute (e.g. Losalt)? Yes 1 No 2	Sherry/Fortified Wines glasses each week
If yes, which brand?	Spirits glasses (singles) each week
	33: Five years ago, how many alcoholic drinks did you have each week?
USE OF SUPPLEMENTS:	Beer or cider pints each week
29: Do you take any vitamins, minerals, fish oils, fibre or	Wine plasses each week
other food supplements? Yes ' No ' *	
If yes, please fill in details below.	
Name and Brand of Supplements How much do you How often do you take these?	Spirits glasses (singles) each week
take at a time Daily Weekly Monthly Less often	SMOKING
1 2 3 4	34: Which one of the following best describes you?
1 2 3 4	I smoke every day 1 I smoke occasionally, but not every day
	Lucad to smake guary day, but do not amake
SPECIAL DIETS:	at all now 3
30: I) Have you changed your diet over the last 12 months? Yes Yes Yes Yes	If you have never smoked, please go to question 37.
If yes, please indicate if the change was for any of the reasons listed below?	35: Do/did you smake?
High Blood Pressure Stomach problems (e.g. ulcer or gastritis)	Cigarettes 1
Bowel problems (e.g. irritable bowel or	Cigars 2
diverticulitis)	A combination of the above 3
Concern over a family history of illness 4 High Blood Cholesterol/Lipids 5	If you currently smalle or used to smalle algoration have made algoration for the above
Overweight/Obesity 6 Diabetes 7	a you currently smoke or used to smoke cigarettes now many doldid you smoke each day?
Allergies (e.g. skin rash) 📃 <sup>8</sup> Other 🛄 <sup>9</sup>	If you currently smoke or used to smoke cigarettes which brand of cigarettes do/did vou usually smoke?
Specify	
13	
	14

36: If you have stopped smoking for what	period of time have y	/ou been a non-sr	noker?				And the second s
1 year or les	3	2-5 years	2	47: In a normal week, do ye to cause sweating or a	ou do any of these activities vigor faster heartbeat?	ously enough	Yes 1 No 2
6-10 year	3	Over 10 years	4	If yes, for how long eac	h week do you do such		
SIZE:		Over to years			vigorous physical activity?	hours	minutes per week
37: Approximately how much did you wei	jh when you were b	orn?					
Ibs or Kg or Don't	(now 📃			ILLNESS:			
				48: Have you ever been tol	d by a doctor that you have, or ha	id, any of the foll	owing conditions?
38: Approximately how much did you wei	gh when you were 20	) years old?		Please tick all which apply a	ind give the age at which each coi	ndition was first	diagnosed.
stones pounds or	Kg or Don	n't Know 🛄		Heart attack, coronary	thrombosis, myocardial infarction	Yes 1 No	<sup>2</sup> at age yrs old
39: Approximately how much do you weig	h at present?				Angina	Yes 1 No	<sup>2</sup> at age yrs old
stones pounds or	Kg or Don	i't Know			Stroke	Yes 1 No	<sup>2</sup> at age yrs old
				Hig	h Blood Pressure (Hypertension)	Yes 1 No	<sup>2</sup> at age yrs old
40: Have you lost more than half a stone in	the last year?		Yes 1 No 2	High Blo	ood Cholesterol, Hyperlipidaemia	Yes 1 No	<sup>2</sup> at age yrs old
(Please ignore weight gained during pl	egnancy.)		Yes 1 No 2		Diabetes	Yes 1 No	<sup>2</sup> at age yrs old
					Gallstones	Yes 1 No	<sup>2</sup> at age vrs old
41: What is your present waist size?	inches or	centimetres or l	Don't Know 1		Polyos in the large intestine	Yes I No	<sup>2</sup> at age wrs old
42: What is your present hip size?	inches or	centimetres or [	Don't Know		Cancar	Voc 1 No	<sup>2</sup> at age vre old
					Concer	103 110 L	
43: What is your present height?	inches or	centimetre	es or Don't Know 1	n yes, what type of can			
44: What size of blouse do you wear?	Size			Any other illnesses or o	operations?	7	
44. What size of bloase do you wear?				Do not include hystered	stomy or breast surgery. These are	covered later in	the questionnaire.
45: What size of skirt do you wear?	Size			Condition/operation/o	lisease		Age first diagnosed
PHYSICAL ACTIVITY							yrs old
46: In a typical week during the last 12 mo	oths, how many hou	rs did you spend	on each of the				yrs old
following activities? Put "0" if none							yrs old
Housework, such as cleaning, washing, coo	king, child care	hours	minutes per week	International and a second sec			
Do-It-Yourself		hours	minutes per week				CIC) yis old
Gardening	In Summer	hours	minutes per week	49: Are you currently recei	ving long-term treatment for any	illness or conditi	on?
	In Winter	hours	minutes per week	If you also should be a		1. 50	Yes 1 No 2
Walking, including to work, shopping & leis	ure In Summer	hours	minutes per week	ii yes, piease give detai	is of treatment. If no please go to	question bu:	
	In Winter		_ minutes per week	Illness or condition	Treatment	Dose	Frequency
Cycling, including to work & leisure	In Summer	hours	minutes per week				
Other physical everying such as less fit	In Winter	hours	minutes per week				
aerobics, jogging, tennis, swimming	In Summer	hours	minutes per week				
	In Winter	hours	minutes per week				
	15				16		

50: Have your mother and/or father ever suffered from cancer or heart attack/heart disease?	Do you have a paid job at present? Yes 1 No 2
Yes 1 No 2 Don't Know 3	If no, how would you describe yourself?
If yes, please give details	Housewife 1 Unemployed 2
	Retired 3 Student 4
	Other <sup>5</sup> describe
	When did you bet have not employment 10 (weet) or Never 1
51: If you have brothers and/or sisters have they ever suffered from cancer or heart attack/beart	
disease?	55: What is your marital status?
Yes 1 No 2 Don't Know 3	Married or living as married
If yes, please describe details	Widowed 3 Single 4
	Separated 5
	If you are not married or living as married, please go to question 57.
	PARTNER'S EMPLOYMENT:
EDUCATION:	56: If married or living as married, has your partner ever had a paid job? Yes 1 No 2 '
52' How old were you when you finished your full time education?	in yes, please answer for your partner's current or most recent job.
	What is/was your partner's job title?
	partner works for make or do?
53: Do you have any of the following qualifications? Tick all applicable	Is/was your partner a Mapager?
CSE A "A" Level, Highers	Supervisor?
GCE "O" Level 2 Teaching diploma, HNC 5	Is/was your partner self-employed? Yes 1 No 2
City & Guilds	Does your partner have a paid job at present? Yes 1 No 2
	If no, how would you describe your partner?
Other describe None of these	House-husband
	Unemployed 3 Retired 4
	Other Statescribe
EMPLOYMENT:	57: Which of these groups would you consider you belong to?
54: Have you ever had a paid job? Yes 1 No 2	White <sup>1</sup> Bangladeshi <sup>2</sup>
If yes, please answer for your current or most recent job	Indian 3 Chinese 4
	Pakistani 🔤 <sup>5</sup> Black - Caribbean 🔤 <sup>6</sup>
what is/was your job title?	Black - other
What do/did you do in your job?	Other
What does/did the organisation	
you work for make or do?	58: How old were you when you had your first menstrual period vears old
Are/were you a Manager? E Foreman/woman?	59: What is the usual length of your menstrual cycle?
Supervisor? 3 None of these? 4	(i.e. from the first day of one period to the first day of the next period e.g. 26 days)?
Are/were you self-employed? Yes 1 No 2	
17	18

60: Have you	ever been pregr	nant?		Yes 1 N	0 2	65: Have you ever used a coil or intra-uterine device (IUD)? Yes 1 N	0 2
Are you p	regnant at the m	noment?	- 00	Yes 1 N	io 2	If yes, do you have a coil or IUD at present? Yes 1 N	0 2
How man	y times have you	u been pregna	nt?	Voc 1 N	2	66: How many "natural" menstrual periods have you had in the last 12 months?	
If you have	e had children	please do to d	uestion 61 If n	int please do tr	o question 63	Do not count bleeding while using the pill or HRT (Hormone Replacement Therapy)	
	o nad officiation, p	prodoci go to q	doodon on an	iot picked go t	, danagin on		
61: Have you	had any children	n?		Yes 1 N	lo 2		
						4 to 5	
If yes, how	v old were you v	when your firs	t child was bor	rn 📃 years		6 to 9	
If yes, how	v many children	have you had	I? Childi	ren			
If none pl	ease go to quest	tion 63.				Not applicable because using the	
Please ca	n you write in ea	ich child's sex	and approxim	ate birthweigh	t.	Pill or HRT or currently pregnant	
			Δτ	oprovimate			
Child	Sex of C	hild	B	irthweight	Child's D.O.B	<ul> <li>When did you last have a "natural" menstrual period? Do not count bleeding while using the HRT (Hormone Replacement Therapy). Record as fully as possible</li> </ul>	pill or
CHILD1:	- 18250 - 1768				19	Date: or age years old Don't kno	ow
CHILD 2:		]			19	68: Have you ever used HRT (Hormone Replacement Therapy) for menopause? Yes 1 N	0 2
CHILD 3:					19		
CHILD 4:					19	If yes, how old were you when you first used HRT?	
CHILD 5:		]			19	For how long altogether have you used HRT? years and months	
62: Did you e	ver breast feed a	any of your chi	ildren?		Yes 1 No	Are you currently using HRT? Yes 1 No 2	
If yes, for	those children y	ou breast-fed,	please descrit	be how long ye	ou continued breast feeding	If no, how old were you when you last used HRT?	
after each	birth, (even only	y occasional b	, preast feeding)	. Tick the appro	opriate box.		
	1-6 days	1-4 weeks	1–3 months	4-6 months	6+ months 12+ months	69. Have you had a hystorectomy 21f no places so to supption 71	
CHILD 1:	1	2	3	4	6		
CHILD 2:		2	3	4	5 6	Yes	
CHILD 3:	1	2	3	4	5 6	70: Have you had an operation to remove one or both your ovaries?	
CHILD 4:	1	2	3	4	5 6	Yes 1 No 2 Don't know	N 3
CHILDE		2	3		5 6	If yes, how old were you? years old	
CHILD 5:						Were one or both ovaries removed? One <sup>1</sup> Both <sup>2</sup> Don't know	N 3
63: Have you	ever seen a doct	tor because of	fertility proble	ems?	Yes 1 No	71. Have you ever had a breast bioney/minor surgery to remove	
lf yes, has	a doctor ever to	old you that yo	u were infertil	e?	Yes 1 No	tissue from your breast for diagnostic purposes)? Yes 1 No 2 Don't know 3	
64: Have you	ever used oral c	ontraceptives	(the pill)?		Yes 1 No	If yes, how old were you (first occurrence)?	
If yes, how	v old were you v	when you first	started to use	the pill?	years old		
For how I	ong altogether d	lid you use the	pill?		years		
Are you c	urrently using th	ne pill?			Yes 1 No		
lf no, how	old were you w	<mark>hen you last</mark> u	ised it?		years old		
			19			20	

72: Can you please give us these other details : DATE OF BI	RTH		
FIRST NAME(s)/FORENAME(s)			Thank you for your help filling out this
OTHER NAME(s)/FORENAME(s)			questionnaire and helping us with our research.
SURNAME AT BIRTH/(MAIDEN NAME)			deceneration and under a second second
CURRENT SURNAME/LAST NAME			Please send your completed questionnaire in the stamped
73: One of the objectives of this study is to astablish the rola	tionship of putrition with the appurcase of	and the second	addressed envelope to:
certain diseases such as cancer which are registered by	the National Health Service (NHS). To de	Wite and the	Amanda Woodhouse,
this we need to know your NHS number if you know it a	and also details about your CP It would be	the blocker fribe	Division of Public Health,
very helpful if you could fill in your NHS purpher (this is	on your medical card). If you were been		The Nuffield Institute for Health,
between 1939 and 1952 this is your Nets number (this is	on your medical card). It you were born		/1-75 Clarendon Koad, Leeds LSZ 9PL.
Example:	nper.		Please phone 0113 233 4862
NHS NUMBER U N A	Z / 4 6 4	Alternation and	if you have any gueries about this guestionnaire.
	Postcode:		
Telephone Number: CONTACTING YOU: 5: If we have any queries about your answers to this question Can you please give us your telephone number	onnaire, we would like to contact you.		
Telephone Number: ONTACTING YOU: 5: If we have any queries about your answers to this question Can you please give us your telephone number Telephone : Dialling Code: Number: Dialling Code: Number:	onnaire, we would like to contact you.		
Telephone Number:         CONTACTING YOU:         75: If we have any queries about your answers to this questi         Can you please give us your telephone number         Telephone :       Dialling Code:         Dialling Code:       Number:         Dialling Code:       Number:         76: We would like to write to you again about the progress or         information about your typical food intake.         In case you change your address and we lose contact with         friends or relatives who would know your new address? F         Contact Name (1):         Contact Aside	Postcode:		
Telephone Number:         CONTACTING YOU:         '5: If we have any queries about your answers to this questic Can you please give us your telephone number         Telephone :       Dialling Code;         Dialling Code:       Number:         Dialling Code:       Number:         Output       Number:         '6: We would like to write to you again about the progress or information about your typical food intake.         In case you change your address and we lose contact with friends or relatives who would know your new address? F         Contact Name (1):       Contact Address:	Postcode:	Plage writ	a pry changes to the details on the label in the snaces below
Telephone Number:         ONTACTING YOU:         5: If we have any queries about your answers to this questic Can you please give us your telephone number         Telephone :       Dialling Code:         Dialling Code:       Number:         Dialling Code:       Number:         S: We would like to write to you again about the progress or information about your typical food intake.         In case you change your address and we lose contact with friends or relatives who would know your new address? F         Contact Name (1):         Contact Address:	Postcode:	Please writ	e any changes to the details on the label in the spaces below
Telephone Number:         ONTACTING YOU:         5: If we have any queries about your answers to this questic Can you please give us your telephone number         Telephone :       Dialling Code:         Dialling Code:       Number:         Dialling Code:       Number:         6: We would like to write to you again about the progress of information about your typical food intake.         In case you change your address and we lose contact with friends or relatives who would know your new address? F         Contact Name (1):         Contact Address:         Telephone Number:	Postcode:	Please writ First name	e any changes to the details on the label in the spaces below
Telephone Number:         CONTACTING YOU:         5: If we have any queries about your answers to this questic Can you please give us your telephone number         Telephone :       Dialling Code:         Dialling Code:       Number:         Dialling Code:       Number:         6: We would like to write to you again about the progress of information about your typical food intake.         In case you change your address and we lose contact with friends or relatives who would know your new address? F         Contact Name (1):         Contact Address:         Telephone Number:         Contact Name (2):	Postcode:	Please write First name Surname	e any changes to the details on the label in the spaces below
Telephone Number:         CONTACTING YOU:         5: If we have any queries about your answers to this questic Can you please give us your telephone number         Telephone :       Dialling Code:         Dialling Code:       Number:         Dialling Code:       Number:         6: We would like to write to you again about the progress of information about your typical food intake.         In case you change your address and we lose contact with friends or relatives who would know your new address? F         Contact Name (1):         Contact Address:         Telephone Number:         Contact Name (2):         Contact Address:	Postcode:	Please write First name Surname Address	e any changes to the details on the label in the spaces below
Telephone Number:         CONTACTING YOU:         '5: If we have any queries about your answers to this questic Can you please give us your telephone number         Telephone :       Dialling Code:         Dialling Code:       Number:         Dialling Code:       Number:         Output       Number:         '6: We would like to write to you again about the progress of information about your typical food intake.         In case you change your address and we lose contact with friends or relatives who would know your new address? F         Contact Name (1):         Contact Address:         Telephone Number:         Contact Name (2):         Contact Address:	Postcode:	Please write First name Surname Address	e any changes to the details on the label in the spaces below